

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

Phosphodiesterase 4 Expression and Proliferation Rates in a Cellular Model of Pulmonary Hypertension

A thesis submitted to the

FACULTY OF BIOMEDICAL AND LIFE SCIENCES

for the degree of

DOCTOR OF PHILOSOPHY

by

Jennifer Elena Millen

Division of Neuroscience & Biomedical Systems Institute of Biomedical and Life Sciences University of Glasgow

October 2004

© Jennifer Millen, 2004

ProQuest Number: 10391050

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10391050

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



i

Declaration

I declare that the work described in this thesis has been carried out by myself unless otherwise cited or acknowledged. It is entirely of my own composition and has not, in whole or in part, been submitted for any other degree.

Jennifer E. Millen

October 2004

Abstract

Pulmonary arterial hypertension (PAH) is characterised by increased vascular resistance which leads to pulmonary artery remodelling and increased smooth muscle cell proliferation (Fishman 2004; Humbert et al., 2004). Chronic hypoxia treated rats (to induce PAH) display alterations in cyclic nucleotide signalling pathways. Both cAMP and cGMP levels are reduced (MacLean et al., 1997), with a corresponding increase in PDE3A/B and PDE5A2 in pulmonary arteries of the chronic hypoxic rat (Murray et al., 2002). The objective of this study was to investigate the expression of PDE4 isoforms in the cellular model of PAH as the PDE4 family represents the major cAMP-hydrolysing activity within these cells (Palmer et al., 1998).

It was shown that in human pulmonary artery smooth muscle cells (hPASMC), the PDE4A10, PDE4A11, PDE4B2 and PDE4D5 isoforms all show significant increases in expression after 14 days exposure of chronic hypoxia (10% O_2). The increase in PDE4 expression did not correspond to an increase in PDE4 activity levels however. Levels of cAMP were tripled and PKA activity was doubled after 7 days in 10% O_2 . The hypoxia induced increase in cAMP was determined to be ERK dependent as use of the MEK inhibitor, UO126, could reduce hypoxic cAMP levels to that of normoxic cells.

The autocrine loop shown previously in aortic smooth muscle cells (Baillie et al., 2001), was demonstrated to be in effect in hPASMC. In this loop, active ERK led to the production of PGE₂ through PLA₂. This increase in PGE₂ stimulated adenylyl cyclase and increased the generation of cAMP. Using the COX-2 inhibitor, indomethacin, completely ablated the rise in cAMP levels in hypoxia. Exogenous PGE₂ mimicked the hypoxia induced rise in cAMP in normoxic cells. Inhibition of ERK also led to a decrease in PDE4 activity through the reduced level of cAMP generated and thus decreased the amount of PKA able to activate PDE4. This effect was not witnessed in hypoxia, suggesting desensitisation to cAMP stimulation had occurred. Indeed, it was observed that hypoxic cells produced less cAMP in response to cAMP agents such as rolipram or PGE₂ than normoxic cells.

Vascular smooth muscle cells have been reported to proliferate in response to hypoxia. Indeed, it was demonstrated in this study that hPASMC do increase their proliferation in response to hypoxia in both serum starved cells and cells cultured in serum. The PKA RII subunit was demonstrated to be essential for normal proliferation of these cells. Both PKA

iii

RI and PKA RII can significantly reduce proliferation in both normoxic and hypoxic cells. Also, the recently discovered cAMP substrate Epac (de Rooij et al., 1998), elicits anti-proliferative effects on serum induced proliferation in hPASMC.

iv

Acknowledgements

My thanks go to both Prof. M. Houslay and Prof. M. MacLean for their supervision, guidance, help with my thesis and all the support they've shown me throughout the last three years.

v

ł

Big thank you to everyone in the Gardiner Lab. and Prof. MacLean's Lab., I've met many people whose friendship has ensured my time there has been enjoyable. I've had such a good laugh with you all (I think because we're all equally nuts!!) Ang, I'll never forget the bad fairy, top of the morning to ya!! Dex – bench buddy, how are you surviving without 7 shots of espresso a day?? Never will I own a sare box that is so full again. Ahmed, one day we'll find a club we both love! Trix, really, you're just a nutcase, thanks for showing me I'm not alone! Martian, I wouldn't have wanted to share the end of two great shows with anyone else! Thank you for your contribution with the siRNA work also. Yorkie!! You're a sweetheart, thank you for all the chats! I look forward to many more now you're officially a 'Shedder'! EVH, I've never laughed so much at baseball! Elaine Huston, keep laughing!! It's soo infectious! George, thanks for all the Georgisms, they always made me chuckle! You guys!

Big T - it wouldn't have been the same without you, you kept me smiling when it was hard! Don't think I've ever drank so much coffee though! Boston was amazing – sorry for all the museums! How much are flats in Berlin?! Although, no double deckers???

I would also like to thank my mum and dad to whom I'm eternally grateful to for all the love and support (not just financially!) they've given me in the last few years and always. Chief and ali-mah-bo also, you always make me smile-you're both mad and I don't know what I'd do without you!

My 'special' friends have ensured the last few years have not been sober ones, thank you all for many good nights in the Shed and many more to come! You're not right, btw!

Rick, I wouldn't have been able to finish this without your support and encouragement. Thank you for everything, you've kept me motivated and upbeat constantly, big roar!!!

Table of Contents

Declaration	ii
Abstract	iii
Acknowledgements	¥
Table of Contents	
List of Figures	xii
List of Tables	xiv
Abbreviations	XV
Chapter 1	
General Introduction	1
1.1 Cyclic nucleotide signalling pathways	2
1.1.1 Cyclic nucleotides	2
I.I.I.I G-Protein Coupled Receptors	
1.1.1.2 Adenylyl Cyclases	
1.1.2 Cellular Targets of cAMP	,4
1.1.2.1 Protein Kinase A	4
1.1.2.2 A-kinase anchoring proteins	5
1.1.2.3 Exchange Protein Directly Activated by cAMP	
1.2The Phosphodiesterase Superfamily	
1.2.1 The PDE1 Family	
1.2.2 The PDE2 Family	
1.2.3 The PDE3 Family	
1.2.4 The PDE4 Family	9
1.2.5 The PDE5 Family	
1.2.6 The PDE6 Family	
1.2.3 The PDF7 Family	11
1.2 % The DDER Family	12
1.2.6 The PDE0 Panity	
1.2.9 The FDE9 family,	ען _{ייי} ייייייייייייייייייייייייייייייייי
1.2.10 The PDE10 Fallity	
(,2,() The TDDTT Pathoy	
1.3 Organisation and Regulation of the PDE 4 Family	
1,3.1 Modular structure of PDE4.	
1.3.2 Regulation of PDE4 activity	
1.3.3 The PDE4A subfamily	
1.3.4 The PDE4B subfamily	
1.3.5 The PDE4C subfamily	
1.3.6 The PDE4D subfamily	
1.3.6.1 Interactions and Targeting of the PDE4D subfamily	
1.3.6.2 Induction of PDE4D by cAMP	
1,4 PDE4 Inhibitors and Disease	
1.4.1 Relaxant effects of PDE4 inhibitors	
1.5 Pulmonary Arterial Hypertension	
1.5.1 Models used in research of PAH	
1.5.2 Pathology of PAH	
1.5.3 Pułmonary artery smooth muscle cells	
1.5.4 Pulmonary artery fibroblasts	
1.5.5 Pulmonary artery endothelial cells	
1.5.5.1 Endothelin-1	

2.....

 Consistent and a state of the second sec second sec

	vii
1.5.5.2 Nitric Oxide	
1.5.5.3 Prostaglandins	
1.5.5.4 Vascular Growth Factors	
1.5.6 Mitogen Activated Protein Kinases	
1.5.6.1 Activation of MAPK	
1.5.6.2 Hypoxia-induced activation of MAPK	
1.5.6.3 HIF-1 and ERK1/2	
1.5.7 Role of 5-Hydroxytryptamine in PAU	
1.5.8 Cyclic nucleotide pathways in PAH	
1,5.8.1 Cyclases	
1.5.8.2 Cyclic nucleotides	
1.5.8.3 CREB.	
1.5.8.4 Phosphodiesterases	

ł

Materials and M	1ethods	
a f Manualian	Call Calture	51
2.1 Mammanan		
2.1.1 IVial	menalice of hr Above, Arture Smooth Mussle Calls	
2.1.1.1 Hu	mail Pullionary Aftery Shooth Muscle Cons	
2, 1, 1, 2 Pas	saging of REASMC material	
2.1.1.5 my	DOXIC NPASIML MODEL	
2.1.1.4 HG	atment of hPASMC with Agomsis/humonors and a second s	
Z.1.2 Transfer	cuon of nPASMC with the Amaxa Nucleoractor	
2.1.2.1 UR	Instection of Sirina constructs	
2.2 Proliferation	Analysis	
2,2.1 DNA Sy	nthesis Assays	53
2.2.1.1	Assay preparation	
2.2.1.2	Addition of [³ H]-Thymidine	
2.2.1.3	Measuring radioactivity	53
2.2 Direk suited	Analysia	53
2.5 Diochemical	Arran years Coll L spoto	53
2.3.1 Ha N	What will be a production	
2.3.1.4 0.2.2 Dot	whole cell tysice production	51
	CE blockrapherszir	54
2.3.3 5173-17	The Electrophotesis and the space above above is	54
2.3.3.1.311	tion and running an approximite cal	54
2.3.3.2 Ca	buing a NuPACE gel	55
2,5.5.5 Ku 2.2 / Waster	Blotting	
2.3.4 Western	stain transfer from acculantide gel to nitrocellulose membrane	
2.3.4.1 PK	seking of nitroeellulaye	. 55
2.3.4.2 Di	munoblotting	55
2.3.4.5 mi 2.3.5 Phosnhi	adjostorase 4 Activity Assay	56
2351 40	tivation of Dower	
2.3.5.1 He	ruple preparation for assay	
2.3.5.2 Oa	termination of PDE3 and PDE4 activity	
236 Imm	unoprecipitation	
2361 Pr	-clearing agarose beads	
2.3.6.2 Bi	ading target protein to antibody	
2.3.7 Cyclic	AMP Assav	
2.3.7.1 As	vonist Treatment	
2.3.7.2 1	sale extraction for assay	
2.3.7.3 Ne	utralisation of the sample	
2.3.7.4 Rc	agent preparation for assay	
2.3.7.5 Sa	mple preparation for assay	
2.3.7.6 Co	nstruction of standard curve	
2.3.7.7 Ca	lculation of cAMP levels	61
2.3.8 PKA A	SSEV	61
2.3.8.1 Ce	Il lysate extraction for PKA assay	
2.3.8.2 Pk	A assay tube pre-incubation	
2.3.8.3	PKA assay reaction	62
2.3.8.4	Calculations for PKA assay data	

2.4 Molecular Biology	63
2.4.1 DNA Production	63
2.4.1.1 Small scale production of DNA	63
2.4.1.2 Large scale production of DNA	64
2,4,1,3. Quantification of DNA & RNA	64
2.4.2 Glycerol Stock Production.	65
2.4.3 Reverse Transcription PCR	65
2.4.3.1 Total RNA extraction	65
2.4.3.2 cDNA synthesis	65
2.4.3.3 PCR reaction	66
2.4.3.4 Semi-Quantitative RT-PCR	
2.3.4.5 Visualisation of PCR Products	66
2.5 Microarray Analysis	67
2.5.1 RNA Extraction	67
2.5.1.1 Extraction of total RNA	67
2.5.1.2 Clean up of total RNA	
2.5.2 Incorporation of Fluorophores	67
2.5.2.1 cDNA synthesis	
2.5.3 Hybridisation	68
2.5.3.1 Incubation of microarray with sample	
2.5.3.2 Washing of microarray	68
2.5.4 Analysis	68

Effect of Chronic Hypoxia on PDE4 in hPASMC	70
3.1 Introduction	71
3.1.1 Pulmonary Arterial Hypertension	71
3.1.1.1 Pulmonary arterial hypertension	71
3.1.1.2 Hypoxia-induced Pulmonary Hypertension	7 2
3.1.2 Cyclic Nucleotide Signalling in Models of Hypoxia-Induced Pulmonary Hypertension	72
3.1.2.1 Cyclic Nucleotides in Hypoxia	72
3.1.2.2 Phosphodicsterases in Hypoxia	73
3.1.2.3 PDE4 in Hypoxia	73

Results

3.2 Characterisation of PDE4 Profile in hPASMC	75
3.2.1 Analysis of PDE4 expression	75
3.2.2. PDF4A profile in hPASMC	
3.2.2.1 Western blot analysis of PDE4A isoforms in hPASMC	75
3.2.2.2 RT-PCR analysis of PDE4A isoforms in hPASMC	
3.2.3 PDE4B profile in hPASMC.	76
3.2.3.1 Western blot analysis of PDE4B isoforms in hPASMC	76
3.2.3.2 KT-PCR analysis of PDE4B isoforms in hPASMC	76
3.2.4 PDE4C profile in hPASMC	77
3.2.4.1 Western blot analysis of PDE4C isoforms in hPASMC	77
3.2.4.2 RT-PCR analysis of PDE4C isoforms in hPASMC	77
3.2.5 PDE4D profile in hPASMC	
3.2.5. Western blot analysis of PDE4D isoforms in hPASMC	77
3.2.5.2 RT-PCR analysis of PDE4D isoforms in hPASMC	77
3.2.6 cAMP-PDF activity in hPASMC	78
3 3 Effect of chronic hypoxia on PDF4 in hPASMC	
3.3.1 Examining the effect of chronic hypoxia on PDE4	
3.3.7 Analysis of PDF4 isoformy expression in chronic hypoxia	
3,3,2 Analysis of 1 DL+ isotoring expression in omotio hypothetic interaction and the second se	70 70
3.3.2.1 FDENK	····· /70
5.5.2, 1,1 western biot analysis of PDB4A	77
3.3.2.1.2 Somi-quantitative KT-PCR analysis of PDE4A	
3.3.2.2 PDE4B	

viii

	1X
3.3.2.2.1 Western blot analysis of PDE4B.	
3.3.2.2.2 Semi-guantitative RT-PCR analysis of PDE4B	
3,3,2.3 PDE4C	80
3.3.2.3.1 Western blot analysis of PDE4C	
3.3.2.4 PDE4D	
3.3.2.4.1 Western blot analysis of PDE4D	
3.3.2.4.2 Semi-quantitative RT-PCR analysis of PDE4D expression	
3.3.3 Western blot analysis of HIF-1a Expression	
3.3.4 Microarray Analysis of normoxic and hypoxic hPASMC	
3.3.5 Effect of chronic hypoxia on cAMP-PDE activity	81
3.3.5.1 Total cAMP-PDE activity	
3.3.5.2 PDE3	
3.3.5.3 PDE4	
3.3.5.4 Immunoprecipitated PDE4 subfamilies	82
3.4 Discussion and Conclusions,	

Effect of Chronic Hypoxia on cAMP Levels in hPASMC	
4.1 Introduction	
4.1.1 Cyclic Nucleotide Signalling in Models of Hypoxia-Induced Pulmonary Hypertension	116
4.1.1.1 Cyclic Nucleotides in Hypoxia	
4.1.1.2 Cyclases in Hypoxia	
4.1.1.3 Attenuated responses in Hypoxia	117

Results

4.1 cAMP signalling in normoxic and hypoxic hPASMC	
4.1.1 cAMP levels in normoxic and hypoxic hPASMC	
4.1.2 PKA Activity in hypoxic hPASMC	
4.1.3 PKA Expression in hypoxic hPASMC	
4.2 Regulation of cAMP levels in hypoxic hPASMC	
4.2.1 cAMP levels in normoxic and hypoxic hPASMC in response to PDIs inhibitors	119
4.2.1.1 Intracellular cAMP levels	
4.2.1.2 Extracellular cAMP levels	
4.2.2 cAMP levels in normoxic and hypoxic hPASMC in response to cAMP effectors	
4.2.2.1 Intracellular cAMP levels	
4.2.2.2 Extracellular cAMP levels	
4.2.3 cAMP levels in normoxic and hypoxic hPASMC in response to U0126	
4.2.4 cAMP levels in normoxic and hypoxic hPASMC in response to PD980589	
4.2.5 Effect of PGE2 and indomethacin on cAMP levels in normoxic and hypoxic hPASMC	
4.3 Discussion and Conclusions	
Chapter 5	
ERK in Hypoxic hPASMC	144
5.1 Introduction	
5.1.1 The ERK pathway	
5.1.1.1 Regulation of ERK activation by cAMP.	

Results

5.2 The autocrine loop and PDE activity in hypoxic hPASMC	
5.2.1 Effect of ERK inhibitors on PDF4 activity	
5.2.1.1 Effect of PD980589 on PDE4 activity	
5.2.1.2 Effect of U0126 on PDE4 activity	
5.2.1.3 Effect of U0126 on ERK phosphorylation	149
5.2.2 PDE Activity in response to PGE2 and indomethacin in hypoxic hPASMC	149

Professional and a strain and a strain a strain and a strain a st

:

	х
5.2.2.1 Total cAMP-PDE activity	
5.2.2.2 PDE4 activity	
5.2.3 Effect of U0126 and Indomethacin on PDE4 expression levels	
5.2.3.1 PDE4A10/4A11	
5.2.3.2 PDE4B2	
5.2.3.3 PDE4D5	
5.3 ERK in normoxic and hypoxic hPASMC	
5.3.1 ERK in normoxic and hypoxic hPASMC	
5.3.1.1 Effect of EGF on pERK levels	
5.3.1.2 Effect of EGF on PDE4 activity levels	
5.4 Knockdown of PDE4D5 in normoxic and hypoxic hPASMC	
5.4.1 Effect of PDE4D5 siRNA on PDE4D5 expression	
5.4.2 Effect of PDE4D5 siRNA on pERK expression	
5.5 Discussion and Conclusions	

Proliferation in hPASMC	
6.1 Introduction	
6.1.1 The Cell Cycle	
6.1.2 Cyclic AMP inhibition of the cell cycle	
6.1.3 Epac	
6.1.4 Phosphodiesterases and proliferation	
6.1.5 Proliferation in Hypoxia laduced Pulmonary Hypertension	

ł

 $\mathcal{A} = \mathcal{A} = \mathcal{A}$

ļ

-

· nu ·····a adar A- a da------

Results

6.2 Basal Proliferation Rates in Normoxic and Hypoxic hPASM cells	
6.2.1 Basal Proliferation Rates	
6.3 Effact of Raising cAMP on Proliferation Rates in Normanic and Hypoxic hPASMC	. 175
6.3.1 Effect of PDE inhibitors	176
6.3.1.1 Effect of Rolinnam	176
6.3.1.2. Effect of Cilostamide	176
6.3.1.3. Effect of Caprinast	176
6.3.1.4 Effect of Rollingan and ("ilostamide	
6.3.2 Rifect of cAMP effectors	177
6.3.2.1 Effect of Isoprotector	
6.3.2.2 Effect of Isoproterenol and Rollingam	
6.3.2.3 Effect of Forskolin	
6.4 Effect of Growth Factors on Proliferation Rates in Normoxic and Hypoxic hPASMC	
6.4.1. Effect of EGF	
6.4.2 Effect of EGF and Rolipram	
6.4.3 TGF-β1	
6.5 Effect of the Src Family Inhibitor, PP2, on Proliferation Rates in Normoxic and Hypoxic hPA	ISMC180
6.5.1 Effect of PP2	
6.5.2 Effect of PP3	
6 6 The cAMP Pathway	
6.6.1 Effect of PKA Agonists	
6.6.]. I PKA RI Agonist	
6.6.1.2 PKA RII Agonist	
6.6.1.3 PKA RI and RII Agonists	
6.6.1.4 Effect of the PKA Agonist, 6-BnZ-cAMP	
6.6.2 Effect of PKA Antagonists and Inhibitor	
6.6.2.1 Effect of the PKA Inhibitor, H89	
6.6.2.2 Effect of PKA RI Antagonist	
6.6.2.3 Effect of the PKA RII Antagonist	
6.6.3 Effect of cAMP analogs	
-	

	xi
6.6.3.1 Effect of 8-Br-cAMP	184
6.6.3.2 Effect of CPT-cAMP	184
6.6.4 Specific EPAC agonist	
6.6.4.1 Effect of 8 -pCPT-2'-O-Me-cAMP	
6.6.5 Effect of cAMP Effectors on the anti-proliferative capacity of 8 -pCPT-2'-O-Me-cAMP	
6.6.5.1 Rolipram	
6.6.5.2 Cilostamide	
6.6.5.3 Rolipram and Cilostamide	185
6.6.5.4 8-br-cAMP	
6.6.5.5 6-BnZ	
6.6.5.6 PKA KI agonist	186
6.6.5.7 PKA RII agonist	186
6.6.5.8 PKA RI and RII agonists	
6.7 Effect of the EPAC Agonist, 8 -pMeOPT-2'-O-Me-cAMP, and a Rap1 Inhibitor on Proliferation	
compared to 8 -pCPT-2'O-Me-cAMP	
6.7.1 Arresting cell cycle using a Rap1 Inhibitor, GGT1-298	
6.7.2 A second EPAC agonist, 8 -pMeOPT-2'-O-Me-cAMP	187
6.8 Discussion and Conclusions	
Chapter 7	
General Discussion and Future Directions	209
References	215

List of Figures

Chapter 1

۲

Figure	1.1 Modular structure of phosphodicsterases	43
Figure	1.2 Splice variants of the PDE4 family	44
Figure	1.3 Autocrine activation of PDE4D5 by ERK	45
Figure	1.4 Venice classification of pultoonary arterial hypertension	47
Figure	1.5 Activation of MAPK	48
Figure	L6 ERK pathway	49
Figure	1.6 ERK pathway	49

Chapter 3

Figure 3.1 Western blot analysis of PDE4A isoforms expressed in hPASMC.	89
Figure 3.2 Primer sequences used for analysis of PDE4A isoforms in bPASMC	90
Figure 3.3 RT-PCR analysis of hPASMC using PDE4A specific primers.	91
Figure 3.4 Western blot analysis of PDE4B isoforms expressed in hPASMC	92
Figure 3.5 Primer sequences used for analysis of PDE4B isoforms in hPASMC	93
Figure 3.6 RT-PCR analysis of hPASMC using PDE4B specific primers	94
Figure 3.7 Western blot analysis of PDE4C isoforms expressed in hPASMC	95
Figure 3.8 RT-PCR analysis of hPASMC using PDE4C specific primers	96
Figure 3.9 Western blot analysis of PDE4D isoforms expressed in hPASMC.	97
Figure 3.10 Primer sequences used for analysis of PDE4D isoforms in hPASMC	98
Figure 3.11 RT-PCR analysis of hPASMC using PDE4D specific primers,	99
Figure 3.12 cAMP-PDE activity in hPASMC	.100
Figure 3.13 Western blot analysis of the effect of chronic hypoxia on PDE4A isoform expression	.101
Figure 3,14a RT-PCR analysis of the effect of chronic hypoxia ou PDE4A isoform expression	,102
Figure 3.14b Densitometry of RT-PCR analysis of the effect of chronic hypoxia on PDE4A isoform	
expression	,103
Figure 3.15 Western blot analysis of the effect of chronic hypoxia on PDE4B isoform expression	.104
Figure 3.16 RT-PCR analysis of the effect of chronic hypoxia on PDE4B isoform expression	.105
Figure 3.17 Western blot analysis of the effect of chronic hypoxia on PDE4C isoform expression	.106
Figure 3.18 Western blot analysis of the effect of chronic hypoxia on PDE4D isoform expression	.107
Figure 3.19a RT-PCR analysis of the effect of chronic hypoxia on PDE4D isoform expression,	.109
Figure 3.19b Densitometry of RT-PCR analysis of the effect of chronic hypoxia on PDE4D isoform	
expression	.109
Figure 3.20 IIIF-1a expression in normoxic and hypoxic hPASMC.	.110
Figure 3.21 Microarray analysis of RNA from normoxic and hypoxic hPASMC	.111
Figure 3.22 Effect of chronic hypoxia on cAMP-PDE activity	.112
Figure 3.23 Effect of chronic hypoxia on PDE4 subfamily activity.	.113
Figure 3.24 Proposed model of cAMP regulation during chronic hypoxia	.114

Chapter 4

Figure 4.1 Effect of chronic hypoxia on intracellular cAMP levels.	128
Figure 4.2 Effect of chronic hypoxia on PKA activity	129
Figure 4.3 Effect of chronic hypoxia on PKA subunit expression.	130
Figure 4.4a Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on intracellular	r
cAMP levels.	131
Figure 4.4b Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on intracellula	r
cAMP levels	132
Figure 4.5a Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on extracellula	ir
cAMP levels	133
Figure 4.5b Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on extracellula	u
cAMP levels	134
Figure 4.6a Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on intracellula	ar
cAMP levels.	135
Figure 4.6b Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on intracellul	ar
cAMP levels	136
Figure 4.7a Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on extracellul	ar
cAMP levels	137

a a a a a a a sea a sea a cara a sea a

÷

Figure 4.7b Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on extracellul	ជវ
cAMP levels	138
Figure 4.8a Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC	139
Figure 4.8b Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC.	140
Figure 4.8c Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC	141
Figure 4.9a Effect of PGE2 and indomethacin on cAMP levels in hypoxic and normoxic	142
Figure 4.9b Effect of PGE2 and indomethacin on cAMP levels in hypoxic and normoxic hPASMC	143

Figure 5.1 cAMP activation of ERK	158
Figure 5.2 Effect of PD980589 on PDE Activity in normoxic and hypoxic hPASMC	159
Figure 5.3 Effect of U0126 on PDE Activity in hypoxic and normoxic hPASMC,	160
Figure 5.4 Effect of U0126 treatment on ERK1/2 phosphorylation.	163
Figure 5.5 Effect of PGE2 and indomethacin on PDE Activity in hypoxic and normoxic hPASMC	162
Figure 5.6 Effect of U0126 and indomethacin on PDE4A10/11 expression levels in hypoxia and normov	cia.
	163
Figure 5.7 Effect of U0126 and indomethacin on PDE4B2 expression levels in hypoxin and normoxia	164
Figure 5.8 Effect of U0126 and indomethacin on PDE4D5 expression levels in hypoxia and normoxia	165
Figure 5.9 Effect of EGF on pERK in normoxic and hypoxic hPASMC	166
Figure 5.10 Effect of EGF on PDE4 activity levels	167
Figure 5.11 PDE4DS knockdown in hPASMC	168
Figure 5.12 Effect of knockdown of 4D5 on pERK expression in normoxic and hypoxic hPASMC	169

Chapter 6

Figure 6.1 Effect of chronic hypoxia on proliferation in hPASMC.	193
Figure 6.2 Effect of PDE inhibitors on proliferation in hPASMC.	. 194
Figure 6.3 Effect of cAMP offectors on proliferation in hPASMC	195
Figure 6.4 Effect of EGI ² on proliferation in hPASMC.	196
Figure 6,5 Effect of TGF-B1 on proliferation in hPASMC.	197
Figure 6.6 Effect of Src inhibitor on proliferation in hPASMC,	198
Figure 6.7 Effect of PKA agonists and inhibitor on proliferation in hPASMC	. 199
Figure 6.8 Effect of PKA antagonists on proliferation in hPASMC.	200
Figure 6.9 Effect of cAMP analogs on proliferation in hPASMC	201
Figure 6.10 Effect of 8 -pCPT-2'-O-Me-cAMP on proliferation in hPASMC	202
Figure 6.11 Effect of PDE inhibitors in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in	
hPASMC.	203
Figure 6.12 Effect of cAMP effectors in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in	
hPASMC.	204
Figure 6.13 Effect of PKA agonists in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in	
hPASMC.	205
Figure 6.14 Effect of Rap1 inhibitor on proliferation in hPASMC	206
Figure 6.15 Inhibition of proliferation in response to EPAC agonists.	207

xiii

List of Tables

Chapter 1

Table 1.1 Summary of adenylyl cyclase isoform distribution	4) 4) 42 46
Chapter 2	40
Table 2.1 Antibodies and Dilutions,	69
Chapter 6	
Table 6.1 Drugs used and their final concentration in the proliferation assays	.08

Abbreviations

AC	adenylyl cyclase
АКАР	A kinase anchoring protein
ASMC	Airway smooth muscle cell
ATP	adenosine trisphosphate
BMPR	Bone morphogenetic protein receptor
Ca ²⁺ /CaM	calcium/ealmodulin
cAMP	cyclic 3'5' adenosine monophosphate
CHO	Chinese hamster ovary
CRE	cAMP response element
CREB	cAMP response element binding protein
Cdk	Cyclin dependent kinase
COPD	chronic obstructive pulmonary disorder
cDNA	complementary DNA
cGMP	cyclic guanosine monophosphate
DMSO	dimethylsulphoxide
DNA	deoxyribonucleic acid
dNTP	deoxynucleotide trisphosphate
DTT	dithiothreitol
ECL	Enhanced chemiluminescence
EDTA	Diaminoethanetetra-acetic acid
EGF	epidermal growth factor
EGTA	Ethylene glycol-bis(β-aminoethyl ether)-N,N,N',N'-tetraacetic acid
Epac	Exchange protein directly activated by cAMP
ERK	Extracellular regulated kinase
ET-1	Endothelin-1
FCS	foetal calf serum
GEF	Guanine nucleotide exchange factor
GPCR	G-protein coupled receptor
G-protein	guanine nucleotide binding regulatory protein
GRK	G-protein receptor specific kinase
HIF	hypoxia inducible factor
HRE	hypoxia response element
br	hour
ІВМХ	isobutylmethylxanthine

ŝ

.

IC_{50}	Concentration of inhibitor required to inhibit half the specific
	activity
IL	interleukin
ſP ₃	inositol-1, 4, 5-triphosphate
K _m	Michealis-Menton constant
kb	kilobase
kDa	kiloDalton
KHEM	potassium (K), HEPES, EGTA, Magnesium
1	litre
LB	Luria-Bertoni
LR	linker region
М	molar
mg	milligram
MAP kinase	mitogen activated protein kinase
MEK	MAPK kinase
min	minute
mRNA	messenger RNA
NO	nitric oxide
NOS	nitric oxide synthase
PA	phosphatidic acid
PAEC	pulmonary arterial endothelial cell
PAF	Pulmonary arterial fibroblast
PAGE	Polyacrylamide gel electrophoresis
PAH	pulmonary arterial hypertension
PASMC	Pulmonary arterial smooth muscle cell
PBS	phosphate buffered saline
PCR	polymerase chain reaction
PDE	phosphodiesterase
PDGF	Platelet derived growth factor
PGE ₂	prostaglandin 2
РКА	protein kinase A
РКС	protein kinase C
PKG	protein kinase G
PLA ₂	phospholipase A ₂
РМА	Phorbol myristate
PPH	primary pulmonary hypertension

xvi

RACK	receptor for activated C kinase
RNA	ribonucleic acid
rpm	revolutions per minute
RT	reverse transcription
SAPU	Scottish antibody production unit
SDS	sodium dodecyl sulphate
sec	second
SH3 domain	Src homology 3 domain
SmBm	smooth muscle basal medium
SmGm	smooth muscle growth medium
TBE	tris/borate/EDTA
TBS	tris buffered saline
TEMED	N,N,N^*,N^* -Tetramethyl-ethylenediamine
TFA	Trifluoroaceatic acid
TGF	Transforming growth factor
VEGF	vascular endothelial growth factor
VSMC	Vascular smooth muscle cell
5-HT	serotonin

...

xvii

General Introduction

1.1 Cyclic nucleotide signalling pathways

1.1.1 Cyclic nucleotides

The cyclic nucleotides, cyclic 3'5' adenosine monophosphate (cAMP) and cyclic 3'5'guanosine monophosphate (cGMP) are intracellular second messengers involved in a myriad of cellular events including metabolism, cell growth and proliferation, gene regulation and vascular tone (Pellegrino & Wang 1997; Dousa 1999). cAMP was discovered in 1958 (Sutherland & Rall 1958) and the second messenger concept was proposed whereby binding of hormones or neurotransmitters, so called 'first messengers', to cell-surface receptors elicited a signal which is transduced via an intracellular mediator (second messenger) such as cAMP.

Adenylyl cyclase catalyses the synthesis of cAMP from ATP and cAMPphosphodiesterases (PDE), which are the only entities able to degrade cAMP, hydrolyse cAMP to the inactive form, 5'AMP. The cognate synthesis and degradation of cGMP exists with guanylyl cyclase providing the synthetic activity of cGMP and cGMP-PDEs being responsible for the degradation and inactivation of cGMP.

The well-characterised target for cAMP action within a cell is the cAMP-dependant kinase, protein kinase A (PKA). Stimulation of PKA results in the phosphorylation of a number of downstream targets leading to alterations in gene expression and cellular processes (Cho-Chung et al., 1995; Shabb 2001). Similarly, cGMP mediates the majority of its effects through cGMP dependant protein kinase (Lohmann et al., 1997). Mammalian cGMP kinase exists in two isoforms, cGMP kinase I and cGMP kinase II which show cell specific distribution and different regulatory roles (Ruth 1999).

1.1.1.1 G-Protein Coupled Receptors

Adenylyl cyclase can be activated directly by forskolin or indirectly through interaction with an activated G-protein coupled receptor (GPCR). These receptors are seven transmembrane spanning proteins that form an α -helix with an extracellular N-terminus, three extracellular loops, three intracellular loops and an intracellular C-terminal tail. The superfamily of GPCRs include receptors for many different ligands and are therefore able to respond to numerous extracellular stimuli including taste, light, odour, neurotransmitters and hormones. GPCRs are responsible for the transduction of these signals into intracellular second messengers able to carry the message inside cells, thus achieving the correct response. Upon ligand binding, the GPCR undergoes a conformational change

allowing interaction and activation of a heterotrimeric guanine nucleotide binding protein (G-protein). This conformational change allows the exchange of GDP for GTP on the G-protein α -subunit which activates the G-protein and promotes the dissociation of the G α -and G_{by}- subunits of the G-protein. The dissociated subunits are then free to act upon their effector protein, e.g. adenylyl cyclase, in a positive or negative fashion. The cAMP signalling pathway is activated by one of the G α_s isoforms and can be inhibited by the G α_i isoforms. Activation of adenylyl cyclase is sustained until the GTP bound to the G α_s is hydrolysed to GDP and the G-protein subunits reassociate.

GPCR signalling is regulated by mechanisms at the level of the ligand, the receptor, the Gprotein and at numerous stages in the signalling pathway. The earliest method of regulation is removal of the agonist from the extracellular fluid by transporters or through degradation. The receptor itself is regulated by feedback mechanisms initiated by activation (reviewed in Bohm et al., 1997, Ferguson 2001). This receptor desensitisation is caused by the uncoupling of the G-protein from the receptor in response and can occur within seconds of activation. The most immediate cause of desensitisation is phosphorylation of the receptor by intracellular serine/threonine kinases (GRKs), which phosphorylate the intracellular loop and C-terminal tail of the GPCR. The G- protein receptor kinases (GRKs) selectively phosphorylate agonist occupied receptors promoting the recruitment of cytosolic binding proteins known as arrestins, which sterically uncouple the receptor from the G-protein (reviewed in Luttrell & Lefkowitz 2002). Arrestin binding is also thought to aid receptor endocytosis. Second messenger dependant kinases such as PKA or PKC can phosphorylate various GPCRs including those without agonist bound, whereupon they also serve to inhibit the interaction between the G-protein and receptor.

A growing number of proteins have been identified (Bohm et al., 1997) that both bind GPCRs and couple GPCRs to G-protein independent signal transduction pathways or affect G-protein specificity and agonist selectivity. These include GRKs, arrestins and A-kinase anchoring proteins (AKAPS).

1.1.1.2 Adenylyl Cyclases

Adenylyl cyclases (ACs) are encoded by multiple genes and are transmembrane proteins. Nine different isoforms have been cloned to date which show tissue-specific expression and differential regulation. AC isoforms from all families are highly expressed in brain tissue, although individual isoforms show distinct patterns of localisation within the brain. AC isoform expression in tissues outside of the brain shows a highly specialised

distribution pattern (Antoni et al., 2000, Hanoune & Defer 2001; summarised in table 1.1). For example, AC2, 3, 5-8 are all expressed in rat pulmonary vascular tissue with AC2, 5 and 8 being of functional importance (Jourdan et al., 2001).

Each AC isoform consists of two hydrophobic domains composed of six transmembrane helices (M1 and M2) and two cytoplasmic domains (C1 and C2) which contain the catalytic core. C1 and C2 share homology between different isoforms whereas the helices do not (Hanoune et al., 1997, Houslay & Milligan 1997).

All AC isoforms are regulated by G-protein subunits with type-specific effects. Group1 isoforms (AC1, 3, 8) are stimulated by calcium and calmodulin. Group2 (AC2, 4, 7) are activated by $\beta\gamma$ subunit and protein kinase C (PKC) phosphorylation. Group 3 (AC5, 6) are inhibited by low concentrations of calcium and Group 4 (AC9) is insensitive to calcium or the $\beta\gamma$ subunit. All isoforms are activated by the Ga_s subunit of the G protein (reviewed in Hanoune & Defer 2001, Cooper 2003; for a summary of AC regulation, see table 1.2). Even within these general subclasses, the cyclases are so different to one another that they have other regulators that act upon each of them in a specific fashion. For example, Ca²⁺/calmodulin can regulate AC in a positive (AC1 and AC8) or negative (AC3, 5 and 6) manner. PKA phosphorylation has also been shown to reduce AC activity by direct phosphorylation (Iwami et al., 1995). PKA phosphorylation affects AC5 and AC6 and causes a 'switching' of receptor coupling from Gas to Gai which exerts an inhibitory effect on AC activity and reduces cAMP synthesis. The diterpene, forskolin, can directly activate all AC isoforms with the exception of AC9.

1.1.2 Cellular Targets of cAMP

The range of action of cAMP is extremely wide with effects reaching cell differentiation, ion channel conductivity and gene transcription amongst others. Targets of cAMP action are cyclic nucleotide gated channels (Kraus-Friedmann 2000) and the Rap-GEFs Epac1 and Epac 2 (exchange protein directly activated by cAMP). The first identified target of cAMP within cells was the cAMP dependent protein kinase A, PKA.

1.1.2.1 Protein Kinase A

The classically known target of cAMP within cells is the cAMP dependent kinase, PKA. PKA is a heterodimeric enzyme consisting of two regulatory (R) and two catalytic (C) subunits. cAMP binds to the R subunit of PKA inducing conformational changes that dissociate the enzyme into an R subunit dimer with four molecules of cAMP bound and

two, free, active C subunits able to phosphorylate target serine/threonine residues (reviewed in Skalhegg & Tasken 2000, PKA targets reviewed in Shabb 2001).

There are two types of PKA which differ in their regulatory subunits, RI or RII, which interact with an identical C subunit. As with AC, PKA exists in several isoforms. Four isoforms of the R subunit have been identified; RIa, RIB, RIIa and RIIB, with several splice variants of RIa reported. The RI isoforms share high homology with each other, are ubiquitously expressed and are activated transiently by low cAMP levels. The RII isoforms also share a high homology with each other, are predominantly expressed in brain, fat, endocrine and reproductive tissues and require persistent levels of high cAMP stimulation for activation (Feliciello et al., 2001). The C subunit also exists in multiple forms; Ca, C β and Cy, with splice variants of both C α and C β reported (Uhler et al., 1986; Showers et al., 1986). A large number of PKA holoenzymes therefore exist with different biochemical properties (Skalhegg & Tasken 2000). Indeed it is known that the PKAI complex ($RI\alpha_2C_2$ or $RI\beta_2C_2$) is generally cytoplasmic whereas the PKAII complex is usually membraneassociated (Cho-Chung et al., 1995, Houslay & Milligan 1997, Skalhegg & Tasken 2000; Murphy & Scott 1998). PKAII is associated to membranes through tethering of the R subunit to A-kinase anchoring proteins (AKAPs) which are found throughout the cell tethered to various subcellular components, thus providing localisation of the PKA activity. PKAII is also known to be subject to autophosphorylation by the C subunit (Skalhegg & Tasken 2000, Shabb 2001; reviewed in Michel & Scott 2002).

1.1.2.2 A-kinase anchoring proteins

PKAII exists associated with a variety of subcellular structures including the plasma membrane, endoplasmic reticulum, microtubules, Golgi and nucleus (summarised in Murphy & Scott 1998, Feliciello et al., 2001). This association exists through an interaction of PKARII with an anchoring protein tethered to the membrane. These AKAPs bind an N-terminal region of the RII protein, at its dimerisation (RII: RII) interface, and contain a unique targeting domain responsible for localising the AKAP/PKAII to a specific subcellular structure (Murphy & Scott 1998, Michel & Scott 2002). A new group of AKAPs has also been identified that can bind both RII and RI (Huang et al., 1997). This anchoring of PKA provides tight regulation of cAMP levels in specific cellular locations. In addition, some AKAPs also act as scaffolding proteins by associating with other kinases and phosphatases. This is true in the case of AKAP79 which scaffolds PKA, PKC and the calcium/calmodulin dependant phosphatase PP2B to regions near the plasma membrane (Dodge & Scott 2000).

1.1.2.3 Exchange Protein Directly Activated by cAMP

PKA-independent actions of cAMP involve an exchange protein that is directly activated by cAMP, Epac (de Rooij et al., 1998). Epac1 and Epac2 are guanine exchange factors for the small Ras like proteins, Rap1 and Rap2 (de Rooij et al., 2000). A third Epac family member, Repac, lacks the regulatory sequences of Epac 1 and 2 but is still able to activate Rap1 and 2 (de Rooij et al., 2000). Epac contains a cAMP binding domain (Epac 2 has two cAMP binding domains (Rehrmann et al., 2002)), a Dishevelled, Egl-10, Pleckstrin (DEP) domain shown to be responsible for membrane targeting of Epac (de Rooij et al., 2000), a Ras-exchanger motif, a Ras-association domain and a guanine nucleotide exchange factor (GEF) domain (Bos 2003). Epac 1 is ubiquitously expressed and displays a cell-cycle dependant localisation. At interphase, Epac1 has been shown to be located at both the mitochondria and nuclear membrane and translocates to the mitotic spindle and centrosomes during metaphase (Qiao et al., 2002). This translocation is not mediated by cAMP as Epac is membrane bound in the presence and absence of cAMP. Epac 2 is predominantly expressed in the brain and adrenal gland.

The catalytic domain of Epac is regulated by a direct interaction between the cAMP binding domain and the GEF domain which acts in an inhibitory fashion. Upon binding cAMP, this inhibition is released by a conformational change in the protein which exposes the GEF domain and allows Epac to activate downstream pathways (de Rooij et al., 2000, Rehrmann et al., 2003).

It has been demonstrated that cAMP activation of Rap1 is through Epac, whereas the cAMP induced inhibition of Raf1 is mediated by PKA (Enserink et al., 2002). Raf1 activates MEK which in turn activates ERK. cAMP has been demonstrated to have both stimulatory and inhibitory effects on the ERK pathway, but it is thought this occurs through PKA activation of Raf1 through Ras and the direct inhibitory effect of PKA on Raf1. As such, the roles of Rap1 only in cell signalling have been attributed to Epac. Epac activation of Rap1 plays a major role in cell adhesion by regulating integrin signalling (Rangarajan et al., 2003). Epac2 has also been reported to be involved in insulin secretion in β -cells (Holz 2004).

1.2 The Phosphodiesterase Superfamily

Hydrolysis of cyclic nucleotides is the only method of inactivation in cells. Cyclic nucleotide hydrolysis is mediated by a large superfamily of phosphodiesterase enzymes

(PDEs). PDEs catalyse the hydrolysis of cAMP and cGMP at their 3'-phosphoester bond to form the corresponding inactive 5' nucleoside monophosphate (5' AMP and 5' GMP). There are currently cleven known families of PDEs with differential tissue distribution, regulatory properties and sensitivity to inhibitors (summarised in table 1.3; reviewed in Francis et al., 1999; Dousa 1999; Soderling & Beavo 2000; Yuasa et al., 2000; Conti 2000; Mehats et al., 2002; Maurice et al., 2003). Each PDE family is encoded for by one to four genes, with multiple protein products arising from alternative splicing and multiple promoters resulting in over 50 mammalian PDE isoforms. Each isoform is expressed in a tissue and cell-specific manner.

All PDEs share the same modular structure, with three functional domains; a conserved catalytic domain of ~270 amino acids near the C-terminus, regulatory regions located at the N-terminus and the C-terminus (*Figure 1.1*). The catalytic domain contains sequences essential for cyclic nucleotide hydrolysis including the common PDE motif; $HD(X_2) H(X_4)$ N which contains consensus metal binding domains (Zn²⁺ and Mg²⁺). The N-terminu of PDEs are highly divergent and are conserved within a family, showing the N-terminus contains each family's specific functional motifs. Regions in this domain include regulatory sequences such as phosphorylation domains and targeting regions. It has also been proposed that the N-terminal region of a PDE contains regions for dimerisation.

PDE inhibitors have been reported as being clinically useful in many disorders (table 1.4 for summary of diseases PDEs are involved in; Torphy & Page 2000). These include disease states such as depression, asthma, pulmonary arterial hypertension (PAII) and erectile dysfunction. Family specific PDE inhibitors have shown promising results in clinical trials and as such, many are attractive targets for drug development (Schmidt et al., 1999; Torphy 1998; Burnouf & Pruniax 2002; Spina 2003; Michelakis et al., 2003; Eddahibi et al., 1998).

1.2.1 The PDE1 Family

The PDE1 family is encoded for by three genes; *PDE1A*, *PDE1B* and *PDE1C* with splice variants of each, and is known to hydrolyse both cAMP and cGMP (reviewed in Sonnenburg et al., 1998; Kakkar et al., 1999). PDE1A and 1B isoforms have a higher affinity for cGMP than cAMP and PDE1C hydrolyses both with equal efficiency. PDE1 is referred to as the calcium/calmodulin (Ca²⁺/CaM) PDE due to the presence of Ca²⁺/CaM binding sites within the PDE1 N-terminal region. Activity of the PDE1 family can be regulated by both Ca²⁺/CaM binding and phosphorylation/dephosphorylation by PKA and

CaM-dependant kinase II. Activation of PDE1 enzymes occurs upon Ca²⁺/CaM binding and thus reduces cAMP. As PDE1A is inhibited by PKA phosphorylation, PDE1 provides a "cross-talk" between calcium and cAMP. PDE1B is inhibited by CaM-dependant kinase II phosphorylation. Both types of phosphorylation reduce the affinity of PDE1 for CaM. It has also been reported that protein kinase C can induce PDE1B. PDE1C has been demonstrated to be essential for human smooth muscle cell (SMC) proliferation (Rybalkin et al., 2002). PDE1C is absent from quiescent arterial SMC and human aorta, but is induced in proliferating arterial SMC. Inhibition of PDE1 by vinpocetine reduces SMC proliferation (Rybalkin et al., 2002).

1.2.2 The PDE2 Family

The PDE2 family are known as the cGMP-stimulated PDEs and hydrolyse both cAMP and cGMP with similar V_{max} , although the affinity for cGMP is two-fold greater. There are three PDE2 isoforms arising from one gene, each containing two specific, noncatalytic cGMP binding domains in their N-terminus. Binding of either cyclic nucleotide induces a conformational change in the PDE2 enzyme, converting it to a more active form. As such, PDE2 provides a means of crosstalk between the two cyclic nucleotide signalling pathways.

PDE2 is predominantly expressed in the adrenal cortex (PDE2A1) and brain and cardiac tissue (PDE2A2). PDE2A3 contains a putative membrane targeting domain. PDE2 activity has been shown to have a role in Ca^{2+} channel regulation, olfactory signalling pathways and aldosterone secretion. Low levels of PDE2 have been detected in vascular smooth muscle cells, and it has been reported that PDE2 activity is involved in hypoxic pulmonary vasoconstriction yet a distinct function has not yet been identified (Haynes et al., 1996). The commonly used inhibitor of PDE2 is *erythro*-9-(2-hydroxyl-3-nonyl) adenine, EHNA (de Jong et al., 1992).

1.2.3 The PDE3 Family

The PDE3 family also hydrolyses both cAMP and cGMP with high affinity (K_m values range from 0.1 to 0.8µM) but with a 2-10 fold higher V_{max} for cAMP (reviewed in Degerman et al., 1997). cGMP in micromolar concentrations binds to the catalytic site of PDE3 and competes with cAMP hydrolysis, hence the PDE3 family being dubbed the cGMP-inhibited PDEs. As the V_{max} for cGMP is so low, cGMP hydrolysis is negligible and PDE3 behaves as a cAMP-specific PDE. PDE3A and PDE3B are also activated by PKA or PKB phosphorylation.

PDE3 is encoded for by two genes; *PDE3A* and *PDE3B* with a higher similarity of PDE3A between human and rat than between human PDE3A and B. This is also true for PDE3B. Alternative start codons give rise to three PDE3A isoforms. PDE3 is widely expressed. In tissues expressing both PDE3A and PDE3B, PDE3A levels are usually dominant. PDE3A is abundant in the cardiovascular system, including vascular smooth muscle, and PDE3B is found predominantly in adipose tissue, hepatocytes and the renal collecting duct epithelium. PDE3 is found in both cytosolic and particulate fractions of cells (Liu & Maurice et al., 1998; Palmer & Maurice 2000). PDE3 has been implicated in a variety of roles including myocardial contractility, smooth muscle relaxation and proliferation of T-lymphocytes and vascular smooth muscle cells. In addition, PDE3B plays a key role in insulin signalling in adipocytes. In response to insulin, PDE3B is phosphorylated and activated by Akt (Kasuga et al., 1999).

In vascular smooth muscle cells, both PDE3A and PDE3B are expressed. Sustained exposure to cAMP elevating agents in vivo increased both PDE3A2 and PDE3B in rat aortic quiescent SMC (Tilley & Maurice 2002). In cultured rat and human aortic SMC, prolonged cAMP elevation increases only PDE3B (Palmer & Maurice 2000). The reasons behind this difference are unclear. PDE3 inhibitors such as cilostamide or milrinone have also been demonstrated to relax arterial tissues and inhibit vascular SMC proliferation (Lindgren et al., 1989; Inoue et al., 2000).

1.2.4 The PDE4 Family

The PDE4 family of enzymes are the most diverse and extensively studied of all PDE families (reviewed in Muller et al., 1996; Salanova et al., 1998; Houslay 2001; Houslay & Adams 2003; for further details, see section 1.3). Four genes encode for PDE4; *PDE4A*, *PDE4B*, *PDE4C* and *PDE4D* which are located on three chromosomes. Alternative mRNA splicing and multiple promoters have led to the discovery of over 20 different PDE4 mammalian isozymes. PDE4 isoforms share 80% sequence identify of their catalytic domain with unique regions shared among PDE4 isoforms of a subfamily. Splice variants differ in their N-terminal region which confers regulatory properties and capability for interactions with other cell structures. PDE4 isoforms are characterised by the presence of two conserved domains within their N-terminus known as Upstream Conserved Region 1 (UCR1) and Upstream Conserved Region 2 (UCR2). PDE4 isoforms can exist as a "long form" with both UCR 1 and UCR2 intact, a "short form" with only UCR2, and "super short forms" with a truncated UCR2 only (*see figure 1.2*). PDE4 is expressed in almost all

cells, with the exception of blood platelets, and displays differential subcellular localisation between isoforms.

Regulation of PDE4 activity can occur through phosphorylation by PKA (Sette & Conti 1996; Ekholm et al., 1997; Hoffmann et al., 1998; Liu & Maurice 1999; MacKenzic et al., 2002) where PKA phosphorylation at Ser⁵⁴ of PDE4D3 and the equivalent residue in the UCR1 of other long forms induces activation of these enzymes. This provides a feedback mechanism for regulating the levels of cAMP.

The p42 MAP kinase, ERK2 also phosphorylates PDE4 isoforms resulting in activation of short forms and inhibition of long forms (Hoffmann et al., 1999; MacKenzie et al., 2000). This is with exception of PDE4A members, which do not act as a substrate for ERK2 (Baillie et al., 2000).

PDE4 is a cAMP-specific enzyme, with a K_m of 1-3µM for cAMP. It is insensitive to cGMP and Ca²⁺/CaM and is specifically inhibited by Rolipram and Ro 20-1724. PDE4 inhibitors are of great interest due to the role PDE4 activity has been shown to play in disease states including asthma, chronic obstructive pulmonary disease (COPD) and depression (reviewed in Spina et al., 1998, Spina 2003 and Burnouf & Pruniaux 2002).

1.2.5 The PDE5 Family

PDE5 enzymes are known as the cGMP binding, cGMP specific PDEs and are characterised by the presence of two cGMP binding domains located N-terminally to the catalytic unit. PDE5 is encoded for by a single gene that gives rise to four splice variants; PDE5A1, PDE5A2, PDE5A3 and PDE5A4, which have differential tissue distribution but share identical catalytic domains. PDE5 is expressed in several tissues including vascular smooth muscle cells, heart and lung (Loughney et al., 1998; Giordano 2001).

Binding of cGMP to both PDE5 cGMP binding domains does not affect PDE5 activity, but does induce a conformational change that allows phosphorylation by PKG, the cGMP dependant kinase, and PKA at Ser^{92} (Thomas et al., 1990; Corbin & Frances 1999). Phosphorylation leads to an increase in PDE5 activity. PDE5 is also activated by Zn^{2+} ions which are essential for catalysis. Specific inhibitors for PDE5 include DMPPO; a compound currently under investigation in models of hypoxia induced pulmonary hypertension due to its vasodilatory properties (Eddahibi et al., 1998); and Sildenafil, a treatment for male erectile dysfunction that has also been shown to reduce the hypoxic

increase in pulmonary arterial pressure (PAP) in models of hypoxia induced pulmonary hypertension (Zhao et al., 2001; Michelakis et al., 2003). PDE5 levels and activity have been reported to increase in hypoxia induced models of pulmonary hypertension and the selective dilation of the pulmonary vasculature by Sildenafil provides a new target for treatment of pulmonary hypertension (Murray et al., 2002).

1.2.6 The PDE6 Family

The PDE6 family are expressed in retinal rods and cones and play a key role in the visual transduction cascade. Due to this, PDE6 are more often referred to as the photoreceptor PDE family. PDE6 is highly specific for cGMP and is the only known family of PDE to be regulated by G-proteins. They exist as hetero- or homotetramers comprised of catalytic subunits (α , α' , β) and inhibitory subunits (γ and δ). The two types of photoreceptor cells, rod and cone, express different forms of PDE γ . The visual cascade is initiated by photons that interact with rhodopsin, resulting in the activation of the G-protein transducin (T $\alpha\beta\gamma$). Active transducin (T α -GTP) dissociates from T $\beta\gamma$ and binds to the inhibitory γ subunits of membrane bound PDE6. This relieves inhibition of PDE6 and decreases cGMP, closing the cGMP-gated channel and resulting in hyperpolarisation of the membrane.

Rod PDE γ is also expressed in lung, kidney, heart, pulmonary smooth muscle and airways. In these systems, Rod PDE γ is phosphorylated by the G-protein receptor kinase (GRK) 2 which regulates the epidermal growth factor (EGF) stimulation of ERK1/2 in human embryonic kidney 293 cells (HEK 293; Wan et al., 2001) This regulation also involves the Src kinase which exists in a complex with PDE γ and upon stimulation with EGF, GRK2 associates with this complex. Src inhibitors block activation of ERK1/2 by EGF as does mutation of the Thr⁶² residue of PDE γ (Wan et al., 2003).

1.2.7 The PDE7 Family

The PDE7 family are high affinity cAMP PDEs, similar to that of PDE4, but are rolipram insensitive (Michaeli et al., 1993). The PDE7 family are comprised of two genes and three individual PDE7 isoforms have been identified, PDE7A1, PDE7A2 and PDE7B1. PDE7A1 is expressed in the immune system, PDE7A2 is highly expressed in skeletal muscle and heart and PDE7B is abundant in the pancreas, brain, heart and skeletal nuscle (Iletman et al., 1999). PDE7A was initially suggested to play a role in T-cell activation and proliferation (Li et al., 1999), however studies using PDE7 knockout mice have disproved this (Yang et al., 2003). Recently, it has been discovered that PDE7 activity can be inhibited by spiroquinazolinones (Lorthiois et al., 2004; Bernardelli et al., 2004).

1.2.8 The PDE8 Family

The PDE8 family are cAMP specific with an affinity for cAMP to rival that of PDE4 and were discovered screening a database of ESTs (expressed sequence tags) to identify PDE ESTs (Soderling et al., 1998; Soderling & Beavo 2000; Hayashi et al., 2002). Two variants of PDE8 exist, PDE8A and PDE8B. PDE8A is highly expressed in testis, eye, liver, kidney, skeletal muscle, embryo, ovary and brain (Soderling et al., 1998). PDE8B is expressed highest in the human thyroid gland, brain and kidney.

PDE8 is unique in that it contains a single PAS domain (Per, Arnt, and Sim proteins) at the N-terminus which regulates protein:protein interactions. Specific functions for the PDE8 isoforms remain to be discovered. PDE8 is, unusually, insensitive to inhibition by the so-called general PDE inhibitor, 3-Tsobutyl-1-methylxanthine, (IBMX).

1.2.9 The PDE9 Family

PDE9 is a cGMP specific PDE expressed in many tissues including spleen, small intestine, kidney, heart and brain tissue (Fisher et al., 1998). Unlike other cGMP PDEs, PDE9 lacks a non catalytic cGMP binding domain. Four variants of PDE9 have been identified but their function is as yet unclear. The only compounds known to date to inhibit PDE9 are the PDE5 and PDE1 inhibiting compounds, zaprinast and SCH518866 (Dousa, 1999).

1.2.10 The PDE10 Family

PDE10 is a dual substrate family with K_m of 0.05µM for cAMP and 3µM for cGMP. The V_{max} for cGMP hydrolysis is approximately five times greater than that for cAMP. This suggests cGMP hydrolysis by PDE10 can be inhibited by cAMP, therefore PDE10 has been referred to as the cAMP-inhibited, cGMP specific PDE (Soderling et al., 1999; Fujishige et al., 1999). Two splice variants of PDE10 have been identified with PDE10A1 and PDE10A2 containing two N-terminally located domains that are similar to the cGMP binding domains of PDE2, PDE5 and PDE6. Unlike the other cGMP specific PDEs however, the cGMP binding domains in PDE10 are not thought to primarily function as such. PDE10 is sensitive to inhibition by IBMX (Soderling & Beavo 2000).

1.2.11The PDE11 Family

The PDE11 family also catalyse the hydrolysis of both cAMP and cGMP and contain a GAF (so named for cGMP binding and stimulated phosphodiesterase, *Anabaena* adenylyl cyclases, and *Escherichia coli* Fh1A) domain at their N-terminus, relating them to the other GAF containing PDEs, PDE2, 5, 6 and 10. There have been four isoforms of PDE11 identified to date (Yuasa et al., 2000) with each containing variations of the GAF domain.

PDE11A1, PDE11A2 and PDE11A3 contain one intact GAF domain and one incomplete GAF domain. PDE11A4 contains two complete GAF domains (Yuasa et al., 2001). PDE11 isoforms also display differential tissue expression with PDE11A3 being specifically expressed in testis and PDE11A4 are predominantly expressed in the prostate. PDE11A1 is abundant in skeletal muscle (Yuasa et al., 2001). No physical functions have been related to any of the PDE11 isoforms as yet.

1.3 Organisation and Regulation of the PDE 4 Family

The Drosophila melanogaster dunce gene was the first gene to be discovered that specifically affected behaviour. Mutations in the *dunce* gene led to defects in learning and memory (Davis & Dauwalder 1991). The mammalian PDE4 family are homologues of the *dunce* gene. Cloning of the *dunce* PDE led to discoveries surrounding the regulation and kinetics of cAMP PDEs and the subsequent cloning of mammalian PDE4 genes. Four mammalian PDE4 genes were discovered located on three chromosomes with a high conservation between species. These four genes each give rise to multiple splice variants with unique N-terminals that are all characterised by their ability to be specifically inhibited by rolipram.

1.3.1 Modular structure of PDE4

The PDE4 family are further characterised by the presence of upstream conserved regions known as UCR1 and UCR2 at their N-terminus. The region that links UCR1 and UCR2 together is called linker region 1 (LR1), and linker region 2 (LR2) connects UCR2 to the catalytic domain (*see figure 1.2*). In contrast to UCR1 and UCR2, the linker regions differ between the different PDE4 families, perhaps conferring isoform-specific functional properties. The alternative mRNA splicing of members of the PDE4 family occurs at two major splice sites, one at the extreme N-termini and the other at the beginning of UCR2. This means both "long" isoforms containing both UCR1 and UCR2 and "short" isoforms with only UCR2 are produced. An additional splice site also generates "super-short" isoforms which have an N-terminally truncated UCR2. UCR1 and UCR2 have no homology to one another, yet are conserved in PDE4 homologs, suggesting they are of functional importance.

Studies have revealed that the C-terminal of UCR1 interacts specifically with the N-terminal of UCR2 (Beard et al., 2000; Lim et al., 1999; MacKenzic et al., 2000). PKA phosphorylation is known to affect this interaction, but some investigators suggest it

strengthens the interaction (Lim et al., 1999), whilst others have shown that it can weaken it (Beard et al. 2000). UCR2 has also been shown to act as a regulatory domain on the catalytic region (Beard et al., 2000; Lim et al., 1999). N-terminally truncated PDE4, with the N-terminus of UCR2 removed, causes an increase in PDE4 activity indicating the role of the UCR2 as an inhibitory domain on the catalytic unit of PDE4. This has been demonstrated to be the case with the PDE4D and PDE4A5 isoforms (Lim et al., 1999; Beard 2002). UCR2 has also been implicated in the targeting of PDE4 isoforms.

1.3.2 Regulation of PDE4 activity

The PDE4 family is regulated through the actions of protein kinases. Long forms of PDE4 contain two PKA phosphorylation sites. In PDE4D3, these are Ser¹³ and Ser⁵⁴. Ser¹³ is located at the unique N-terminal of the isoform and Ser⁵⁴ is located within UCR1. PKA phosphorylation elicits activation of the PDE4 enzyme (Alvarez et al., 1995; Sette & Conti 1997; Hoffmann et al., 1998; Beard et al., 2000; Oki et al., 2000). Only the Ser⁵⁴ residue has been shown to be essential for activation by PKA phosphorylation in the PDE4D3 isoform by mutation of Ser⁵⁴ to Ala⁵⁴ (Sette & Conti 1996). Replacing Scr⁵⁴ with Asp⁵⁴ or Glu⁵⁴ can mimic enzyme activation (Hoffmann et al., 1998). The PKA consensus sequence, RRESF, is also found within the UCR1 region of other PDE4 long forms, suggesting all can be activated by PKA phosphorylation. Other work has shown that this is the case for PDE4A8, PDE4B1, PDE4C2 and PDE4D5 as well as PDE4D3 (MacKenzie et al., 2002). PKA phosphorylation induces a conformational change in the PDE4 enzyme which increases the affinity of UCR1 for UCR2. This relieves the auto-inhibitory actions of UCR2 on the catalytic domain and allows PDE4 to bind cAMP (Lim et al., 1999). Although PKA does phosphorylate Ser¹³ in PDE4D3 unlike other PDE4 long forms, the functional consequences of this are still unclear (MacKenzie et al., 2002).

The extracellular kinase (p42), ERK2 is activated by dual phosphorylation (Prowse & J.ew 2001) and phosphorylates the long form PDE4D3 at Ser⁵⁷⁹. This leads to a reversible 75% reduction in PDE4D3 enzyme activity (Hoffmann et al., 1999). More recent work has revealed that although isoforms encoded by the PDE4A, 4B, 4C and 4D genes all contain the ERK2 docking sites, KIM and FQF, (MacKenzie et al., 2000), only PDE4B, 4C and 4D act as substrates for ERK2 (Baillie et al. 2000). ERK2 phosphorylation of PDE4 long forms is transient, as it causes a localised increase in cAMP and thus activation of PKA. PKA phosphorylation can negate the ERK2 inhibition of PDE4 long forms, (Baillie et al., 2000; Hoffmann et al., 1999) introducing feedback regulation. Studies have revealed that whereas ERK2 phosphorylation of PDE4 long forms causes inhibition of enzyme

activation, ERK2 phosphorylation of PDE4 short forms surprisingly induces activation (Baillie et al., 2000). Thus there are opposing effects on ERK by cAMP. Inhibition of ERK is achieved through the Raf-1 isoform whereas ERK activation by cAMP is achieved through the B-Raf isoform. The cAMP signalling pathway is subject to many points of "cross-talk" between itself and the ERK pathway (Houslay & Kolch 2000). In human acrtic smooth muscle cells, ERK stimulation feeds into an autocrine loop that results in activation of a PDE4 long form (Baillie et al., 2001, figure 1.3). Activated ERK caused an increase in PGE₂ which was released from the cell and activated adenylyl cyclase. This induced a rise in cAMP levels with a concomitant increase in PKA activity and resultant PDE4D5 phosphorylation. This phosphorylation ablated the effect of ERK phosphorylation and a net activation of PDE4D5 was observed.

1.3.3 The PDE4A subfamily

The PDE4A gene is located at chr19p13.1 and gives rise to seven different isoforms (Houslay et al., 1995; reviewed in Houslay 2001). It was originally thought that the inactivation of cyclic nucleotides was the only function of phosphodiesterases, but research suggests PDEs also play a role in signal compartmentalisation. The 'super-short' PDE4A isoform, PDE4A1, (also known as RD1), is exclusively membrane-associated, a feature unique among the PDE4 isoforms. It was initially revealed that the first twenty-five residues of the N-terminus of PDE4A1 are essential in membrane targeting (Pooley et al., 1997; Houslay et al., 1995; Houslay et al., 1998). N-terminally truncated isoforms were found solely in the cytosol. Further work revealed that PDE4A1 was specifically targeted towards the Golgi complex in FTC cell lines (Pooley et al., 1997). Displacing PDF4A1 from the membrane requires non-ionic detergents indicating that the binding involved hydrophobic interactions. Investigation by Baillie et al., 2002 revealed that PDE4A1 is inserted into the membrane in a manner due solely to an 11-residue helical module within the N-terminus that also confers selectivity for interaction with phosphatidic acid (PA). Interaction with PA has previously been shown to alter PDE4 activity. The module identified has been named the tryptophan anchoring phosphatidic acid selective binding domain 1, (TAPAS-1). Insertion is Ca^{2+} dependent and irreversible.

The human PDE4A4B, initially known as pde46, is a homologue of the rat PDE4A5 and is a 'long form' PDE. PDE4A4B selectively binds SH3 domains e.g. of Src family kinases and this binding alters PDE4A4B sensitivity to rolipram through a conformational change in the catalytic unit (McPhee et al., 1999; MacKenzie & Houslay 2000). The SH3 binding domains located in the N-terminus and UCR2 region also act to target the PDE4A4B/4A5

isoform to the plasma membrane where it localises in membrane ruffles (Beard et al., 2002). Perinuclear localisation of PDE4A4B/4A5 has been attributed to the N-terminal of UCR2. During apoptosis in Rat-1 fibroblasts and PC12 cells, the SH3 domain is cleaved by caspase 3. PDE4A4B/4A5 loses its targeting ability and is found uniformly throughout the cell (Huston et al., 2000). Binding of rolipram to PDE4A4/4A5 also affects intracellular distribution, inducing foci formation (Terry et al., 2003). Another 'long form' PDE4A, rat PDE4A8 (also known as rpde39), does not have a human homologue and has been shown to be exclusively expressed in the testes and hepatocytes (Bolger et al., 1996). PDE4A7 is a catalytically inactive isoform which is N- and C-terminally truncated (Johnston et al., 2004). The inactivity of 4A7 is not through inhibitory actions of the N- or C-terminal domains as removal of either of these does not allow PDE4A7 to hydrolyse cAMP. The PDE4A10 isoform was originally cloned from a rat olfactory lobe cDNA library (Rena et al., 2001). PDE4A10 is another PDE4A 'long form' which migrates at approximately the same size as PDE4A4 and the newly characterised, PDE4A11 (Tm3) on SDS-PAGE (~121kDa) PDE4A10 is conserved between human, mouse and rat and displays a predominantly perinuclear localisation in COS7 cells (Rena et al., 2001).

1.3.4 The PDE4B subfamily

The human PDE4B gene is located at chr1 and it provides four PDE4B isoforms. PDE4B2 is a short form PDE, subject to activation by ERK2, and PDE4B1, 4B3 and 4B4 are all long forms (reviewed in Houslay 2001). All PDE4Bs are found in both particulate and cytosolic fractions of cells. It is well established that PDE4 is the predominant PDE in leukocytes and has been attributed with a role in inflammation as it is involved in the activation and proliferation of inflammatory cells (Banner et al., 1999; Banner et al., 2000; Ogawa et al., 2000). In addition, PDE4 inhibitors significantly reduce TNF- α and IL- β release from LPS-stimulated macrophages (Kambayashi et al., 1995, Timmer et al., 2002). PDE4B has been singled out in inflammation as it is the predominant PDE4 in neutrophils and monocytes. In particular, the PDE4B2 isoform is the sole PDE4B variant expressed in these cells and is induced by lipopolysaccharide (LPS) in a manner inhibited by interleukin (IL)-10 and IL-14 (Wang et al., 1999).

Induction of PDE4B2 has also been observed in cells exposed to IL-1 β (Oger et al., 2002) or a combination of IL-1 β and TNF- α (Hakonarson et al., 2001). The increase in PDE4B2 has been reported to be due to the raised cAMP levels in response to IL-1 β treatment. Exposing myometrial cells to IL-1 β raised PGE₂ prior to induction of PDE4B2 and this induced an increase in PDE4 activity. IL-1 β caused an initial increase in cAMP levels,
peaking at twelve hours, before PDE4 activity increased to a peak at eighteen hours and cAMP levels were reduced. The non-specific inhibitor of PGE₂ synthesis, indomethacin, was observed to completely block the rise in cAMP and the increased PDE4 activity (Oger et al., 2002). PDE4B2 induction in myometrial cells has also been observed directly in response to cAMP raising agents (Mchats et al., 1999) or naturally in near pregnant and pregnant myometrial strips as PDE4 inhibitors can block spontaneous myometrial contraction (Mehats et al., 2002).

1.3.5 The PDE4C subfamily

The PDE4C gene is located at chr19p13.2 and is reported to give rise to three long forms of PDE4C (Obernolte et al., 1997). There is little research on the PDE4C subfamily, although it is known that the PDE4C2 isoform is subject to regulation by ERK2 phosphorylation. Upon ERK2 phosphorylation at Ser⁵³⁵, activity of PDE4C is inhibited (Baillie et al., 2000).

1.3.6 The PDE4D subfamily

The PDE4D gene is located at chromosome 5 and through two major and one minor splice points (*see figure 1.2*), nine splice variants have been reported so far (Bolger et al., 1997; Wang et al., 2003; Gretarsdottir et al., 2003). Only one of these is a short form, PDE4D1 with a weight of 68 kDa. The super-short form PDE4D2 has an N-terminal truncation and is the same weight as PDE4D1. PDE4D6 is also another super short form with a predicted weight of 59kDa. PDE4D3, PDE4D4, PD4D5, PDE4D7, PDE4D8 and PDE4D9 are all long form PDE4Ds with weights of 95, 119, 105, 103, 96 and 95kDa respectively (weights of PDE7, 8 and 9 are predicted). The similarity in sizes on SDS-PAGE of PDE4D1 with PDE4D2; PDE4D3 with PDE4D8, 9; PDE4D5 with PDE4D7 and indeed PDE4A4 with PDE4A10 and PDE4A11 highlights the necessity of correct identification of PDE4 isoforms by means other than SDS-PAGE.

The function of PDE4D has been investigated in knock-out mice models. Studies on $PDE4D^{-4}$ mice have shown that PDE4D plays a major role in growth and contraction. These transgenic mice display a lower than normal weight due to a reduction in muscle and bone mass and decreased weight of internal organs (Jin et al., 1999). The airways of mice deficient in PDE4D are also refractory to muscarinic cholinergic stimulation (Hansen et al., 2000). Further studies revealed this was due to a five-fold increase in sensitivity to prostanoid stimulation and an enhanced cAMP synthesis. The tracheal response to the contractile agonist carbachol was reduced in transgenic mice and could be mimicked by

17

a , ,

treating wild-type mice with rolipram. Inactivation of PDE4B had no effect on tracheal contraction. The PDE4D^{-/-} phenotype could be reversed by indomethacin (Mehats et al., 2003).

1.3.6.1 Interactions and Targeting of the PDE4D subfamily

In COS7 cells, PDE4D1 and 4D2 were found solely in the cytosol while PDE4D3, 4D4 and 4D5 were located in both the cytosolic and particulate fractions (Bolger et al. 1997). This confirms the involvement of the unique N-terminal in targeting as both 4D1 and 4D2 are short forms and therefore do not possess it. Translocation of the 4D3 isoform from the particulate fraction of cells to the soluble fraction has been seen in vascular smooth muscle cells upon stimulation of both the ERK and PKA pathways (Liu & Maurice 1999).

Temporal compartmentalisation of the PDE signal has been shown in FRTL-5 cells (Jin et al. 1998). Upon stimulation with thyroid stimulating hormone (TSH), PDE4D long forms were activated at membranous structures almost immediately while 4D2 was activated in the cytosol after a period of time. PDE4D long forms are targeted towards the Golgi/centrosome and filamentous structures within these cells.

In addition to PKA and ERK phosphorylation, PDE4D3 is also subject to regulation by PA. Accumulation of endogenous PA, through inhibiting the actions of PA degraders, induced an increase in PDE4 activity with a resultant decrease in cAMP and PKA activity (Nemoz et al., 1997). In FRTL5 cells, PDE4D3 was demonstrated to specifically bind PA in a unique binding site in the N-terminal region which contains the PKA phosphorylation site (Grange et al., 2000).

As with PDE4A4B/4A5, PDE4D4 has also been shown to contain an SH3 domain that predisposes PDE4D4 to bind selective SH3 domain proteins (Beard et al., 1999). Binding of Src or Lyn SH3 domains are prevalent, with no detectable effect on the catalytic activity of PDE4D. It is postulated that this interaction is involved in targeting of the PDE4D isoform.

PDE4D5 specifically has been demonstrated to interact with the receptor for activated kinase (RACK1). This interaction does not affect activity of the enzyme, but does affect sensitivity to rolipram (Yarwood et al., 1999). No physiological role for this interaction has yet been uncovered.

It has been recently shown that all PDE4 isoforms can interact with the scaffolding proteins, β -Arrestin1/2. Indeed, it has been demonstrated that PDE4D3 and PDE4D5 are targeted to activated GPCR along with β -Arrestin, aiding the localised degradation of cAMP (Perry et al., 2002) and receptor desensitisation. The interaction of PDE4 and β -Arrestin is also involved in the "switching" of GPCR from the G_s subunit to the G_i subunit of the G-protein. This occurs upon PKA phosphorylation of the receptor and leads to ERK activation within the cell. By using a catalytically inactive mutant of PDE4D5 which displaces active PDE4D5 from β -Arrestin, it was demonstrated that with inactive PDE4D, β_2 -adrenoceptor can induce higher levels of PKA and increase the levels of receptor activated ERK (Baillie et al., 2003). PDE4 thus plays a key role in β_2 -adrenoceptor regulation.

The scaffold protein myomegalin was identified as an interacting protein for PDE4D through yeast two hybrid analyses using PDE4D3 as bait (Verde et al., 2001). This interaction was demonstrated to localise PDE4D3 to Golgi/centrosomal areas and occurs through the N-terminus of UCR2 in PDE4D (Verde et al., 2001). As UCR2 is conserved in PDE4 long forms, it is possible that other PDE4 isoforms also interact with myomegalin. The UCR2 of PDE4D3 has also been shown to interact with AKAP450, targeting PDE4D3 to the centrosomal region in Sertoli cells (Tasken et al., 2001). UCR2 is not the only putative targeting mechanism of PDE4. As each isoform differs only at their N-terminus and LR, it was proposed that these regions conferred isoform-specific targeting. Indeed, it has been demonstrated that the muscle-selective AKAP, mAKAP, interacts with the unique N-terminus of PDE4D3 and not PDE4D5, targeting it to perinuclear regions in cardiac cells (Dodge et al., 2001).

1.3.6.2 Induction of PDE4D by cAMP

Intronic promoters have been identified controlling the expression of the PDE4D1 and PDE4D2 isoforms. Within these promoters, cAMP response elements (CRE) have been identified that allow cAMP response element binding protein (CREB; reviewed in Montminy 1997; Shaywitz & Greenberg 1999; Mayr & Montminy 2001) to bind and induce expression. This was demonstrated to be the case in Sertoli cells, where raising cAMP levels upregulated the expression of PDE4D1 and PDE4D2 (Vicini & Conti 1997). A CRE site has also been uncovered in the PDE4D5 promoter and cAMP-dependant induction of PDE4D5 has been observed in human airway smooth muscle cells (Le Jeune et al., 2002).

1.4 PDE4 Inhibitors and Disease

It has long been known than an increase in cAMP levels can suppress the immune response and induce smooth muscle cell relaxation (Schudt et al., 1999; Torphy 1998; Torphy et al., 1999; Wong & Koh 2000; Essayan 2001). Studies revealed that the PDE4 family is responsible for the majority of cAMP hydrolysis in immune and inflammatory cells as well as pulmonary and airway smooth muscle cells.

For these reasons, there has been interest in the development of PDE4 inhibitors for use in such diseases as asthma and chronic obstructive pulmonary disorder, (COPD; Spina et al., 1998; Landells et al., 2001; Spina 2003; Grootendorst et al., 2003; PDE4 inhibitors reviewed in Burnouf & Pruniaux 2002). Indeed, theophylline, a non-specific PDE inhibitor, has been used in the treatment of asthma for over 70 years. Unfortunately, nonspecific PDE inhibitors exhibit emetic properties due to raising cAMP levels in the emetic centres within the central nervous system. For this reason, PDE4 isozyme specific inhibitors are being developed and tested. These inhibitors demonstrated that PDE4A/4B inhibitors inhibited TNFa release from monocytes and attenuated the proliferation of T lymphocytes while PDE4D inhibitors had no effect on these responses (Manning et al., 1999). PDE4C is absent in cells involved in the immune response. Other disease states for which PDE4 inhibitors have been considered include depression, rheumatoid arthritis, eczema and multiple sclerosis (Doherty 1999; Essayan 1999; Houslay et al., 1998; Schmidt et al. 1999; Schudt et al. 1999; Souness & Rao 1997; Torphy 1998; Torphy et al., 1999; Wong & Koh 2000). A new PDE4 inhibitor-cilomilast, is in development as a treatment for COPD and asthma (Giembycz 2001).

Recent research has implicated PDE4D7 and PDE4D9 isoforms in ischemic stroke (Gretarsdottir et al., 2003). A decreased level of PDE4D7 and PDE4D9 was observed in EBV transformed B-cell lines from affected patients which correlated to an increased risk for the disease. The authors proposed that PDE4D is involved in atherosclerosis underlying ischemic stroke due to the previous research on PDE4 inhibitors. In vascular smooth muscle cells, PDE4 inhibitors have been demonstrated to significantly potentiate the antimigratory effects of forskolin (Palmer et al., 1998; Goncharova et al., 2003) and attenuate proliferation (Pan et al., 1994; Johnson-Mills et al., 1998). PDE3 and PDE4 inhibitors used together act synergistically to dramatically reduce proliferation (Pan et al., 1998).

1.4.1 Relaxant effects of PDE4 inhibitors

It is well known that raising cAMP levels in vascular smooth muscle cells initiates relaxation (Murray 1990), an effect also seen with PDE inhibition. Studies have indicated that PDE4 and PDE3 inhibition exert the most potent relaxant effect. Indeed, in human intralobar pulmonary arteries, all PDE inhibitors tested induced a concentration dependant relaxation of which rolipram produced the most potent (Bardou et al., 2002). Other studies have confirmed the relaxant effect of PDE4 inhibitors, and the synergistic effect when used together with PDE3 inhibitors (Wagner et al., 1997; Eckly-Michel et al., 1997; Goirand et al., 2001; Pauvert et al., 2002). PDE4 inhibitors also potentiate the effect of cAMP elevating agents in rat aortic smooth muscle cell (Tilley & Maurice 2002). The vasorelaxant effect of PDE4 inhibitors has been demonstrated to require a functional endothelium present (Komas et al., 1991).

The vasorelaxant properties of PDE3 and PDE4 inhibitors are of considerable interest to researchers regarding therapies for cardiovascular disease. In contrast with the systemic circulation, the pulmonary circulation contracts in response to hypoxia. Chronic hypoxia is associated with several pulmonary diseases such as COPD and can be caused by vascular injuries, thrombosis and ischemic cardiovascular disease. Sustained chronic hypoxia leads to the development of pulmonary arterial hypertension (PAH, see section 1.5 for further details). The pulmonary vasoconstriction witnessed in hypoxia can be opposed by the use of PDE4 and PDE3 inhibitors (Pauvert et al., 2002; Goirand et al., 2001). In PAH, pulmonary artery relaxation is significantly attenuated in response to vasodilators. Studies investigating the responses of isolated pulmonary artery rings to agonists that induce relaxation in rat models of PAH, have demonstrated that PDE4 and PDE3 inhibitors can improve the response to vasodilators in the PAH rat (Wagner et al., 1997).

1.5 Pulmonary Arterial Hypertension

Pulmonary hypertension (PH) was previously grouped into two categories; primary pulmonary hypertension (PPH) or secondary pulmonary hypertension. Classification of the disease depended on the absence or presence of identifiable causes or risk factors. In 2003, this classification was revised to take into account the identification of genetic factors and is now grouped into three categories, idiopathic PAH, familial PAH and PAH related to risk factors or associated conditions (*see figure 1.4 for 'Venice classification'*).

Pulmonary arterial hypertension is a progressive and usually fatal disorder characterized by a maintained increase of pulmonary arterial pressure (PAP) caused by occlusion of the small pulmonary arteries (reviewed in Humbert et al., 2004). When the mean PAP exceeds 25mmHg at rest and 30mmHg during exercise, this is diagnosed as PAH. Idiopathic and familial PAH is an uncommon disorder with an estimated occurrence of 1-2 cases per million people and is more commonly seen in females of 20-40 years of age. The symptoms of PAH are extremely variable and the age of onset differs dramatically between individuals. This generally leads to the disease being diagnosed in its latter stages. The variable age of onset and reduced penetrance of the disease also indicates environmental factors play a role in the development of PAH. Current treatment strategies include prostanoids such as beraprost, (Badesch et al., 2004), nitric oxide, endothelin receptor antagonists like bosentan (Channick et al., 2004; see section 1.5.5.1 for further details), vasodilators and in extreme cases, lung transplantation. PDE inhibitors are also included in the list of emerging therapies for pulmonary hypertension (Ghofrani et al., 2004).

In 1999, candidate gene mapping revealed a gene locus linked to familial PPH on chromosome 2q33 (Deng et al., 2000). Through further studies on this locus, known as PPH1, it has been indicated that mutations within a gone at this locus causes familial PPH (The International PPH Consortium, 2000; Machado et al., 2001). This gene is the bone morphogenetic protein type II receptor (BMPR-II). Germline mutations in BMPR2 are also found within at least 26% cases of sporadic PAH (Thomson et al., 2000; Machado et al., 2001). To date, 46 unique mutations within the BMPR2 gene have been identified in cases of PAH (Machado et al., 2001). The mutations are dispersed throughout the gene except for in three of the thirteen exons, 5, 10 and 13. The nature of the mutations range from frameshift to nonsense mutations, resulting in truncated transcripts and variable nucleotide changes. It has been suggested that the inheritance mechanism of PPH is haploinsufficiency (Machado et al., 2001). This would indicate the pathway mediated through BMPR-II to be of critical importance in maintenance and repair of the pulmonary vasculature. BMPR2 encodes the transforming growth factor- β (TGF- β) type II receptor BMPR-II which is expressed in the endothelium and smooth muscle cells of the pulmonary and systemic vasculature.

Hypoxia-induced pulmonary hypertension is aggravated by the increased amount of 5-HT delivered to the lung in rats and this can be blocked with a 5-HT antagonist (Eddahibi et al., 1997). This occurs through a hypoxia-induced increase in 5-HT transporter (5-HTT)

levels (Eddahibi et al., 1999). A study of patients with PAH revealed a link between the variant of 5-HTT expressed and susceptibility to PAH (Eddahibi et al., 2003).

PAH occurs more commonly as a consequence of serious disorders such as COPD, HIV infection or left ventricular failure. Additionally, secondary PAH can develop due to the use of appetite suppressants such as fenfluramine or aminorex phentermine and in response to sustained hypoxia caused by cardiovascular disease, vascular injury or due to high altitude. The prevalence of secondary PAH is difficult to estimate due to the large number of causes overlapping with symptoms of more common disease states preventing a clear diagnosis.

The most frequent initial stimuli of PAH are mechanical obstruction to pulmonary flow and hypoxia. Acute hypoxia causes a reversible PAP increase whereas chronic exposure to hypoxia induces alterations in the small pulmonary vessels and renders the resultant increase in PAP irreversible.

1.5.1 Models used in research of PAH

The most common animal model of PAH is the chronic hypoxie rat (Rabinovitch et al., 1979). Male Wistar rats are exposed to low oxygen (10%O₂) by reducing the atmospheric pressure in a hypobaric chamber. A rise in PAP is observed after three days of hypoxia as well as right ventricular hypertrophy. After two weeks in hypoxia, hypertrophy of the pulmonary artery occurs (Vender 1994). A more recently utilised rat model is the Fawn hooded rat. The fawn hooded rats are used for their inherited sensitivity to the development of idiopathic PAH (Stelzner et al., 1992). Recently, other transgenic mice have provided valuable new models for study. For example, mice expressing a dominant-negative form of the BMPRII gene, dnBMPRII mice, display elevated PAP and pulmonary arterial muscularisation when the mutation is activated after birth (West et al., 2004). These mice confirm that a loss of function of the BMPRII gene is sufficient to produce a pulmonary hypertensive phenotype. Also, mice that overexpress the 5-HTT transporter, 5-HTT+ mice (see section 1.5.7 for further details), exhibit increased right ventricular pressure compared with wild-type mice (MacLean et al., 2004). Under conditions of hypoxia, the 5-HTT+ mice display increased remodelling compared to wild-type.

Cells can easily be grown in a hypoxic environment, mimicking conditions that lead to the development of PAH. Pulmonary vasoconstriction is an inherent property of pulmonary vascular smooth muscle cells (Voelkel & Tuder 1997) and a direct effect of hypoxia on

smooth muscle cells has been witnessed. After 1 week of hypoxia, the proliferation rates of PASMC from distal human pulmonary arteries were observed to increase (Yang et al., 2002) and isolated PASMC cells have also been observed to contract in response to hypoxia (Murray 1990). Human PASMC (hPASMC) have therefore been utilised as a cellular model of PAH (Murray et al., 2002; Murray et al., 2003).

1.5.2 Pathology of PAH

PAH is associated with right ventricular hypertrophy and changes in the structure and function of smooth muscle cells, fibroblasts and endothelial cells leading to a sustained increase in PAP and vasoconstriction. These changes, along with the heterogeneity among cell phenotypes, contribute to the vascular remodelling (reviewed in Mandegar et al., 2004; Humbert et al., 2004), altered tone, thrombosis and vasoreactivity witnessed in PAH. Although the entire pulmonary vascular tree undergoes histological alterations, it is the smaller vessels which are primarily affected. Vascular smooth muscle cells proliferation rate increases, resulting in medial hypertrophy and arteriolar muscularisation, fibrosis, thrombosis. Endothelial cells increase in proliferation causes intimal thickening and the formation of plexiform lesions. The dysfunction of inflammatory cells is also considered to play a role in the development of PAH. All of these problems contribute to the remodelling of the pulmonary vascular tree. Medial hypertrophy is mainly due to abnormalities observed in pulmonary artery smooth muscle cell (PASMC) function, namely an increase in proliferation and migration (Yuan & Rubin 2001). Another characteristic of some forms of PAH is the formation of plexiform lesions arising from the concentric endothelial cell proliferation and smooth muscle cells migration. These lesions occur distally to obliterative intimal lesions.

Pathophysiological studies have highlighted the importance of several mediators in the development of PAH, namely, Endothelin-1 (section 1.5.5.1; Giaid et al., 1993), nitrie oxide (section 1.5.5.2; Giaid et al., 1995), prostacyclin (1.5.5.3; Christman et al., 1992; Tuder et al., 1999) and serotonin (5-HT, section 1.5.7; Herve et al., 1995; MacLean et al., 2000).

1.5.3 Pulmonary artery smooth muscle cells

Increased smooth muscle cell proliferation and hypertrophy have been implicated in the development of PAH (Yuan & Rubin 2001) and leads to muscularisation of the pulmonary resistance vessels. Hyperplasia is characteristic of the larger vessels, whereas the small vessels undergo more extensive remodelling (Humbert et al., 2004). The pulmonary artery smooth muscle cells are normally present in a quiescent and contractile state, but injury to

the vessel wall activates cells into a proliferative, migratory and secretory phenotype. This phenotypical change has been reported to be dependent upon the growth factors transforming growth factor β (TGF- β), and basic fibroblast growth factor (bFGF; Boudreau et al., 1991). The smooth muscle cells then migrate through to the intimal layer and can invade the lumen of the vessel forming concentric lesions. Thrombin induced hypertrophy has been demonstrated to occur through activation of a set of immediate early genes (IEG), whereas PDGF induced proliferation occurs through an alternative set of IEG (Rothman et al., 1994). IEG expression in both instances appears to be regulated by extracellular calcium, PKC and sodium (see below for further details).

In cultured PASMC, hypoxia inhibits the voltage gated potassium channels in PASMC causing a depolarisation of the membrane and a resultant increase in intracellular calcium (Platoshyn et al., 2001; Limsuwan et al., 2001; reviewed in Strange et al., 2002). The rise in Ca²⁺ occurs through many different mechanisms including activation of voltage dependent calcium channels, the production of inositol 1,4,5-triphosphate (IP₃) thus facilitating the release of Ca^{2+} from intracellular stores, and the increase of Ca^{2+} levels by reversing Na⁺/ Ca²⁺ exchange (Madden et al., 2001). It has also been reported that the maintenance of Ca²⁺ levels is aided by the inhibition of Na⁺/ Ca²⁺ exchange (Wang et al., 2000). A reduction in K^+ channel expression and function has been observed in PASMC from PAH patients (Yuan et al., 1998a; Yuan et al., 1998b; review of hypoxic effects on K^+ channels, Raj & Shimoda 2002) as well as in PASMC cultured in hypoxia (Wang et al., 1997) or exposed to serum from patients with PAH (Limsuwan et al., 2001). The rise in intracellular calcium also leads to vasoconstriction and activation of PKC. PKC is a calcium dependant enzyme that stimulates DNA synthesis in systemic artery smooth muscle cells and PASMC will not proliferate in response to hypoxia unless PKC is activated. Recent studies using a PKC-ɛ knockout mouse model confirmed the role of PKC in hypoxic vasoconstriction as the transgenic mice displayed reduced hypoxic pulmonary vasoconstriction (Littler et al., 2002). As well as increased proliferation and hypertrophy, hypoxia also decreases apoptosis of PASMC.

PASMC exposed to hypoxia also display an increase in Rho-kinase activity that has been reported to contribute to hypoxia-induced pulmonary vasoconstriction (Wang et al., 2001). This is mediated by Rho-kinase increasing myosin light chain phosphorylation. Use of the Rho-kinase inhibitor Y-27632 was able to ablate this effect. In addition, Rho-kinase activation causes an inhibition of myosin phosphatase, enabling PASMC to sustain constriction (Wang et al., 2003).

It has been shown that vascular smooth muscle cells display marked heterogeneity with several functionally distinct phenotypes (Frid et al., 1997). Each distinct population of cells expresses different cytoskeletal and contractile proteins and channels and display different growth rates and responses to growth factors. In hypoxia, there is a redistribution of these cell types, altering the phenotype and responses of cach pulmonary artery. This heterogeneity of smooth muscle cells adds complexity to an already complex interplay of many different factors involved in the response to chronic hypoxia and the development of PAH.

1.5.4 Pulmonary artery fibroblasts

An increase in extracellular matrix proteins and pulmonary arterial fibroblasts (PAF) is observed in PAH due to sustained hypoxia. This causes vascular hardening maintaining pulmonary vasoconstriction. Increased fibroblast proliferation has been shown to exceed that of PASMC and pulmonary artery endothelial cells (PAEC) in hypoxic animal models (Belknap et al., 1997). Hypoxia acts through Ga_{Vo} and G_q coupled P_{2Y} activation to promote fibroblast proliferation (Stenmark et al., 2002). Molecular studies have revealed that constitutive p38 MAPK (section 1.5.6) activity enhances PAF proliferation in remodelled vessels (Welsh et al., 2001).

Similar to PASMC, PAF also display marked heterogeneity (Stenmark et al., 2002). Responses to hypoxia are unique among the sub populations of cells, with some even displaying a reduction in DNA synthesis. Populations of PAF with hypoxia induced changes in proliferative and matrix-producing phenotypes also display the expression of smooth muscle α -actin, suggesting that some PAF transdifferentiate into myofibroblasts (Stenmark et al., 1995).

1.5.5 Pulmonary artery endothelial cells

Endothelium-dependant mediators have been implicated in the modulation of hypoxic pulmonary vasoconstriction. Indeed endothelial dysfunction has long been considered to have a key role in the development of PAH. The endothelium releases vasoconstrictors such as endothelin and vasodilators including prostacyclins and NO. An imbalance between vasodilators and vasoconstrictors and an increase in PAEC proliferation is apparent in PAH.

1.5.5.1 Endothelin-1

Endothelin-1 (ET-1; Yanigasawa et al., 1988; Masaki et al., 1991) is a potent vasoconstrictor produced by both pulmonary and systemic endothelial cells. It is released upon stimulation by a range of agents including epinephrine, TGF- β and IL-1 and also by shear stress or hypoxia. Once released, ET-1 binds to type A receptors (ET_A) found mainly in smooth muscle cells or type B receptors (ET_B) found on endothelial cells. ET_A receptors are generally thought to regulate vasoconstriction by facilitating the production of IP₃, while ET_B receptors of endothelial cells lead to vasodilation through the release of nitric oxide (NO; Eddahibi et al., 1991). However, the distribution of receptors displays anatomical heterogeneity with ET_A mediating vasoconstriction in the large arteries, while in the smaller resistance arteries atypical ET_B receptors regulate vasoconstriction (McCulloch et al., 1998). No effect has been observed through the binding of ET-1 to endothelial cells, hence the contributory effect of ET-1 to sustained vasoconstriction. ET-1 also contributes to vascular remodelling through increasing the proliferation of PASMC via ET_A receptors (Janakideva et al., 1992; Zamora et al., 1993).

ET-1 can also indirectly increase cGMP levels through NO release and directly stimulate or inhibit the synthesis of cAMP in pulmonary arteries. The ET-1 induced increase in cGMP occurs in the larger pulmonary arteries alone through the ET_A receptor. The increased synthesis of cAMP by ET-1 is observed in the small resistance arteries and could be blocked by an ET_B receptor antagonist. ET-1 mediated reduction in cAMP levels is mediated in part by the ET_A receptor and occurs in the main pulmonary arteries. All these effects were endothelium independent (Mullaney et al., 2000). Under chronic hypoxic conditions, basal levels of cAMP are reduced in all but the small resistance arteries of the rat where cAMP levels are mildly increased (Mullaney et al., 1998). ET-1 induced cAMP responses were lost in the hypoxic rat. Reduction of cAMP in the large pulmonary arteries was no longer observed in hypoxic rats, instead, a significant increase of cAMP was observed. In the resistance arteries of the hypoxic rat, ET-1 was no longer able to increase cAMP (Mullaney et al., 1998).

ET-1 levels have been shown to be raised in systemic and pulmonary circulation in patients with primary and secondary PAH to a level sufficient enough to induce vasoconstriction (Stewart et al., 1991; Cody et al., 1992; Cacoub et al., 1997). The role of ET-1 in PAH (reviewed in MacLean 1998; MacLean 1999) has been investigated further through the use of chronic hypoxic rat models. In lungs from the chronic hypoxic rat, ET-1 levels and ET_A receptor expression increase (Li et al., 1994). Further investigation into these alterations in

the lung uncovered an increase in vasoconstriction mediated by the ET_A receptor in the large and small pulmonary arteries (MacLean et al., 1995; McCulloch et al., 1998). ET receptor antagonists have thus been developed to investigate their therapeutic potential in PAH. To date, the ET_A and mixed ET_A / ET_B receptor antagonists tested have proven to be effective at hindering the development of and even reversing PAH in chronic hypoxic rat models (DiCarlo et al., 1995; Chen et al., 1997; Underwood et al., 1998; reviewed in MacLean 1999). Currently, a mixed ET-1 receptor antagonist, bosentan, is in use as a treatment for PAH patients (Channick et al., 2004).

1.5.5.2 Nitric Oxide

Nitric oxide (NO) is a well characterised molecule with functions including inhibition of smooth muscle growth and constriction (Singh & Evans 1997). The effects of NO are mediated partly through the soluble guanylyl cyclase receptor which increases cGMP in cells and, in turn, increases PKG activity. PKG reduces the intracellular concentration of Ca^{2+} and hyperpolarises the membrane through inhibition of voltage and receptor Ca^{2+} channels and activation of K⁺ channels respectively. Smooth muscle relaxation is also induced through the activation of myosin light chain phosphatase by cGMP. Independently of cGMP, NO signalling in hypoxic conditions has been demonstrated to induce modifications in enzymatic activity and, more recently, activate the sarcoplasmic reticulum Ca^{2+} -ATPase (SERCA) causing a decrease in intracellular calcium (Mingone et al., 2003).

NO is produced in endothelial cells during the conversion of L-arginine to L-citrulline which is catalysed by nitric oxide synthase (NOS). There are three known types of NOS, nNOS or type I NOS which is secreted by neurones. Type II NOS or iNOS is inducible and secreted by various cells in response to cytokine stimulation. Type III NOS or eNOS is a Ca^{2+} -dependant, constituitively expressed isoform secreted by endothelial cells. eNOS deficient mice display mild PAH in a normal atmosphere and have an increased susceptibility to hypoxia induced pulmonary hypertension (Fagan et al., 1999). Similar studies in mice with disrupted iNOS or nNOS genes suggest these isoforms have a minor role in determining PAP in mice.

Results from studies investigating NO levels in hypoxic conditions and in PAH have produced conflicting results. In hypoxic rat lungs, eNOS and iNOS mRNA and protein levels have been reported to significantly increase (Igari et al., 1998). Immunocytochemical staining revealed the de novo eNOS and iNOS proteins appear in the small and medium sized arteries of hypoxic rats whereas these proteins appear limited to large arteries in normoxic rats (LeCras et al., 1996). Hypoxia has also been shown to induce iNOS in a HIF-1 dependant manner in PAEC and PASMC (Palmer et al., 1998). Lungs from patients with PAH also display an increased amount of eNOS, especially in the plexiform lesions (Mason et al., 1998). Lung NO production in PAH patients has been reported by several groups to increase (Archer et al., 1998; Forrest et al., 1999).

A number of studies have reported a reduction in NO levels in PAH. Both acute and chronic hypoxia were observed to attenuate NO production in the main pulmonary artery of the rat with a resultant decrease in intracellular levels of cGMP (Shaul et al., 1993; MacLean et al., 1996). In addition, pulmonary arteries from patients with PAH have been observed to exhibit reduced expression of eNOS (Giaid & Saleh 1995). More recent work on pulmonary arteries from the chronic hypoxic rat have shown that although endothelial NO production in response to carbachol was reduced, total NO protein was unaltered, indicating hypoxia affects eNOS activity at the post-translational level (Murata et al., 2002) Such results were the reason for use of inhaled NO therapy in PAH.

NO has been shown to inhibit ET-1 induced pulmonary vasoconstriction (Lang & Lewis .1991) and attenuate vascular remodelling (Horstman et al., 1998). Inhaled NO has proven beneficial to patients with short-term and reversible PAH, but not for patients requiring to use it over long periods due to its irritant properties and short action. Among patients treated with inhaled NO, the effects are varied and it is not always successful as a treatment. Underlying this could be the cause for opposing results from different research groups. As such, the definitive role of NO in PAH remains to be determined.

1.5.5.3 Prostaglandins

Arachodonic acid (AA) is metabolised through the cyclooxygenase and lipooxygenase pathways to form prostaglandins and leukotrines. Metabolites of arachodonic acid such as prostacyclin (PGI₂) and thromboxane (T_xA_2) elicit vasoactive effects in the pulmonary circulation (reviewed in Christman, 1998). PGI₂ is produced by endothelial cells and is a potent vasodilator and an inhibitor of platelet aggregation. PGI₂ activation of prostacyclin receptors stimulates synthesis of cAMP, leading to vasodilation and reduced DNA synthesis in vascular smooth muscle cells. The net effect of this is a reduction in pulmonary vascular resistance. T_xA_2 is produced by platelets and is a vasoconstrictor.

Depending on experimental conditions, hypoxia has been reported to both increase and decrease prostacyclin release. Patients with severe PAH have an imbalance in production

of PGI_2 and T_xA_2 and a reduction in expression of PGI_2 synthase (Christman et al., 1992; Tuder et al., 1999). An overexpression of PGI_2 synthase has been reported to protect mice against hypoxia induced pulmonary hypertension (Geraci et al., 1999). Conversely, PGI_2 has been reported to increase in response to seven days hypoxia (Shaul et al., 1991). ET-1 can also induce PGI_2 production in rat lung (Barnard et al., 1991).

In most instances, the effects of PGI_2 are mediated by the production of cAMP as responses can be mimicked in cultured cells using isoprenaline and forskolin, and potentiated using the general PDE inhibitor, IBMX.

Clinical studies investigating the use of prostacyclin analogues as a treatment in PAH revealed they increase survival rate (reviewed in Badesch et al 2004). However, short halflifes and side effects compromise the benefits, therefore new prostacyclin analogues are under development and testing. Subcutaneous delivery of Treprostinil has beneficial effects on exercise and hemodynamics, however side effects including nausea, headaches and pain and erythema at the site of delivery. In Europe, another prostacyclin analogue, Iloprost, has recently been approved for the treatment of PAH and is delivered through inhalation. Inhaled iloprost has little effect on the systemic circulation, providing selective vasodilation of the pulmonary circulation as does inhaled nitric oxide, but only iloprost studies indicate an increased survival rate.

1.5.5.4 Vascular Growth Factors

Many vascular growth factors are secreted by PAEC and a number of these have been observed to be increased in PAH. The most well characterised example is the vascular endothelial growth factor (VEGF). Although VEGF is not secreted by endothelial cells under physiological conditions, it is present in plexiform lesions of PAH patients (Archer & Rich 2000). The only known target of VEGF is the endothelial cells themselves. In hypoxic conditions, the VEGF gene is stimulated almost instantly due to binding of the hypoxia inducible factor -1 (HIF-1; reviewed in Semenza 2000) to a hypoxia response element (HRE) in the promoter region (Gerber et al., 1997). VEGF is also known to be upregulated in PAH (Tuder et al., 1995).

Endothelial cells under hypoxic conditions are also known to release a platelet-derived growth factor (PDGF) which induces vasoconstriction and stimulates PAF migration and proliferation (Peacock et al., 1993; Faller et al., 1999). PDGF is also capable of inducing VEGF. Another growth factor upregulated in PAH is TGF- β (Acrot et al., 1993). In

vascular cells, the predominant effects of TGF- β are growth inhibition, cell differentiation and stimulation of collagen synthesis. Other growth factors elevated in PAH include; bFGF (Arcot et al., 1995); IGF-1 (Perkett et al., 1992) and EGF (Gillespie et al., 1989). The increase in secreted growth factors from endothelial cells can lead to increased proliferation of PASMC through feeding into the mitogen activated protein kinases (MAPK) pathways (Xiao 1993).

1.5.6 Mitogen Activated Protein Kinases

Mitogen activated protein kinases (MAPK) are activated by a wide range of stimuli including growth factors, hormones, extracellular matrix components, GPCR agonists, cellular stress and cytokines (reviewed in Gutkind 2000; Tibbles & Woodgett 1999; Wildman et al., 1999). MAPK are split into groups according to their function. The best characterised MAPK are the extracellular regulated kinases I and 2, also known as p44 and p42 respectively. The functions of the ERK1/2 are extremely diverse and include regulation of cell proliferation (reviewed in Stork & Schmitt 2002), differentiation and tumoregenesis. Other well characterised MAPK are; the c-Jun NH₂-terminal kinases also known as JNK/SAPK, and the p38 MAPK family. JNK and p38 MAPK are involved in the cellular response to stressors. MAPK are activated through phosphorylation by a MAPK kinase (MAPK), each family of MAPK have specific MAPKK and MAPKKK and upstream activators (*figure 1.4*).

1.5.6.1 Activation of MAPK

The classical MAPK pathway of ERK1/2 is now well characterised (*figure 1.5*). Binding of epidermal growth factor (EGF) to its tyrosine kinase receptor (RTK) leads to the phosphorylation of the receptor itself and other substrates. Phosphorylated sites act as docking sites for adaptor proteins such as Grb2 which contains a Src homology 2 (SH2) domain and two SH3 domains or Shc which possesses a phosphotyrosine binding (PTB) domain, one SH2 and one SH3 domain. She acts as a substrate for the EGF receptor and upon phosphorylation binds the SH2 domain of Grb2, resulting in the recruitment of SOS. SOS induces the exchange of GDP bound to Ras for GTP and initiated the MAPKKK cascade. Once activated, ERK1/2 translocates to the nucleus and phosphorylates key transcription factors that ultimately regulate the expression of genes essential for processes such as cell proliferation. GPCRs can also influence cell growth and differentiation through activation of the ERK1/2 pathway (reviewed in Gutkind 2000). This occurs in the same manner as RTKs. In contrast, activation of the JNK pathway occurs through GPCRs

and not RTKs (Coso et al., 1995). How this occurs is as yet unclear as is the activation of the p38 pathway by GPCRs.

1.5.6.2 Hypoxia-induced activation of MAPK

Numerous studies have revealed hypoxia induced activation of the MAPK cascades in many different cell types (reviewed in Minet et al., 2000a). PAF and PAEC have been reported by several groups to display a marked activation of the ERK1/2 pathway under hypoxic conditions (Minet et al., 2000b; Scott et al., 1998; Welsh et al., 2001). In addition, pulmonary arteries from chronic hypoxic rats also show an activation of ERK1 and 2 that peaks at day seven (Jin et al., 2000). Increased phospho-ERK1/2 was observed in both large and small pulmonary arteries along with the activation of JNK and p38.

The p38 family and JNKs have also been reported to display an increased activity in PAF exposed to acute hypoxia for up to 30 hours (Scott et al., 1998). Activation of p38 peaked at 6 hours and 24 hours while JNK activation peaked at 6 hours before returning to basal levels. p38 activity is also increased in PAF, but not in aortic fibroblasts, under chronic hypoxic conditions (Welsh et al., 2001). The transient increase in activation can be explained by the induction of MAPK phosphatases in hypoxia or in response to MAPK activation (Laderoute et al., 1999; Ward et al., 1994). This feedback control could be to avoid apoptosis which has been shown to occur with sustained MAPK activation.

1.5.6.3 HIF-1 and ERK1/2

Hypoxia inducible factor -1 (HIF-1) is a transcription factor responsible for the induction of expression of glycolytic enzymes, metabolic enzymes, proteins involved in proliferation and proteins involved in vascular biology (reviewed in Semenza 1999; Huang & Bunn 2003). HIF-1 can be induced by various hormones, serum and hypoxia. HIF-1 is composed of two subunits, HIF-1 α and HIF-1 β . Both subunits are constituitively expressed, however HIF-1 α is rapidly degraded under normal conditions by proteasomal degradation. Under hypoxic conditions however, degradation of the HIF-1 α subunit is attenuated allowing a higher amount of functional dimers to form (Sutter et al., 2000). Thus, research has focused on HIF-1 α as the regulator of the hypoxic response. HIF-1 activates target genes by translocating to the nucleus and binding to a hypoxia response clement (HRE) in the promoter region. HIF-1 α and HIF-1 β knockout mice models results in embryonic lethality. Partially HIF-1 α deficient mice display a marked reduction in vascular remodelling in response to hypoxia (Yu et al., 1999).

HIF-1 α is highly phosphorylated *in vivo* which allows the stabilisation of the protein. Phosphorylation of HIF-1 α is mediated by an ERK1/2 dependant pathway (Minet et al., 2000b). The other well known MAPK, JNK and p38 MAPK, do not phosphorylate HIF-1 α (Richard et al., 1999). ERK1/2 phospborylation has been shown to promote the transcriptional activity of HIF-1 α (Richard et al., 1999). The effect of HIF-1 α on gene expression is also regulated by p53 which inhibits HIF-1 stimulated transcription (Blagosklonny et al., 1998). Other regulators of HIF-1 include Jab1 and p300/CBP (Bae et al., 2002; Kallio et al., 1998).

1.5.7 Role of 5-Hydroxytryptamine in PAH

The use of appetite suppressants such as aminorex has been determined to be a significant risk factor for the development of PAH (Fishman 1999). These drugs belong to a vast class of amphetamine and epinephrine like drugs that act by increasing local and circulating 5-HT levels. This increase is achieved by the inhibition of 5-HT transporters (5-HTT), release of platelet 5-HT and preventing 5-HT clearance by inhibiting monoamine oxidase (MAO).

5-HT is a vasoconstrictor of the pulmonary circulation and a co-mitogen of vascular smooth muscle cells in culture. 5-HTT is also involved in 5-HT signalling through activation of NAD(P)H oxidase to produce reactive oxygen species (ROS) and ERK1/2 activation which is involved in 5-HT induced smooth muscle cells hyperplasia or hypertrophy (Lee et al., 1999). 5-HT levels have been reported to increase under several conditions that lead to the development of PAH (reviewed in MacLean et al., 2000). Levels of the 5-HTT are also observed to increase in lung tissue and pulmonary arteries from PAH patients with a resultant enhancement of PASMC proliferation in response to 5-HT (Eddahibi et al., 2001). This has also been observed in hypoxic PASMCs (Eddahibi et al., 1999). The increased expression of the 5-HTT is also associated with a polymorphism of the 5-HTT gene disruption are less prone to develop PAH than control mice (Eddahibi et al., 2000). 5-HTT inhibitors have been observed to attenuate hypoxia induced PAH.

Vasoconstriction induced by 5-HT occurs through the 5-HT_{1B/1D} and 5-HT_{2A} receptors depending on vascular tone and in a species specific manner. In humans, vasoconstriction is mediated by the 5-HT_{1B} receptor (MacLean 1999). In rat pulmonary arteries, vasoconstriction usually occurs through the 5-HT_{2A} receptor, however in the chronic hypoxic rat model, vasoconstriction is mediated by the 5-HT_{1B} and 5-HT_{2A} receptors

(MacLean et al., 1996). 5-HT_{1B/1D} receptors are coupled to the $G\alpha_i$ protein and induce contraction by decreasing cAMP levels. Contraction through the 5-HT_{2A} receptors occurs through stimulating the $G\alpha_q$ pathway, leading to an increase in intracellular calcium and activation of PKC.

1.5.8 Cyclic nucleotide pathways in PAH

1.5.8.1 Cyclases

Functionally important isoforms of adenylyl cyclase found in the rat lung are AC2, AC-6 and AC-8 (Jourdan et al., 2001). AC-2 is potently stimulated by PKC (Jacobowitz & Tyengar 1994) and is insensitive to calcium. AC-6 is inhibited by both PKA and PKC, as well as by calcium (Lai et al., 1999; Iwami et al., 1995; Cooper et al., 1995). AC-8 is stimulated by calcium binding and PKC has no effect (Cali et al., 1994). Taken together, these regulatory properties of the AC isoforms found in the lung allows for a tight regulation of cAMP levels, able to tailor stimulation or inhibition when required in response to specific effectors. Under chronic hypoxic conditions, it has been demonstrated that AC activity measured under GTP, forskolin or isoproterenol stimulation is reduced in rat hearts (Pci et al., 2000; Hrbasova et al., 2003). Hypoxia has also been demonstrated to affect guanylyl cyclase. Soluble GC levels and activity have been reported to increase in chronic hypoxic rats (Li et al., 1999).

1.5.8.2 Cyclic nucleotides

Many of the effects of the pathways mentioned exert their effects through cyclic nucleotides. Stimulation of the cAMP or cGMP pathways in smooth muscle leads to relaxation while inhibition of these nucleotides would lead to contraction of smooth muscle. This is mediated by the cAMP- and cGMP-dependent kinases, PKA and PKG.

An increase in either cAMP or cGMP is known to inhibit vascular smooth muscle cell proliferation (reviewed by Koyama et al., 2000; Hayashi et al., 2000). Through the actions of PKA, cAMP attenuates proliferation by antagonising mitogenic pathways and causes a cell cycle block by arresting cells in the G_1 phase and also in the G_2/M phase (Kronemann et al., 1999; Stewart et al., 1999; Van Oirschot et al., 2001). Growth factors such as PDGF induce the ERK pathway in early G_1 phase of the cell cycle. Several studies have indicated the regulation of ERK activation by cAMP (reviewed in Schmitt & Stork 2001). cAMP can inhibit growth factor mediated activation of ERK and also suppress activity of the ERK pathway (Bornfeldt & Krebs 1999; Yu et al., 1997). It is believed this is achieved through the antagonism of Raf activation. However, inhibition of the ERK1/2 pathway is not

always required for cAMP induced growth arrest (Balmanno et al., 2003). Inhibition of ERK by cAMP is cell-specific, in certain cell types cAMP can lead to the activation of ERK. In cells expressing COX-2, activation of the ERK cascade acts as a negative regulator of proliferation through the production of cAMP (Bornfeldt and Krebs 1999).

Raising cGMP levels also attenuates smooth muscle cell proliferation, although not as potently as cAMP (Yu et al., 1997). In contrast to the cAMP-induced block of the cell cycle, it has been observed that cGMP delays G_1/S transition in smooth muscle cells (Fukumoto et al., 1999). This indicates the pathways governed by cAMP and cGMP involved in proliferation are distinct.

In addition to its inhibitory effects on proliferation, cAMP also inhibits migration of smooth muscle cells (Goncharova et al., 2003; Newman et al., 2003). It is unclear how this is achieved, although a link between PKA and inhibition of cell migration is clear in several studies, although this effect is again cell-specific (Sun et al., 2002; Kaufman et al., 2002). In addition, cAMP inhibits the synthesis of some extracellular matrix proteins (Kaji et al., 1996) and also, as mentioned previously is involved in the role of smooth muscle cell contraction/relaxation. It appears that raising cAMP levels is able to inhibit most of the processes involved in the response to vascular injury in smooth muscle cells, making cAMP signalling pathways an attractive therapeutic target in cardiovascular disease such as PAH. Indeed it has been shown that the chronic hypoxic rat displays reduced levels of both cyclic nucleotides in all pulmonary arteries with the exception of the resistance arteries (MacLean et al., 1996). The differing effects of hypoxia on the cyclic nucleotide levels in separate arteries can be explained by the heterogeneity of the smooth muscle cells in each branch. It should be noted that other studies have indicated a rise in cGMP levels in chronic hypoxic rats compared to controls (Cohen et al., 1996). An increase in cyclic nucleotides would be beneficial under hypoxia, aiding to restore the normal tone of the pulmonary circulation. In fact, it has been suggested that hypoxia initially leads to an increase in cAMP through PGI₂ activation, yet after prolonged exposure to hypoxia, desensitisation occurs and pathways are initiated to remove cAMP. In addition, it has been shown that PKA can activate PDE3A and PDE4 in smooth muscle whilst inhibiting AC6, altogether reducing cAMP levels (Murthy et al., 2002).

1.5.8.3 CREB

The cAMP response element binding protein (CREB) is activated by PKA and is responsible for cAMP effects on gene expression (reviewed in Mayr & Montminy 2001; Shaywitz & Greenberg 1999). PKA activates CREB through phosphorylation on Ser¹³³. Phosphorylated CREB is then able to bind to cAMP responsive elements (CRE) in the promoter region of cAMP inducible genes.

CREB has also been reported to be phosphorylated at Ser¹³³ by PKG (Pilz et al., 2003), Akt (Shaywitz et al., 1999), and PKC (Sacki et al., 1999) amongst other kinases (reviewed in Johannessen et al., 2004). Other sites in CREB also act as phosphoacceptor sites including Ser¹¹⁷ and Ser¹²¹ including hypoxia induced kinases (Taylor et al., 2000).

Hypoxia has been shown to lead to a rapid phosphorylation of CREB at Ser¹³³ in a manner more potent than that of forskolin in PC12 cells (Beitner-Johnson & Millhorn 1998). CREB phosphorylation was not mediated through PKA or any of the pathways known to activate CREB. In addition, the phosphorylation of CREB was sustained for 24 hours which is in contrast with the typical rapid phosphorylation/dephosphorylation of CREB. The authors of this study suggested that the maintained CREB phosphorylation implicated a role for CREB in the cells adaptive response to hypoxia. Indeed, it is already known that the CREB coactivator CBP/p300 interacts with HIF-1 α and participates in the transcriptional regulation of hypoxia induced genes (Arany et al., 1996). In smooth muscle cells, active CREB is apparent in proliferating cells and reduced in those that are proliferation-resistant (Klemm et al., 2001). Indeed, it was reported that CREB levels were reduced in hypoxia where smooth muscle cells are known to exhibit increased proliferation. The activity of CREB was not measured however. These studies implicate CREB in the hypoxic response and a further investigation of CREB regulation and function could aid in the understanding of hypoxia regulated gene expression.

1.5.8.4 Phosphodiesterases

Phosphodiesterase (PDE) activity is critical for the tight regulation of cyclic nucleotide levels. An increase in PDE activity reduces the levels of cyclic nucleotides and thus the effect of cAMP or cGMP on smooth muscle functioning. The lung is known to express all PDEs except the retinal specific PDE6 (Polson & Strada 1996; Soderling & Beavo 2000; Yuasa et al., 2000; Koyama et al., 2001). This presents the possibility that the actions of cyclic nucleotides in the pulmonary circulation are dependent on the activities of the PDE families. Indeed, it has been demonstrated that both cAMP-PDE and cGMP-PDE activity increase in the pulmonary arteries of the chronic hypoxic rat (MacLean et al., 1997). Within the pulmonary artery, PDE1, PDE2, PDE3, PDE4 and PDE5 activities have all been demonstrated (Pauvert et al., 2002). Inhibitors of each of these families were also demonstrated to relax precontracted vascular smooth muscle preparations (Pauvert et al., 2002).

A role for PDE1C in vascular smooth muscle cells has already been mentioned (Rybalkin et al., 2002) and it is also recognised that PDE1C isoform expression in vascular smooth cells differs between species (Palmer & Maurice 2000). This is also thought to be the case with PDE2 expression in vascular smooth muscle cells.

PDE2 displays low activity levels and although PDE2 inhibitor use uncovered a role for PDE2 in hypoxic pulmonary constriction in rat lung in one study (Haynes et al., 1996), a regulatory function for PDE2 in vascular smooth muscle cell function has not been reported.

Both genes of the PDE3 family are expressed in vascular smooth muscle cells and have been demonstrated to be induced in response to cAMP-elevating agents in vivo (Tilley & Maurice 2002). However, maintained increases in cAMP levels in cultured rat and human aortic SMC only increased levels of PDE3B (Palmer & Maurice 2000). Chronic hypoxia has been observed to increase PDE3 activity in rat pulmonary vessels (MacLean et al., 1997). This increase can be explained by the induction of both PDE3A and PDE3B in chronic hypoxic rat pulmonary arteries (Murray et al., 2002). In the same study, a cAMPmediated increase of the PDE3A isoform in hypoxic hPASMC was demonstrated.

The use of PDE3 inhibitors on vascular SMC has been demonstrated to inhibit proliferation (Johnson-Mills 1998), migration (Palmer et al., 1998) and act as a vasodilator (Bardou et al., 2001). Indeed, cilostamide attenuates hypoxia induced PAH (Phillips et al., 2000). Unfortunately, PDE3 inhibitors also affect the systemic circulation and could lead to harmful side effects if employed as a treatment. Thus, newly developed PDE3 inhibitors are currently under investigation as putative therapies for cardiovascular diseases. Milrinone and SCA40 have been shown to be potent relaxants of preconstricted pulmonary arteries (Jeffrey & Wallstall 1998). Milrinone has also been reported to significantly decrease PAP in the hypoxic dog and heart failure patients (Jaski et al., 1985; Kato et al., 1998).

In rat and human vascular SMC, PDE5A1 and PDE5A2 are expressed (Murray et al., 2002; Rybalkin et al., 2002). PDE5 is well known to be involved in the regulation of smooth muscle cell relaxation. Under conditions of chronic hypoxia, PDE5 expression and activity are observed to increase in rat pulmonary arteries and hPASMC (Murray et al., 2002), specifically the PDE5A2 isoform. In other animal models of PAII, this result is also observed (Hansen et al., 1998; Black et al., 2001). Consistent with these studies, inhibitors of PDE5 are observed to inhibit proliferation and migration of vascular SMC, enhance NO-protection against vascular leakage and act in a pulmonary specific manner (Osinski et al., 2001; Schutte et al., 2000; Eddahibi et al., 1998). All of these mechanisms can attenuate the development of PAII. Indeed, the PDE5 inhibitor sildenafil has been shown to be effective at improving the prognosis for PAH patients with no effect on the systemic pressure (Zhao et al., 2001; Michelakis et al., 2003). In the chronic hypoxic rat, treatment with sildenafil prior to hypoxic exposure protects against the rise in PAP and inhibits vascular remodelling (Sebki et al., 2003).

PDE4 inhibitors have been developed for use in inflammatory conditions such as COPD due to their ability to suppress the activity of immune and inflammatory cells and reduce airway smooth muscle hypertrophy and hyperplasia (Vignola 2004; Schmidt et al., 1999; Essayan 1999; Spina 2003). Recently, new PDE4 inhibitors, cilomilast (Ariflo) and roflumilast, have been developed for use in the treatment of COPD and asthma (Giembcyz 2001; Compton et al 2001; Underwood et al., 1998). Investigation into the role of PDE4 identified PDE4A and PDE4B as the major regulators of inflammatory cell functions with 4D also mediating in part (Manning et al., 1999). PDE4 and PDE3 inhibitors are able to cause relaxation of bronchial rings and when used in tandem, a much greater bronchorelaxant effect is seen than when used separately. Airway SMC proliferation has also been shown to be regulated by cAMP levels (Tomlinson et al., 1995). PDE4 inhibitors are known to display similar results in vascular SMC and therefore have been considered for use in pulmonary disorders. Indeed PDE4 inhibitors are also capable of inhibiting SMC migration (Palmer et al., 1998) and proliferation (Chen et al 2002, Ogawa et al 2002). In both proliferation and migration studies, PDE4 inhibitors or PDE3 inhibitors used alone have mild effects, when used together they synergise to potentiate their effect (Palmer et al., 1998). Due to the success of clinical trials using PDE4 inhibitors as a treatment for COPD, the potential for PDE4 inhibitors in modulating the response to hypoxia is apparent. It is as yet unknown what effect, if any, hypoxia has on PDE4 in vascular SMC. However, a differential expression of PDE4 in cases of COPD has been observed (Barber et al., 2004). It would therefore be beneficial to uncover the PDE4 isoforms expressed in

pulmonary arterial cells and to distinguish isoform-specific functions. In this study, I set out to identify the PDE4 isoforms expressed in human pulmonary artery smooth muscle cells and investigate their role in the altered cyclic nucleotide levels apparent in hypoxia. In addition, the cAMP regulation of proliferation was investigated.

Adenylyl Cyclase Isoform	High Expression	Potential Function
I	Brain	Circadian rhythm, learning, memory
II	Brain, lung	Synaptic plasticity, cell proliferation arrest
111	Olfactory epithelium, pancreas	Odorant stimulation
IV	Widespread	
v	Brain, heart,	
VI	Heart, kidney, widespread	Cell proliferation
VII	Widespread	Ethanol dependancy
VIII	Brain, pancreas	Synaptic plasticity
IX	Pituitary, widespread	

Table 1.1 Summary of adenylyl cyclase isoform distribution.

This table summarises the current knowledge of specific adenylyl cyclase isoform distribution throughout the body and potential associated functions.

Adenylyl Cyclase Isoform	Gsa	Gsβγ	Ca2+	РКС
I	+	-	+	0
Ш	+	?	0	+
III	+	0	?	?
IV	+	+	0	?
V	+ '	0	-	+ (ξ)
VI	+	0	-	?
VII	+	+	0	+
VIII	+	?	+	?
IX	+	?	0 (inhibited by calcineurin)	?

Table 1.2 Summary of adenylyl cyclase regulation

This table summarises the regulation of each adenylyl cyclase isoform. Question marks represent modes of regulation that have not been fully investigated for that isoform. Stimulation is represented by + and inhibition by -. 0 signifies no effect on that isoform.

PDE family	Nucleotide specificity	Inhibitors	Regulation
PDE1	cAMP/cGMP	Nicardipine, vinpocentine, zaprinast	(+) Ca ²⁺ /calmodulin (-)PKA/PKG, CamKII
PDE2	cAMP/cGMP	EHNA	(+) cGMP, PKC
PDE3	cAMP	Cilostamide, cilostazol, milrinone, amrinone	(+) PKA, PKB (-) cGMP
PDE4	cAMP	Roflumilast, Rolipram, Ro-20-1724, cilomilast, denbufylline, BAY 19-8004	(+)PKA, ERK, PA (-)ERK, caspases
PDE5	cGMP	Sildenafil, DMPPO, vardenafil, tadalafil Zaprinast	(+) cGMP, PKA, PKG (-) caspases
PDE6	cGMP	Zaprinast, dipyridamole, sildenafil, DMPPO	(+)Transducin (-) cGMP
PDE7	cAMP	Dipyrimidamole	(+/-) PKA
PDE8	cAMP	Dipyridamole	PAS domain
PDE9	cGMP	Zaprinast	
PDE10	cAMP/cGMP	Dipyramidole	(-) cAMP
PDE11	cAMP/cGMP	Dipyramidole, zaprinast	

Table 1.3 Summary of phosphodiesterase family regulation

This table summarises substrate specificity and regulatory mechanisms for each of the phosphodiesterase families. Compounds known to inhibit each family are also shown, not all are selective. Stimulation is represented by +, and inhibition by -.



Figure 1.1 Modular structure of phosphodiesterases

All phosphodiesterase families share a common modular structure with a unique Nterminal involved in regulation, a catalytic domain responsible for nucleotide hydrolysis which is conserved throughout all PDEs and a unique C-terminal domain.



Figure 1.2 Splice variants of the PDE4 family

The PDE4 family are characterised by the presence of upstream conserved regions known as UCR1 and UCR2 at their N-terminus. The region that links UCR1 and UCR2 together is called linker region 1 (LR1), and linker region 2 (LR2) connects UCR2 to the catalytic domain. The linker regions differ between the different PDE4 families. Alternative mRNA splicing of members of the PDE4 family gives rise to "long" isoforms containing both UCR1 and UCR2 and "short" isoforms with only UCR2 are produced. An additional splice site also generates "super-short" isoforms which have an N-terminally truncated UCR2.



Figure 1.3 Autocrine activation of PDE4D5 by ERK

ERK activation inhibits PDE4D5 activity through direct phosphorylation. In human aortic smooth muscle cells however, ERK leads to the autocrine production of PGE_2 which stimulates adenylyl cyclase. cAMP levels then rise and PKA is activated and can then phosphorylate PDE4D5. Due to the rapidity and magnitude of ERK dependant PKA phosphorylation of ERK, activation overall is achieved (Baillie et al., 2001).

PDE Family	Disease states in which PDE plays a role	References
PDE1	Parkinsons's disease Tumorigenesis	Kakkar et al., 1999; Kakkar et al., 1997
PDE3	РАН	Wagner et al., 1997; MacLean et al., 1997; Murray et al., 2002
PDE4	Asthma, COPD, depression, malignant gliomas, PAH, acute lymphoblastic leukemia, ischemic stroke	Torphy 1998; Schmidt et al., 1999; Schudt et al., 1999; Torphy et al., 1999; Spina 2003; Gale et al., 2002; Nemoz et al., 1985; Grootendorst et al., 2003; Chen et al., 2002; Ghoframi et al., 2004; Hatzelmann & Schudt 2001; Wagner et al., 1997;Goirand et al., 2001; Ogawa et al., 2002; Gretarsdottir et al., 2003
PDE5	Lung ischemia reperfusion injury, PAH	Schutte et al., 2000; Sebkhi et al., 2003; Michelakis et al., 2003; Eddahibi et al., 1998; Goirand et al., 2001; MacLean et al., 1997; Murray et al., 2002

Table 1.4 Summary of disease states PDEs are reported to play a role

This table summarises some of the diseases PDE inhibitors are therapeutic in or diseases in which the PDE family has been implicated. References for the named disorders are also included.

Revised Clinical Classification of Pulmonary Hypertension (Venice 2003)

1. Pulmonary arterial hypertension (PAH)

- 1.1 Idiopathic (IPAH)
- 1.2 Familial (FPAH)
- 1.3 Associated with (APAH):
 - 1.3.1 Collagen vascular disease
 - 1.3.2 Congenital systemic-to-pulmonary shunts
 - 1.3.3 Portal hypertension
 - 1.3.4 HIV infection
 - 1.3.5 Drugs and toxins
 - 1.3.6 Other
- 1.4 Associated with significant venous or capillary involvement
 - 1.4.1 Pulmonary veno-occlusive disease (PVOD)
 - 1.4.2 Pulmonary capillary hemangiomatosis (PCH)
- 1.5 Persistent pulmonary hypertension of the newborn

2. Pulmonary hypertension with left heart disease

- 2.1 Left sided atrial or ventricular heart disease
- 2.2 Left sided valvular heart disease

3. Pulmonary hypertension associated with lung diseases and/or hypoxemia

- 3.1 Chronic obstructive pulmonary disease
- 3.2 Interstitial lung disease
- 3.3 Sleep-disordered breathing
- 3.4 Alveolar hypoventilation disorders
- 3.5 Chronic exposure to high altitude
- 3.6 Developmental abnormalities

4. Pulmonary hypertension due to chronic thrombotic and/or embolic disease

- 4.1 Thromboembolic obstruction of proximal pulmonary arteries
- 4.2 Thromboembolic obstruction of distal pulmonary arteries
- 4.3 Non-thrombotic pulmonary embolism
- 5. Miscellaneous

Figure 1.4 Venice classification of pulmonary arterial hypertension.

With the variety of new disorders proven to lead to pulmonary arterial hypertension, the previous classification of PAH was considered to be outdated. Thus, in the Venice 2003 meeting, the classification was revised. Adapted from Simonneau et al., 2004.



Figure 1.5 Activation of MAPK

The ERK, JNK and p38 cascades are the classic examples of MAPK cascades. Upon stimulus, a MAP kinase kinase kinase, MAPKKK, activates its MAP kinase kinase MAPKK. This leads to activation of the MAP kinase which can then relocate to the nucleus and regulate gene expression through its action on transcription factors.



Figure 1.6 ERK pathway

The ERK pathway can be activated through the actions of a stimulated G-protein coupled receptor or activated tyrosine kinase receptor. The intracellular cascades have different points of cross talk where they can 'finely-tune' the ERK response following stimulation. Red arrows signify the cAMP pathway involved in ERK activation.

Chapter 2

Materials and Methods

2.1 Mammalian Cell Culture

2.1.1 Maintenance of hPASMC

2.1.1.1 Human Pulmonary Artery Smooth Muscle Cells

Human Pulmonary Artery Smooth Muscle Cells (hPASMC) are obtained commercially from Cambrex Biowhittaker and are derived from 1st and 2nd order pulmonary arteries. Each vial of cryopreserved cells came with a certificate of analysis showing guaranteed viability of cells, presence of smooth muscle cell markers and negative viral detection. hPASMC were maintained in Smooth Muscle Growth Medium (SmGm, Cambrex) which had been optimized for the growth of smooth muscle cells. The SmGm was prepared using Smooth Muscle Basal Medium (SmBm) supplemented with 5% Foetal Calf Serum (FCS), gentamycin GA-1000 (50µg/ml) penicillin/streptomycin (100 units/ml), human fibroblast growth factor-B (2ng/ml), human EGF (0.5 ng/ml) and insulin (5µg/mf). These were all available from Cambrex in single use aliquots as a SmGm-2 bullet kit. Cells were seeded into 25cm² flasks at the recommended seeding density of 3500cells/cm² using the following calculations:

Max. area that can be plated = No. of cells available/Recommended seeding density

Max. no. flasks prepared = Max. surface area that can be plated/Growth area of flask

Flasks were prepared with 1ml SmGm-2/5cm² growth area of flask at 37°C in a 5% CO₂ atmosphere for 30 minutes prior to seeding. The day after seeding, and every other day thereafter, the SmGm-2 was replenished to remove unattached cells and waste from the cells. As the cells became more confluent, 2ml SmGm-2/5cm² growth area of flask was used. Cells were maintained in a Heraues CO₂ incubator set at 37°C in an atmosphere of 5% CO₂. The cells were examined daily for any signs of stress, for example atypical morphology. Each vial of cells arrived at passage 3 and were used at passage 5-7 for all experiments. Cambrex guarantee normal morphology and proliferation rates of cells until passage 10.

2.1.1.2 Passaging of hPASMC

Cells were passaged at 85% confluency. This was achieved by rinsing the cells in SmBm to remove all traces of serum. To detach the cells from the culture vessel, 2mls/100mm plate of Trypsin/EDTA (145mM NaCl, 5.4mM KCl, 8.2mM Na₂HPO₄, 1.5mM KH₂PO₄, 42mM Trypsin, 10mM Glucose and 0.7mM EDTA) was added to the culture flask which

is subset of the production during the following sector and when

modelines for the set of the relevant of the rest

was then left at room temperature. After 5 minutes incubation with Trypsin, the culture vessel was rapped gently to detach any remaining adherent cells. Complete cell detachment was checked by microscopic examination. The cells were then collected in 2mls/100mm plate of fresh SmGm-2 and centrifuged at 700rpm for 3 min in the MSE Mistral 1000 swinging bucket centrifuge. The Trypsin/SmGm-2 solution was aspirated off the cells before resuspending them in prewarmed SmGm-2. The cells were then plated out at the optimal seeding density of 3500 cells per cm³ of growth area. SmGm-2 was replaced every other day.

2.1.1.3 Hypoxic hPASMC Model

At the desired passage number and confluency, the culture vessels were split into two groups following replenishment of SmGm-2. One group was returned to the normoxic incubator whilst the other group were transferred to a hypoxic incubator (Wolf Laboratories – Galaxy CO₂ Galaxy incubator with 2-99% oxygen control) at an atmosphere of 5% CO₂ (550 bar), 10% O₂ balance N₂ (300 bar). Cells were initially maintained in the hypoxic incubator for 24 hours – 14 days to determine the optimal time point to investigate signalling. At all times, the hypoxic cells were treated identically to the control cells.

2.1.1.4 Treatment of hPASMC with Agonists/Inhibitors

Prior to treating hPASMC with various drugs, the cells were quiesced for 24 hours by replacing the growth medium with SmBm.

2.1.2 Transfection of hPASMC with the Amaxa Nucleofactor

hPASMC were transfected using a nucleofector and human aortic smooth muscle cell nucleofector kit (Amaxa) according to manufacturer's instructions. Briefly, hPASMC were trypsinsed as in section 2.1.1.2 and 1×10^6 cells/transfection were resuspended in 100µl nucleofector solution. This was placed into a cuvette and inserted into the nucleofector that was set at programme U-25. Once the programme was finished, the cells were gently transferred into 3 wells of a 6-well culture plate containing SmGm and left at 37°C in an atmosphere of 5% CO₂ for 4-24 hours.

2.1.2.1 Transfection of siRNA constructs

Using the nucleofector as described in section 2.1.2, siRNA samples were transfected into hPASMC. The siRNA samples were diluted to 60nM prior to use and after nucleofection, the cells were gently transferred into 3 wells of a 6-well culture plate containing SmGm and left at 37° C in an atmosphere of 5% CO₂ for 24 hours.
2.2 Proliferation Analysis

2.2.1 DNA Synthesis Assays

2.2.1.1 Assay preparation

hPASMC were plated out into 96-well tissue culture plates at a seeding density of 1000 cells/well 5 days prior to the DNA synthesis assay. Cells were quiesced in SmBm for 24 hours prior to treatments. Smooth muscle growth medium containing 5% FCS was reintroduced after this time.

2.2.1.2 Addition of [⁸H]-Thymidine

Twenty hours after additions, 0.5μ Ci/well [³H]-Thymidine (Amersham) was added to the plates. The cells were placed in their respective incubators for a further four hours to allow the tritiated Thymidine to incorporate before the plates were placed at -20°C to stop proliferation.

2.2.1.3 Measuring radioactivity

The 96-well tissue culture plates were allowed to thaw at room temperature for two hours before using a Betaplate 96-well harvester (Wallac) to harvest the plates onto glass fibre filter mats (Wallac). These were then placed into sample bags (Wallac) before being placed into a 1205 Betaplate Liquid Scintillation Counter (Wallac) and [³H]-Thymidine incorporation, expressed as counts per minute (cpm), was measured.

2.3 Biochemical Analysis

2.3.1 Harvesting Cell Lysate

2.3.1.1 Whole cell lysate production

Cells were harvested on ice and all buffers were pre-chilled. Cell media was removed from plates and the cells were rinsed in ice-cold PBS. The cells were drained before scraping into a 1.5ml eppendorf in 350µls / 100mm plate complete KHEM buffer (50mM KCl, 10mM EGTA, 1.92mM EGTA, 59mM HEPES-KOH pH 7.2 plus complete, EDTA-free protease inhibitor cocktail tablets (Roche) and 1% Triton X-100. The cells were homogenized by passing them through a 26 gauge needle 12 times whilst on ice. The cells were then centrifuged at 13000rpm at 4°C for 3 minutes (Biofuge fresco, Heraeus) to remove cellullar debris. The supernatant was then removed, aliquoted and used for analysis or snap frozen in dry ice before being stored at -80°C.

2.3.2 Determination of protein concentration (Bradford assay)

Protein assays were carried out in 96 well microtitre plates. $2\mu 1$ of sample was added to $48\mu 1$ dH₂O and 200 $\mu 1$ Bio-rad reagent (diluted 1/5 with distilled water). Each sample was analysed in triplicate. Protein concentration of samples was calculated by measuring the absorbances and constructing a standard curve using known concentrations of Bovine Serum Albumin (BSA). This was achieved by using the Revelation package on the computer, connected to an MRX microtitre plate reader, which read absorbance at a wavelength of 590nm. Protein concentrations were determined by plotting the standard curve and using least squared regression analysis to obtain the line of best fit. The equation of the line was used to determine the protein concentration of the samples.

2.3.3 SDS-PAGE Electrophoresis

2.3.3.1 Sample preparation for SDS-PAGE electrophoresis

Lysates were obtained from 100mm dishes scraped into 350µl of cKHEM + 1% Triton X-100 buffer and spun at 13000 rpm for 3 mins at 4°C. The supernatant was kept and used for protein measurement before boiling in Hannah sample buffer (260 mM Tris/HCl (pH6.7), 55.5% Glycerol, 8.8% SDS, 0.007% Bromophenol blue, 11.1% 2-mercaptoethanol, Laemmli et al., 1970), for 5 minutes to denature the proteins.

2.3.3.2 Casting and running an acrylamide gel

When using over 50µg protein, a Bio-Rad Protean II system was used. 16cm thick plates were rinsed with distilled water and ethanol prior to use. The apparatus was assembled according to the manufacturer's instructions. A resolving gel containing the appropriate percentage of acrylamide (determined by the molecular weight of the protein of interest), usually 10% unless otherwise stated, was cast between the two plates of the gel apparatus (10% 29:1 acrylamide:N.N'-methylenebisacrylamide mix, 375mM Tris/HCl (pH8.8), 0.1% SDS, 0,1% Ammonium persulphate, 0.06% N,N,N',N',-tetramethylethylenediamine (TEMED)). This was overlaid with water to ensure an even gel and allowed to polymerise at room temperature for one hour. The water was then removed and a 5% stacking gel was poured, (5% 29 :1 acrylamide:N,N'-methylenebisacrylamide mix, 125mM Tris/HCl (pH6.8), 0.1% SDS, 0.1% Ammonium persulphate, 0.1% TEMED). A comb was carefully inserted immediately between the plates. The stacking gel was allowed to polymerise for 30 minutes before the comb was removed and the wells were washed with tank buffer (192mM Glycine, 25mM Tris, 0.15% SDS) to remove any unpolymerised acrylamide. The gels were then placed in a running tank containing tank buffer in both upper and lower reservoirs. Bio-Rad prestained broad range precision protein markers were loaded into the

first well to allow determination of protein weight by gel electrophoresis migration. Prepared samples were loaded into the wells and the gels were run at the appropriate current (8mA overnight) until the dye front reached the bottom of the gel.

2.3.3.3 Running a NuPAGE gel

When using less than 50µg protein, the Novex NuPAGE gel systems were used with Invitrogen pre-cast 4-12% Tris-Glycine gels. Samples were prepared as in 2.3.3.1. Tank buffer used was either MOPS buffer or MES buffer depending on the size of the protein wishing to visualize. Gels were run for 45 minutes at 200V.

2.3.4 Western Blotting

2.3.4.1 Protein transfer from acrylamide gel to nitrocellulose membrane

Following electrophoresis, the gels were placed into a transfer cassette. Under transfer buffer (192mM Glycine, 25mM Tris, 20 % Methanol), a piece of foam was overlaid with a piece of Whatmann 3MM filter paper. The gel was placed on top of this and overlaid with a piece of Protran nitrocellulose paper (Schleicher & Schuell). The nitrocellulose was covered with another piece of Whatmann filter paper and finally another piece of foam. During assembly, care was taken to ensure air bubbles were excluded from all layers. The cassette was closed and placed into a Hoefer transfer tank filled with transfer buffer with the nitrocellulose side of the cassette to the positive electrode. The proteins were transferred for 0.06 amps overnight or 0.6 amps for 1.5 hours.

2.3.4.2 Blocking of nitrocellulose

Once the proteins had been transferred onto the nitrocellulose as in section 2.3.4.1., the membrane was washed with distilled water followed by visualisation of the transferred proteins with Ponceau S stain (0.1% Ponceau S, 3% Trichloroaceatic acid). The Ponceau S stain was added to the nitrocellulose for a few minutes until the protein became stained, the nitrocellulose membrane was then washed with water and rinsed with Tris-buffered saline (TBS-tween 20) (137mM NaCl, 20mM Tris/HCl (pH7.6), 0.1% tween20) before immunological detection of protein. The unoccupied protein binding sites on the nitrocellulose were blocked with 5% skimmed milk powder (Marvel) in TBS-tween20 for 1 hour at room temperature with gentle agitation.

2.3.4.3 Immunoblotting

After blocking, the appropriate primary antibody was added at the relevant dilution (see table 2.1) in 2% skimmed milk powder in TBS-tween20. This incubation was carried out for 1-2h at room temperature or overnight at 4°C with vigorous agitation. The membrane

was washed three times in TBS-tween20 for 5 minutes before being incubated with the appropriate diluted secondary antibody mix for 1 hour at room temperature with vigorous agitation. The secondary antibody (Sigma) was a horse-radish peroxidase (HRP) conjugated anti-immunoglobulin (IgG) antibody directed against the primary antibody. This antibody was diluted 1:5000 in 2 % skimmed milk powder in TBS-tween20. The nitrocellulose was finally washed 3-5 times in TBS-tween20 for 5 minutes. To detect immunoreactive bands, the membrane was incubated with ECL reagents (Amersham) for a minute with gentle rocking. A piece of x-ray film (Kodak) was exposed to the membrane in a darkroom for the appropriate amount of time and then developed using a Kodak X-omat.

2.3.5 Phosphodiesterase 4 Activity Assay

Phosphodiesterase activity was measured using a modification of the two step procedure of Thompson and Appleman [Thompson and Appleman., 1971], as described previously by Marchmont and Houslay [Marchmont and Houslay, 1980]. The first step is the hydrolysis of the ³H-cyclic nucleotide (8 position of the adenine or guanine ring) by the phosphodiesterases, which generates labelled nucleotide monophosphate. In the second step, incubation with snake venom, (which has 5'-nucleotidase activity), converts the mono-phosphate ring to the corresponding labelled nucleoside. The unhydrolysed cyclic nucleotide was separated from the nucleoside by batch binding of the mixture to Dowex IX8-400 anion exchange resin. This binds only the charged nucleotides and leaves behind the uncharged nucleosides.

2.3.5.1 Activation of Dowex

To activate the Dowex 1X8 -400, 41 of 1M NaOH was added to 400g of the resin, the mixture was stirred for 15 min at room temperature. The resin was allowed to settle and the supernatant was poured off. The resin was then washed 30 times with 41 distilled water. The resin was then resuspended in 41 1M HCl and stirred for 15 min at room temperature before being allowed to settle by gravity. The resin was then washed 5 times with 41 distilled water until required.

2.3.5.2 Sample preparation for assay

Into 1.5ml eppendorf tubes an appropriate amount of cell lysate was placed (30-60 μ g protein depending on the concentration of the sample) and made up to a volume of 50 μ l with PDE assay diluting buffer (20mM Tris/HCl, pH7.4). Each sample tube was repeated in triplicate. All tubes were set up on ice, and remained on ice until all components for the

assay had been added. To each tube 50µl of 2µM cAMP containing 3µCi [³H]cAMP in 20mM Tris/HCl /10mM MgCl₂ pH7.4 was added, the tubes vortexed and incubated at 30°C for 10 min. After this time, the tubes were boiled for 2 min to inactivate any PDE present and then cooled on ice for 15 minutes. 25µl of Jmg/ml snake venom in 20mM Tris/HCl, pH 7.4 was added to each tube, mixed by vortexing and incubated at 30°C for 10 minutes. The tubes were then cooled on ice for 15 minutes and 400µl Dowex/ethanol/water, in a 1:1:1 ratio, was added to the tubes, vortexed and incubated on ice for at least 20 min. The tubes were vortexed again, the dowex removed by centrifugation at 13000 rpm for 3 min (Heraeus bench top centrifuge) and 150µl of the clear supernatant added to 1ml Opti-scint scintillation fluid and counted on a Wallac 1409 liquid scintillation counter.

2.3.5.3 Determination of PDE3 and PDE4 activity

The PDE3 family is specifically inhibited by the drug cilostamide (Hidaka et al., 1979) and the PDE4 family by rolipram (Wachtel, 1982). PDE inhibitors were dissolved in 100% DMSO as 10mM stocks and diluted in PDE assay dilution buffer for use in assay. The residual levels of DMSO do not affect PDE activity at the concentrations used (Spence et al., 1995). Measurement of PDE activity with and without cilostamide (10 μ M) and with and without rolipram (10 μ M) present gave the contribution of PDE3 and PDE4 respectively.

2.3.6 Immunoprecipitation

2.3.6.1 Pre-clearing agarose beads

Protein G beads were used when the target protein was immuno-precipitated with a monoclonal antibody or a polyclonal antibody raised in a sheep and protein A beads were used when the target protein was immuno-precipitated with any other polyclonal antibodies. 25 μ l of the appropriate beads were washed in 200 μ l cKHEM buffer and recovered by centrifugation at 13000 rpm for 2 min at 4 °C in a refrigerated bench-top centrifuge. 300 μ g of protein from cell lysate, diluted to a total of 500 μ l with lysis buffer supplemented with protease inhibitors, was added to the washed beads and incubated with end-over-end rotation at 4 °C, for 30 min.

2.3.6.2 Binding target protein to antibody

After pre-clearing the beads were collected by centrifugation at 13000 rpm at 4 °C for 2 min (Heraeus refrigerated bench-top centrifuge), the supernatant was removed and placed

into a fresh eppendorf tube. To this fresh tube 7 μ l of the polyelonal antibody or 3 μ l monoclonal antibody was added and the tube rotated end-over-end at 4 °C for at least 2 h, to enable the protein and antibody to bind. The solution was then transferred into a tube containing 60 μ l pre-washed protein beads (A or G as required) and rotated end-over-end overnight at 4 °C. The beads were isolated from solution by centrifugation at 13000 rpm at 4 °C for 2 min (Heraeus refrigerated bench-top centrifuge) and washed to remove any non-specifically bound protein. The beads were washed firstly in lysis buffer supplemented with 500 mM NaCl, secondly with lysis buffer supplemented with 0.1 % NP-40 in 10 mM Tris pH 7.4. The final wash was carried out in 20 mM Tris (pH 7.4) to prepare samples for use in a PDE assay.

2.3.7 Cyclic AMP Assay

Intracellular cAMP levels were measured using a modification of the procedure of Savage. (Savage, 1995). The assay involves the measurement of cAMP by the use of the cAMPbinding protein, cyclic AMP-dependant protein kinase. (Brown, 1972) Briefly, following stimulation of the cells, they were lysed to release the cytosolic cylic uncleotides. After neutralization, the lysate was incubated with [5', 8' $-^{3}$ H] cyclic AMP and cyclic AMP binding protein, allowing competition of labelled and unlabelled cyclic AMP for a limited number of binding sites on the binding protein. Activated charcoal was then added to the sample to bind any free cyclic nucleotide and the charcoal pelleted by centrifugation. The radioactivity of the supernatant was then measured. A standard curve was generated by incubating a range of known concentrations of cAMP with the fixed amounts of binding protein and radioactive cyclic AMP. It was then possible to determine cAMP concentrations from the radioactivity counted by reference to the standard curve.

2.3.7.1 Agonist Treatment

Cells were plated out in 6-well plates at the optical seeding density of 3500cells/cm² four days prior to treatment. Cells were serum starved overnight prior to agonist stimulation. Agonists were usually made to a stock solution of 10mM in 100% DMSO and were diluted in SmBm to a desired concentration prior to addition to cells. Each treatment was performed in triplicate in each assay.

2.3.7.2 Lysate extraction for assay

To halt agonist treatment, the media was aspirated from the cells and 100µl of 2% Perchloric acid was added directly onto the cells. The media was kept to be used for measuring extracellular cAMP levels. Following incubation on ice for 15 minutes, the cells were scraped into eppendorfs and the precipitated protein pelleted by centrifuging the

samples in a microfuge at 13000rpm for 3 mins. Two wells from both normoxic and hypoxic plates were lysed into 100μ l cKHEM + 1% Triton X-100 as in section 2.3.3.1. to be used for protein quantification.

2.3.7.3 Neutralisation of the sample

The supernatant from section 2.3.6.2 was added to 5μ l universal indicator solution (BDH) and it turned pink. The samples were neutralised by gradual addition of 2M KOH, 0.5M Triethanolamine until the solution turned light green, indicating a pH of 7-7.5. Centrifugation at 13000rpm for 3 minutes pelleted the potassium perchlorate precipitate.

2.3.7.4 Reagent preparation for assay

cAMP assay buffer: 50mM Tris/4mM EDTA kept at 4°C.

The following solutions were prepared fresh on the day of the assay.

cAMP Standards: Stock of 1mM cAMP in cAMP assay buffer diluted 1:100 and an initial dilution of 32pmol/50µ1 generated. A series of two-fold dilutions from this concentrated standard generated the dilutions used in the assay.

[5', 8' - ³ H] cyclic AMP:	8.7μ1 [5', 8' - ³ H] cAMP
	10mls cAMP assay buffer.

Binding Protein :	10mg Bovine Crude Fraction	
	125mg BSA	
	25mls cAMP assay buffer.	

Diluted 1:5 in cAMP assay buffer for use on day of assay.

Charcoal Solution:	2% w/v activated charcoal
	1% w/v B S A
	30mls cAMP assay buffer.

Charcoal solution was stirred on ice for 20 minutes prior to use in assay.

2.3.7.5 Sample preparation for assay

A standard curve was obtained for each assay by including a set of tubes containing known quantities of cAMP ranging from 0.00625 and 16pmol. The standards tubes were set up in duplicate on ice with the unknown samples in the order as follows (with binding protein being added last):

Tube No	cAMP (pmol/50µl)	Buffer (µl)	[³ H]-cAMP (μl)	Binding Protein (µl)
1-2	Blank	200	100	0
3-4	Zero	100	100	100
5-6	0.0625	50	100	100
7-8	0.125	50	100	100
9-10	0.25	50	100	100
11-12	0.5	50	100	100
13-14	1	50	100	100
15-16	2	50	100	100
17-18	4	50	100	100
19-20	8	50	100	100
21-22	16	50	100	100
23 onwards	Unknown	50	100	100

The first two tubes with no binding protein were included as a blank to determine the amount of cAMP that remained after nucleotide binding by charcoal. Tubes 3 and 4 indicated the maximum cAMP bound in the absence of unlabelled cAMP and tubes 5 to 22 were prepared with known concentrations of unlabelled cAMP to aid in the construction of a standard curve to allow calculation of unknown cAMP values.

Tubes 23 onwards were set up as shown using samples prepared as in 2.3.6.1 or media from the cells. The tubes were vortexed and incubated on ice for 2 hours. After this time, 25µl charcoal solution was added to each tube. The tubes were vortexed and centrifuged at 13000rpm for 3 minutes. 300µl supernatant from each tube was added to 1ml Opti-scint scintillation fluid, vortexed until clear and counted on a Wallac 1409 liquid scintillation counter.

2.3.7.6 Construction of standard curve

Microsoft Excel was used to plot standards (pmol cAMP vs 1/DPM-mean blank) and generate a formula for the best fit curve.

2.3.7.7 Calculation of cAMP levels

After generating the best fit formula for the standard curve, this was used with the unknown samples to calculate their cAMP concentrations. Results were calculated as pmol cAMP/ng protein and expressed as mean percentage normoxic basal \pm S.E., with normoxic basal = 100%.

2.3.8 PKA Assay

2.3.8.1 Cell lysate extraction for PKA assay

Cells were grown in both normoxia and hypoxia for seven days, after which time the cell media was removed, the cells washed with PBS and drained thoroughly. The monolayer of cells was scraped into 500µl extraction buffer (5mM EDTA, 50mM Tris, pH 7.5) and homogenized by drawing through a 13 gauge needle 12 times. The cell debris was then removed by centrifugation for 2 mins at 13000 rpm at 4°C (Heraeus refrigerated bench-top centrifuge) and the supernatant was used for a PKA assay.

2.3.8.2 PKA assay tube pre-incubation

For each cell lysate, the following were set up on ice in 1.5 ml eppendorf tubes. The cell extract was added last and the samples were incubated on ice for 20 min to allow the inhibitor to bind PKA.

Tube	Cell Extract	Diluent	4x PKA Inhibitor	4x PKA Activator
A	10µ1	20µ1	0µ1	Οµ1
В	10µ1	10µ1	10µ1	0u]
C	10µ1	10µl	0µ1	10µ1
D	10µ1	0µ1	10µ1	10µ1

PKA Assay Diluent: 50mM Tris, pH7.5.

4 x PKA Inhibitor: 4µM PKI(6 -22) amide, 50mM Tris, pH7.5.

4 x PKA Activator: 40µM cAMP, 50mM Tris, pH7.5.

4 x PKA substrate: 200μM Kemptide, 400μM ATP, 40mM MgCl₂, 1 mg/ml BSA, 50mM Tris, pH7.5.

2.3.8.3 PKA assay reaction

To 1ml of the 4 x PKA substrate, 6000 μ Ci/mmol of [γ -³²P]ATP was added whilst maintained on ice. To the first assay tube, (1A), 10µl of radioactive 4 x PKA substrate solution was added. The sample was mixed gently before being placed in a waterbath at 30°C for 10 minutes. 20 seconds after substrate addition to the first tube, 10 µl of the radioactive substrate was added to the next tube, (1B), which was then mixed and incubated. This method of addition of substrate to tubes every 20 seconds continued until substrate had been added to all samples and they were incubating at 30 °C. After the 10 min incubation of the first tube, 2011 of the reaction mix was removed and spotted onto a pre-marked piece of ion exchange phosphocellulose paper P81 (Whatman). This was carried out for all tubes after they had each undergone the 10 min incubation with the substrate. The phosphocellulose pieces were then placed into a large beaker containing 1 %(v/v) phosphoric acid (H₃PO₄) and washed for 3 min with slight agitation. The waste acid was removed and the acid wash repeated. The phosphocellulose was then washed twice in dH₂O before being placed in 1.5 ml eppendorf tubes, to which 1 ml scintillation fluid was added. The ³²P incorporated into the peptide bound to the phosphocellulose was counted on the Wallac 1409 liquid scintillation counter. Two separate vials were set up with 10µl of the radioactive 4 x PKA substrate added to 1ml scintillation fluid to enable the determination of the total counts from the substrate solution.

2.3.8.4 Calculations for PKA assay data

<u>Step I.</u>

 $\frac{Total \ counts}{4nmolATP} x \frac{1nmol}{1000 \ pmol} = cpm/pmol \ phosphate$

Step 2.

 $\frac{(cpm/phosphocellulose)x2}{STEP1} = total pmol peptide - incorporated phosphate$

<u>Step 3.</u>

 $\frac{STEP \ 2}{10 \min} = pmol/\min / assay \ tube$

Step 4.

 $\frac{STEP \ 3}{0.01ml \ extract} = pmol/\min/ml$

<u>Step 5.</u>

Tube A – Tube B = pmol/min activated PKA

Tube C - Tube D = pmol/min total PKA

<u>Step 6.</u>

 $\frac{pmol/\min\ activated\ PKA}{pmol/\min\ total\ PKA} x100 = \%\ activated\ PKA$

2.4 Molecular Biology

To prevent any contamination all glassware, tubes, tips, buffers and media used for molecular biology were sterilised.

2.4.1 DNA Production

2.4.1.1 Small scale production of DNA

5mi LB growth media (580mM NaCl, 0.5 % (w/v) BactoYeast Extract, 1 % (w/v) Bacto-Tryptone pH7.5) supplemented with antibiotics in a sterile 25 ml universal tube was spiked with a pipette tip from a glycerol stock of transformed cells containing the DNA of interest. The culture was incubated at 37 °C overnight with agitation and the cells harvested the next day by centrifugation at 3000 rpm for 5 min (Heraeus bench-top refrigerated centrifuge). The DNA was extracted from the cells using the QIAprep spin miniprep kit according to the manufacturer's instructions. Briefly: The bacterial pellet was resuspended in 250µl Buffer P1 (resuspension buffer), to which 250µl Buffer P2 (lysis buffer) was added. The solutions were mixed and incubated at room temperature for 5 minutes. 350µl of Buffer N3 (neutralisation buffer) was then mixed with the lysed cells and the solution was centrifuged for 10 min at 13000 rpm. The supernatant was added to the QIAprep spin column and centrifuged at 13000 rpm for 1 min. The flow through was discarded and the column washed with 750µl Buffer PE. The column was centrifuged for 1 min prior to the elution of the bound DNA to ensure removal of all buffer. 50µl dH₂O was placed onto the resin in the column and incubated for 5 minutes and the DNA was eluted

from the column by centrifugation for 1 min at 13000 rpm (Heraeus bench-top refrigerated centrifuge).

2.4.1.2 Large scale production of DNA

400ml LB growth media supplemented with antibiotics in a 21 flask was spiked with a pipette tip from a glycerol stock of transformed cells containing the DNA of interest. The culture was incubated at 37°C overnight with agitation and harvested the next day by centrifugation at 5000g for 10 minutes using the JA-14 rotor in the Beckman refrigerated contrifuge. The DNA was extracted from the cell pellet using the Promega Wizard Maxiprep kit according to the manufacturer's instructions. Briefly: The bacterial pellet was resuspended in 15ml resuspension solution, to which 15ml lysis buffer was added. The solutions were mixed gently and inverted for 15 minutes, 15ml of neutralisation buffer was then added to the lysed cells and the mixture was gently inverted three times. The cell lysate was clarified by centrifugation of the solution at 13000 g for 15 minutes at room temperature and then passed through muslin to separate out clumps. 0.6 x volume Isopropanol was added and mixed before centrifugation at 13000g for 1 hour at room temperature. The DNA pellet was then washed and resuspended in 2mls TE buffer. 10mls DNA purification resin was added and mixed with the resuspended DNA pellet. The resin/DNA mix was then passed through a Maxicolumn by attached to a vacuum manifold and applying a vacuum. To wash the columns, 25mls column wash solution was added and drawn through by vacuum. The resin was rinsed with 5mls of 80% ethanol. To ensure all ethanol was removed, the maxicolumn was centrifuged at 1300g for 5 minutes and the resin was then dried by applying a vacuum for 5 minutes. Following this, 1.5mls preheated (70°C) nuclease-free water was added to the maxicolumn to elute the DNA into a 50ml screw cap tube. The water was left on the column for 5 minutes at room temperature before centrifugation at 1300g for 5 minutes.

2.4.1.3. Quantification of DNA & RNA

DNA and RNA concentrations were quantified by a WPA Lightwave spectrophotometer blanked with distilled water. 5µi DNA or RNA was diluted to 1ml with distilled water and absorbance measurements were taken at 260nM and 280nM. The concentration of nucleic acid was then calculated using the following approximations:

An absorbance reading of 1 at 260 nM corresponds to; 50µg/ml double stranded DNA 40µg/ml single stranded RNA Therefore, [DNA] $(\mu g/ml) = A_{260} \times 50 \times 200$ (Dilution Factor) [RNA] $(\mu g/ml) = A_{260} \times 40 \times 200$ (Dilution Factor).

The ratio between the absorbance measurements at 260 μ M and 280 μ M provided an indication of the purity of the nucleic acid. In solution, pure DNA or RNA typically have A₂₆₀:A₂₈₀ ratios of between 1.8 and 2.

2.4.2 Glycerol Stock Production

A 500µl sample from an overnight culture of transformed cells was taken aseptically and placed into a sterile screw top cryovial to which 500µl sterile 80% glycerol was added. The sample was mixed well and stored at --80°C for further use.

2.4.3 Reverse Transcription PCR

2.4.3.1 Total RNA extraction

For isolation of total RNA from hPASMC, a Qiagen RNeasy kit was used. Cells were trypsinised as described in section 2.1.1.2, pelleted and the medium aspirated. The cell pellet was then either frozen at -80°C for use at a later date or resuspended in 600 μ I Buffer RLT containing 10 μ I/ml of 14.5M β -Mercaptoethanol. Lysates were then loaded onto QIA shredders to ensure maximum homogenization of sample and the manufacturer's instructions for the kit were followed.

In order to prevent potential contamination from genomic DNA, an incubation step with 4 units DNase at room temperature for 15 minutes was included. The clean up protocol was also used to ensure maximal purity. Total RNA was eluted in RNase free H₂O and stored at -20° C.

To determine the concentration and purity of RNA, the absorbency of the sample was measured at 260nm and 280nm in a spectrophotometer. (WPA, Lightwave).

2.4.3.2 cDNA synthesis

First strand synthesis was carried out in each reaction using 2µg total RNA catalysed by the enzyme Superscript II reverse transcriptase (Invitrogen). The reaction was primed using 500ng of Oligo(dT)15 (Promega), in a final volume of 20µl. The reverse transcription mixture contained 1µl of 10mM stock dNTP mix, 4µl 5X First strand buffer, 2µl 0.1M DTT, 40 units of RNasin Ribonuclease inhibitor and 200 units of Superscript II RNase H⁻ Reverse Transcriptase or 1µl H₂O(molecular biology grade) as a negative control for PCR. The mixture was incubated at 42°C for 50 minutes before heating to 70°C for 15 minutes to terminate the reaction. Control reactions were carried out without enzyme.

2.4.3.3 PCR reaction

The PCR reaction was set up using Platinum Pfx DNA Polymerase (Invitrogen). 1µg cDNA (generated in 2.4.3.2) was added to 2.5µl of 10X Pfx amplification buffer, 0.75µl of 10mM dNTPs, 0.5µl 50mM MgSO₄, 0.75µl 10µM of each of sense and antisense primers for detection of the transcript of interest, 0.75µl 10µM of sense and antisense primers for cyclophylin, 0.25µl of Platinum Pfx DNA Polymerase and made up to a total volume of 25µl with RNase- free water. Tubes were also set up as blanks using the control reactions without enzyme from section 2.4.3.2.

The PCR conditions used were as follows:

95°C for 2 minutes,
94°C for 30 secs,
*58/60°C for 30 secs,
40 cycles
72°C for 1 minute,
72°C for 10 minutes,
4°C forever.

*The annealing temperature for 4D5 was 58°C whereas it was 60°C for all other isoforms.

2.4.3.4 Semi-Quantitative RT-PCR

The PCR reaction was set out as in section 2.4.3.2 in triplicate. One tube per sample was removed after 30 cycles, one after 35 cycles and another was left to undergo another 5 cycles. To ensure even comparison, there was no final extension step of 72°C for 10 minutes.

2.3.4.5 Visualisation of PCR Products

DNA was visualised using agarose gel electrophoresis, the percentage of agarose used in the gel was dependent on the size of DNA fragment to be identified, usually 2 %. For a 2 % agarose gel 2 % agarose was dissolved in 1 x TBE (45mM Tris/IICl, 1mM EDTA, 0.9M Boric Acid) by heating until the agarose dissolved. To this, 0.01 % ethidium bromide was added, which enabled visualisation of the DNA under a UV light source. The molten agar was poured into the gel apparatus, set up according to the manufacturer's instructions and allowed to set completely. The comb and end stoppers were removed, the gel tank filled with 1x TBE and the samples loaded into the lanes. All DNA samples were diluted 6:1 in 6x sample buffer (0.25 % Bromophenol Blue, 0.25 % xylene cyanol blue, 30 % glycerol in H_2O). In order to size the DNA fragments, the molecular size marker XIV (Promega) was loaded alongside the samples. The gel was run at 100 volts until the dye front moved sufficiently through the gel. The gel was removed from the tank and the DNA observed under UV light.

The range of separation of linear DNA molecules according to agarose concentration;

Percentage Gel	Size of Fragment (kb)
0.9	0.5 -7
1.2	0.4 -6
1.5	0.2 -3
2	0.1 -2

2.5 Microarray Analysis

2.5.1 RNA Extraction

2.5.1.1 Extraction of total RNA

Total RNA from both normoxic and hypoxic samples was isolated as in section 2.4.3.1 using the Qiagen RNeasy kit. RNA concentration was determined and samples were stored at '80°C until needed.

2.5.1.2 Clean up of total RNA

RNA was cleaned up following the add-on protocol of the Qiagen RNeasy kit to increase purity and remove potential contaminants.

2.5.2 Incorporation of Fluorophores

2.5.2.1 cDNA synthesis

Two reaction tubes were set up together, one with either normoxic or hypoxic RNA and Cy3 or Cy5 –dCTP. cDNA was synthesised using $25\mu g$ of total RNA mixed with $12\mu g$ random hexamers and incubated at 70°C for 10 minutes to allow the primers to anneal. First strand cDNA synthesis was performed using 400U Superscript II (Invitrogen) and 40 μ M of one of Cy-3 or Cy-5. This reaction was mixed and incubated at 25°C for 10 minutes then 42°C for five hours before stopping the reaction then cleaning up the cDNA using a QIAquick column.

2.5.3.1 Incubation of microarray with sample

Both sets of cDNA were mixed together with a hybridisation buffer and incubated at 95°C for 3 minutes before being placed on ice for 1 minute. This mixture was then placed on the microarray and covered with a coverslip. The microarray was then rotated gently at 42°C for 24 hours.

2.5.3.2 Washing of microarray

The microarray was washed gently with SSC buffer three times by being placed inside a 50ml centrifuge tube with buffer and placed horizontal on a rotator set at the lowest setting. The microarray was then dried by brief centrifugation.

2.5.4 Analysis

Analysis of microarray was carried out by Dr. Ian McPhee. A Genetic Microsystems 418 array scanner was used to obtain the data with Imagene 5.0 to determine the values. Gene sight light 3.2 was used to normalise the data and calculate ratios.

2.6 Statistical Analyses

Data was analysed by two-way ANOVA with Bonferroni's post test using GraphPad Prism version 4.00 for Windows, GraphPad Software, San Diego California USA, to analyse raw data. Where only two groups of data were to be analysed, an unpaired t-test was performed, and *p<0.05 was considered to be significant.

Antibody	Company	Dilution	Incubation Time	Secondary
PDE4A,B,C,D	ICOS	1:10000	1 hour	Anti-mouse IgG
PDE4A,B,C,D	SAPU	1:5000	1 hour	Anti-goat IgG
4A4	Genosys	1:5000	2 hours	Anti-rabbit IgG
4A10	Genosys	1:1000	2 hours	Anti-rabbit IgG
pERK1/2	Cell Signalling	1:1000	Overnight	Anti-mouse IgG
ERK1/2	Cell Signalling	1:1000	Overnight	Anti-mouse IgG
ΡΚΑ RΙα	BD Transduction	1:1000	1 hour	Anti-mouse IgG
PKA RΠα	BD Transduction	1:1000	1 hour	Anti-mouse IgG
ΡΚΑ RIIβ	BD Transduction	1:2000	1 hour	Anti-mouse IgG
α - tubulin	Sigma	1:10000	1 hour	Anti-mouse IgG

Table 2.1 Antibodies and Dilutions

This table shows the antibodies used for western blotting throughout this study and the conditions determined to get an optimal immunoreactive signal.

Chapter 3

Effect of Chronic Hypoxia on PDE4 in hPASMC

3.1 Introduction

3.1.1 Pulmonary Arterial Hypertension

3.1.1.1 Pulmonary arterial hypertension

Pulmonary arterial hypertension (PAH) is characterised by increased vascular resistance which leads to pulmonary artery remodelling and increased smooth muscle cell proliferation (Fishman 2004; Humbert et al., 2004). Normal mean adult pulmonary arterial pressure is ~14mm Hg whereas patients with pulmonary hypertension have a mean PAP exceeding 25mmHg at rest and 30mmHg during exercise. Pulmonary hypertension can occur without apparent cause, previously known as Primary Pulmonary Arterial Hypertension (PPAH) It is now known that PAH can be due to mutations in the bone morphogenetic protein receptor II, BMPR2, or 5-HT transporter gene (The International PPH Consortium et al., 2000; Thomson et al., 2000; Machado et al., 2001; Eddahibi et al., 2003). Hence these forms have been classified as familial pulmonary arterial hypertension (FPAH; table 1.5). These cases only account for 6% of all PPAH patients. The occurrence of PPAH in the population is rare, with only two cases per million. PPAH can occur in men, women and children of all ages though it is most commonly seen in females between 20 and 40 years old. PAH occurs more often secondary to pulmonary vascular disorders, such as chronic obstructive pulmonary disorder (COPD), and exposure to hypoxic conditions. This form of PAH is now classified as PAH'PAH with lung disease and /or hypoxemia' and is relatively common but is underdiagnosed due to the large number of causes. As such, an estimate of the prevalence of secondary PAH is difficult to obtain. PAH is usually progressive and fatal.

There is currently no cure for PAH. Current treatments for PAH include anticoagulants, diuretics, calcium channel blockers and vasodilators. Vasodilators used include inhaled nitric oxide (NO), ACE inhibitors or prostacyclin. So far, only treatment with prostacyclin analogues such as iloprost have improved survival. Heart and/or lung transplantation is considered in extreme cases. Promising results have been found with sildenafil, a PDE5 inhibitor, which has been shown to significantly reduce pressure in severe PPAH patients (Michelakis et al., 2003; Prasad et al., 2000). It has even been shown to be more effective than inhaled NO in decreasing PAP and reducing pulmonary vascular resistance (Michelakis et al., 2002). The responses to medication are varied between patients so it can take time before a successful drug therapy is achieved. If a cure is to be found for PAH,

then the underlying cellular and molecular changes, along with the physiology of the disease, must be determined.

3.1.1.2 Hypoxia-induced Pulmonary Hypertension

Hypoxia causes high altitude-induced PAH and is also thought to play a role in PAH associated with COPD. The mechanism underlying this is currently unclear. Animal and cellular models have been utilised to investigate PAH, and to suggest possible treatment targets.

Chronic hypoxia can lead to the structural remodelling of pulmonary arteries. In contrast to systemic arteries, the pulmonary arteries are distinct in that they constrict in response to hypoxia (Voelkel & Tuder, 1997). This is required for ventilation-perfusion matching in which pulmonary capillary blood flow is automatically adjusted to match alveolar ventilation.

It is known that vasoconstriction is an intrinsic property of pulmonary vascular smooth muscle cells (Voelkel and Tuder 1997) and it has been shown that hypoxia can have a direct effect on smooth muscle cells. After 1 week of hypoxia, proliferation rates of PASMC from distal human pulmonary arteries were observed to increase (Yang et al 2002). The cells were also observed to contract in response to hypoxia (Murray 1990). Due to this, hypoxic PASMC have been used as a cellular model of PAH.

3.1.2 Cyclic Nucleotide Signalling in Models of Hypoxia-Induced Pulmonary Hypertension

3.1.2.1 Cyclic Nucleotides in Hypoxia

The second messengers' cAMP and cGMP have been shown to play vital roles in the regulation of vascular smooth muscle tone (Murray et al., 1990b). cAMP and cGMP relax smooth muscle through the activation of protein kinase A (PKA) and protein kinase G (PKG) respectively. PKA and PKG can induce relaxation by acting on downstream targets of Ca2+ mobilisation (Shabb 2001; Marin et al., 1998). Activation of these kinases is achieved by either stimulating adenylyl or guanylyl cyclase or through inhibiting the action of cyclic nucleotide phosphodiesterases (PDEs). These pathways have been of interest to researchers to uncover the mechanisms underlying PAH (MacLean et al., 1996). Investigation of the hypoxic Wistar rat model uncovered that the levels of both cyclic nucleotides are seen to decrease, with a corresponding increase in PDE activity, in response to two weeks of $10\% O_2$ (MacLean et al., 1997).

3.1.2.2 Phosphodiesterases in Hypoxia

In the rat pulmonary arterial circulation, the predominant cyclic nucleotide phosphodiesterases are from the PDE1, PDE3, PDE4 and PDE5 families (Rabe et al., 1994), although PDE2 is also present (MacLean et al., 1997).

Investigations into the expression of these PDEs in hypoxia have revealed that their profiles are altered. PDE activity was examined in pulmonary arteries from control and hypoxia-induced pulmonary hypertensive rats (MacLean et al., 1997). cAMP-PDE activity was increased in first-branch and intrapulmonary arteries from the hypoxic rats. cGMP-PDE activity was also increased in the first, intrapulmonary and main pulmonary arteries of the hypoxic rat. Investigation of these increases attributed it to an increase in PDE3, PDE1 and PDE5 activities (MacLean et al 1997).

Using human PASMC and rat pulmonary artery homogenates, it was discovered that PDH3 and PDE5 expression levels and activity were increased after two weeks exposure at 10% O_2 (Murray et al., 2002). The increase in the PDE3A isoform in hPASMCs was observed to be PKA-dependant and could be mimicked by cAMP analogs (Murray et al., 2002). There was also an increase in PDE3 and PDE5A2 mRNA and protein levels in pulmonary arteries from the chronic hypoxic rat model (Murray et al., 2002).

The role of PDE inhibitors as a therapy for PHT is under investigation. PDE inhibitors have been shown to reduce smooth muscle cell proliferation, migration and induce vasodilation of the pulmonary circulation (Koyama et al., 2001; Ghofrani et al., 2004; Wagner et al., 1997; Bardou et al., 2002; Goirand et al., 2001; Palmer et al., 1998; Osinski & Shror 2000). Zaprinast, rolipram, cilostamide and theophylline can all oppose pulmonary vasoconstriction seen in hypoxia (Pauvert et al., 2002, Goirand et al., 2001). PDE3 or 4 inhibitors can also improve responses to β -adrenergic agents that are attenuated in hypoxia. When both PDE3 and PDE4 inhibitors are used, they act synergistically to potentiate this effect (Wagner et al., 1997). Rolipram has been observed to be more potent than PDE3 or PDE5 inhibitors at relaxing human intralobar pulmonary arteries (Bardou et al., 2002).

3.1.2.3 PDE4 in Hypoxia

PDE4 is a cAMP-specific, rolipram-inhibited PDE that is known to be the major hydrolysing enzyme of cAMP in vascular smooth muscle cells. There are eighteen known human PDE4 isoforms with differing subcellular locations and modes of regulation (reviewed in Houslay 2001; further details in general introduction, sections 1.2.4 and 1.3). General PDE4 inhibitors are currently being tested as a putative therapy for such diseases as asthma and COPD (Giembycz 2001; Spina 2003). However, little is known about the effect hypoxia has on PDE4 isoforms in vascular smooth muscle cells. In this chapter, 1 set out to investigate PDE4 expression and activity in hPASMC under normoxic and hypoxic conditions.

Results

3.2 Characterisation of PDE4 Profile in hPASMC

All methods used in this chapter are described in full in section 2, materials and methods.

3.2.1 Analysis of PDE4 expression

To determine the PDE4 profile in hPASMC, western blotting was employed as described in section 2.3.4. PDE4A, 4B, 4C and PDE4D antibodies have been raised against peptide sequences from the C-terminal of each PDE4 family. This has generated subfamilyspecific isoforms as the C-terminal is only conserved among subfamily members, i.e. all PDE4A isoforms but not any PDE4B, PDE4C or PDE4D isoforms (Huston et al, 1996; Mackenzie et al., 1998). These PDE4 subfamily specific antibodies were used to probe hPASMC lysates to identify the PDE4 family isoforms present within these cells. The results gained were further specified and confirmed by RT-PCR analysis.

3.2.2. PDE4A profile in hPASMC

3.2.2.1 Western blot analysis of PDE4A isoforms in hPASMC

To identify whether PDE4A isoforms were expressed in hPASMC, cell lysates were immunoblotted with a 4A monoclonal antibody (Materials and Methods 2.3.4) raised against the sequence EEFVVAVSHSS, which is found at the C-termini of all known active human PDE4A isoforms. This detected a single PDE4 splice variant that co-migrated with recombinant PDE4A4 (Genbank accession L20965) and PDE4A10 (Genbank accession AF073745) isoforms, having an apparent Mwt of 120kDa when analysed by SDS-PAGE (Figure 3.1). PDE4A11 (Genbank accession AF069488) is also known to migrate at this weight. Thus, isoform specific antibodies were employed to identify this band. The PDE4A4 specific antibody did not react with hPASMC lysate indicating the absence of PDE4A4 in these cells. The specificity of this PDE4A4 specific antibody was confirmed by detecting PDE4A4 recombinant protein, but not PDE4A10 or PDE4A1. The specificity of the PDE4A10 antibody was confirmed by the detection of the PDE4A10 recombinant protein and not PDE4A4 or PDE4A1. This PDE4A10 specific antibody identified a band at 120kDa, confirming the presence of the PDE4A10 isoform in hPASMC. A PDE4A11 antibody has not yet been successfully generated so it was impossible to confirm the absence or presence of PDE4A11 by western blotting.

3.2.2.2 RT-PCR analysis of PDE4A isoforms in hPASMC

RT-PCR was employed to identify the120kDa immunoreactive species detected through western blotting (Materials and Methods 2.4.3). As the long human PDE4A isoforms, PDE4A4, PDE4A10 and PDE4A11, all migrate at this weight on an SDS-PAGE gel, isoform specific primers were used to investigate transcript expression of these three splice variants. Primers to PDE4A1 and PDE4A7 were also used to investigate if transcripts for these isoforms were present or not. This was because PDE4A7 is truncated at the C-terminus and there is no antibody available to detect it. Also, in the case of PDE4A1, it is possible that it could be expressed at such low levels that western blotting would be unable to detect it. RT-PCR analysis of total RNA identified the presence of PDE4A7, PDE4A10 and PDE4A11 transcripts in hPASMC but not those for PDE4A4 (*Figure 3.3*).

3.2.3 PDE4B profile in hPASMC

3.2.3.1 Western blot analysis of PDE4B isoforms in hPASMC

To identify if any PDE4B isoforms are expressed in hPASMC, cellular lysates were immunoblotted with a PDE4B specific polyclonal antibody (Materials and Methods 2.3.4) raised against the sequence DPENRDSLGETDIDIATED, which is found at the C-terminus of all known, active PDE4B isoforms. This antisera detected a single PDE4B splice variant that co-migrated with the recombinant PDE4B2 standard (Genbank accession L20971) This PDE4B2 standard recombinant protein migrated at 80 kDa on SDS-PAGE (*Figure 3.4*).

3.2.3.2 RT-PCR analysis of PDE4B isoforms in hPASMC

As a means of confirming the presence of the PDE4B2 isoform, RT-PCR was employed to identify transcripts for the single immunoreactive species observed at 80kDa through western blotting (Materials and Methods 2.4.3). Primers specific to PDE4B1, PDE4B2 and PDE4B3 were also used to investigate transcript expression of these three known human PDE4B splice variants. RT-PCR analysis of total RNA using such isoform specific primers identified the presence of PDE4B2 transcripts but failed to detect those for PDE4B1 and PDE4B3, thus confirming the absence of expression of the PDE4B1 and PDE4B3 hours in hPASMC (*Figure 3.6*).

3.2.4 PDE4C profile in hPASMC

3.2.4.1 Western blot analysis of PDE4C isoforms in hPASMC

To identify whether PDE4C is expressed in hPASMC, cellular lysates were immunoblotted (Materials and Methods 2.3.4) with a polyclonal PDE4C specific antibody, which detects a sequence located in the extreme C-termini of all known active PDE4C isoforms. This detected a single reactive species which co-migrated with recombinant PDE4C2 (Genbank accession U66346), which migrated with an apparent molecular weight of 80kDa on SDS-PAGE (*Figure 3.7*)

3.2.4.2 RT-PCR analysis of PDE4C isoforms in hPASMC

As a means of confirming the presence of the PDE4C2 isoform, RT-PCR was employed to identify the reactive species observed at 75kDa through western blotting (Materials and Methods 2.4.3). Primers specific to PDE4C2 were used to investigate transcript expression. RT-PCR analysis of total RNA using isoform specific primers identified the presence of PDE4C2 transcripts in hPASMC, thus confirming the western blot data that identified PDE4C2 expression (*Figure 3.8*).

3.2.5 PDE4D profile in hPASMC

3.2.5.1 Western blot analysis of PDE4D isoforms in hPASMC

To identify if any PDE4D isoforms are expressed in hPASMC, cellular lysates were immunoblotted with a PDE4D specific monoclonal antibody (Materials and Methods 2.3.4) raised to a peptide corresponding to the sequence; TQDSESTEIPLDEQVEE, located at the C-termini of all known PDE4D isoforms. This detected three immunoreactive bands. These co-migrated with recombinant PDE4D1/2 (Genbank accession U50157, U50158 respectively), PDE4D3 (Genbank accession L20970) and PDE4D5 (Genbank accession S:1059276), whose molecular weights are 68 kDa, 95kDa and 105kDa, respectively, on SDS-PAGE (*Figure 3.9*).

3.2.5.2 RT-PCR analysis of PDE4D isoforms in hPASMC

To confirm the presence of the PDE4D1/2, PDE4D3 and PDE4D5 in hPASMC, RT-PCR was employed to identify the reactive species observed through western blotting (Materials and Methods 2.4.3). Primers specific to PDE4D1, PDE4D2, PDE4D3 and PDE4D5 were used to identify transcript expression. Primers to PDE4D4 were also used to determine if this isoform was present at such low levels that western blotting would be unable to detect it. RT-PCR analysis of total RNA using isoform specific primers identified transcripts for

PDE4D1, PDF4D2, PDF4D3 and PDE4D5, thus confirming their expression in hPASMC, but failed to identify transcripts for PDE4D4, confirming the inability to detect this isoform immunologically in hPASMC (*Figure 3.11*).

3.2.6 cAMP-PDE activity in hPASMC

PDE3 and PDE4 are known to be the main cAMP hydrolysing enzymes expressed in vascular smooth muscle cells (Palmer et al., 1998). In order to determine how much of this is attributable to PDE4 within hPASMC, a cAMP-PDE activity analysis (Materials and Methods 2.3.5) was carried out in the presence and absence of either the PDE4 selective inhibitor rolipram (Nemoz et al., 1985) or the PDE3 selective inhibitor, cilostamide. Briefly, the PDE activity of cell lysates was assayed in the presence of 1 μ M cAMP in either the absence of presence of 10 μ M of the PDE4-specific inhibitor rolipram or the PDE3-specific inhibitor cilostamide. Analysis revealed that the main cAMP hydrolysing enzyme in hPASMC, contributing 53 ± 7% (SEM; n = 6) of the total cAMP-PDE activity, is the rolipram-sensitive fraction, namely PDE4 activity. It was found that a further 25 ± 8% (SEM; n=3) of the activity is due to PDE3 activity (*Figure 3.12*). Total cAMP-PDE activity was 66.5 ± 4.7 pmol/min/mg protein (SEM; n=6), PDE4 activity was 37.1 ± 5.3 pmol/min/mg protein (SEM; n=3).

3.3 Effect of chronic hypoxia on PDE4 in hPASMC

3.3.1 Examining the effect of chronic hypoxia on PDE4

Previous research into models of pulmonary hypertension has indicated that cAMP signalling is altered. Thus, once the PDE4 profile and activity had been determined in hPASMC, I set out to investigate whether these were altered by exposing the cells to hypoxic conditions. Initially, the cells were maintained in a 10% oxygen (balance N_2) environment for a period of two weeks as done before (Murray et al 2002) and to determine the optimal time point to study the cells.

3.3.2 Analysis of PDE4 isoforms expression in chronic hypoxia

Cells were placed under hypoxic conditions (Materials and Methods 2.1.1.3) with a corresponding set of normoxic controls. Cells were harvested for lysate or total RNA as required and examined for PDE4 expression compared to normoxic.

3.3.2.1 PDE4A

3.3.2.1.1Western blot analysis of PDE4A

Cells were harvested and probed with the PDE4A subfamily antibody as described in materials and methods 2.3.4. PDE4A expression was seen to alter throughout the two weeks of hypoxia (*Figure 3.13*). The PDE4A isoform which migrated at 120 kDa on SDS-PAGE, and was thus the combined PDE4A10+PDE4A11 band, was seen to increase to 209 \pm 58% (SEM; n=3) of basal levels after 24 hours exposure to hypoxia and this increase was maintained throughout the 14 days of incubation. Normoxic PDE4A10/4A11 only increased to 102 \pm 63% (SEM; n=3) of basal levels.

3.3.2.1.2 Semi-quantitative RT-PCR analysis of PDE4A

In order to gain insight into whether the increase in PDE4A was due to either PDE4A10 or PDE4A11, 1 undertook semi-quantitative RT-PCR analysis on total RNA isolated from cells maintained under hypoxic conditions for seven days and the corresponding normoxic controls (Materials and Methods 2.4.3.3). Primers specific for PDE4A10 and PDE4A11 were used and PCR was carried out on total RNA isolated from seven day normoxic and hypoxic cells. Primers for cyclophilin were also included to use as a control. It was seen that an increase in both PDE4A10 and PDE4A11 (*Figure 3.14a*) after seven days of hypoxia was confirmed. PDE4A10 levels were seen to increase to $138 \pm 15.9\%$ (SEM; n=3) as compared with normoxic. PDE4A11 increased to $124 \pm 8\%$ (SEM; n=3) of normoxic levels (*Figure 3.14b*).

3.3.2.2 PDE4B

3.3.2.2.1 Western blot analysis of PDE4B

The PDE4B subfamily specific antibody was used to probe cell lysates (Materials and Methods 2.3.4) collected over a two week period of exposure to 10% oxygen to examine any effect on the expression of the PDE4B2 isoform. The expression of PDE4B2 changed throughout the two weeks exposure with a transient increase at day 5-7 to $250 \pm 64 \%$ (SEM; n=3) of basal levels and then this gradually decreased to $120 \pm 2\%$ over a further seven days of hypoxia (*Figure 3.15*).

3.3.2.2.2 Semi-quantitative RT-PCR analysis of PDE4B

Total RNA from cells collected after seven days hypoxia and normoxia was used for semiquantitative RT-PCR with PDE4B2 specific and cyclophilin primers to examine the increase observed in PDE4B2 seen through western blotting (Materials and Methods 2.4.3.3). Transcript levels of PDE4B2 at day seven were seen to be $125 \pm 16\%$ (SEM; n=3) that of normoxic controls (*Figure 3.16*). This increase was considered insignificant in an unpaired t-test.

3.3.2.3 PDE4C

3.3.2.3.1 Western blot analysis of PDE4C

Lysates from hypoxic and normoxic cells were collected over the two week period and probed with the PDE4C specific antibody (Materials and Methods 2.3.4). PDE4C protein expression was observed to remain at basal levels throughout the two weeks with only a $104 \pm 3\%$ (SEM; n=3) increase detected at day 7 (*Figure 3.17*).

3.3.2.4 PDE4D

3.3.2.4.1 Western blot analysis of PDE4D

To examine the protein levels of PDE4D within hypoxic and normoxic cells, the cells were harvested and the cell lysate probed with the PDE4D specific antibody (Materials and Methods 2.3.4). A gradual increase in PDE4D5 protein levels, peaking at day five to seven, was observed (*Figure 3.18*). This increase was maintained for the duration of the fortnight. Levels of PDE4D5 were seen to rise to 172 ± 19 % (SEM; n=3) at day 7 and 177 \pm 17% (SEM; n=3) at day 14 of basal levels. PDE4D1/4D2 and PDE4D3 were not observed to change.

3.3.2.4.2 Semi-quantitative RT-PCR analysis of PDE4D expression

Total RNA from seven day hypoxic and normoxic cells was used for RT-PCR with PDE4D1/2, PDE4D3 and PDE4D5 specific primers (Materials and Methods 2.4.3.3). PDE4D1/2 levels were seen to alter slightly, with an increase of $115 \pm 8\%$ of PDE4D1 and $116 \pm 7\%$ of PDE4D2 (SEM; n=3) of normoxic levels. PDE4D3 was confirmed to significantly increase after seven days hypoxia to $115 \pm 1\%$ of basal levels (*Figure 3.19a, b*).

3.3.3 Western blot analysis of HIF-1α Expression

Cells were placed under hypoxic conditions (Materials and Methods 2.1.1.3) with a corresponding set of normoxic controls for seven days to reach the time point witnessed with maximal change in PDE expression. Cell lysate was used to perform western blotting with an anti-HIF-1 α antibody to identify a key example of the expression of hypoxia regulated genes at this time point (*Figure 3.20*). An anti-human monoclonal HIF-1 α antibody used in previous studies was procured from BD Pharmingen (Hanze et al., 2003)

and identified an immunoreactive band at 120kDa in hypoxic samples only. This concurs with the known molecular weight for HIF-1 α .

3.3.4 Microarray Analysis of normoxic and hypoxic hPASMC

Total RNA from normoxic and seven day hypoxic cells was used for hybridisation onto custom made cell-signalling microarrays from MWG Biotech. RNA was hybridised onto the microarray chips and analysis was carried out by Dr. Ian McPhee of Scottish Biomedical. The chips contained 'spots' of hybridisation for specific PDE4 isoforms and other signalling effectors including PKA isoforms, Rap and ERK. Spots for general PDE4 families were also included. As the normoxic and hypoxic cDNA samples had been synthesised from RNA, they had incorporated different fluorophores. In different experiments, this labelling of samples was reversed to allow for naturally differing intensities of the fluorophores. After hybridisation, fluorophores were excited and intensity values were taken for each fluorophore for each spot on the chip. After normalisation, this data was then interpreted as a ratio of normoxic over hypoxic. By corresponding these values to a 'map' of the chip, changes in gene expression could be identified.

The changes are listed in figure 3.21. It was observed that the increase seen through western blotting and RT-PCR in PDE4A10/4A11, PDE4B2 and PDE4D5 could also be seen by microarray analysis as the general PDE4A and PDE4B, along with the specific PDE4D5 spots produced values indicative of an increase in expression in hypoxia. Other PDE isoforms observed to increase were the PDE5A3, PDE8A, PDE9A and PDE3. Strangely however, the general PDE5A spot indicated an overall decrease in PDE5A isoforms. The largest increase in expression of a PDE was of the PDE3B isoform, this isoform has previously been reported to increase in hypoxic hPASMC (Murray et al., 2002). Modulation of the cAMP pathway appears to occur in the hypoxic samples as the PKA isoforms and Rap1b also show a change in expression is increased in hypoxia. As there was no indication of an RIα isoform decrease in hypoxia, this indicates an overall increase in the PKARI subunit of PKA. Also noteworthy were increases in p53, NFκB and iNOS, all of which are already known to increase in hypoxia (Graeber et al., 1994; Chiarugi et al., 1998).

3.3.5 Effect of chronic hypoxia on cAMP-PDE activity

With the observed increase in PDE4 expression in hypoxic hPASMC, I then set out to determine cAMP-PDE activity in normoxic and hypoxic cells to see if this was altered in hypoxic cells by the change in PDE4 expression.

3.3.5.1 Total cAMP-PDE activity

Lysates from hypoxic and normoxic cells were assayed for cAMP PDE activity with 1μ M cAMP as substrate as described in section 2.3.5. Surprisingly, there appeared to be no significant increase in total cAMP-PDE activity (*Figure 3.22*). Normoxic total cAMP-PDE activity was 66.5 ± 4.7 pmol/min/mg protein (SEM; n=6), whilst a value of 70.8 ± 4.7 pmol/min/mg protein (SEM; n=6) was noted after 7 days hypoxia.

3.3.5.2 PDE3

PDE enzymatic activity of seven day hypoxic cell lysates was assayed in the presence of 10 μ M of the PDE3-specific inhibitor cilostamide in order to determine PDE3 activity (Materials and Methods 2.3.5). As can be seen in figure 3.22 no evident change in PDE3 activity was observed. Normoxic PDE3 activity was 16.9 ± 0.7 pmol/min/mg protein (SEM; n=3) whilst a value of 19.6 ± 6.2 pmol/min/mg (SEM; n=3) was observed after 7 days hypoxia.

3.3.5.3 PDE4

Cell lysates from seven day hypoxic and normoxic cells were assayed for PDE4 enzymatic activity. This was achieved by measuring total cAMP-PDE activity at 1µM cAMP substrate concentration in the presence and absence of 10µM of the PDE4-specific inhibitor, rolipram (Materials and Methods 2.3.5). Again, little difference in PDE4 activity was noted (*Figure 3.22*). Normoxic PDE4 activity was $37.2 \pm 5.3 \text{ pmol/min/mg}$ protein (SEM; u=6), which increased to $44.3 \pm 3.4 \text{ pmol/min/mg}$ protein after 7 days hypoxia (SEM; n=3).

3.3.5.4 Immunoprecipitated PDE4 subfamilies

The total PDE4 activity is represented by four subfamilies within hPASMC. As each subfamily, and indeed individual isoforms, are subject to different modes of regulation, measuring total PDE4 activity would not reveal any change in specific PDE4 subfamily activities. Therefore, I used cell lysates from seven day hypoxic and normoxic cells to assay specifically for the different PDE4 subfamily enzymatic activities. This was achieved by selectively immunopurifying the respective PDE4 subfamilies, as described in section 2.3.6, and using the resulting samples in a cAMP-PDE assay. An increase in the PDE activity for all four PDE4 subfamilies, including PDE4C, was observed (*Figure 3.23*). However, whilst the apparent increase in PDE4A, PDE4B and PDE4C activities was not significant that for the PDE4D subfamily was significantly increased (p<0.01) to 172% of normoxic PDE4D activity. Thus the key target for hypoxia-induced increase in PDE4 activity is due to the PDE4D sub-family, but such an effect can be partially masked due to

the activities of the three other PDE4 subfamilies, which are little if at all changed in hypoxia.

3.4 Discussion and Conclusions

Exposing vascular smooth muscle cells to hypoxia has been shown to alter many cell signalling pathways including cAMP and cGMP signalling (MacLean et al., 1997; Murray et al., 2002). The decrease observed in cyclic nucleotides in hypoxia has been attributed to increased hydrolysis by PDEs as cAMP-PDE and cGMP-PDE activities have been reported to increase in the pulmonary arteries of the hypoxic rat (MacLean et al., 1997). The cGMP-PDE PDE5 and the cAMP-PDE PDE3 have previously been reported to increase during chronic hypoxia in the pulmonary arteries of the rat and in hPASMC (Murray et al., 2002). Although PDE4 activity represents a major cAMP-hydrolysing activity within these cells, their expression and activity have not been fully investigated.

Each cell type has a specific profile of cAMP-PDE isoforms, which is presumed to enable differential regulation of cAMP signalling (Houslay & Milligan 1997; Houslay 1998). It has previously been established that the major cAMP-PDEs present within vascular smooth muscle cells are PDE3 and PDE4 (Maurice et al., 2003). Prior research has investigated PDE3 expression and shown that the PDE3A and PDE3B isoforms are present within hPASMC (Murray et al 2002). Investigation into the PDE4 profile showed conclusively that PDE4A7, PDE4A10, PDE4A11, PDE4B2, PDE4C, PDE4D1, PDE4D2, PDE4D3 and PDE4D5 are expressed in hPASMC. PDE4 activity contributes approximately 53% of the total cAMP-PDE activity within hPASMC with PDE3 being responsible for another 25% under basal/resting conditions.

Once I had established the PDE4 profile within hPASMC, I set out to investigate if this profile was altered in chronic hypoxia, initially through western blotting and then confirmation by RT-PCR. This enabled me to discover that the level of expression of the PDE4 isoforms, PDE4A10, PDE4A11, PDE4B2 and PDE4D5 were all increased to varying degrees over 14d of exposure to 10% O₂. The increase in both PDE4A10 and PDE4A11 was due to increased transcription with RT-PCR identifying a 138% and 124% increase in PDE4A10 and PDE4A11 respectively. This was the largest increase observed of the PDE4 isoforms increased in hypoxia. The increased transcription of PDE4A10/PDE4A11 is suggestive of a hypoxic transcription factor such as HIF-1 being responsible for the upregulation of these PDE4A isoforms. As PDE4A10 and PDE4A11 are recently characterised PDE4A isoforms, their functional attributes have not yet been determined. It has been hypothesised that PDE4A10 could be involved in the inflammatory

response due to the presence of possible sites for GATA within its promoter. This site is known to interact with the H_2O_2 -activated USF (Andrews 2000) as well as CREB and NF κ B. Indeed, acute hypoxia has been observed to increase GATA DNA binding activities (Jamali et al., 2004). It is known however that the PDE4A subfamily is not subject to regulation by ERK (Baillie et al. 2000). This means PDE4A10 and PDE4A11 activities are not susceptible to inhibition by the increased active ERK reported in hypoxic vascular smooth muscle cells (Minet et al., 2000b; Scott et al., 1998; Welsh et al., 2001).

The increase in the PDE4B2 isoform represented the largest increase in protein levels among the altered PDE4 isoforms with an increase to 250% of basal levels. Although RT-PCR identified an increase in PDE4B2 transcription to 125 % of basal, this increase was not deemed significant by an unpaired t-test due to the large standard error of the mean. PDE4B2 has been established as having a key role in the regulation of inflammatory responses (Ma et al., 1999; Wang et al., 1999; Oger et al., 2002). Myometrial contractility occurs due to intrauterine infections which induce an inflammatory response involving an increase in the levels of IL-1 β . An increase in PDE4B2 expression has been reported to occur in human smooth muscle myometrial cells exposed to $1L-1\beta$ due to the increased cAMP levels produced through induction of PGE_2 (Oger et al., 2002), thereby acting in a feedback loop to regulate cAMP levels. IL-1 β increased cAMP to maximal levels after 12 hours and this induced an increase in PDE4 activity, increasing the hydrolysis of cAMP and returning cAMP levels to normal after a further 12 hours. An increase in PDE4B2 has also been observed directly in response to cAMP raising agents in human myometrial cells (Mehats et al., 1999). This suggests that the increase in PDE4B2 could be due to an increase of cAMP in hypoxia. In hypoxic rat pulmonary arteries however, a decrease in cAMP levels is observed (MacLean et al. 1997). This occurred after 14 days of hypoxic exposure. The increase in PDE4B2 protein levels observed here was transient as protein levels returned to normal after 14 days, suggesting a similar feedback loop to that shown in the myometrial cells (Figure 3.24). Hypoxia could initially induce an increase in cAMP levels through inflammatory mediators and cause an increase in PDE4B2 activity, reducing cAMP levels to below that of normal and alleviate the induction of PDE4B2 expression.

Although previous studies have shown cAMP prevents the induction and maintenance of contraction in smooth muscle cells (Silver 1985), PDE4 inhibitors can block spontaneous myometrial contraction (Mehats et al., 2002). During this spontaneous contraction, PDE4B2 levels increase, suggesting a role for PDE4B2 in this response (Mehats et al.,

2002). Therefore, this is suggestive of a remodulation of cAMP signalling during this contraction and infers PDE4B2 could be initially involved in the hypoxic pulmonary vasoconstrictive response seen in models of PAH.

PDE4D isoforms are involved in the maintenance of airway smooth muscle tone (Mehats et al., 2003). PDE4D knockout mice are refractory to muscarinic cholinergic stimulation and show increased sensitivity to PGE_2 (Hansen et al., 2000). Therefore an alteration in PDE4D could affect smooth muscle contraction and refaxation, suggesting the increased expression of PDE4D5 could also be involved in the hypoxic pulmonary vasoconstrictive response. As PDE4D5 increased levels are maintained whereas the PDE4B2 levels return to normal, it is possible that PDE4D5 is involved in the maintenance of contraction, while PDE4B2 is involved in the initial phase of contraction.

PDE4D5 levels have been reported to increase in human airway smooth muscle in response to cAMP-elevating agents (Le Jeune et al., 2002). This response was shown to be due to a cAMP-response element (CRE) within the promoter region of PDE4D5. This again is suggestive of an initial rise in cAMP levels in hypoxia which cause an increase in PDE4D5 expression. Indeed, the PDE4D5 increase was shown to be due, at least in part, to increased transcription as RT-PCR identified a 115% increase in transcript levels after 7 days hypoxia. PDE4D1 and PDE4D2 also contain a CRE site within their promoters. It was observed that their transcript levels increased to 115% and 116% respectively, although no notable increase in PDE4D1/PDE4D2 protein levels was observed. The effect of hypoxia on these PDE4D, cAMP-inducible isoforms, is again suggestive of an increase in cAMP levels in hypoxia.

In this study, the increase in PDE4 expression was not mirrored by a significant increase in total PDE4 activity after seven days of hypoxic exposure. Examining PDE4 subfamily activities, an increase in PDE4D activity to 172% of normoxic levels was observed. Total PDE3 activity was also unaffected, but has previously been reported to be increased after 14 days hypoxia in hPASMCs in a manner mimicked by cAMP-raising agents and blocked by a PKA inhibitor (Murray et al., 2002). In the aforementioned study (Mehats et al., 2002), the observed increase in PDE4 activity was not an immediate effect and required 18 hours exposure to $11-1\beta$, not reaching maximal levels until after 24 hours. These combined results lend credence to the hypothesis that cAMP signalling undergoes a gradual remodulation during chronic hypoxia, with cAMP levels initially rising and a concomitant increase in PDE4 activity. After prolonged exposure to high cAMP levels however, the PDE

activities increase to a sufficient level to reduce cAMP and attenuate the effect of cAMP-raising agents (Wagner et al., 1997) alleviating the induction of PDEs by cAMP (*Figure 3.24*). As many other pathways are altered in hypoxia, it is possible that after cAMP no longer regulates the expression of PDEs, other factors can affect their regulation and indeed maintain the increase in some PDEs.

To confirm the activation of hypoxia induced pathways in the seven day hypoxic cells, the presence of the transcription factor HIF-1 α was investigated. HIF-1 α is constituitively expressed within cells, but is degraded within minutes under normal conditions (Sutter et al., 2000). Under hypoxic conditions however, degradation of the HIF-1 α subunit is attenuated allowing HIF-1 to act upon hypoxia inducible genes containing HIF-1 binding sites. Using a commercial HIF-1 α antibody, the presence of HIF-1 α was confirmed in hypoxic hPASMC, indicating the cellular responses to hypoxia were activated.

To further confirm the increase in PDE4 protein levels and investigate other signalling pathways, microarray analysis was utilised to examine hypoxia dependant changes in expression. Increases in p53, NFkB and iNOS, all of which are already known to increase in hypoxia, were observed (Graeber et al., 1994; Chiarugi et al., 1999; Igari et al., 1998; Palmer et al., 1998). Microarray analysis indeed confirmed an increase in PDE4D5, PDE4A10 and PDE4A11 expression in hypoxia. The increase observed in the PDE3B isoform has previously been identified in hypoxic hPASMCs (Murray et al., 2002).

The increase in the RI β subunit of PKA also indicated an alteration in response to cAMP within hypoxic cells. PKARI has previously been shown to be a little overexpressed in normal cells upon stimulation of proliferation, but is constituitively expressed in tumours (McDaid., et al 1999). Vascular smooth muscle cells are known to proliferate in response to hypoxia (Yuan & Rubin 2001). This indicates upregulation of PKARI β could be involved in the proliferative response of vascular smooth muscle cells to hypoxia.

However, the results from the microarray analysis are only indicative of possible alterations in gene expression as the analysis was only performed twice and problems with microarray analysis have to be taken into account (Watson et al., 2000). Variations between chip to chip, even spot-to spot, allow for problems when considering background levels to normalise data. With tightly regulated genes, this problem is emphasised and small, significant changes are difficult to confirm. Thus, enough replicate microarray hybridisations require to be performed to confirm the data collected.

The data presented in this chapter indicates that the cellular response to chronic hypoxia is a gradual process, with certain signalling pathways being altered immediately, and others occurring over an extended period due to alterations in gene expression and activity. The results suggest cAMP levels are increased immediately, with CRE-regulated genes being induced including PDE4B2, PDE4D1, PDE4D2 and PDE4D5. This increase in cAMP-PDE isoforms reduces cAMP over time and is responsible for the desensitisation of the cAMP pathway in hypoxia (*Figure 3.24*). Unfortunately, as cAMP-PDE activity was not observed to increase after seven days and a time point of fourteen days was not examined, this would need further work to explore in detail.


Figure 3.1 Western blot analysis of PDE4A isoforms expressed in hPASMC.

hPASMC cells were harvested in cKHEM buffer and cell lysate was made as described in section 2.3.1.1. 40μ g protein was ran out on an 8% SDS-PAGE alongside 5μ g recombinant protein standards of PDE4A isoforms PDE4A1 (79kDa), PDE4A4 (125kDa) and PDE4A10 (121kDa). This was then transferred to nitrocellulose and probed with an antibody directed against the c-terminus of PDE4A. As PDE4A4, PDE4A10 and PDE4A11 all run at similar weights on a gel, (125, 121 and 126 kDa respectively), isoform specific antibodies directed against the n-terminus of PDE4A4 (*Panel A*), PDE4A10 (*Panel B*), and the c-terminus of PDE4A to produce a general 4A antibody (*Panel C*) were used to distinguish which isoform was being expressed in hPASMC. The molecular weight markers are indicated. A band of approximately 120kDa (indicated in bold), thus corresponding to the PDE4A10/4A11 isoforms, was detectable in hPASMC lysate.

89

	PRIMER NAME	PRIMER SEQUENCES (5' - 3')	FRAGMENT LENGTH (bp)	
	4A1 F	TTCTTCTGCG AGACCTGCTC TAAGC	344	
4A1	4A1 R	GGT CTT CAC CCC AAA TCG GGG AAT		
	4A4 F	CGG AAA GGA GCC TGT CTC TG	257	
4A4	4A4 R	AGT GCC ATG GAA GGA CGA GG		
117	4A7 F	GAT AAT GGT GCT TCC TTC AGA CCA AGG C	212	
4A7	4A7 R	CCA GCA GTA CGT GGG TGG ACT GCA GCA	348	
4410	4A10 F	AGA TCT GTC AGC TTC GAG GCA G	281	
4A10	4A10 R	AGT GAG AAG TTG CTA CGG ACG C		
4A11	4A11 F	GCC TAG GCC GCA TCC CGG AGC TGC AAC T 350		

Figure 3.2 Primer sequences used for analysis of PDE4A isoforms in hPASMC.

A table indicating primer sequences and fragment size expected for of human PDE4A splice variants, hsPDE4A1 (Genbank accession U97584), hsPDE4A4 (Genbank accession L20965), hsPDE4A7 (Genbank accession U18088), hsPDE4A10 (Genbank accession AF073745), and hsPDE4A11 (Genbank accession AF069488).



Figure 3.3 RT-PCR analysis of hPASMC using PDE4A specific primers.

RNA was isolated from hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2., followed by RT-PCR using primers specific for the PDE4A isoforms, hsPDE4A1, hsPDE4A4, hsPDE4A7, hsPDE4A10 and hsPDE4A11 (*see figure 3.2 for primer sequences and accession numbers*). The PCR products were ran out on a 2% agarose gel along side a negative control. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. Primer products of approximately 280bp, 310bp and 390bp were detected with PDE4A10, PDE4A11 and PDE4A7 primer sets respectively (indicated in bold).



Figure 3.4 Western blot analysis of PDE4B isoforms expressed in hPASMC.

hPASMC cells were harvested in cKHEM buffer and cell lysate was made as described in section 2.3.1.1. 40µg protein was ran out on an 8% SDS-PAGE alongside 5µg recombinant protein of PDE4B isoforms PDE4B1 (104kDa), PDE4B2 (80kDa) and PDE4B3 (103kDa). This was then transferred to nitrocellulose and probed with an antibody directed against the common PDE4B c-terminus region. A band of approximately 70kDa (indicated in bold) could be detected. As this was the same weight as the PDE4B2 protein standard ran on the gel, this band was assumed to be PDE4B2 in hPASMC lysate.

	PRIMER NAME	PRIMER SEQUENCES (5' - 3')	FRAGMENT LENGTH (bp)
101	4B1 F	AAA GCA GGA GTG TGA TGA CGG	226
481	4B1 R	CGT TGT CAA AGG CAG TGT GGT	236
402	4B2 F	CCT TGA GAT GGC AAA GCA CTC	264
482	4B2 R	AAT CAC AGT GGT GCT CTG CCT	264
402	4B3 F	AAA AGC ATT CGG CAG CGT C	214
463	4B3 R	TCG ACA TCG CCT TTG GTG A	214

Figure 3.5 Primer sequences used for analysis of PDE4B isoforms in hPASMC.

A table indicating primer sequences and fragment size expected for the human PDE4B splice variants, hsPDE4B1 (Genbank accession L20966), hsPDE4B2 (Genbank accession L20971), and hsPDE4B3 (Genbank accession U85048).



Figure 3.6 RT-PCR analysis of hPASMC using PDE4B specific primers.

RNA was isolated from hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2, followed by RT-PCR using primers specific for the PDE4B isoforms, hsPDE4B1, hsPDE4B2 and hsPDE4B3 (*see figure 3.5 for primer sequences and accession numbers*). The PCR products were ran out on a 2% agarose gel along side a negative control. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. A primer product of approximately 250bp was detected with the PDE4B2 primer set (indicated in bold).



Figure 3.7 Western blot analysis of PDE4C isoforms expressed in hPASMC.

hPASMC cells were harvested in cKHEM buffer and cell lysate was made as described in section 2. 3.1.1. 40µg protein was run out on 8% SDS-PAGE alongside 5µg recombinant protein of the PDE4C isoform PDE4C2 (80kDa). This was then transferred to nitrocellulose and probed with an antibody directed against the c-terminus of PDE4C. A band of approximately 85kDa (indicated in bold) could be detected.

A

	PRIMER NAME	PRIMER SEQUENCES (5' - 3')	FRAGMENT LENGTH (bp)	
4C	4C F	ATG GAT GGT AAA GCC CTT TGG CTC TTG G	05	
	4C R	GTC TCC CTA AAT GGG TGG GAA AGT GAA G	95	



Figure 3.8 RT-PCR analysis of hPASMC using PDE4C specific primers.

Panel A, a table indicating primer sequences and fragment size expected of the transcript of the human PDE4C splice variant, hsPDE4C2 (Genbank accession U66346).

Panel B, RNA was isolated from hPASMC, subjected to first strand cDNA synthesis as described in section 2.4.3.2 and followed by RT-PCR using primers specific for the PDE4C isoform, hsPDE4C2. The PCR products were ran out on a 2% agarose gel along side a negative control. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. A primer product of approximately 110bp was detected with the PDE4C2 primer set (indicated in bold).



Figure 3.9 Western blot analysis of PDE4D isoforms expressed in hPASMC.

hPASMC cells were harvested in cKHEM buffer and cell lysate was made as described in section 2.3.1.1. 40µg protein was ran out on an 8% SDS-PAGE alongside 5µg recombinant protein of PDE4D isoforms PDE4D1 (68kDa, PDE4D2 runs at a similar weight, 67kDa), PDE4D3 (95kDa), PDE4D4 (119kDa) and PDE4D5 (105kDa). This was then transferred to nitrocellulose and probed with an antibody directed against the c-terminus of PDE4D. The molecular weight markers are indicated. Bands of approximately 100kDa, 89kDa and 65kDa (indicated in bold) could be detected in hPASMC lysate.

В	PRIMER NAME	PRIMER SEQUENCES (5' - 3')	FRAGMENT LENGTH (bp)
4D1/2	4D1/2 F	ATA TGA AGG AGC AGC CCT CAT G	221 & 307
	4D1/2 R	CCA GAC CGA CTC ATT TCA GAG A	
4D3	4D3 F	GCG AAC ATG ATG CAC GTG AA	292
	4D3 R	TGG CCA AGA CCT GAG CAA AT	
4D4	4D4 F	AGA AAT CCA GGA TGT CCT GGC	- 347
	4D4 R	TGC TAG GTG CTC GAT CTT GCA	
4D5	4D5 F	TGC CAG CTG TAC AAA GTT GAC C	212
	4D5 R	TTC TCG GAG AGA TCA CTG GAG A	212

Figure 3.10 Primer sequences used for analysis of PDE4D isoforms in hPASMC.

A table indicating primer sequences and fragment size expected for the human PDE4D splice variants, hsPDE4D1/2 (Genbank accession U50157, U50158 respectively), hsPDE4D3 (Genbank accession L20970), hsPDE4D4 (Genbank accession L20969), and hsPDE4D5 (Genbank accession S:1059276).



Figure 3.11 RT-PCR analysis of hPASMC using PDE4D specific primers.

RNA was isolated from hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2., followed by RT-PCR using primers specific for the PDE4D isoforms, hsPDE4D1, hsPDE4D2, hsPDE4D3, hsPDE4D4 and hsPDE4D5 (*see figure 3.11 for primer sequences and accession numbers*). The PCR products were ran out on a 2% agarose gel along side a negative control. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. Primer products of approximately 200bp, 220bp, 300bp and 300bp were detected with the PDE4D1, PDE4D5, PDE4D3 and PDE4D2 primer sets respectively (indicated in bold).



Figure 3.12 cAMP-PDE activity in hPASMC.

Cells were harvested in cKHEM + 1% Triton and the lysates were assayed for total cAMP-PDE Activity, PDE4 Activity, determined as the rolipram sensitive fraction, and PDE3 Activity, determined as the cilostamide sensitive fraction. Phosphodiesterase assay method was described in section 2.3.5. Results are expressed as *panel A*, mean +/- S.E. of 3 independent experiments in pmol/min/mg protein and *panel B*, percentage of total cAMP-PDE activity.



Figure 3.13 Western blot analysis of the effect of chronic hypoxia on PDE4A isoform expression.

hPASMC were split into two groups, one was treated as normal and the other was maintained in a 10% O_2 environment for a period of two weeks. Both sets were treated identically throughout. Cells were harvested from both groups at days throughout the time course and 40µg protein from each time point was used for western blotting.

Panel A, blots were probed with a pan PDE4A antibody. Results shown are hypoxic samples. *Panel B*, densitometry results with normoxic and hypoxic time points compared directly expressed as a percentage of the respective basal density.

Blot shown is representative of three independent experiments and *, ** denotes significance (p<0.05, p<0.01 respectively)

A



Figure 3.14a RT-PCR analysis of the effect of chronic hypoxia on PDE4A isoform expression.

RNA was isolated from both normoxic and 7 day hypoxic hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2. This was followed by RT-PCR using Superscript II and primers specific for the PDE4A isoforms (*see figure 3.2*). The PCR products were ran out on a 2% agarose gel along side negative controls. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. The PDE4A PCR products identified are shown and are representative of three independent experiments.



A

B

C

Figure 3.14b Densitometry of RT-PCR analysis of the effect of chronic hypoxia on PDE4A isoform expression.

Quantity-one software was used to measure the density of each band which was then normalised against the density of the corresponding cyclophilin band. *Panel A*; densitometric analysis of PDE4A7 band expressed as a percentage of the normoxic band present at 35 cycles. *Panel B*; densitometric analysis of PDE4A10 band expressed as a percentage of the normoxic band present at 35 cycles. *Panel C*; densitometric analysis of PDE4A11 band expressed as a percentage of the normoxic band present at 35 cycles. *Results* shown are mean \pm S.E. of three independent PCR reactions and * denotes significance (p<0.05).

103



Figure 3.15 Western blot analysis of the effect of chronic hypoxia on PDE4B isoform expression.

hPASMC were split into two groups, one was treated as normal and the other was maintained in a 10% O_2 environment for a period of two weeks. Both sets were treated identically throughout. Cells were harvested from both groups at days throughout the time course and 40µg protein from each time point was used for western blotting.

Panel A, blots were probed with a pan PDE4B antibody. Results shown are hypoxic samples. *Panel B*, densitometry results with normoxic and hypoxic time points compared directly expressed as a percentage of the respective basal density. Blot shown is representative of three independent experiments and * denotes significance (p<0.05).

104



Figure 3.16 RT-PCR analysis of the effect of chronic hypoxia on PDE4B isoform expression.

RNA was isolated from both normoxic and 7 day hypoxic hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2. This was followed by RT-PCR using Superscript II and primers specific for the individual PDE4B isoforms (*see figure 3.5*). The PCR products were ran out on a 2% agarose gel along side negative controls. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. *Panel A*, PDE4B PCR products, *panel B*, densitometric analysis of PDE4B2 band expressed as a percentage of the normoxic band present at 40 cycles. The results shown are indicative of three independent experiments.

A



Figure 3.17 Western blot analysis of the effect of chronic hypoxia on PDE4C isoform expression.

hPASMC were split into two groups, one was treated as normal and the other was maintained in a 10% O_2 environment for a period of two weeks. Both sets were treated identically throughout. Cells were harvested from both groups at days throughout the time course and 40µg protein from each time point was used for western blotting.

Panel A, blots were probed with a pan PDE4C antibody. Results shown are hypoxic samples. *Panel B*, densitometry results with normoxic and hypoxic time points compared directly expressed as a percentage of the respective basal density.

106



Figure 3.18 Western blot analysis of the effect of chronic hypoxia on PDE4D isoform expression.

hPASMC were split into two groups, one was treated as normal and the other was maintained in a 10% O_2 environment for a period of two weeks. Both sets were treated identically throughout. Cells were harvested from both groups at days throughout the time course and 40µg protein from each time point was used for western blotting.

Panel A, blots were probed with a pan PDE4D antibody. Results shown are hypoxic samples. *Panel B*, densitometry results with normoxic and hypoxic time points compared directly expressed as a percentage of the respective basal density.

Blot shown is representative of three independent experiments and ** denotes significance (p<0.01).

107



Figure 3.19a RT-PCR analysis of the effect of chronic hypoxia on PDE4D isoform expression.

RNA was isolated from both normoxic and 7 day hypoxic hPASMC and subjected to first strand cDNA synthesis as described in section 2.4.3.2. This was followed by RT-PCR using Superscript II and primers specific for the PDE4D isoforms (*see figure 3.11*). The PCR products were ran out on a 2% agarose gel along side negative controls. When the negative control indicated contamination, the PCR result was discarded. DNA marker sizes are indicated. The PDE4D PCR products identified are shown and are representative of three independent experiments. PDE4D3 results are not shown due to only n=1 without contamination.



Figure 3.19b Densitometry of RT-PCR analysis of the effect of chronic hypoxia on PDE4D isoform expression. Densitometric analysis of results shown in figure 3.19a. Quantity-one software was used to measure the density of each band which was then normalised against the density of the corresponding cyclophilin band. *Panel A*; densitometric analysis of PDE4D1 band expressed as a percentage of the normoxic band present at 35 cycles. *Panel B*; densitometric analysis of PDE4D2 band expressed as a percentage of the normoxic band present at 35 cycles. *Panel B*; densitometric analysis of PDE4D5 band expressed as a percentage of the normoxic band present at 35 cycles. *Results* shown are mean \pm S.E. of three independent PCR reactions and *** denotes significance (p<0.001).

Densitometry of PDE4D1



Figure 3.20 HIF-1a expression in normoxic and hypoxic hPASMC.

Seven day hypoxic and normoxic hPASMC were harvested as described in section 2.3.1.1 and used for western blotting with an anti-HIF-1 α antibody. HIF-1 α is 120kDa (Hanze et al., 2003), and a band was detected at 121kDa in hPASMC exposed to seven days hypoxia, but not in normoxic cells. Blot shown is representative of three independent experiments.

Gene	Ratio= norm/hyp	Number of spots changed/number of spots analysed
PDE5A	1.5	4/4
PDE4A	0.9	4/6
PDE4A11	0.9	3/4
PDE5A3	0.9	5/6
PDE8A	0.9	4/6
PDE4B	0.8	6/6
ERK1	0.8	4/6
ERK2	0.8	4/4
PKAR1B	0.8	5/6
iNOS	0.8	3/6
NFĸB	0.8	6/6
p53	0.8	6/6
PDE4D5	0.7	4/6
PDE9A	0.7	2/2
PDE3B	0.7	5/6
Rap1B	0.7	5/6

Figure 3.21 Microarray analysis of RNA from normoxic and hypoxic hPASMC. Total RNA from normoxic and seven day hypoxic cells was used for hybridisation on a custom microarray chip (MWG Biotech). Results shown are from two separate chips, with reverse labelling of samples.

A Total cAMP-PDE Activity



Figure 3.22 Effect of chronic hypoxia on cAMP-PDE activity.

Normoxic and seven day hypoxic cells were harvested in cKHEM + 1% Triton and the lysate was assayed for cAMP-PDE activity as in section 2.3.5. *Panel A*, total cAMP-PDE activity, means \pm S.E. of 6 independent experiments. *Panel B*, PDE4 activity, means \pm S.E. of 6 independent experiments. *Panel C*, PDE3 activity, means \pm S.E. of 3 independent experiments. All results are expressed in pmol/min/mg protein.



Figure 3.23 Effect of chronic hypoxia on PDE4 subfamily activity.

PDE4A, PDE4B, PDE4C and PDE4D activities were immunoprecipitated as described in section 2.3.6. Samples were then used in a cAMP-PDE assay as described in section 2.3.5. Results shown are mean \pm S.E. of 6 independent experiments. Results were calculated in pmol/min/mg protein and converted to a percentage of normoxic activity for each PDE4 subfamily. Significance is denoted by ** (p<0.01).



Figure 3.24 Proposed model of cAMP regulation during chronic hypoxia

Above is a model of a proposed theory to explain the desensitisation of the cAMP pathway in hypoxia and during prolonged increased cAMP. Hypoxia causes a rise in cAMP levels, possibly through prostacyclins which are known to increase in hypoxic conditions. This rise in cAMP levels induces PDE expression. cAMP levels continue to rise to maximum rapidly, preventing increases in PDE to have an impact. Once at maximum, cAMP levels remain steady until PDE expression and activity reaches a 'threshold' point where the amount of cAMP hydrolysed is more than the amount generated. This causes a reduction in cAMP levels and attenuates cAMP generation in response to exogenous cAMP elevating agents. Chapter 4

Effect of Chronic Hypoxia on cAMP Levels in hPASMC

4.1 Introduction

4.1.1 Cyclic Nucleotide Signalling in Models of Hypoxia-Induced Pulmonary Hypertension

4.1.1.1 Cyclic Nucleotides in Hypoxla

It is well established that raising intracellular levels of cAMP and cGMP can lead to smooth muscle relaxation and inhibit VSMC proliferation (Koyama et al., 2001; Rybalkin & Bornfeldt 1999; Osinski & Shror 2000; Indolfi et al., 2001; Boyton & Whitfield 1983, Hayashi et al., 2000; Kronemann 1999; Hofmann 2000; Hakonarson & Grunstein 1998). Raising cAMP levels is known to be more effective at inhibiting VSMC proliferation than elevating cGMP (Murray 1990; Koyama et al 2000). The effects of these cyclic nucleotides are thought to occur through the activation of protein kinase A (PKA) and protein kinase G (PKG) which phosphorylate contractile proteins and channels. For example, PKA acts on the vasodilator-stimulated phosphoprotein, or VASP, which promotes actin nucleation and binds to actin filaments *in vitro* and associates with stress fibres in cells (Shabb 2001). PKA phosphorylation has a negative effect on VASP interaction with actin filaments (Harbeck et al., 2000). PKA and PKG can induce relaxation by acting on downstream targets of Ca2+ mobilisation (Shabb 2001; Marin et al., 1998).

It has been demonstrated that the chronic hypoxic rat displays reduced levels of both cyclic nucleotides in all pulmonary arteries with the exception of the resistance arteries (MacLean et al., 1996). It should be noted that other studies have indicated a rise in cGMP levels in chronic hypoxic rats compared to controls (Cohen et al., 1996). An increase in cyclic nucleotides would be beneficial under hypoxia, aiding to restore the normal tone of the pulmonary circulation. In fact, it has been suggested that hypoxia initially leads to an increase in cAMP through PGI_2 activation, yet after prolonged exposure to hypoxia, desensitisation occurs and pathways are initiated to remove cAMP.

4.1.1.2 Cyclases in Hypoxia

Studies of guanylyl cyclase activity have observed that soluble guanylyl cyclase activity is increased in hypoxia with a corresponding increase in cGMP levels (Li et al., 1999). In the rat lung, the adenylyl cyclase isoforms AC2, AC3 and AC5/6 predominate in pulmonary VSMC, although the isoforms AC 2, AC5 and AC8 are the most functionally important. Looking at these isoforms in hypoxia in the rat, there was no change of expression observed, (Jourdan et al 2001) although there are reduced responses to adenylyl cyclase activators such as forskolin. Under chronic hypoxic conditions, it has been demonstrated

that adenylyl cyclase activity measured under GTP, forskolin or isoproterenol stimulation is reduced in rat hearts (Pei et al., 2000; Hrbasova et al., 2003). Other studies have demonstrated no effect of hypoxia on AC activity however (Pei et al., 2000).

4.1.1.3 Attenuated responses in Hypoxia

It is well established that responses to agents known to stimulate cAMP production such as β -adrenergic agonists are diminished in hypoxia (Hrbasova et al., 2003). Hearts from chronic hypoxic animals indeed display a reduction in β -adrenoceptors and a desensitisation of adenylyl cyclase (Leon-Velarde et al., 2001; Voelkel et al., 1981) However, other studies have demonstrated an increase in β -adrenoceptors and unchanged adenylyl cyclase activity in isolated cardiac myocytes in response to chronic hypoxia (Li et al., 1996), highlighting the different effects of chronic hypoxia due to experimental differences. This is also apparent on studies investigating the effect of hypoxia on G_s and G_i signalling. G_s protein levels have been reported to be decreased or unchanged and G_i protein levels have reported to be increased or unaltered (Hrbasova et al., 2003; Leon-Velarde et al., 2001; Pei et al., 2000).

Precontracted pulmonary arteries relaxed by the addition of the β -adrenoceptor agonist, isoproterenol, demonstrated β -adrenergic relaxation is significantly attenuated in chronic hypoxic rat pulmonary arteries compared (Wagner et al., 1997). In addition, forskolinstimulated relaxation is also attenuated in the chronic hypoxic rat (Wagner et al., 1997). As no change in adenylyl cyclase activity has been observed in hypoxia (Shaul et al., 1990), it was hypothesised that there was an increased hydrolysis of cAMP. Indeed, this attenuation was alleviated by the addition of either PDE3 or PDE4 inhibitors (Wagner et al., 1997).

As I observed PDE4 levels to increase in hypoxia without any marked increase in total PDE4 activity in the previous chapter, I set out to determine cAMP levels in both normoxic and hypoxic cells and the response of these levels to various agonist and inhibitors. All assays were carried out after seven days of hypoxia. This was in order to gain insight into why there was so little change in PDE4 activity in hypoxia at this time point by uncovering other pathways that can affect upon cAMP and also regulate PDE4 expression and activity in hPASMC. Also, the response to cAMP-elevating agents in hypoxic hPASMC has not yet been investigated.

117

ŝ

i.

Results

4.1 cAMP signalling in normoxic and hypoxic hPASMC

All methods used in this chapter are described in full in section 2, materials and methods. Results are calculated as pmol cAMP per ng protein and expressed as a percentage of basal, with basal figures equal to 100%. This normalisation was done due to the variability in absolute amounts of cAMP noted in the various preparations of smooth muscle cells. This could be due to a wide variety of factors including differences caused by serum, PGE1-mediated autocrine stimulation of adenylyl cyclase, cell numbers and cell cycle. However, I have also given an approximate value for cAMP levels, which gives an indication of the magnitude of the level experienced.

4.1.1 cAMP levels in normoxic and hypoxic hPASMC

Cells were maintained in hypoxia for seven days before being harvested, whilst the corresponding controls were maintained under normoxic conditions. These were then used for analysis of intracellular cAMP levels. cAMP levels were measured using a modification of the procedure of Savage, (Savage, 1995), as described in section 2.3.7. Intracellular levels of cAMP in hypoxia were observed to be elevated to three times that of normoxic levels with normoxic cAMP levels = 5.5 ± 2.2 pmol/ng protein and hypoxic cAMP levels = 15.8 ± 1.3 pmol/ng protein, n=3 (*Figure 4.1*).

4.1.2 PKA Activity in hypoxic hPASMC

With this unexpected increase in cAMP levels in hypoxia, other methods were employed to confirm this observation independently. In doing this I chose to evaluate protein kinase A (PKA) activity as this enzyme provides a prime downstream signalling system that is activated by clevation of intracellular cAMP. Seven day hypoxic and normoxic hPASMC were collected and used for analysis of PKA activity levels as described in section 2.3.7. As can be seen in figure 4.2, under normoxic conditions the percentage of PKA activity activated was 5.5 ± 4.5 % of the total. In marked contrast to this, under hypoxic conditions the percentage of the total PKA activity that was activated had increased to 23.4 ± 6.8 % of the total (n=2). These data are entirely consistent with an increase in cAMP levels occurring in hypoxia.

an rhathhren

4.1.3 PKA Expression in hypoxic hPASMC

Seven day hypoxic and normoxic cells were harvested and used for western blotting to identify if PKA subunit expression was altered along with cAMP signalling in hypoxia. Commercial antibodies for PKA RI α , PKA RII α and PKA RII β were employed to detect protein levels of the different PKA subunits. No antibodies are available for the detection of PKA RI β as yet. Comparing normoxic and hypoxic cell lysates, there was no change in the observed PKA subunits expression levels.(*Figure 4.3*).

4.2 Regulation of cAMP levels in hypoxic hPASMC

After identifying an increase in intracellular cAMP levels, the effect of cAMP pathway effectors was investigated to study the regulation of cAMP with an altered basal level. Previously, cAMP responses have been reported to be attenuated in models of chronic hypoxia (Wagner et al., 1997).

4.2.1 cAMP levels in normoxic and hypoxic hPASMC in response to PDE inhibitors

PDE inhibitors were employed to determine PDE regulation of cAMP levels in normoxic and hypoxic hPASMC. Thus 7d hypoxic and normoxic cells were treated with PDE inhibitors for 20 min prior to being harvested for analysis of intracellular cAMP levels. Rolipram and cilostamide were used to inhibit PDE4 and PDE3 activity, respectively and the non-specific PDE inhibitor IBMX was also used. After this treatment, not only were cells harvested for assay of intracellular cAMP, but the growth medium that the cells were maintained in was aspirated to examine extracellular cAMP levels. cAMP levels were measured as described in section 2.3.7.

4.2.1.1 Intracellular cAMP levels

Intracellular levels of cAMP were seen to increase significantly in response to rolipram (p<0.001), cilostamide (p<0.001) and IBMX treatment (p<0.001) in normoxic cells only (*Figure 4.4a*). Treating hypoxic cells with the inhibitors for the same length of time was observed to have no effect over hypoxic basal (*Figure 4.4a*). Rolipram increased normoxic cAMP levels to 16.4 pmol cAMP/ng protein, cilostamide increased cAMP levels to 14.5 pmol cAMP/ng protein and IBMX increased cAMP levels to 24.7 pmol cAMP/ng protein (*Figure 4.4b*). The increased levels of cAMP that occurred in normoxic cells in response to PDE4 or PDE3 inhibition was therefore approximately equal to that of hypoxic basal levels; 15.8 pmol/ng protein.

Comparing the effect of PDE inhibitors on normoxic samples against hypoxic samples, there was no significant difference between the cAMP levels reached in normoxia or hypoxia (*Figure 4.4b*).

4.2.1.2 Extracellular cAMP levels

As intracellular cAMP levels in hypoxic hPASMC were not witnessed to rise in response to PDE inhibition, it was hypothesised that the basal levels of cAMP in hypoxia were maximal and extra cAMP generated might have been extruded from the cell. Thus, I set out to determine extracellular levels of cAMP in response to PDE inhibition also.

Extracellular cAMP levels in response to hypoxia were seen to rise to approximately double that of the normoxic extracellular cAMP levels (*Figure 4.5a, b*). Basal normoxic extracellular cAMP levels were 12 pmol cAMP /ng protein, and hypoxic extracellular cAMP levels were 25.3 pmol cAMP /ng protein (*Figure 4.5a*). In normoxic cells, both cilostamide and IBMX were observed to increase the extracellular cAMP levels to three times that of basal with an increase to 25.9 pmol cAMP /ng protein and 24.7 pmol cAMP /ng protein respectively (*Figure 4.5b*), reaching approximately the same levels as hypoxic cells under the same treatment. Rolipram appeared to have no effect on normoxic extracellular cAMP levels over basal. In hypoxic cells, rolipram, cilostamide and IBMX appeared to have no effect on the extracellular cAMP levels compared to basal (*Figure 4.5a*).

Comparing the effects of rolipram, cilostamide and IBMX on hPASMC cultured in normal conditions on cAMP extracellular levels against the equivalent hypoxic samples showed no significant difference in cAMP levels between normoxic and hypoxic cells, regardless of treatment. This indicates that the hypoxic cells were not generating any extra cAMP in response to PDE inhibition.

4.2.2 cAMP levels in normoxic and hypoxic hPASMC in response to cAMP effectors

Seven day hypoxic and normoxic cells were pre-treated with the non-selective PDE inhibitor, IBMX, for 20 min prior to being treated with either the β -adrenoceptor agonist isoproterenol or the adenylyl cyclase stimulator forskolin for 5 mins or isoproterenol only for 5 min. The cells were then harvested for analysis of intracellular cAMP levels. The growth medium the cells were maintained in was also aspirated to examine extracellular cAMP levels. cAMP levels were measured as described in section 2.3.7.

4.2.2.1 Intracellular cAMP levels

Intracellular levels of cAMP in normoxic cells were seen to increase in response to isoproterenol, IBMX and isoproterenol and IBMX and forskolin treatment (p<0.001) to 3.6 times, 4.4 times and 13.3 times that of normoxic basal levels respectively (*Figure 4.6a*). Treating hypoxic cells did not have the same effect (*Figure 4.6a*). Instead; isoproterenol only increased hypoxic cAMP levels to 1.6 times that of hypoxic basal. Using both isoproterenol and IBMX together raised hypoxic cAMP levels to 1.7 times that of hypoxic basal levels and IBMX used in conjunction with forskolin raised cAMP levels to only 2.9 times that of hypoxic basal (*Figure 4.6a*).

Isoproterenol increased normoxic cAMP levels to 20.2 pmol cAMP /ng protein and hypoxic cAMP levels to 22.4 pmol cAMP /ng protein. IBMX and isoproterenol increased normoxic cAMP levels to 25.8 pmol cAMP /ng protein and hypoxic cAMP levels to 27.6 pmol cAMP /ng protein. However, IBMX and forskolin increased normoxic cAMP levels to 81.8 pmol cAMP /ng protein whereas they increased hypoxic cAMP levels to only 59.3 pmol cAMP /ng protein (*Figure 4.6b*).

The increased levels of cAMP in response to β -adrenergic stimulation was therefore approximately equal in both normoxic and hypoxic cells with no significant difference between them (*Figure 4.6b*), although hypoxic basal levels are higher than that of normoxic basal levels. Inhibition of PDE activity potentiated the effect of isoproterenol, but, again, there was no significant difference between the normoxic and hypoxic cAMP levels. Maximally raising cAMP levels with the adenylyl cyclase stimulator, forskolin, and at the same time inhibiting PDE activity, served to raise normoxic cAMP levels to approximately 81.8 pmol cAMP/ng protein. In hypoxic cells, cAMP was only elevated to 59.3 pmol cAMP/ng protein in response to IBMX and forskolin treatment. Therefore, the previously reported attenuation of the cAMP response in the chronic hypoxic rat is also apparent in chronically hypoxic hPASMC (Wagner et al., 1997).

Comparing the effect of isoproterenol, isoproterenol and IBMX, and forskolin with IBMX, there was no significant difference between the cAMP levels reached in normoxic or hypoxic cells (*Figure 4.6b*).

4.2.2.2 Extracellular cAMP levels

Treating normoxic and hypoxic hPASMC with isoproterenol had no significant effect on extracellular cAMP levels compared to their respective basal cAMP levels (*Figure 4.7a, b*)

This was also the case for treatment with IBMX and isoproterenol or IBMX or forskolin. Although these treatments did induce a rise in normoxic extracellular cAMP levels, this increase was not significant over basal. Comparing the extracellular cAMP levels in response to these stimulations, it was also observed that there was no significant difference between normoxic and hypoxic extracellular cAMP levels.

4.2.3 cAMP levels in normoxic and hypoxic hPASMC in response to U0126

It had been previously reported that active ERK within smooth muscle cells can feed into the PGE₂ pathway and lead to an increase of cAMP levels through activating adenylate cyclase (Baillie et al., 2001; see figure 1.3). ERK inhibitors were then used to determine if this pathway was in effect in hPASMC. U0126 is an ERK inhibitor (Duncia et al., 1998) that works by inhibiting MEK1 (IC₅₀ = 72nM) and MEK2 (IC₅₀ = 58nM). As it was used at 10 μ M, both MEK1 and MEK2 were fully inhibited. Normoxic and hypoxic cells were treated with U0126 for a period of either two or forty-eight hours and cells immediately assayed for levels of intracellular cAMP. In this set of assays, basal normoxic cAMP levels were 192.7 pmol cAMP/ng protein and untreated hypoxic cAMP levels were 1531.8 pmol cAMP/ng protein (893% of basal; *Figure 4.8a*).

It was discovered that U0126 had no significant effect on cAMP levels in normoxic cells. In hypoxic cells however, cAMP levels were significantly reduced to $434 \pm 138\%$ of basal levels (p<0.05). This effect was further potentiated after 48 hrs with a reduction in cAMP to $315 \pm 135 \%$ of basal (p<0.01; *figure 4.8a, b*). By comparing normoxic and hypoxic values against each other, it was observed that there was no longer any significant difference in cAMP levels following treatment with U0126 (*Figure 4.8b*). In addition, when comparing the hypoxic cAMP levels in response to U0126 with normoxic basal cAMP levels, there was no significant difference (*Figure 4.8c*).

4.2.4 cAMP levels in normoxic and hypoxic hPASMC in response to PD980589

PD980589 is another ERK inhibitor that works by inhibiting MEK1 (IC50 = 4 μ M) and MEK2 (IC₅₀ = 50 μ M) and as such was also used to investigate the effect of inhibiting ERK on cAMP levels to confirm the results seen with U0126 (Alessi et al., 1995). By using PD980589 at 20 μ M, only MEK1 was fully inhibited. In this set of assays, basal normoxic cAMP levels were 192.7 pmol/ng protein and basal hypoxic cAMP levels were 1531.8 pmol/ng protein (893% of basal; *Figure 4.8a*).

122

It was discovered that PD980589 had no effect on cAMP levels in normoxic cells after 2 hrs and 48 hours treatment (*Figure 4.8a,b*) In hypoxic cells, PD980589 reduced cAMP levels significantly to $319 \pm 96\%$ of basal levels (p<0.01) and this reduction was maintained after 48 hours at 293 \pm 83% (p<0.01; *figure 4.8a,b*). By comparing normoxic and hypoxic values against each other, it was observed that as with U0126 treatment, there was no longer any significant difference following treatment with PD980589 (*Figure 4.8b*). In addition, when comparing the hypoxic cAMP levels in response to PD980589 with normoxic basal cAMP levels, there was no significant difference (*Figure 4.8c*).

4.2.5 Effect of PGE₂ and indomethacin on cAMP levels in normoxic and hypoxic hPASMC

As inhibition of the ERK pathway reduced cAMP levels in hypoxic cells to the equivalent of that in normoxia, it was proposed that the autocrine loop was indeed in effect in these cells. In order to address this experimentally, hPASMC were exposed to either PGE₂ or indomethacin to directly stimulate adenylyl cyclase or inhibit the actions of COX-2 and thus reduce production of endogenous PGE₂. Normoxic and hypoxic cells were treated with 1 μ M PGE₂ and 10 μ M indomethacin for a period of 10 mins or 2 hours respectively and then intracellular cAMP levels were assayed. In this set of assays, basal normoxic cAMP levels were 81.5 pmol/ng protein and basal hypoxic cAMP levels were 1416.9 pmol/ng protein (1738% of basal; *Figure 4.9b*).

It was discovered that PGE₂ treatment raised normoxic cAMP levels to $1787 \pm 131\%$ of basal (p<0.001; *figure 4.9a, b*). In hypoxic cells however, PGE₂ had no significant effect on cAMP levels with only a reduction to $1538 \pm 29\%$ of basal (*Figure 4.9a, b*), suggesting AC function is compromised in hypoxic cells. Treatment with the COX inhibitor, indomethacin, reduced cAMP levels to $212 \pm 5\%$ that of basal levels in normoxic cells. cAMP levels in hypoxic indomethacin treated cells were reduced to $304 \pm 38\%$ of basal, a level at which there was no significant difference between normoxic and hypoxic indomethacin treated samples (*Figure 4.9b*).

4.3 Discussion and Conclusions

It was observed in the previous chapter that exposing hPASMC to hypoxic conditions altered their PDE4 profile when compared to normoxic hPASMC. The increased expression of PDE4 did not result in higher levels of cAMP hydrolysis however as no change in PDE4 activity or indeed total cAMP-PDE activity was observed in response to hypoxia. I then chose to investigate cAMP levels in response to cAMP effectors to see if they could induce different effects within hypoxia and normoxia and investigate if desensitisation was apparent in hypoxic hPASMC.

Hypoxia was observed to induce a three-fold increase in basal levels of cAMP in hypoxic hPASMC. This is in contrast with previous studies that have reported a reduction in cAMP in hypoxic rat lung (MacLean et al., 1997). However, the increase in cAMP levels in hypoxia was confirmed by an approximate two-fold increase in the activity of the cAMPdependant kinase, PKA, after 7 days hypoxia. This increased activity was not due to an increase in PKA RIa, PKA RIIa or PKA RIIB expression. It is conceivable that as PKA RIB protein levels were not investigated, then this PKA isoform might possibly be increased in hypoxia. It has also been indicated in studies in mice with PKA RI or RII 'knocked-out' that adaptive changes occur in the other subunit to compensate, thus it may be difficult to see any change of expression (Veugelers et al., 2003). The difference between these studies can be explained by the different time point of hypoxia studied. MacLean et al. measured cAMP levels after 14 days of hypoxia and it is possible that cAMP levels are modulated throughout this time course. Indeed, the effect of hypoxia on PDE3 is suggestive of an initial increase in cAMP levels as the increase in the PDE3A isoform observed in hPASMCs is PKA-dependant and can be mimicked by cAMP analogs (Murray et al., 2002). Also, results in chapter 3 demonstrate that the cAMP-inducible PDE4 isoforms present within hPASMC display an increased expression after 7 days hypoxia.

PDE inhibitors have been previously shown to reduce smooth muscle cell proliferation and induce vasodilation of the pulmonary circulation (Koyama et al., 2001; Ghofrani et al., 2004; Wagner et al., 1997; Bardou et al., 2002; Goirand et al., 2001; Palmer et al., 1998; Osinski & Shror 2000), both of which are characteristic responses to elevated cAMP. The
response to the inhibition of each PDE family differs in magnitude. For example, rolipram has been observed to be more potent than PDE3 or PDE5 inhibitors at relaxing human intralobar pulmonary arteries (Bardou et al., 2002). Thus, the levels of cAMP generated in response to PDE inhibition were measured. In normoxic hPASMC, PDE4, PDE3 and general PDE inhibition elevated cAMP levels to approximately three times that of basal. Surprisingly, PDE inhibition did not appear to elevate cAMP over hypoxic basal in hypoxic cells. As responses to PDE inhibition are still apparent in hypoxic models (Wagner et al., 1997), it was hypothesised that the basal levels of cAMP in hypoxic cells represented a threshold level above which, excess cAMP was extruded from the cells. Measuring extracellular cAMP levels were equal in basal hypoxic cells and hypoxic cells with PDE inhibition. Intriguingly however, it was demonstrated that whereas the extracellular levels of cAMP in normoxic cells in response to cilostamide and IBMX were increased, extracellular cAMP levels.

The levels of cAMP in response to isoproterenol and forskolin was measured in normoxic and hypoxic hPASMC to determine if the attenuated responses to these agents in hypoxic models (Leon-Velarde et al., 2001; Voelkel et al., 1981; Wagner et al., 1997) is due to a reduced production of cAMP. Indeed, this was demonstrated to be the case. In normoxic cells, isoproterenol alone and isoproterenol with IBMX, increased cAMP levels to approximately four times that of basal. IBMX and forskolin treated cells maximally increased cAMP to thirteen times basal levels. In hypoxic cells however, isoproterenol with and without IBMX only managed to increase cAMP levels to approximately double that of hypoxic basal cells. The amount of cAMP generated in response to IBMX and forskolin was also reduced in hypoxic cells as it was only increased to three times that of hypoxic basal levels. Although hypoxic basal levels are higher than that of normoxic basal cAMP levels, the increased cAMP levels in response to isoproterenol and forskolin stimulation was greater in normoxic cells. The reduced cAMP in response to isoproterenol and forskolin stimulation in hypoxic cells was not due to excess cAMP being extruded from the cells as extracellular cAMP levels were approximately equal in normoxic and hypoxic cells.

ERK is known to regulate PDE4 enzymatic activity. It has been previously shown that ERK inhibits the activity of the long forms PDE4B1, 4C2 and 4D5 as they have an ERK docking motif within their catalytic domain (MacKenzie et al., 2000). PDE4A enzymes are

unaffected as they lack the ERK phosphorylation motif (Baillie et al. 2000). As PDE4D5 is seen to increase too in hypoxia, I hypothesised that due to the known increase in pERK in hypoxia it may become inhibited constitutively by being phosphorylated by ERK. I thus investigated the effects of ERK inhibitors on cAMP levels to see if this could perhaps release the PDE4D5 from inhibition and allow it to hydrolyse cAMP within hypoxia. It was observed that the ERK inhibitors PD985089 and U0126 reduced hypoxic cAMP levels to half of hypoxic basal levels. This reduced level of cAMP was not significantly different to normoxic basal levels. This indicates ERK regulates the increased level of cAMP witnessed initially in hypoxia. However, ERK inhibition of PDE4D5 is transient due to the resultant PKA phosphorylation of PDE4D5 overriding any effect of ERK, indicating another mechanism for ERK regulation of cAMP levels.

Previous research into PDE4 in smooth muscle cells revealed an autocrine loop whereby active ERK fed into the PGE₂ pathway leading to a rise in cAMP, activating PKA thus activating PDE4D5 and decreasing cAMP levels (Baillie et al 2001; see figure 1.3). It is known that levels of PGE₂ and active ERK are increased in hypoxia (Shaul et al., 1991; Jin et al., 2000), also COX-2 expression is observed to increase in response to sustained hypoxia (Bradbury et al., 2002; Yang et al., 2002). Thus, I set out to investigate if this is how ERK affects cAMP levels in hypoxia by examining cAMP levels in response to PGE₂ and indomethacin. I observed that PGE₂ increased normoxic cAMP levels to that of basal hypoxic cells and indomethacin reduced hypoxic cAMP levels to that of basal normoxic cells. In hypoxic cells, the response to exogenous PGE₂ was desensitised, with no increase in cAMP levels over basal. This indicates that the autocrine effect observed previously in human aortic smooth muscle cells (Baillie et al., 2001) was in effect within my cells. Thus an increased level of cAMP could be due to an increase in endogenous PGE observed previously in hypoxia (Bradbury et al., 2002).

The results shown in this chapter demonstrate desensitisation to cAMP-elevating agents in hypoxia is due to reduced cAMP production in response to G_s stimulation. Desensitisation was most likely caused by the prolonged exposure to increased cAMP induced by 7 days of hypoxia as maintained exposure of cells to cAMP analogs or AC stimulators has previously been shown to cause desensitisation (Moon et al., 2002). In addition, the known elevation of ERK in hypoxia (Jin et al., 2000) I have shown here is intimately involved in mediating the hypoxia-induced increase of cAMP. This presumably occurs through the actions of ERK on PLA₂ and the generation of PGE₂, with consequent autocrine effect

(Pinelli et al., 1999), coupled perhaps with the ability (Baillie et al., 2000) of ERK to modify the activity of members of the PDE4 family.



Figure 4.1 Effect of chronic hypoxia on intracellular cAMP levels.

Cell lysates were assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7. Normoxic and hypoxic samples show 5.5 ± 2.2 and 15.8 ± 1.3 pmol cAMP/ng protein respectively. Results are expressed as mean +/- S.E. of 3 independent experiments as pmol cAMP per ng protein and * denotes significance (p<0.05) between normoxic and hypoxic cAMP levels.



Figure 4.2 Effect of chronic hypoxia on PKA activity.

Cell lysates were assayed for PKA activity in normoxic and seven day hypoxic hPASMC as described in section 2.3.7. Normoxic and hypoxic samples show 5.5 ± 4.5 % and 23.4 ± 6.8 % activated PKA respectively. Results are expressed as mean +/- S.E. of 2 independent experiments as percentage of activated PKA.

N H kDa RIα 50 RII a 50 RII ß 50 8000-Normoxic 7000 □ Hypoxic Density (INT/mm2) 6000-5000 4000-3000-2000-1000 0 Rild RIG. RIIP



Panel A, seven day hypoxic and corresponding normoxic cells were harvested as described in section 2.3.1.1 and used for western blotting with antibodies raised against the specific subunits of PKA. All isoforms are 50kDa in weight so blots were done separately. The blots shown are representative of three individual experiments.

Panel B, densitometry analysis of bands for each subunit expressed as mean \pm S.E. of 3 independent blots as INT/mm².

A

B



Cilostamide	3 +/- 0.4	1 +/- 0.1
IBMX	3.2 +/- 0.4	1.0 +/- 0.06

Figure 4.4a Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on intracellular cAMP levels.

Cells were pre-treated with 10μ M rolipram, 10μ M cilostamide or 10μ M IBMX for 20 min and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E. of fold normoxic basal. *Panel B*, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples.

Results shown are mean \pm S.E. of three independent experiments and *** denotes significance (p<0.001).



	Normoxic	Hypoxic
Basal	5.5	15.8
Rolipram	16.4	18.3
Cilostamide	14.5	15.7
IBMX	24.7	19.3

Figure 4.4b Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on intracellular cAMP levels.

Cells were pre-treated with 10μ M rolipram, 10μ M cilostamide or 10μ M IBMX for 20 min and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, approximate pmol cAMP/ng protein

Results shown are mean \pm S.E. of three independent experiments and * denotes significance (p<0.05).



B

A

15	Normoxic	Hypoxic
Basal	1	1
Roli	0.8 +/- 0.1	1.1 +/- 0.1
Cilo	2.9 +/- 1	1 +/- 0.03
IBMX	2.7 +/- 1	1 +/- 0.07

Figure 4.5a Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on extracellular cAMP levels.

Cells were pre-treated with 10µM rolipram, 10µM cilostamide or 10µM IBMX for 20 minutes and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP +/- S.E. of fold normoxic basal. *Panel B*, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples.

Results shown are mean +/- S.E. of three independent experiments.

 $\begin{array}{c}400\\300\\\hline\\300\\\hline\\200\\\hline\\100\\\hline\\100\\\hline\end{array}$

Roli

	Normoxic	Hypoxic
Basal	12	25.3
Roli	11.4	27
Cilo	18.5	25.9
IBMX	16.9	24.7

Cilo

IBMX

Figure 4.5b Effect of PDE inhibitors in chronic hypoxia compared with normoxic levels on extracellular cAMP levels.

Cells were pre-treated with 10μ M rolipram, 10μ M cilostamide or 10μ M IBMX for 20 min and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, approximate pmol cAMP/ng protein

Results shown are mean +/- S.E. of three independent experiments.

A

Extracellular cAMP (% normoxic

B

0

Basal

■ Normoxic ⊐ Hypoxic



B

	Normoxic	Hypoxic
Basal	1	1
Isoproterenol	3.6 +/- 1	1.6 +/- 0.6
IBMX + Iso	4.4 +/- 1	1.7 +/- 0.4
IBMX + Forsk	13.3 +/- 1.6	2.9 +/- 0.5

Figure 4.6a Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on intracellular cAMP levels.

Cells were pre-treated with 10μ M IBMX for 20 minutes prior to stimulation with 10μ M forskolin or 10μ M isoproterenol for 5 minutes, or only stimulated for 5 minutes with isoproterenol and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E. of fold normoxic basal.

Panel B, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples. Results shown are mean \pm S.E. of three independent experiments and *** denotes significance (p<0.001).



Figure 4.6b Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on intracellular cAMP levels. Cells were pre-treated with 10μ M IBMX for 20 minutes prior to stimulation with 10μ M forskolin or 10μ M isoproterenol for 5 minutes, or only stimulated for 5 minutes with isoproterenol and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, *approximate* pmol cAMP/ng protein.

Results shown are mean \pm S.E. of three independent experiments.

137

A



	Normoxic	Hypoxic
Basal	1	1
Iso	2 +/- 0.1	1 +/- 0.1
Iso + IBMX	2.9 +/- 0.7	1.1 +/- 0.1
Forsk + IBMX	2.1 +/- 0.8	1 +/- 0.1

Figure 4.7a Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on extracellular cAMP levels.

Cells were pre-treated with 10μ M IBMX for 20 minutes prior to stimulation with 10μ M forskolin or 10μ M isoproterenol for 5 minutes, or only stimulated for 5 minutes with isoproterenol and were immediately assayed for extracellular cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP +/- S.E. of fold normoxic basal. *Panel B*, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples.

Results shown are mean +/- S.E. of three independent experiments.



Figure 4.7b Effect of cAMP effectors in chronic hypoxia compared with normoxic levels on extracellular cAMP levels.

Cells were pre-treated with 10μ M IBMX for 20 minutes prior to stimulation with 10μ M forskolin or 10μ M isoproterenol for 5 minutes, or only stimulated for 5 minutes with isoproterenol and were immediately assayed for extracellular cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, approximate pmol cAMP/ng protein.

Results shown are mean +/- S.E. of three independent experiments.

A



B

	Normoxic	Hypoxic
Basal	1	1
U0126 2 hrs	1.6 +/- 0.5	0.5 +/- 0.2
UO126 48 hrs	1.4 +/- 0.4	0.4 +/- 0.2
PD980589 2 hrs	1.4 +/- 0.5	0.4 +/- 0.1
PD980589 48 hrs	1.6 +/- 0.6	0.4 +/- 0.1

Figure 4.8a Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC.

Cells were treated with 10μ M U0126 or 20μ M PD980589 for 2 and 48 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP +/- S.E. of fold normoxic basal. *Panel B*, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples.

Results shown are mean +/- S.E. of four independent experiments and *, ** denotes significance (p<0.05, p<0.01 respectively).



Figure 4.8b Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC.

Cells were treated with 10μ M U0126 or 20μ M PD980589 for 2 and 48 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, approximate pmol cAMP/ng protein.

Results shown are mean +/- S.E. of four independent experiments and ** denotes significance (p<0.01).



Figure 4.8c Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC.

Cells were treated with 10µM U0126 or 20µM PD980589 for 2 and 48 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7. This graphs shows hypoxic levels of cAMP in response to ERK inhibition compared to normoxic basal cAMP levels.

Results shown are mean +/- S.E. of four independent experiments.



Figure 4.8c Effect of MEK inhibitors on cAMP levels in hypoxic and normoxic hPASMC.

Cells were treated with 10µM U0126 or 20µM PD980589 for 2 and 48 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7. This graphs shows hypoxic levels of cAMP in response to ERK inhibition compared to normoxic basal cAMP levels.

Results shown are mean +/- S.E. of four independent experiments.

2000 1500 1000 500 0 Normoxic Normoxic Hypoxic

B

A

	Normoxic	Hypoxic
Basal	1	1
PGE2	17.9 +/- 1.3	0.9 +/- 0.03
Indomethacin	2.1 +/- 0.05	0.2 +/- 0.03

Figure 4.9a Effect of PGE2 and indomethacin on cAMP levels in hypoxic and normoxic hPASMC.

Cells were treated with $1\mu M PGE_2$ for 10 mins and $10\mu M$ indomethacin for 2 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP +/- S.E. of fold normoxic basal. *Panel B*, results expressed as mean \pm S.E. of fold basal for both normoxic and hypoxic samples.

Results shown are mean +/- S.E. of two independent experiments and *** denotes significance (p<0.001).



Figure 4.9b Effect of PGE2 and indomethacin on cAMP levels in hypoxic and normoxic hPASMC.

261.5

247.8

Cells were treated with 1μ M PGE₂ for 10 mins and 10μ M indomethacin for 2 hours and were immediately assayed for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.7.

Panel A, results graphed as mean cAMP \pm S.E of fold normoxic basal with normoxic against hypoxic analysis. *Panel B*, approximate pmol cAMP/ng protein.

Results shown are mean +/- S.E. of two independent experiments.

Indomethacin

Chapter 5

ERK in Hypoxic hPASMC

5.1 Introduction

5.1.1 The ERK pathway

The mitogen activated protein kinases (MAPK) are activated by a wide range of stimuli including growth factors, hormones, extracellular matrix components, GPCR agonists, cellular stress and cytokines (reviewed in Gutkind 2000; Tibbles & Woodgett 1999; Wildman et al., 1999; see general introduction for further details). The best characterised MAPK are the extracellular regulated kinases 1 and 2, also known as p44 and p42 respectively. The involvement of ERK1/2 in cell proliferation is well established (reviewed in Stork & Schmitt 2002).

ERK1/2 is activated by the binding of e.g. EGF to its tyrosine kinase receptor (RTK) which results in the activation of the Ras – Raf – MEK1/2 – ERK1/2 pathway (reviewed in Kolch 2000). It was originally thought that ERK1/2 could also be activated by cAMP through Epac activating Rap1 and thus B-Raf, leading to MEK1/2 activation and finally the activation of ERK (*figure 1.5*). However, recent work has demonstrated that Rap1 activation is PKA-independent and does not act through B-Raf to lead to the activation of ERK, instead Rap1 is now known to activate 'inside-out' signalling to integrins (Bos et al., 2003). This presents a model whereby cAMP activation of ERK via Raf is mediated through PKA and activation of Epac leads to integrin activation which has also been demonstrated to activate ERK (*figure 5.1*). ERK is activated specifically by dual phosphorylation (Prowse & Lew 2001) and, upon activation, moves to the nucleus to act upon its transcription factors and becomes dephosphorylated (Volmat et al., 2001). ERK remains inactive in the nucleus until it translocates back to the activating MEK in the cytoplasm (Peyssonnaux, et al., 2001).

5.1.1.1 Regulation of ERK activation by cAMP

Regulation of the ERK cascade by the cAMP pathway provides vital crosstalk between hormone and growth factor signalling (reviewed in Houslay & Kolch 2000). ERK can be inhibited or activated by cAMP in a cell specific manner (reviewed in Stork & Schmitt 2002).

Inhibition occurs through the actions of PKA. PKA blocks Raf-1 activation and inhibits Raf-1 activity (Cook & McCormick 1993; Mischak et al., 1996), thus blocking Rasdependent signals to ERK. PKA has been demonstrated to act indirectly through phosphorylation of Src upon Rap1 to inhibit ERK (Ribeiro-Neto et al., 2002; Schmitt & Stork 2002b). Another PKA-mediated inhibition of ERK is the activation of Akt (PKB) which inhibits Raf-1 (Zimmerman & Moelling 1999) and thus prevents activation of ERK. cAMP can also inhibit ERK through activation of MAP kinase phosphatases (MKP; Burgun et al., 2000)

cAMP can lead to the activation of ERK. In cell types that express B-Raf, for example endothelial cells (Wojnowski et al., 1997), PKA activates B-Raf, leading to activation of ERK. PKA activation of Rap1 through Src can therefore lead to activation as well as inhibition of ERK. Indeed, ERK activation by PKA has been shown to require members of the Src family of kinases (Lindquist et al., 2000). The recently discovered cAMP substrate, Epac (de Rooij et al., 1998) may also activate Rap1 independently of PKA.

5.1.1.2 Regulation of cAMP levels by ERK

ERK is also capable of regulating cAMP levels through the actions of the cAMP-specific PDE4 family. Although isoforms encoded by the PDE4A, PDE4B, PDE4C and PDE4D genes contain the ERK2 docking sites, KIM and FQF (MacKenzie et al., 2000), PDE4A does not act as a substrate for ERK (Baillie et al., 2000). Phosphorylation of PDE4 long forms by ERK2 has been demonstrated to lead to a 75 % reduction in enzyme activity (Hoffmann et al., 1999). This ERK-mediated inhibition of PDE4 is transient as the result is a rise in cAMP levels, and the consequential activation of PKA. PKA phosphorylation of PDE4 negates the ERK2 phosphorylation (Baillie et al., 2000; Hoffmann et al., 1999), hence the reversal of inhibition of PDE4 activity.

ERK can also lead to activation of the long form, PDE4D5 (Baillie et al., 2001, figure 1.3). This is apparent in human aortic smooth muscle cells whereby activated ERK caused an increase in PGE_2 which was released from the cell and activated adenylyl cyclase. This induced a rise in cAMP levels with a concomitant increase in PKA activity and resultant PDE4D5 phosphorylation. This phosphorylation ablated the effect of ERK phosphorylation and a net activation of PDE4D5 was observed.

ERK2 phosphorylation of PDE4 short forms results in their activation (Baillie et al., 2000). This differential regulation of PDE4 activity by ERK2 is due to the upstream conserved region 1 (UCR1) which is only found in PDE4 long forms and contains the PKA phosphorylation site. Intriguingly, the super short form PDE4D2, which lacks UCR1 and

has a truncated UCR2, is inhibited by ERK2 phosphorylation, indicating the requirement of the N-terminal portion of UCR2 for activation of PDE4 short forms (MacKenzie et al., 2000). As PDE4D2 also lacks the PKA phosphorylation site, it is not subject to feedback regulation by PKA.

5.1.1.3 ERK in hypoxia

The activation of ERK in hypoxia has been characterised in pulmonary artery fibroblasts, pulmonary artery endothelial cells and in pulmonary arteries from the chronic hypoxic rat (Minet et al., 2000b; Scott et al., 1998; Welsh et al., 2001; Jin et al., 2000). ERK1/2 phosphorylation has been shown to promote the transcriptional activity of HIF-1a (Richard et al., 1999) and therefore the transcription of the vascular endothelial growth factor, VEGF.

In this chapter, I set out to investigate ERK expression specifically in hypoxic hPASMC. In chapter 4, I observed ERK to play a key role in the regulation of hypoxic cAMP levels in hPASMC. As ERK is known to achieve this regulation through the PDE4 family, I thus also set out to investigate the effect of ERK on PDE4 activity and expression in an attempt to uncover the hypoxia mediated rise in cAMP. Also, I investigated the effect of PDE4 on active ERK expression by employing siRNA technology.

Results

5.2 The autocrine loop and PDE activity in hypoxic hPASMC

Results from the previous chapter indicated an autocrine loop whereby active ERK can lead to an increase in cAMP levels through the production of PGE_2 and stimulation of adenylyl cyclase. In a previous study, this mechanism has been shown to override inhibition of PDE4D5 by ERK and lead to net activation of PDB4D5 in human aortic smooth muscle cells due to the rapidity of the response and the magnitude of the resultant PKA phosphorylation of PDE4D5 (Baillie et al., 2001).

5.2.1 Effect of ERK inhibitors on PDE4 activity

In order to determine the effect of ERK on overall PDE4 activity in these cells, cells were treated with the MEK inhibitors, 20μ M PD980589 and 10μ M U0126 for either 30 min or 48 h and then harvested for use in a cAMP-PDE activity assay.

5.2.1.1 Effect of PD980589 on PDE4 activity

Treating cells with 20μ M PD980589 for 30 mins or 48 hours had no significant effect on total cAMP-PDE activity or PDE4 activity in normoxic or hypoxic hPASMC (*Figure 5.2*).Normoxic basal PDE4 activity was 9.0 ± 1.3 pmol/min/mg protein, 30 minutes treatment reduced this to 8.0 ± 0.5 pmol/min/mg protein and 48 hours treatment reduced this to 6.4 ± 0.8 pmol/min/mg protein (SEM; n=2). Hypoxic cells displayed a similar trend with basal PDE4 activity being 9.0 ± 0.2 pmol/min/mg protein, 30 mins treatment reduced this to 7.5 ± 1.8 pmol/min/mg protein and 48 hours reduced PDE4 activity to 6.7 ± 0.1 pmol/min/mg protein (SEM; n=2).

5.2.1.2 Effect of U0126 on PDE4 activity

Treating cells with 10µM U0126 for 30 min or 48 hours again had no significant effect on either total cAMP-PDE activity or PDE4 activity in either normoxic or hypoxic hPASMC (*Figure 5.3*). Normoxic basal PDE4 activity was 9.6 \pm 0.5 pmol/min/mg protein , 30 minutes treatment with 10µM U0126 had no significant effect on PDE4 activity, whereas 48 hours treatment with 10µM U0126 significantly reduced PDE4 activity to 6 \pm 0.5 pmol/min/mg protein (SEM; n=4; p<0.05). Hypoxic basal PDE4 activity was 7.7 \pm 1.1 pmol/min/mg protein and 30 mins treatment with 10µM U0126 had no significant effect on this. 48 hours of treatment with 10μ M U0126 induced a small, but insignificant increase in PDE4 activity to 9 ± 0.3 pmol/min/mg protein (SEM; n=4).

These results suggest little effect of ERK on total PDE4 activity. However, ERK differentially regulates long, short and super short forms which are all present in hPASMC. PDE4B2 and PDE4D1, whose expression increases in hypoxia, can potentially be activated by ERK phosphorylation, so inhibition of ERK might be expected to reduce their activity if they were so ERK phosphorylated. PDE4D2 and PDE4D5, whose expression is also increased in hypoxia, can be inhibited upon phosphorylation by ERK. Thus inhibiting ERK might be expected to alleviate this and thereby activate these isoforms if they were so ERK phosphorylated. Hence, ERK inhibition might be expected to result in contrasting effects on the activity of various PDE4 isoforms in hPASMC. Indeed, this may be complicated further if the actions of ERK are compartmentalised and only certain PDE4 isoforms are available for phosphorylation by ERK.

5.2.1.3 Effect of U0126 on ERK phosphorylation

To ensure that ERK was fully inhibited for the full 48 h, cells were treated with U0126 for 30 min and 48 h and prepared for gel electrophoresis. The proteins were transferred onto nitrocellulose and probed with anti-pERK to detect levels of active ERK. Figure 5.3 shows that U0126 does inhibit ERK activity after 30 minutes and 48 hours treatment as the densitometry results compared with basal levels show. (*Figure 5.4*)

5.2.2 PDE Activity in response to PGE₂ and indomethacin in hypoxic hPASMC

As the cAMP assay results in the previous chapter indicate that active ERK feeds into an autocrine pathway involving generation of PGE₂, I set out to investigate the effect of PGE₂ and the COX inhibitor, indomethacin on cAMP-PDE activity in normoxic and hypoxic hPASMC. PGE₂ directly stimulates adenylyl cyclase to increase cAMP while indomethacin inhibits the actions of COX and thus reduces production of endogenous PGE₂ in response to elevated ERK. In order to determine the effect of PGE₂ and indomethacin on overall PDE4 activity in these cells, cells were treated with 1µM PGE₂ for 10 mins and 10µM indomethacin for 2 hours and then harvested for use in a cAMP-PDE activity assay.

5.2.2.1 Total cAMP-PDE activity

As can be seen in panel A of fig 5.5, both PGE_2 and indomethacin had no effect on hypoxic or normoxic total cAMP-PDE activity.

5.2.2.2 PDE4 activity

Seven day hypoxic and normoxic cells were treated with 1μ M PGE₂ for 10 min or 10 μ M indomethacin for 2 hours. In figure 5.5, panel B, the results are shown. Basal PDE4 activity was 5.7 ± 0.2 pmol/min/mg protein and 6 ± 0.2 pmol/min/mg protein in normoxic and hypoxic cells respectively. Addition of PGE₂ significantly reduced PDE4 activity in normoxic cells to 4.6 ± pmol/min/mg protein (p<0.05). In hypoxic cells however, an insignificant increase in PDE4 activity to 7.2 ± pmol/min/mg protein was observed (SEM; n=2). An increase in PDE4 activity was expected as PGE₂ activates adenylyl cyclase. The COX-2 inhibitor indomethacin caused a decrease in both normoxic and hypoxic PDE4 activity being reduced to 2.2 ± 1.5 pmol/min/mg protein (SEM; n=2). These results are indicative of an autocrine effect in hPASMC, and suggest that this loop is enhanced in hypoxia through increase PDE4 activity in analyses are required.

5.2.3 Effect of U0126 and Indomethacin on PDE4 expression levels

As U0126 and indomethacin were both observed to normalise hypoxic intracellular cAMP levels, their effect on the expression levels of the PDE4 isoforms previously observed to increase in hypoxia was investigated. Hypoxic and normoxic cells were treated with 10μ M U0126 or 10μ M indomethacin for 24 hours and harvested as described in section 2.3.1.1 for use in western blotting.

5.2.3.1 PDE4A10/4A11

Lysates from hypoxic and normoxic cells treated with 10µM U0126 and 10µM indomethacin for 24 hours and were run out on gels alongside basal controls and a PDE4A10 protein standard, then transferred to nitrocellulose and probed with an anti-PDE4A antibody to investigate PDE4A10/4A11 expression levels. The resulting blots were scanned for densitometry analysis. As can be seen in figure 5.6, panel A and B, U0126 and indomethacin had no significant effect on normoxic or hypoxic protein levels of PDE4A10/4A11, although U0126 did increase normoxic PDE4A levels slightly.

5.2.3.2 PDE4B2

Normoxic and hypoxic lysates from cells treated with 10μ M U0126 or 10μ M indomethacin, as in section 5.2.3.1, were used for gel electrophoresis alongside a 4B2 protein standard and basal controls. The resulting membranes after transfer were probed with an anti-4B antibody. The blots were used for a densitometry analysis. U0126 and indomethacin were found to have no significant effect on normoxic or hypoxic PDE4B2 protein levels after 24 hours (*Figure 5.7*).

5.2.3.3 PDE4D5

Cells maintained in hypoxia for seven days and their corresponding normoxic controls were treated with 10μ M U0126 or 10μ M indomethacin for 24 hr sand used for gel electrophoresis as in section 5.2.3.1. After transfer, the membranes were probed with an anti-PDE4D specific antibody and it was revealed that U0126 and indomethacin had no significant effect on PDE4D5 protein levels. There was a slight decrease in PDE4D5 levels, but this was apparent in both normoxic and hypoxic cells (*Figure 5.8*).

5.3 ERK in normoxic and hypoxic hPASMC

ERK is known to inhibit PDE4B, PDE4C and PDE4D long form activities within cells by causing the phosphorylation of their catalytic unit (Baillie et al., 2000). In contrast to this, ERK is known to activate the PDE4B2 short form by such phosphorylation (Baillie et al., 2000). Numerous studies have revealed a hypoxia-dependant induction of the ERK1/2 pathway in many cell types (Minet et al 2000a). In the chronic hypoxic rat, ERK1 and 2 are shown to be increased in pulmonary arteries (Jin et al., 2000). Due to this and the results from the previous chapter which indicate the involvement of ERK in the regulation of cAMP levels, the effect of activating ERK directly through the use of the epidermal growth factor (EGF) was investigated to determine the effect of ERK activation in hPASMC on PDE4 activity and also the effect of raising cAMP on ERK.

5.3.1 ERK in normoxic and hypoxic hPASMC

5.3.1.1 Effect of EGF on pERK levels

Seven day hypoxic and normoxic cells were treated with 50pg/µl EGF for up to 20 min and then harvested for analysis by gel electrophoresis. After transferring, the membranes were probed with anti-pERK. This revealed that basal levels of pERK in hypoxic cells were not significantly increased compared to normoxic levels although levels of pERK2 were slightly elevated in hypoxia. EGF stimulation increased normoxic levels of pERK1 and 2 to approximately 6 times that of basal and continued to increase until they reached approximately 8 times that of basal levels after 10 minutes of EGF stimulation (*Figure 5.9*). In hypoxic cells, phosphorylation of ERK1 and 2 peaked at approximately 17 times that of hypoxic basal levels at 5 mins stimulation. After 10 mins, ERK phosphorylation was reduced, although at approximately 10 times that of hypoxic basal levels (*Figure 5.9*). This demonstrates that a more rapid and potentiated phosphorylation of ERK occurs in hypoxia than in normoxia.

5.3.1.2 Effect of EGF on PDE4 activity levels

Cell lysates of normoxic cells treated with 50pg/µl EGF for both 4 min and 10 min were used in a cAMP-PDE activity assay in the presence of 10µM rolipram to determine the effect of raising active ERK within normoxic hPASMC on PDF4 activity levels. Assay results showed that PDE4 activity reached 2.5 times that of basal levels after 4min of EGF stimulation and this increased level was maintained up to 10min (*Figure 5.10*). This indicates ERK acts to increase PDE4 activity in hPASMC, presumably through the autocrine loop whereby active ERK leads to active PKA which can phosphorylate and negate any inhibition of ERK on PDE4D5. Also, ERK would directly activate 4B2 and 4D1 present in hPASMC.

5.4 Knockdown of PDE4D5 in normoxic and hypoxic hPASMC

To investigate the relationship determined previously between 4D5 and pERK (Baillie et al., 2001), siRNA technology was employed (Schutze et al., 2004). The discovery of a natural method of gene silencing has provided an invaluable tool in gene functioning studies (Dykxhoorn et al., 2003), although the mechanism by which siRNA works is still unclear.

5.4.1 Effect of PDE4D5 siRNA on PDE4D5 expression

Normoxic hPASMC were transfected with 60nM of the PDE4D5 siRNA construct and harvested after 24 hours incubation as in section 2.3.1.1 for western blotting to examine levels of PDE4D5 expression. This construct worked successfully to knock down levels of endogenous PDE4D5 as can be seen in figure 5.11 with over 80% reduction in PDE4D5 protein levels.

5.4.2 Effect of PDE4D5 siRNA on pERK expression

Seven day hypoxic and normoxic cells transfected with the PDE4D5 siRNA construct were also harvested for analysis of basal pERK expression. It was observed that knocking out PDE4D5 resulted in an increase of basal pERK expression within normoxic cells to approximately double and a decrease in hypoxic cells to 20% of basal hypoxic levels (*Figure 5.12*). This demonstrates that the cross-talk between cAMP and ERK is modified in hypoxia with a reversal of effect.

Many thanks to Dr. M. Lynch, Gardiner Lab who created siRNA constructs.

5.5 Discussion and Conclusions

The results in chapter 4 demonstrate the regulation of cAMP levels through the MAPK, ERK. This has previously been hypothesised to occur through the production of PGE_2 and the consequential AC stimulation (Baillie et al., 2001). Indeed, it is known that ERK, COX-2 and PGE₂ display increased levels in hypoxia (Jin et al., 2000; Scott et al., 1998; Bradbury et al., 2002), contributing to the increased cAMP observed in hPASMC in chapter 4. PDE4 is also involved in this loop as ERK and the cAMP-dependant kinase, PKA, both act upon PDE4 with opposing effects and thus impact upon the regulation of cAMP levels. The increased PDE4D5 discussed in chapter 3 is subject to inhibition by the increased ERK levels in hypoxia. The increased levels of ERK are transient however (Jin et al., 2000; Scott et al., 1998), as hypoxia has also been observed to stimulate MAPK phosphatase 1 (MKP-1; Laderoute et al., 1999). ERK phosphorylation peaks at day 7 (Jin et al., 2000; Scott et al., 1998), the time point studied throughout this investigation. Maximal levels of active ERK would therefore inhibit PDE4D5 activity (Baillie et al., 2000; Hoffmann et al., 1999) and result in a maintained increase of cAMP levels generated by PGE₂, This increase would lead to increased levels of active PKA and activation of PDE4D5 as PKA phosphorylation negates the inhibitory effect of ERK (Baillie et al., 2000). Thus cAMP levels would be reduced and, along with the hypoxia-induced activation of MKP-1 and resultant decrease in ERK activity, PDE4D5 would continue to reduce cAMP levels below that of normal cells.

I thus initially set out to investigate the effect ERK has on PDE4 activity in hPASMC by employing two different ERK inhibitors. Both U0126 and PD980589 inhibit ERK activation by inhibiting its upstream kinases, MEK. U0126 was used at a concentration that inhibited both MEK1 and MEK2, whereas PD980589 was used at a concentration that inhibited MEK1 only. Examining the effects of ERK inhibition on PDE4 activity, inhibiting MEK1 only had no significant effect in either normoxic or hypoxic cells. However, inhibition of both MEK1 and MEK2 induced a significant reduction in PDE4 activity after 48 hours in normoxic cells. This could possibly be due to a reduced generation of PGE₂, and thus reduced cAMP levels, decreased PKA activity and therefore decreased PKA activation of PDE4. This indicates ERK is required to maintain a balanced effect with PKA on PDE4 activity. In hypoxic cells, this effect was not observed. Instead, there was no significant effect on PDE4 activity after 48 hours of U0126 treatment. This could occur due to the desensitised response to PGE₂ on cAMP levels in hypoxic hPASMC observed in chapter 4. This would result in reduced activation of PKA compared to normoxic cells and thus ERK inhibition of PDE4 activity could balance the reduced level of PKA activation of PDE4. This would mean that in normoxic cells, ERK would lead to the activation of PDE4 by PKA and reduce cAMP levels. Indeed activation of ERK by EGF was observed to elevate PDE4 activity to three times that of basal in hPASMC. Raised cAMP levels would remain increased in hypoxic cells due to the inhibition of PDE4 by ERK. This could also be responsible for the lack of increased PDE4 activity in hypoxia demonstrated in chapter 3.

To further examine this effect on PDE4 activity, the effect of either PGE₂ or indomethacin treatment on PDE4 activity was measured. Exogenous PGE₂ had no significant effect on PDE4 activity in hypoxic cells, but induced a small reduction in PDE4 activity in normoxic cells (p<0.05). This contrasts with the earlier results indicating ERK leads to activation of PDE4 through PGE_2 in normoxic cells. However, the amount of PGE_2 added to the cells is far greater than the amount of endogenous PGE₂ produced though ERK activity. As a result, the effects of exogenous PGE_2 will be greatly enhanced compared to that witnessed normally in the cell through ERK. Therefore it is possible that the increased PKA activity caused by this amount of PGE₂ has a greater effect on its substrates. As well as PDE4, the vast range of PKA substrates includes GPCRs. Upon PKA phosphorylation, the "switching" of GPCR from the G_s subunit to the G_i subunit of the G-protein occurs and this leads to increased levels of receptor activated ERK and reduced cAMP. This increased ERK could then override PKA phosphorylation of PDE4 and result in net inhibition. In hypoxic cells, the lack of effect of PGE_2 on PDE4 activity could be explained by the desensitisation of AC mentioned above. Indeed, it was observed in chapter 4 that PGE_2 failed to increase cAMP levels over hypoxic basal levels.

Treating hPASMC with the COX-2 inhibitor indomethacin induced a significant decrease in PDE4 activity in both normoxic and hypoxic cells. This is due to the lack of PGE_2 generated by ERK because of COX-2 inhibition and therefore a decrease in cAMP (as observed in chapter 4) and PKA activity. This would mean PDE4 would be regulated solely by basal levels of ERK and thus an inhibition of PDE4 activity would be apparent. With the increased levels of active ERK in hypoxia, indomethacin would then be expected to have a greater inhibitory effect on PDE4 activity than in normoxic cells. However, the increased basal levels of PKA activity could affect this, and reduce the effect of ERK on PDE4. As ERK and COX-2 inhibition by U0126 and indomethacin reduce cAMP levels in hPASMC, it was hypothesised that this could affect the increase in PDE4 expression in hypoxia. As such, normoxic and hypoxic hPASMC were treated with U0126 or indomethacin and examined for levels of PDE4A, PDE4B and PDE4D expression. U0126 and indomethacin had no effect on PDE4A, PDE4B or PDE4D expression, however, treatment with these compounds was only for 24 hours and a more chronic treatment could display an alteration in PDE4 expression.

ERK activation can occur through a number of direct and indirect pathways (see general introduction for more details; reviewed in Peyssonnaux & Eychene 2001). As such, the activation of ERK by EGF in normoxic and hypoxic hPASMC was investigated. In normoxic cells, EGF induced a gradual activation of ERK, reaching an increase of eight times that of basal levels after ten minutes. In hypoxic cells, the activation of ERK is potentiated and more rapid, peaking after 5 minutes at 17 times that of hypoxic basal levels, and falls again to 10 times that of hypoxic basal levels after 10 minutes.

Following on from the effect of ERK on cAMP and PDE4 activity, siRNA constructs to effectively 'knock-out' PDE4D5 in normoxic and hypoxic hPASMC were used to investigate any effect of PDE4D5 on active ERK. It was observed that a successful reduction of PDE4D5 expression induced increased levels of activated ERK in normoxic hPASMC. This could be due to the increased PKA mediated phosphorylation of the β -adrenoceptor and the resulting increase in receptor activated ERK by G_s to G_i switching. Indeed, it has been observed in cardiomyocytes that rolipram increases levels of ERK through this mechanism (Baillie et al., 2002), although this effect was witnessed with β -adrenoceptor stimulation.

In hypoxic cells, silencing PDE4D5 expression resulted in a significant reduction of pERK levels (p<0.01). This indicates PDE4D5 is essential for phosphorylation of ERK in hypoxic hPASMC and suggests the mode of activation of ERK is altered in hypoxia. ERK can be either inhibited or activated by elevated cAMP levels, usually in a cell-specific manner (reviewed in Houslay & Kolch 2000; Stork & Schmitt 2002). PKA is capable of acting upon Raf, which results in an inhibition of ERK activation in some cell types (Mischak et al., 1996; Cook et al., 1993) and an activation in others. Therefore the effect of cAMP on the ERK pathway is dependent upon the predominant Raf isoform within a cell. The results presented here suggest cAMP activates the ERK pathway in normoxic hPASMC and inhibits ERK activation in hypoxic hPASMC, possibly by switching from

signalling through B-Raf signalling in normoxia to Raf-1 signalling in hypoxia. However, this would contrast with the increase of ERK witnessed in hypoxia (Jin et al., 2000; Scott et al., 1998).

Other studies have demonstrated a switch from cAMP inhibition of proliferation to cAMP stimulated proliferation. In normal human kidney cells, cAMP acts in an anti-proliferative manner. However, in cells derived from polycystic kidney patients, cAMP stimulates proliferation (Yamaguchi et al., 2004). This was due to a disruption in calcium mobilisation that alleviated B-Raf/ERK inhibition through the reduction of Akt.

It has been demonstrated that Rap1 does not always activate B-Raf, (Zwartkruis 1998) and this is dependent on the B-Raf isoform expressed. Thus, if B-Raf activity is altered in hypoxia, this could explain the activation of ERK observed. It could be that PDE4D5 aids the activation of B-Raf by Rap1 and knocking out PDE4D5 prevents this, by increasing cAMP and resulting in an inhibition of ERK, possibly through PKA mediated activation of Src.

The results presented in this chapter show a delicate balance of ERK and PKA phosphorylation exists to regulate PDE4 activity within the autocrine loop demonstrated in smooth muscle cells (Baillie et al., 2001) and indicates a complex method of cAMP regulation, dependant on various factors and tailored by the cell in a concentration dependant manner. In hypoxia, modifications to this pathway appear to result in a confusion of signals attempting perhaps to return to normal signalling.



Figure 5.1 cAMP activation of ERK

Following generation of cAMP, PKA and Epac are activated. PKA can lead to the activation of ERK through B-Raf, or inhibition of ERK through c-Raf-1. Thus, ERK responses to cAMP are dependant on Raf isoforms present within the cell. Activation of Epac has no impact upon PKA signalling to ERK, instead it leads to the activation of integrin signalling (Bos et al., 2003). Integrin-mediated adhesion has been demonstrated to lead to the partial activation of the epidermal growth factor receptor, resulting in the activation of ERK (Bill et al., 2004).

A Total cAMP-PDE activity



Figure 5.2 Effect of PD980589 on PDE Activity in normoxic and hypoxic hPASMC.

Cells were treated with 20µM PD980589 for 30 mins or 48 hours and were immediately harvested to be used in a cAMP-PDE activity assay for cAMP levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.5. *Panel* A, total cAMP-PDE activity results graphed as mean cAMP \pm S.E. *Panel B*, PDE4 activity results graphed as mean \pm S.E. *Panel C*, results expressed as mean \pm S.E. in pmol/min/mg in tabular form. Means \pm S.E. of 2 independent experiments


Figure 5.3 Effect of U0126 on PDE Activity in hypoxic and normoxic hPASMC.

Cells were treated with 10 μ M U0126 for 30 mins or 48 hours and were immediately harvested to be used in a cAMP-PDE activity assay for PDE activity in normoxic and seven day hypoxic hPASMC as described in section 2.3.5. *Panel* A, total cAMP-PDE activity results graphed as mean cAMP ± S.E. *Panel B*, PDE4 activity results graphed as mean ± S.E. *Panel C*, results expressed as mean ± S.E. in pmol/min/mg in tabular form. Means ±S.E. of 4 independent experiments.



Figure 5.4 Effect of U0126 treatment on ERK1/2 phosphorylation.

Seven day normoxic and hypoxic hPASMC were treated with 10µM of the MEK inhibitor, U0126, for differing lengths of time to ensure ERK1/2 phosphorylation was inhibited. Cells were harvested as in section 2.3.1.1 and subjected to gel electrophoresis before being transferred onto a nitrocellulose membrane and probed with and anti-phospho-ERK1/2 antibody. The membrane was then stripped prior to probing with total ERK2 antibody. The blot shown is indicative of three separate experiments. Densitometry analysis of the three experiments is shown below the blot.

A Total cAMP-PDE Activity



Figure 5.5 Effect of PGE₂ and indomethacin on PDE Activity in hypoxic and normoxic hPASMC

4.6 +/- 0.4

7.2 +/- 1.5

Cells were treated with $1\mu M PGE_2$ for 10 mins and $10\mu M$ indomethacin for 2 hours and were harvested to assay for cAMP-PDE activity levels in normoxic and seven day hypoxic hPASMC as described in section 2.3.5. Panel A, total cAMP-PDE activity results graphed as mean cAMP ± S.E. Panel B, PDE4 activity results graphed as mean ± S.E. Panel C, results expressed as mean ± S.E. in pmol/min/mg in tabular form. Means ± S.E. of 2 independent experiments. Significance is as compared with basal and is denoted by *, ** (p<0.05, p<0.01 respectively.)



Figure 5.6 Effect of U0126 and indomethacin on PDE4A10/11 expression levels in hypoxia and normoxia.

Cells were treated with 10μ M U0126 and 10μ M indomethacin for 24 hours and were harvested as described in section 2.3.1.1 20µg protein of each sample was used for SDS-PAGE alongside a basal control sample on a 4-12% gel. *Panel A*, The proteins were transferred to nitrocellulose and probed with a PDE4A antibody. *Panel B*, the densitometry results as compared to normoxic samples expressed in INT/mm². Blots shown are representative of two independent experiments.



Figure 5.7 Effect of U0126 and indomethacin on PDE4B2 expression levels in hypoxia and normoxia.

Cells were treated with 10μ M U0126 and 10μ M indomethacin for 24 hours and were harvested as described in section 2.3.1.1 20µg protein of each sample was used for SDS-PAGE alongside a basal control sample on a 4-12% gel. *Panel A*, the proteins were transferred to nitrocellulose and probed with a PDE4B antibody. *Panel B*, the densitometry results as compared to normoxic samples expressed in INT/mm². Blots shown are representative of two independent experiments.



Figure 5.8 Effect of U0126 and indomethacin on PDE4D5 expression levels in hypoxia and normoxia.

Cells were treated with 10μ M U0126 and μ M indomethacin for 24 hours and were harvested as described in section 2.3.1.1. 20μ g protein of each sample was used for SDS-PAGE alongside a basal control sample on a 4-12% gel. *Panel A*, the proteins were transferred to nitrocellulose and probed with a PDE4D antibody. *Panel B*, the densitometry results as compared to normoxic samples expressed in INT/mm2. Blots shown are representative of two independent experiments.



Fig 5.9 Effect of EGF on pERK in normoxic and hypoxic hPASMC.

Seven day hypoxic and normoxic hPASMC were treated with $50ng/\mu l$ EGF for five or ten minutes and also pre-treated with 10μ M rolipram for 40 min prior to EGF treatment for five minutes. These cells were then collected as described in section 2.3.1.1 and 30μ g lysate used on a 10% agarose gel. *Panel A*, After transferring onto nitrocellulose, an anti-phospho ERK1/2 antibody was used to identify the phosphorylated forms of ERK. *Panel B*, the densitometry results as compared to normoxic samples expressed in INT/mm². Results not shown for rolipram due to differences of n=2. Levels of pERK shown are representative of four independent experiments.



Figure 5.10 Effect of EGF on PDE4 activity levels. Cells were stimulated with $50ng/\mu l$ EGF for four and ten minutes prior to being immediately harvested for use in a cAMP-PDE activity assay as described in section 2.3.5. Results shown are mean \pm S.E. of three independent assays.



Figure 5.11 PDE4D5 knockdown in hPASMC

Normoxic and 7d hypoxic hPASMC were nucleofected with 60nM PDE4D5 siRNA. After nucleofection, the cells were transferred into 6 well plates and incubated for 24 hours. Cells were then harvested as described in section 2.3.1.1 20µg protein of each sample was used for SDS-PAGE alongside a basal control sample on a 4-12% gel. *Panel A*, the proteins were transferred to nitrocellulose and probed with an antibody directed against the c-terminus of PDE4D. *Panel B*, the densitometry results as compared to normoxic samples expressed as a percentage of the control PDE4D5 band.



Figure 5.12 Effect of knockdown of 4D5 on pERK expression in normoxic and hypoxic hPASMC

Normoxic and 7d hypoxic hPASMC were nucleofected with 60nM PDE4D5 siRNA. After nucleofection, the cells were transferred into 6 well plates and incubated for 24 hours. Cells were then harvested as described in section 2.3.1.1 20µg protein of each sample was used for SDS-PAGE alongside a basal control sample on a 4-12% gel. *Panel A*, the proteins were transferred to nitrocellulose and probed with a phospho-ERK antibody. *Panel B*, the densitometry results as compared to control samples expressed in INT/mm² and ** denotes significance (p<0.01).

Chapter 6

Proliferation in hPASMC

6.1 Introduction

6,1,1 The Cell Cycle

The eukaryotic cell cycle is a tightly regulated process. It is divided into four phases; Gap1 (G_1) phase, S phase, Gap 2 (G_2) phase and M phase. The restriction point at which the cell becomes committed to completing the cell cycle is within the G₁ phase. During G₁ phase, the cell cycle is under the influence of extracellular stimuli which it requires for the induction of necessary proteins to pass the restriction point. Following this, and entering into the S phase, DNA synthesis occurs. The second Gap phase, G₂, is where the cell prepares for cell division by confirming DNA duplication is completed and the DNA is undamaged. Once this checkpoint is passed, the cell enters into the M phase where mitosis occurs. When cells have exited the cell cycle they are quiescent and are said to be in G₀.

The cell cycle is regulated by complexes of cyclin dependant kinases and cyclins. The cyclin dependant kinases (cdk) are expressed at similar levels throughout the cell cycle and cyclins are expressed and degraded throughout the cell cycle at appropriate points to allow the cycle to continue. The cyclin-dependant kinases therefore undergo a sequential activation and inactivation through the cell cycle, providing regulatory phosphorylation.

6.1.2 Cyclic AMP inhibition of the cell cycle

The ability of cAMP to regulate the cell cycle was first reported 29 years ago (Pastan et al., 1975). Whilst it has been shown that raising the intracellular level of cAMP can promote the proliferation of certain cell types such as pituitary cells and thyrocytes, (Yonehara et al., 2001; Iacovelli et al., 2001) this action has also been noted to inhibit proliferation in most cell types, including endothelial cells, NIH 3T3 cells, Rat-1 fibroblasts and smooth muscle cells (D'Angelo 1997; Schmitt & Stork 2001; Osinki et al., 2000).

In vascular smooth muscle cells cAMP attenuates proliferation induced by vascular injury (Indolfi et al., 1997). It also prevents cells from entering S phase by arresting them within the G₁ phase of the cell cycle. This is achieved by various means. For example, cAMP has been shown to reduce levels of cyclin D1 and cyclin D3 by increasing their degradation (Kronemann et al., 1999; Stewart et al., 1999; Van Oirschot et al., 2001). This decrease in cyclin D levels results in a reduced phosphorylation of Rb (Boucher et al., 2001) and therefore reduces availability of proteins required for the G₁/S phase transition. Such a reduction of cyclin D also increases the amount of free p27^{*Klp1*} protein which can then inhibit cyclin E/cdk2 (L'Allemain et al., 1997). An overexpression of cyclin D1 in cells

can result in abnormal cell cycle control as seen in cancer cells (Musat et al., 2004). This is seen to be the case in breast and lung cell carcinomas, as well as in tumour cell lines (Bos et al., 2004; Yuan et al., 2004).

 $p27^{Kip1}$ binds to active cdk complexes resulting in their inactivation and a G1 block in the cell cycle (Kato et al., 1994, L'Allemain et al., 1997). Interestingly, increases in the cell cycle inhibitor proteins $p21^{Cip1}$ (Hayashi et al., 2000, Bauer et al., 2001) and $p27^{Kip1}$ (L'Allemain et al., 1997, Van Oirschot et al., 2001) have also been shown to be elicited by increased levels of cAMP.

cAMP can also in certain circumstances block growth factor activation of the extracellular regulated kinases, ERK 1 and 2. This can be achieved through PKA mediated inhibition of Raf-1 (Cook & McCormick 1993; Mischak et al., 1996). PKA has also been demonstrated to act indirectly through the phosphorylation of Src upon Rap1 to inhibit ERK (Ribeiro-Neto et al., 2002). However, the Src family of kinases have also been implicated in activating ERK1/2 through a process involving cAMP (Schmitt & Stork 2002). Activation of ERK1/2 promotes the expression of cyclin D1, cyclin E, cyclin A and the degradation of p27^{*Kip1*} (Lavoie et al., 1996; Stork & Schmitt 2002). This allows the cell to pass through the G₁/S phase transition (Schmitt & Stork 2001). However, inhibition of the ERK1/2 pathway is not a prerequisite for cAMP induced growth arrest (Balmanno et al., 2003).

6.1.3 Epac

Until recently, the main target of cAMP within cells was thought to be protein kinase A (PKA). In 1998 however, an exchange protein directly activated by cAMP (Epac) was uncovered as another major target of cAMP within cells (deRooij et al., 1998). Epac was uncovered by investigating a PKA-insensitive activation of Rap1 by cAMP. Epac is a guanine nucleotide exchange factor for the Ras like small GTPases Rap1 and Rap2. Although cAMP is well established to play a role in cell cycle control, it has been previously attributed to act solely through the actions of PKA. Recent studies show that there are PKA independent mechanisms involved in cAMP regulation of the cell cycle (Cass et al., 1999, Fujita et al., 2002). It has been shown that endogenous Epac localisation is dependent on cell cycle. During interphase, Epac is mainly localised to the nuclear membrane and mitochondria in COS-7 cells. Epac disassociates from these membranes in metaphase and localises to the mitotic spindle and centrosomes. Once the cell cycle is completed, however, Epac returns to the nuclear membrane (Qiao et al., 2002). Initial work on the cAMP-ERK pathway suggested the Epac activation of Rap1 lead to the activation of

ERK through B-Raf; however, recent studies have shown that Rap1 does not activate B-Raf (Bos et al., 2003). Instead, Rap1 is involved in integrin signalling which can activate the EGF receptor, leading to ERK activation (Bill et al., 2004). Notwithstanding this, the effect of Epac-mediated cAMP effects on cellular proliferation has not been investigated to date.

6.1.4 Phosphodiesterases and proliferation

In many disease states where excessive cellular proliferation is a hallmark, an increase in PDE activity and consequential decrease in cAMP levels has been observed (Savini et al., 1995; Vignola 2004; Schmidt et al., 1999; Essayan 1999; Spina 2003). Indeed, cAMP-specific PDE inhibitors have been shown to attenuate proliferation in various cell types, including vascular smooth muscle cells and glioma cells by inducing the cell cycle inhibitor proteins, $p21^{Cip1}$ and $p27^{Kip1}$ (Chen et al., 2002, Ogawa et al., 2002). Stimulation of proliferation itself has also been shown to induce the expression of PDE1C in human smooth muscle cells (Rybalkin et al., 2003). Using umbilical cord blood mononuclear cells, the PDE4 selective inhibitor rolipram has been shown to have an inhibitory effect on proliferation whilst specific inhibitors of PDE3, PDE5 and PDE1 did not, indicating both compartmentalisation of cAMP actions and specificity for cAMP over cGMP (Banner et al., 2000).

In A-172 glioma cells, rolipram has been observed to induce $p21^{Cip1}$ and $p27^{Kip1}$ proteins and to decrease the activity of cdk2, leading to a G₁ block (Chen et al., 2002). In human acute lymphoblastic leukaemia cells, rolipram, but not selective inhibitors for either PDE3 or PDE1, suppressed growth. The resultant increase in cAMP caused by PDE4 inhibition led to an induction of p53 and $p21^{Cip1}$, G₁ and G₂/M cell cycle arrest and increased apoptosis (Ogawa et al., 2002).

In human aortic smooth muscle cells, both the PDE4 selective inhibitor, rolipram and the PDE3 selective inhibitor, CI-930 attenuated proliferation by reducing the number of cells entering S phase (Johnson-Mills et al., 1998). Indeed, using both PDE4 and PDE3 inhibitors together produced an additive inhibitory effect on proliferation. It was interesting to note that whilst using either inhibitor alone did not appear to raise cAMP levels significantly, it was only when they were used in tandem was there a substantial increase in intracellular cAMP.

6.1.5 Proliferation in Hypoxia Induced Pulmonary Hypertension

Vascular smooth muscle cell growth and hypertrophy are being intensely studied in an effort to uncover the mechanisms governing thickening of the pulmonary arteries in PHT and other chronic lung disorders (Voetkel & Tuder 1997). This thickening has been demonstrated (Heath 1993) to be due to progressive muscularisation of the pulmonary arteries due to smooth muscle cell hypertrophy and hyperplasia. Vascular smooth muscle is normally contractile and quiescent, but stress or injury to the vessel wall results in a conversion of the cells to a migratory, secretory and proliferative phenotype (Heath 1993). Hyperplasia plays an important part in hypertension as shown by a significant increase in smooth muscle cell proliferation in pulmonary vessels from chronic hypoxic animals (Meyrick & Reid 1978, Wohrley et al., 1995). The normal aortic smooth muscle cell growth in rat is 0.01%/day which increases to 1%/day in hypertensive rats. This thickening of the pulmonary artery contributes to the rise in pulmonary artery pressure witnessed in PHT (Rich et al., 1987).

The three cell types of the pulmonary artery; endothetial cells, smooth muscle cells and fibroblasts can all show increased proliferation when maintained in hypoxic conditions compared to those in normoxia (Stenmark et al., 2002; Voelkel et al., 1997; Cool et al., 1995; reviewed in Humbert et al., 2004). The mechanism by which hypoxia increases the proliferation of vascular cells is unknown, although it is known that many of the growth factors which induce proliferation in vascular smooth cells are upregulated in hypoxia e.g. ET-1, PDGF, VEGF and 5-HT (Brij & Peacock 1998).

In this chapter, I set out to investigate whether the increased rate of proliferation in hypoxic hPASM cells compared to normoxic hPASM cells could be affected by compounds known to interfere with cAMP signalling. Thus, for example, I used agonists and antagonists to allow for selective perturbation of signalling through the RI and RII isoforms of PKA as it has suggested from studies done on HL-60 cells that RI is growth stimulatory while RII is growth inhibitory (Cho-Chung et al., 1993). I also employed two novel Epac agonists, 8 - pCPT-2'-O-Me-cAMP and 8 -pMeOPT-2'-O-Me-cAMP so as to, for the first time, evaluate the putative role that Epac might play in mediating aspects of the cAMP regulation of the cell cycle. I also used selective inhibitors of PDE4 and PDE3, so as to appreciate any role of these key cAMP degrading enzymes. The methods employed in this chapter are described in full in section 2.2, with a table of drug concentrations at the end of this chapter.

Results

6.2 Basal Proliferation Rates in Normoxic and Hypoxic hPASM cells

All proliferation assays were carried out after seven days of hypoxia to evaluate the effect of compounds in normoxic cells compared to cells with the PDE4 profile maximally altered.

To determine the basal proliferation rates in both seven day hypoxic and corresponding normoxic controls, DNA synthesis was measured by [³H]-thymidine incorporation as described in section 2.2. Briefly, cells were cultured in 96-well plates in both normoxic and hypoxic conditions as normal until day five of hypoxic exposure. The cells were quiesced overnight in serum free medium and on day six, serum was reintroduced to the medium in one set of plates, while another was refreshed with serum free medium. After 20 hours, 0.5 μ Ci [³H]-thymidine was added to each well and the cells were harvested after a further 4 hours to measure radioactivity.

In all results in this chapter, results are presented as a percentage with 100% being equal to mean normoxic basal proliferation rate. Two-way ANOVA with Bonferroni's post test was performed using GraphPad Prism version 4.00 for Windows, GraphPad Software, San Diego California USA, to analyse raw data. Where only two groups of data were to be analysed, an unpaired t-test was performed, and *p<0.05 was considered to be significant.

6.2.1 Basal Proliferation Rates

It was observed that hypoxia induced a 0.5 fold increase in proliferation over basal in scrum-free cells, however this increase was not significant. This increase was larger in cells with serum re-introduced. With serum, hypoxia caused a significant increase in proliferation (p<0.001). Scrum induced proliferation was doubled in hypoxic cells compared to normoxic proliferation (*Figure 4.1, Panel A and B*).

6.3 Effect of Raising cAMP on Proliferation Rates in Normoxic and Hypoxic hPASMC

It has been previously reported that an increase in cAMP levels exerts an anti-proliferative effect in vascular smooth muscle cells (Kronemann et al., 1999; Koyama et al 2000;

Hayashi et al., 2000; Bornfeldt & Krebs 1999; Indolfi et al., 2001). The extent of this effect has not been measured in hypoxic hPASM cells specifically however.

6.3.1 Effect of PDE inhibitors

The effect of PDE inhibition on proliferation in normoxic and seven day hypoxic cells was investigated using the selective PDE4 inhibitor, rolipram; the selective PDE3 inhibitor, cilostamide; a mixed PDE1 and PDE5 inhibitor, zaprinast and also using both rolipram and cilostamide together as such actions have been observed to potentiate the effect of either inhibitor alone (Johnson-Mills et al., 1998).

6.3.1.1 Effect of Rolipram

Treating serum-starved cells with rolipram for 24 hours had no significant effect on normoxic or hypoxic cells in serum-free medium (*Figure 6.2, Panel A*), although it was observed to reduce the rate of proliferation in hypoxic cells to basal levels seen in normoxic cells.

Rolipram treatment of cells with serum re-introduced, again had no significant effect on proliferation in either normoxic or hypoxic cells.

6.3.1.2 Effect of Cilostamide

Cilostamide did not significantly reduce proliferation in normoxic serum starved cells. However, in contrast to this, cilostamide significantly reduced proliferation (p<0.05) in hypoxic serum starved cells, (62 % of hypoxic basal levels). This reduced rate was equivalent to 90% of normoxic basal levels (*Figure 6.2, Panel A*).

Cilostamide had a significant inhibitory effect on serum-induced proliferation in both normoxic and hypoxic cells. Proliferation was significantly reduced following cilostamide addition to 38% (p<0.05) in normoxic cells and 56% (p<0.05) in hypoxic cells of respective serum induced control rates. Thus, hypoxic proliferation was reduced to almost that of normoxic serum induced proliferation (*Figure 6.2, Panel B*). The increased effect of cilostamide was consistent with the data in chapter 4 which showed cilostamide increased cAMP levels to 3 times that of basal in normoxic cells, whereas it didn't significantly affect cAMP levels in hypoxia.

6.3.1.3 Effect of Zaprinast

Evaluating serum-starved cells treated with zaprinast, there was no significant effect on normoxic or hypoxic cells after 24 hours treatment (*Figure 6.2, Panel A*). With cells

grown in serum-containing medium for the same 24 hours, there was still no significant effect (*Figure 6.2, Panel B*).

6.3.1.4 Effect of Rolipram and Cilostamide

Using both PDE4 and PDE3 inhibitors together has previously been shown to potentiate the effect of either single inhibitor in studies on human coronary artery smooth muscle cells (Johnson-Mills et al., 1998). Indeed, treating normoxic and hypoxic serum starved cells with both rolipram and cilostamide together induced a larger inhibition of proliferation than with either inhibitor alone. This effect was significant in hypoxic cells (p<0.01) where proliferation was reduced to 50% of hypoxic basal proliferation rate (*Figure 6.2, Panel A*). In normoxic cells however, this effect was not significant.

The combined inhibition of PDE3 and PDE4 activity had the largest inhibitory effect of any of the inhibitors on the proliferation status of both normoxic and hypoxic cells grown with serum for the last 24 hours. Under conditions for serum-stimulated proliferation, it was observed that rolipram and cilostamide acted synergistically to reduce proliferation rates to 45% in normoxic cells and to 30% (p<0.01) in hypoxic cells of respective serum induced control rates (*Figure 6.2, Panel B*).

6.3.2 Effect of cAMP effectors

Another way of raising cAMP within the cell is through activating adenylyl cyclase directly with an adenylyl cyclase activator such as forskolin, or indirectly through a beta-adrenergic agent such as isoproterenol. As such, the effect of both isoproterenol and forskolin on proliferation was investigated in hPASM cells.

6.3.2.1 Effect of Isoproterenol

24h treatment with isoproterenol had no significant effect on proliferation in normoxic or hypoxic cells, whether in serum free or serum containing medium (*Figure 6.3, Panet A and B*).

6.3.2.2 Effect of Isoproterenol and Rolipram

In contrast to using either isoproterenol or rolipram alone, when used together for the 24h treatment, then a highly noticeable inhibition of proliferation was evident. This inhibition was significant in hypoxic serum induced proliferation where proliferation was reduced to 20% (p<0.05) in hypoxic cells of respective serum-induced control rates (*Figure 6.3, Panel*)

.....

B). The effect on serum starved normoxic and hypoxic cells and normoxic serum induced proliferation was not significant.

Rolipram and isoproterenol have no effect on proliferation when used alone, thus these results indicate that a 'threshold' level of cAMP activation is required to be breached in order for it to inhibit proliferation in hPASMC.

6.3.2.3 Effect of Forskolin

Introducing forskolin alone to the cells had no significant effect under either hypoxic or normoxic conditions in either serum starved cells or cells with serum re-introduced (*Figure 6.3, Panel A and B*). Furthermore, adding the non-selective PDE inhibitor, 3-isobutyl-1-methylxanthine, (IBMX), along with forskolin appeared to increase the rate of proliferation of control levels in serum-treated normoxic cells, although not significantly. In hypoxic serum-induced proliferation, the combination of forskolin together with IBMX acted in an anti-proliferative manner, where proliferation was reduced to 20% (p<0.05) of the hypoxic serum induced control rate (*Figure 6.3, Panel B*).

These results indicate that indeed a threshold level of cAMP is required for cAMP to act in an inhibitory fashion on proliferation in hPASMC. However, it was observed in the chapter 4 that the cAMP produced in hypoxic cells in response to tBMX and forskolin is less than that in normoxic cells. Thus, this is suggestive of IBMX acting through a different mechanism to override the proliferative mechanism induced in hypoxia.

6.4 Effect of Growth Factors on Proliferation Rates in Normoxic and Hypoxic hPASMC

Epidermal growth factor (EGF) has been thoroughly studied in a wide range of cell types and is well known to serve as a growth factor and mitogen. In chapter 5, EGF was shown to activate PDE4 in hPASMC, thus it was used in this study to investigate any effect on proliferation.

Transforming Growth Factor- β_1 (TGF- β_1) has been under the spotlight in PHT research since uncovering the role of inherited mutations in the *BMPR2* gene in the development in PAH (see general introduction section 1.5; Thomson et al., 2000; Machado et al., 2001). TGF- β is an important regulator of cellular differentiation and proliferation in the lung(Vicencio et al., 2002) and has been shown to be both a stimulator of cell growth and an inhibitor, depending on cell type. In rat aortic vascular smooth muscle cells, TGF- β_1 decreased serum-induced proliferation (Grainger et al., 1994). Therefore, I used TGF- β_1 in these proliferation assays to see if it can impact upon hPASMC proliferation also.

6.4.1. Effect of EGF

EGF had no effect on serum starved proliferation in either normoxic or hypoxic cells (*Figure 6.4, Panel A*). In cells cultured in serum, 50 pg/µ1 medium EGF induced a 25% increase in proliferation compared to the normoxic serum induced control rate of proliferation. In hypoxic cells, this increase was even greater with an increase of 50% in proliferation (p<0.01) compared to the hypoxic serum induced control proliferation rate (*Figure 6.4, Panel B*). Therefore, EGF requires a serum-induced pathway to clicit a proliferative effect in hPASMC.

6.4.2 Effect of EGF and Rolipram

Using EGF or rolipram has been demonstrated to have no effect on serum starved proliferation in hPASMC. It was also shown here that when used together, EGF and rolipram do not affect proliferation of serum starved cells (*Figure 6.4, Panel A.*)

Serum-induced proliferation of normoxic hPASMC was not affected by the addition of EGF and rolipram. EGF alone has previously been shown to elicit a 25% increase in serum induced proliferation in normoxic cells and rolipram caused a 40% reduction in hPASMC proliferation in normoxic serum-induced proliferation (6.3.3.1). Thus, it appears here that when used together they negate the effect of each other. In hypoxic cells, rolipram alone was previously shown to reduce serum-induced proliferation by 28% and EGF caused a 50% increase. Again, when used together, EGF and rolipram appear to negate the effect of one another on hypoxic serum-induced proliferation as no effect is seen compared to hypoxic serum-induced control rate of proliferation (*Figure 6.4, Panel B*).

6.4.3 TGF-β1

TGE- β_1 appeared to have no effect on proliferation in either normoxic or hypoxic hPASM cells that had been serum starved (*Figure6.5, Panel A.*). Serum-induced proliferation of normoxic and hypoxic hPASMC is also unaffected by TGF- β_1 (*Figure 6.5, Panel B*).

6.5 Effect of the Src Family Inhibitor, PP2, on Proliferation Rates in Normoxic and Hypoxic hPASMC

The Src family of kinases have previously been implicated in activating ERK1/2, and has also been shown to be required for cAMP activation of Rap1 in fibroblast cells(Schmitt & Stork 2002). PP2 is a potent inhibitor of the Src family kinases which also inhibits many other kinases including CSK, LCK and SAPK 2a/p38.

I also investigated here a compound called PP3, which inhibits selectively the EGF receptor tyrosyl kinase but not the activity of Src family tyrosyl kinases, which are inhibited selectively by PP2. These two compounds allow the potential to gain insight into the relative roles of the EGFR tyrosyl kinases and SRC family tyrosyl kinases in proliferation of the hPASM cells under normoxic and hypoxic conditions.

6.5.1 Effect of PP2

PP2 elicited no significant effect on proliferation in normoxic serum starved cells. In hypoxia however, PP2 reduced proliferation to 38% (p<0.05) of the basal proliferation rate in normoxic serum starved cells (*Figure 6.6, Panel A*). PP2 had no effect on normoxic serum induced proliferation, but reduced hypoxic proliferation to 25% (p<0.01) of hypoxic serum-induced proliferation control rate (*Figure 6.6, Panel B*).

6.5.2 Effect of PP3

PP3 had no significant effect on proliferation in normoxic or hypoxic serum starved cells (*Figure 6.6, Panel A*). PP3 also had no significant effect on serum induced proliferation in normoxic or hypoxic cells (*Figure 6.6, Panel B*).

These results indicate that Src is involved in the increased hypoxic proliferation rate of hPASMC. The Src family of kinases have been demonstrated to activate Rap1 (Schmitt and Stork 2002b) which can activate B-raf and activate ERK. Indeed, ERK activation by PKA has been demonstrated to require Src (Lindquist et al., 2000). This suggests the increased PKA activity shown in chapter 4 in hypoxic hPASMC acts on Src to activate the ERK pathway through Rap1-B-raf. This would also explain the increased levels of ERK witnessed in hypoxia (Jin et al., 2000; Scott et al., 1998).

6.6 The cAMP Pathway

As mentioned previously (see general introduction section 1.5.8.2; reviewed by Koyama et al., 2000; Hayashi et al., 2000), it is known that cAMP acts as an anti-proliferative agent within vascular smooth muscle cells. Although no significant effect was observed using the AC stimulator, forskolin alone or the beta-adrenergic agent, isoproterenol alone on proliferation, there was an inhibition of proliferation using the combined PDE inhibitors rolipram and cilostamide. It is well known that cAMP signalling is compartmentalised in cells and that for effects to be seen then increased cAMP levels must occur in a functionally relevant compartment. One way to achieve this and to evaluate whether actions are actually due to cAMP is to swamp the cell with a cAMP analogue. When cAMP elicited effects are observed, it can be determined if this is through specific PKA isoforms or EPAC, using selective agonists and antagonists. As PKA activity was observed to increase in hypoxic hPASMC in chapter 3, specific PKA isoform agonists and antagonists were employed to investigate the effect of this on hPASMC proliferation.

6.6.1 Effect of PKA AgonIsts

PKA is a heterodimer consisting of two regulatory and two catalytic subunits. Previous research has indicated differing roles in growth regulation effected by the RI and RII subunits. Specific agonist pairs for the PKA RI and PKA RII subunits were employed to investigate any differing effect on proliferation by cAMP signalling through these two PKA isoforms.

6.6.1.1 PKA RI Agonist

The cAMP analogs 8-PIP-cAMP and 8-IIA-cAMP were used to activate PKA RI selectively. In serum starved cells, the PKA RI agonist had no significant effect on proliferation in normoxic or hypoxic cells (*Figure 6.7, Panel A*). The PKA RI agonist also had no effect on normoxic serum induced proliferation. However, it did reduce hypoxic serum induced proliferation to 50% (p<0.001) of the control hypoxic serum induced proliferation rate (*Figure 6.7, Panel B*).

6.6.1.2 PKA RII Agonist

The cAMP analogs 8-PIP-cAMP and 8-MBC-cAMP were used to activate PKA RII selectively. The PKA RII agonist had no significant effect on proliferation in normoxic serum starved cells (*Figure 6.7, Panel A*). It did reduce hypoxic proliferation to 47%

(p<0.05) of the basal hypoxic proliferation rate in serum starved cells. Serum-induced proliferation was reduced to 57% (p<0.001) in hypoxic cells compared to the hypoxic control serum induced proliferation rate (*Figure 6.7, Panel B*). There was no effect on normoxic serum-induced proliferation however.

6.6.1.3 PKA RI and RII Agonists

Using both the PKA RI and PKA RII agonists together had no significant effect on normoxic or hypoxic proliferation in serum starved cells (*Figure 6.7, Panel A*). Normoxic serum induced proliferation was reduced to 68% (p<0.05) of normoxic control serum induced proliferation rate. This effect was greater in hypoxic cells, where these agonists used together reduced serum induced proliferation to 35% (p<0.001) of the hypoxic control proliferation rate (*Figure 6.7, Panel B*).

These results indicate PKA RI:RII exists in a balanced activation state within hPASMC, and if this is altered in favour of either subunit, then this elicits an inhibitory effect on proliferation. In particular, PKA RII appears to have a significant role in the increased hypoxic proliferation rate seen in hPASMC as activating this subunit reduces hypoxic proliferation to that of normoxic cells, in both serum starved cells and those with serum reintroduced. Agonising the PKA RI subunit was also capable of returning the hypoxic serum induced proliferation rate to that of normal, but only when the serum response pathways were in effect, similar to the treatment with forskolin and IBMX (6.3.2.3).

6.6.1.4 Effect of the PKA Agonist, 6-BnZ-cAMP

The non-selective PKA agonist, 6-BnZ-cAMP, which activates both RI and RII forms of PKA, was used to confirm the results witnessed using the RI and RII agonists in combination. In serum starved cells, 6-BnZ had no significant effect on proliferation in normoxic or hypoxic cells (*Figure 6.7, Panel A*). 6-BnZ also had no significant effect on normoxic serum induced proliferation, but did reduce hypoxic serum induced proliferation to 71% (p<0.01) of the hypoxic control serum induced proliferation rate (*Figure 6.7, Panel B*).

This again confirms the role of PKA in the hypoxic induced proliferation witnessed in hPASMC as it has no effect on normoxic proliferation, whether in the presence or absence of serum. The magnitude of the effect induced by the PKA RI and RII agonists together compared to 6-BnZ suggests the separate agonist somehow synergise to enhance their effect whereas 6-BnZ perhaps has a more 'real' effect on PKA.

6.6.2 Effect of PKA Antagonists and Inhibitor

6.6.2.1 Effect of the PKA Inhibitor, H89

H89 has been used by various investigators to inhibit protein kinase A selectively. However it is now known that H89 is also capable of inhibiting various other kinases, including Rho kinase (Davies et al., 2000). H89 had no significant effect on normoxic serum induced proliferation or on proliferation in serum starved normoxic cells (*Figure 6.7, Panel A and B*). Intriguingly, H89 induced a similar reduction in proliferation in hypoxic cells to that of the PKA agonist, although with a more potent effect. In serum starved hypoxic cells, II89 reduced proliferation to 49% (p<0.05) of basal hypoxic proliferation, while it reduced hypoxic serum induced proliferation to 64% (p<0.001) of the hypoxic serum induced proliferative control rate (*Figure 6.7, Panel A and B*).

The results obtained using PKA agonists and the PKA inhibitor, H89, suggest PKA is involved in a balance to maintain the increased hypoxic proliferation as any alteration to PKA signalling ultimately leads to a reduction in proliferation, decreasing levels to that of normoxic proliferation or lower. Thus, antagonists of the PKA RI and PKA RII were employed to investigate if this also occurs with one of the subunits inhibited.

6.6.2.2 Effect of PKA RI Antagonist

The PKA RI antagonist had no significant effect on proliferation in hPASMC whether in normoxic, hypoxic cells or in the presence or absence of serum (*Figure 6.8, Panel A and B*).

6.6.2.3 Effect of the PKA RII Antagonist

The PKA RII antagonist reduced proliferation in serum starved cells, although this effect was only significant in hypoxic cells with proliferation reduced to 37% (p<0.05) of the hypoxic basal proliferation rate (*Figure 6.8, Panel A*). The PKA RII antagonist completely ablated serum induced proliferation in normoxic and hypoxic cells, reducing proliferation rates to < 5% in both normoxic and hypoxic cells (*Figure 6.8, Panel B*). This highly significant reduction in serum induced proliferation is indicative of the cells undergoing apoptosis when PKA RII is inhibited, thus PKA RII appears to be essential for the normal growth of hPASMC.

6.6.3 Effect of cAMP analogs

The majority of downstream effects of cAMP have been attributed to the cAMP dependant kinase, PKA. However, in 1998, another major cAMP target was uncovered in cells (de

Rooij et al., 1998). This cAMP-GEF was an Exchange Protein Activated by cAMP, (Epac), and has recently been shown to be involved in cell adhesion (Rangarajan et al., 2003). Localisation of Epac within cells is cell-cycle dependant. As such, two cAMP analogs were employed in this assay. Both of these analogs activate PKA and Epac (Enserink et al., 2002). 8-bromo-cAMP has a higher affinity for PKA than Epac, whilst another, CPT-cAMP, is known to have a higher affinity for Epac. These cAMP analogs were used to determine if PKA or EPAC had the larger role in regulation of proliferation in hPASMC under normoxic and hypoxic conditions.

6.6.3.1 Effect of 8-Br-cAMP

The cAMP analog 8-Br-cAMP was observed to have no significant effect on the proliferation of hPASMC (*Figure 6.9, Panel A and B*).

6.6.3.2 Effect of CPT-cAMP

In serum starved cells and cells cultured with serum, the cAMP analog, CPT-cAMP, induced a highly significant reduction in proliferation of normoxic and hypoxic cells (*Figure 6.9, Panel A and B*). Cells appeared morphologically normal after treatment in all cases. In normoxic and hypoxic serum starved cells, CPT-cAMP reduced proliferation to 44% and 10% (p<0.01) of basal rates respectively. Serum induced proliferation was reduced to approximately 17% (p<0.001) in both normoxic and hypoxic cells when compared to their respective serum induced proliferation control rate.

6.6.4 Specific EPAC agonist

Although CPT-cAMP has a higher affinity for EPAC over PKA, there is still some activation of PKA reported in the use of this analog. At the time of this study, there was originally one cAMP analog altered to specifically activate EPAC available. This was 8 - pCPT-2'-O-Me-cAMP (Enserink et al., 2002). This agonist was used in the proliferation assay to confirm the results observed with CPT-cAMP and to investigate this anti-proliferative effect further.

6.6.4.1 Effect of 8 -pCPT-2'-O-Me-cAMP

It was observed that the EPAC agonist, 8 -pCPT-2'-O-Me-cAMP, had no significant effect on proliferation in normoxic or hypoxic serum starved cells (*Figure 6.10, Panel A*).

8 -pCPT-2'-O-Me-cAMP reduced scrum-induced proliferation to approximately 50% (p<0.01) in normoxic cells and 36% (p<0.001) in hypoxic cells when compared with their respective control rate of serum induced proliferation (*Figure6.10, Panel B*).

6.6.5 Effect of cAMP Effectors on the anti-proliferative capacity of 8 - pCPT-2'-O-Me-cAMP

In an attempt to understand the effect of the EPAC agonist on proliferation, PDE inhibitors and other cAMP effectors were used in conjunction with the agonist to see if they could recover the cells from this or further potentiate the effect.

6.6.5.1 Rolipram

Including rolipram with the EPAC agonist for 24 hours potentiated the effect seen with rolipram alone or the 8 -pCPT-2'-O-Me-cAMP alone. In serum starved hypoxic cells, this combined treatment reduced proliferation to 65% in normoxic cells and 56% (p<0.01) in hypoxic cells (*Figure 6.11, Panel A*).

Treatment with rolipram and 8 -pCPT-2'-O-Me-cAMP reduced serum induced proliferation to 45% in normoxic cells and 27% (p<0.001) in hypoxic cells (*Figure 6.11*, *Panel B*). Again this was greater than the effect seen with either compound used alone.

6.6.5.2 Cilostamide

Using the PDE3 inhibitor cilostamide with 8 -pCPT-2'-O-Me-cAMP again potentiated the effect of using either compound alone. Normoxic proliferation was reduced to 73% in serum starved cells, although this was not significant to treatment with either cilostamide or the Epac agonist alone. Hypoxic proliferation in serum starved cells was significantly affected with a decrease to 47% (p<0.01) of basal hypoxic proliferation (*Figure 6.11, Panel A*). Serum induced proliferation was reduced to 34% in normoxic cells and 34% (p<0.001) in hypoxic cells when compared to the respective control serum induced proliferation rates (*Figure 6.11, Panel B*).

6.6.5.3 Rolipram and Cilostamide

Using both rolipram and cilostamide with 8 -pCPT-2'-O-Me-cAMP potentiated the effect seen with 8 -pCPT-2'-O-Me-cAMP alone. However, this effect was not as great as that seen when using rolipram and cilostamide alone. In serum-starved cells, the combined treatment had no effect on normoxic proliferation, but reduced hypoxic proliferation to 62% (p<0.05) of hypoxic basal proliferation (*Figure 6.11, Panel A*). The combined treatment also reduced normoxic proliferation to 30% of control serum induced proliferation, although this was not significant. In hypoxic cells with serum reintroduced, this effect was greater, with a reduction to 22% (p<0.001) of hypoxic control serum induced proliferation (*Figure 6.11, Panel B*).

6.6.5.4 8-br-cAMP

8-bromo-cAMP used with 8 -pCPT-2'-O-Me-cAMP had no effect on the anti-proliferative capacity of 8 -pCPT-2'-O-Me-cAMP alone (*Figure 6.12, Panel A and B*).

6.6.5.5 6-BnZ

The PKA agonist, 6-BnZ, slightly potentiated the inhibitory effect of 8 -pCPT-2'-O-MecAMP when used together in all conditions (*Figure 6.12*, *Panel A and B*).

6.6.5.6 PKA RI agonist

Normoxic serum starved cells treated with both the EPAC agonist and a specific PKA RI agonist did not have any significant effect on proliferation when compared to 8 -pCPT-2'-O-Me-cAMP used alone. However, the combined effect was increased in hypoxic serum starved cells compared to using either compound alone (*Figure 6.13, Panel A*). Hypoxic serum starved proliferation was reduced to 55% (p<0.05) of basal hypoxic proliferation.

The combined effect of the Epac agonist and PKA RI agonist did not have any effect over that of 8 -pCPT-2'-O-Me-cAMP alone in serum induced proliferation in normoxic and hypoxic cells (*Figure 6.13, Panel B*).

6.6.5.7 PKA RII agonist

Using the PKA RII agonist in conjunction with the Epac agonist slightly reduced the antiproliferative effect of using either agonist alone in serum starved normoxic and hypoxic cells, although not significantly (*Figure 6.13, Panel A*). There was no effect observed, however, in cells with serum reintroduced (*Figure 6.13, Panel B*).

6.6.5.8 PKA RI and RII agonists

Using both the RI and RII agonists, along with the EPAC agonist, potentiated the antiproliferative effect of using either both the RI and RII agonists or the Epac agonist alone on proliferation in serum starved cells. This effect was significant (p<0.01) in hypoxic serum starved cells where proliferation was reduced to 50% of hypoxic basal (*Figure 6.13*, *Panel A.*)

Using both the RI and RII agonists along with the Epac agonist had no significant effect on normoxic serum induced proliferation when compared with the Epac agonist alone. However, it did potentiate the effect of the RI and RII agonist or the Epac agonist observed in hypoxic serum induced proliferation, further reducing it to 17% (p<0.001) of the hypoxic control serum induced proliferation rate (*Figure 4.13, Panel B*).

6.7 Effect of the EPAC Agonist, 8 -pMeOPT-2'-O-Me-cAMP, and a Rap1 Inhibitor on Proliferation compared to 8 -pCPT-2'-O-Me-cAMP

6.7.1 Arresting cell cycle using a Rap1 Inhibitor, GGTI-298

GGTI-298 (Calbiochem) is a GGTase I inhibitor that inhibits the processing of Rap1A without affecting H-Ras. It is also capable of arresting cells in the G_0/G_1 phase of the cells cycle, but it has previously been reported that using GGTI-298 can reduce the inhibitory effects on proliferation seen with cAMP-elevating agents in RPE cells (Hecquet et al., 2002). As can be seen in figure 6.14, the Rap1 inhibitor reduced both normoxic and hypoxic proliferation rates to 30% and 20% respectively of control levels. Including the EPAC agonist along with this inhibitor had no significant effect on normoxic or hypoxic proliferation over that of the Rap1 inhibitor alone (*Figure 6.14*). Epac has been demonstrated to activate Rap1 (Enserink et al., 2002), therefore these results suggest the anti-proliferative effect of the Epac agonist is independent of Rap1 as the Rap1 inhibitor can not preserve the proliferative capacity of hPASM cells when used with the Epac agonist. However, as the Rap1 inhibitor reduces proliferation significantly when used alone, this can not be assumed.

6.7.2 A second EPAC agonist, 8 -pMeOPT-2'-O-Me-cAMP

As a second novel EPAC agonist became available towards the end of this study, I decided to use this to confirm the anti-proliferative effect of the first EPAC agonist and to compare the potency of both. Biolog reported this new agonist to have a higher activity and membrane permeability compared with 8 -pCPT-2'-O-Me-cAMP. Anti-proliferative activities were compared in cells with serum re-introduced.

In normoxic and hypoxic cells, 8 -pCPT-2'-O-Me-cAMP had an IC_{50} of approximately 50µM. The IC_{50} for 8 -pMeOPT-2'-O-Me-cAMP was different to this. In normoxic cells, it had an IC_{50} of 1mM and in hypoxic cells, an IC_{50} of 500µM. Although both agonists elicit an anti-proliferative effect on both normoxic and hypoxic cells, the original agonist, 8 -pCPT-2'-O-Me-cAMP was far more potent at inhibiting proliferation in normoxic and hypoxic hPASMC (*Figure 6.15*).

6.8 Discussion and Conclusions

Exposing vascular smooth muscle cells to hypoxia has been shown to increase their proliferative capacity as witnessed in the rat model of hypoxia induced pulmonary hypertension(Meyrick & Reid 1978, Wohrley et al., 1995) I have shown that human pulmonary artery smooth muscle cells are also useful as cellular model of pulmonary hypertension for proliferation studies as they double their serum-induced proliferation rate when exposed to $10\% O_2$ for seven days.

Previous studies have established that raising cAMP within smooth muscle cells acts in an anti-proliferative manner (reviewed in Bornfeldt & Krebs 1999) The adenylyl cyclase stimulator, forskolin has been shown to achieve this, in part by reducing serum stimulated cyclin D1 and cyclin A levels in vascular smooth muscle cells (Kronemann et al., 1999) Raising cAMP through PDE4 inhibition can also elicit an antiproliferative effect as shown in A172 glioma cells and human acute lymphoblastic leukaemia cells (Chen et al., 2002, Ogawa et al., 2001) In human aortic and pulmonary arterial smooth muscle cells, PDE3 and PDE4 inhibitors have previously been shown to attenuate proliferation (Johnson-Mills et al., 1998, Indolfi et al., 1997). These effects have not been investigated in hypoxia however.

Rolipram did not have a significant effect on hPASMC proliferation, which contrasts with previous studies demonstrating rolipram can significantly reduce the proliferation of smooth muscle cells (Pan et al., 1994). However, other studies suggest there is no decrease in proliferation (Souness et al., 1992; Osinski and Schror, 2000). Cilostamide however, did significantly reduce proliferation of hPASMC as observed previously (Tsuchikane et al., 1999) with a greater effect observed in normoxic cells than in hypoxic cells. This can perhaps be explained by the desensitisation of the cAMP pathway demonstrated to occur in hypoxia in chapter 4. Using both of these inhibitors together, produced a supra-additive effect as observed in previous studies on smooth muscle cells (Johnson-Mills et al., 1998; Pan et al., 1994).

Raising cAMP levels with the β_2 -adrenoceptor agonist, isoproterenol, appeared to have no effect on cell proliferation in hypoxia or normoxia. However, an action of isoproterenol was uncovered when cells were pre-treated with the PDE4 inhibitor, rolipram. Prior work with Albuterol, a β_2 -adrenoceptor agonist has shown (Stewart et al., 1999) it inhibits

thrombin-stimulated proliferation when used prior to the restriction point. This it seemingly does by reducing cyclin D1 levels. Such inhibition could, however, be alleviated by treating cells with either the ERK inhibitor, PD98059 or the proteasome inhibitor MG132 (Stewart et al., 1999). Here I showed that treating cells with isoproterenol and rolipram together produced a highly significant inhibition of proliferation in both normoxic and hypoxic cells, indicating that the cells either immediately exited the cell cycle or underwent apoptosis. This suggests the pathway induced by β_2 -adrenoceptor stimulation is under great influence by PDE4. Indeed, it has been shown that PDE4 is recruited in a complex with β -Arrestin to the β_2 -adrenoceptor upon stimulation of the receptor and plays a role in the phosphorylation of the receptor, one consequence of which is to regulate G_s to G₁ switching in HEK 293 cells (Baillie et al., 2002, Perry et al., 2002).

There has been no indication that the role of PDE4 in mediating G-protein switching atfects cell proliferation, however it is known that G-protein coupled receptors (GPCRs) play a role in cell growth. Mutations affecting $G\alpha_8$ are known to be present in different types of tumours (Chen & Iyengar 1994). It is also possible that PDE4 plays a role in the interaction of β -Arrestin 2 with the oncoprotein Mdm2. Mdm2 associates with β -Arrestin 2 when a GPCR is stimulated, interfering with it's self ubiquitination and thus p53 degradation, allowing an increase in p53 activity (Wang et al., 2003). p53 is a tumour suppression protein capable of inducing growth arrest and apoptosis. An increase in p53 might help to explain the results seen using both rolipram and isoproterenol together suggesting that PDE4 perhaps enhances the binding of β -Arrestin 2 to Mdm2, thus reducing p53 ubiquitination.

Similarly, forskolin had no effect on cell proliferation unless the non-selective PDE inhibitor, IBMX, was first used. In normoxic cells, this treatment unexpectedly increased the rate of DNA synthesis, whereas in hypoxic cells, this rate was reduced to 20% of the control hypoxic serum induced proliferation rate. Stimulation of cAMP with forskolin alone has previously been reported to result in an attenuation of serum induced proliferation in vascular smooth muscle cells (Kronemann et al., 1999). This was not observed with the hPASMC. In certain conditions however, stimulation of cAMP levels can result in increased DNA synthesis in neonatal but not adult pulmonary SMC(Guldemeester et al., 1998). The use of forskolin and IBMX together has also been shown to inhibit serum-induced proliferation previously in pancreatic cancer cells (Boucher et al., 2001).

EGF is a well known mitogen and in this study was shown to induce a significant increase in serum induced proliferation but, interestingly, not in serum free cells. This suggests that EGF requires an additional input, supplied by a serum factor, to exert its effects on these cells. This increase caused by EGF in serum-treated cells was unaffected by rolipram in normoxic cells but was reduced to control levels in hypoxic cells, indicating the effects of EGF and rolipram can negate one another, EGF acts on the Ras/Raf/MEK/ERK pathway to activate cell growth and proliferation, therefore these results suggest rolipram exerts an inhibitory effect on this pathway. Another growth factor, TGF- β_1 has been used in studies of activation of ERK and cell proliferation as it is an important regulator of cellular proliferation in the lung (Vicencio et al., 2002) and has been shown to be both a stimulator of cell growth and an inhibitor, depending on cell type. The results are, however, conflicting. In epithelial cells, TGF- β_1 has been reported to both induce a G1 block in the cell cycle with a corresponding inhibition of ERK (Howe et al., 1993) and also to activate ERK and promote cell proliferation (Hartsough & Mulder 1995). In hPASMC, TGF- β_1 had no effect on proliferation.

Src has been demonstrated to be essential in both PC12 cells and NIH 3T3 fibroblasts to witness forskolin-stimulated ERK activation (Klinger et al., 2002). It has also been demonstrated in both NIH3T3 cells and mouse embryonic fibroblasts that Src phosphorylation by PKA is required for cAMP inhibition of ERK and cell proliferation (Schmitt & Stork 2002). Using the Src inhibitor, PP2, in hPASMC significantly reduced hypoxic serum induced proliferation and also hypoxic proliferation in serum starved cells. There was no significant effect on normoxic cells under either condition. This suggests there is a pathway not mediated by the serum response factor that is altered in hypoxic cells and requires Src for the increased proliferation observed in hypoxia.

The anti-proliferative actions of cAMP on cell proliferation have routinely been attributed to the cAMP-dependant kinase, PKA. PKA has been demonstrated to increase the levels of p27Kip1 and decrease the levels of cyclins D1 and D3 (L'Allemain et al., 1997, Kronemann et al., 1999, Stewart et al., 1999, Van Oirschot et al., 2001). This results in a G1 block in the cell cycle. As mentioned previously, PKA can also exert an effect upon ERK activation, which can lead to either cell proliferation or differentiation or apoptosis (Schmitt & Stork 2002). PKA is a heterodimer consisting of two regulatory and two catalytic subunits. Activation of PKA requires binding of cAMP to the regulatory subunits which releases the catalytic subunits. Opposing functions for the two regulatory isoforms of PKA in the cell cycle have been observed (reviewed in Cho-Chung et al., 1995). Indeed,

the balance of the RI/RII ratio is critical as alterations can lead to tumour formation (Stergiopoulos & Stratakis 2003), PKA RI is a little overexpressed in normal cells upon stimulation of proliferation, yet is constitutively overexpressed in cancer cells and is associated with a poor prognosis in different human cancers (Miller et al., 1993, McDaid et al., (999). Inhibition of PKA results in a significant reduction of cellular proliferation in hypoxic hPASMCs only, whether serum starved or not. Activation of PKA is also capable of reducing cell proliferation, although only significantly in hypoxic serum induced proliferation. This effect appears to be predominantly due to the PKA RII subunit as the PKA RII agonist elicits a greater effect than the RI agonist. These effects are only significant in hypoxia, suggesting an alteration in PKA signalling in hypoxia. This is confirmed by the significant reduction of serum induced proliferation by the general PKA agonist, 6-BnZ, in hypoxia, but not in normoxic cells. As stimulation of total PKA has no effect on normoxic serum induced proliferation, this suggests an alteration of PKA RI/RII ratio in hypoxic cells. When the specific subunit agonists are used together however, it does not mirror the effect of a general PKA agonist. They appear to synergise and reduce scrum induced proliferation more dramatically, PKA RI inhibition has no significant effect on hPASMC proliferation, although it does appear to cause a slight increase in normoxic serum starved cells. RII inhibition results in a highly significant reduction of cell proliferation in both normoxic and hypoxic cells. These results suggest that PKA RII is essential for normal cell proliferation and is involved in the hypoxia-induced increase of proliferation. It should be noted that the PKA RI subunit was also demonstrated to induce a small, albeit not statistically significant, increase in normoxic proliferation in serum starved cells. This suggests the PKA RI subunit can act in a growth inducing manner, which is regulated by the PKA RII subunit and, vice versa, the PKA RII subunit can act in an inhibitory manner which is regulated by PKA RI. This again is indicative of an alteration in the RI to RII ratio in hypoxic cells. In chapter 3 however, western blotting of the PKA RIa, RIIa and RIIB subunits did not show any change in expression. PKA RIB expression was not investigated.

PKA is not the only effector of cAMP signalling in cells. Epac is another target for cAMP signalling in mammalian cells, although its effects on proliferation have not yet been investigated. Using the cAMP analog, CPT-cAMP, it was clear this activated something different to the cAMP analog, 8-bromo-cAMP as it produced a more potent inhibition of proliferation. Indeed, it was later discovered that CPT-cAMP has a higher affinity for Epac than PKA. Using the novel Epac agonists, 8 -pCPT-2'-O-Me-cAMP and 8 -pMeOPT-2'-O-Me-cAMP, it was revealed for the first time that Epac is capable of potently inhibiting

proliferation. Epac appears to require the serum response to have an impact as this inhibition only occurs in serum induced proliferation. This is similar to the effect of the Src inhibitor PP2. Thus, this suggests that Epac affects proliferation in a manner similar to that of Src inhibition. It is known that both Epac and Src activate the GTPase Rap1. Activation of Rap1 has been associated with both the cAMP mediated activation of ERK1 and thus cell proliferation (Ribeiro-Neto et al., 2002) and the cAMP mediated inhibition of ERK and cell proliferation (Schmitt & Stork 2001). The Epac agonist used in this study has previously been shown to have no direct effect upon ERK activity in NIH3T3 cells stably transfected with Epac1 and CHO ceils (Enserink et al., 2002). However, it is possible that Epac could act indirectly on ERK activity. The divergent roles of Rap1 in cAMP mediated cell proliferation are indicative of numerous stimuli acting on Rap1. As Epac is already known to activate Rapl, it seems possible that the levels of activation of Epac could regulate the PKA directed activation of Rap1. Indeed, increasing cAMP levels in the cell through PDE inhibition, raising PKA activation by addition of a cAMP analog or direct PKA stimulation, or even specific PKA regulatory subunit activation, showed only slight potentiation or alleviation of this effect. This is suggestive of a cAMP threshold level at which Epac is activated and elicits its effect rather than a concentration dependant effect. It also appears that the level of PKA cannot influence the effect of Epac once activated. Thus, Epac perhaps acts as an 'off' switch for PKA mediated signalling through Rap1.

In this chapter, I have presented results that suggest a role for PKA RI:RII imbalance in hypoxia-induced proliferation and demonstrate a new role for Epac in cell proliferation.



Figure 6.1 Effect of chronic hypoxia on proliferation in hPASMC.

Normoxic and hypoxic cells were cultured in 96-well plates for five days prior to serum starvation overnight. One set of cells was maintained in serum free medium, (*Panel A*), and another had serum re-introduced (*Panel B*). After twenty hours, $[^{3}H]$ -thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 4 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *** (p<0.001).



Figure 6.2 Effect of PDE inhibitors on proliferation in hPASMC.

Cells were treated as in figure 6.1 and all PDE inhibitors were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *, ** (p<0.05, p<0.01 respectively).



Long the second second

Figure 6.3 Effect of cAMP effectors on proliferation in hPASMC.

Cells were treated as in figure 6.1 and all additions were made when the serum free medium was replenished, *(Panel A)*, or serum was re-introduced to the cells *(Panel B)*. After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by * , ** (p<0.05, p<0.01 respectively).


Figure 6.4 Effect of EGF on proliferation in hPASMC.

Cells were treated as in figure 6.1. $50pg/\mu l$ medium of EGF was added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *** (p<0.001).



Figure 6.5 Effect of TGF- β_1 on proliferation in hPASMC.

Cells were treated as in figure 6.1. TGF- β_1 was added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate.



Figure 6.6 Effect of Src inhibitor on proliferation in hPASMC.

Cells were treated as in figure 6.1. PP2 and PP3 were added when the serum free medium was replenished, *(Panel A)*, or serum was re-introduced to the cells *(Panel B)*. After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 4 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *** (p<0.001).



Figure 6.7 Effect of PKA agonists and inhibitor on proliferation in hPASMC.

Normoxic

As in figure 6.1, agonists and inhibitors were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Hypoxic

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *** (p<0.001).



Figure 6.8 Effect of PKA antagonists on proliferation in hPASMC.

Cells were treated as in figure 6.1. Antagonists were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *, *** (p<0.05, p<0.001 respectively).



Figure 6.9 Effect of cAMP analogs on proliferation in hPASMC

Cells were treated as in figure 6.1. 8-bromo-cAMP or CPT-cAMP were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by **, *** (p<0.01, p<0.001 respectively).



Figure 6.10 Effect of 8 -pCPT-2'-O-Me-cAMP on proliferation in hPASMC.

Cells were treated as in figure 6.1. 8 -pCPT-2'-O-Me-cAMP was added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*) After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by **, *** (p<0.01, p<0.001 respectively).



Figure 6.11 Effect of PDE inhibitors in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in hPASMC.

As in figure 6.1, 8 -pCPT-2'-O-Me-cAMP plus inhibitors were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *, **, *** (p<0.05, p<0.01, p<0.001 respectively).



Figure 6.12 Effect of cAMP effectors in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in hPASMC.

As in figure 6.1, 8 -pCPT-2'-O-Me-cAMP and other additions were made when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *, *** (p<0.05, p<0.001 respectively).



Figure 6.13 Effect of PKA agonists in combination with 8 -pCPT-2'-O-Me-cAMP on proliferation in hPASMC.

As in figure 6.1, 8 -pCPT-2'-O-Me-cAMP and agonists were added when the serum free medium was replenished, (*Panel A*), or serum was re-introduced to the cells (*Panel B*). After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *, **, *** (p<0.05, P<0.01, p<0.001 respectively).



Cells with serum reintroduced

Figure 6.14 Effect of Rap1 inhibitor on proliferation in hPASMC.

Cells were treated as in figure 6.1. GGTI-298 and the EPAC agonists were added when the serum was re-introduced to the cells. After twenty hours, [³H]-Thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means \pm S.E. of 3 independent experiments expressed as a percentage of the normoxic basal proliferation rate. Significance is denoted by *** (p<0.001).



8-pCPT-2'-O-Me-cAMP B 100 1nM Proliferation (% maximum) 10nM 75 100nM 1µM 50 10µM 100µM 25 0 Normoxic Hypoxic

Figure 6.15 Inhibition of proliferation in response to EPAC agonists.

Cells were treated as in figure 6.1. 8-pCPT-2'-O-Me-cAMP, (*Panel A*), or 8-pMeOPT-2'-O-Me-cAMP, (*Panel B*), was added at varying concentrations when serum was reintroduced to the cells. After twenty hours, [³H]-thymidine was added to all plates for a further four hours before being harvested as described in section 2.2 to measure incorporation rates as a measure of DNA synthesis.

Results shown are means +/- S.E. of 3 independent experiments as percentage inhibition compared to 1nM treatments which showed no effect on proliferation. Significance is denoted by *, *** (p<0.05, p<0.001 respectively).

Drug	Final Concentration
Rolipram	10µM
Cilostamide	10µM
Zaprinast	10µM
IBMX	10µM
Isoproterenol	10µM
Forskolin	10µM
EGF	50pg/µ1
TGF –β ₁	10ng/µ1
PP2	10µM
PP3	10µM
PKA RI agonists; 8-PIP-cAMP 8-HA-cAMP	10μM 5μM
PKA RII agonists; 8-PIP-cAMP 8-MBC-cAMP	10μM 5μM
H89	10µM
PKA RI antagonist; Rp-8-Cl-cAMPS	10µM
PKA RII antagonist; Rp-8-CPT-cAMPS	10µM
6-BnZ-cAMP	10µM
8-bromo-cAMP	100µM
CPT-cAMP	10µM
8 -pCPT-2'-O-Me-cAMP	10µM
GGTI-298	10µM
8-pMeOPT-2'-O-Me-cAMP	10µM

Table 6.1 Drugs used and their final concentration in the proliferation assaysThis table lists the compounds used throughout this chapter and the concentration used atin the proliferation assays.

Chapter 7

General Discussion and Future Directions

The cyclic nucleotides cAMP and cGMP have well characterised functions including roles in cell proliferation, differentiation, inflammation and maintenance of vascular tone (Murray 1990; Koyama et al., 2001; Pellegrino & Wang 1997; Dousa 1999; Torphy et al., 1999; Wong & Koh 2000; Essayan 2001). The cell tailors its response to these second messengers through a complex interplay of the synthesis and degradation mechanisms available for cAMP and cGMP. For example, there are nine forms of adenylyl cyclase (AC; Antoni et al., 2000, Hanoune & Defer 2001) capable of regulating the synthesis of cAMP and eleven families of phosphodiesterases, all able to efficiently degrade cAMP or cGMP (Francis et al., 1999; Dousa 1999; Soderling & Beavo 2000; Yuasa et al., 2000; Conti 2000; Mehats et al., 2002; Maurice et al., 2003).To add to this regulation, it is also well documented that compartmentalised responses to cAMP exist within a cell due to localised cAMP production, anchored cAMP substrates and targeted PDE isoforms. The net result of this is a highly specialised form of signalling, enabling the cell to tailor the response to cAMP and act in the desired manner.

The superfamily of phosphodiesterases is well established as a crucial component in the regulation of cAMP signalling. Differential tissue distribution and regulatory properties of PDEs add a further complexity to cAMP signalling, along with the generation of multiple splice variants. In addition, the expression of PDE isoforms has been demonstrated to be altered in response to hypoxia (Murray et al., 2002) and COPD (Barber et al., 2004). For example, in the pulmonary arteries of the chronic hypoxic rat model of pulmonary arterial hypertension (PAH), increased levels of the PDE3 and PDE5 expression and activity (Murray et al., 2002) have been demonstrated to be responsible for the reduction in cAMP and cGMP levels witnessed after 14 days exposure to 10% O₂ (MacLean et al., 1997). This alteration in PDE expression was also observed in the cellular model of PAH, hypoxic human pulmonary artery smooth muscle cells (Murray et al., 2002). Thus, the reduced levels of cyclic nucleotides have been attributed to an increased hydrolysis induced by hypoxia.

The PDE4 family of enzymes represent a major part of the cAMP hydrolysing activity within vascular smooth muscle cells (Palmer et al., 1998) but their activity and expression has not previously been investigated in hypoxia. PDE4 inhibitors are considered therapeutic targets for cardiovascular diseases such as COPD and asthma (Spina et al., 1998; Landells et al., 2001; Spina 2003). PDE4 inhibitors have also been considered for

use in treating PAH, as they are capable of synergising with PDE3 inhibitors to produce an enhanced relaxation of arteries and potentiated inhibition of smooth muscle cell proliferation (Pan et al., 1994; Palmer et al., 1998; Chen et al 2002, Ogawa et al 2002). I thus set out to determine if PDE4 has a role in the altered signalling apparent in models of PAH.

The main finding of the work presented here demonstrates that specific PDE4 isoforms are indeed affected by hypoxia. Increased PDE4 expression was detectable in hypoxic hPASMC after 7 days of hypoxic treatment, although there was no discernable affect on overall PDE4 activity. Indeed, it was also discovered that, in contrast with previous studies, cAMP levels are elevated following 7 days hypoxia treatment in hPASMC. Taken together, it was concluded from these results that the response to chronic hypoxia is a gradual one, with an initial increase in cAMP in an attempt to restore normal signalling, but instead causing the increased expression of the PDE3A isoform (Murray et al., 2002) and cAMP-inducible PDE4D1, PDE4D2, PDE4D5 isoforms. The elevated levels of cAMP-PDE isoforms possibly then leads to the marked reduction in cAMP and the desensitisation of further cAMP stimulation as observed previously (MacLean et al., 1997; Wagner et al., 1997). The elevated levels of PDE4B2 could also be due to the increased cAMP as an increase in PDE4B2 has been observed in human myometrial cells directly in response to cAMP raising agents (Mehats et al., 1999). The increased PDE4A10 and PDE4A11 expression highlights the importance of uncovering the regulatory properties and specific roles of these isoforms within a cell.

The increased expression of PDE4B2 and PDE4D5 also indicates a role for these isoforms in the hypoxic vasoconstrictive response in the pulmonary circulation. Previous studies have implicated increased levels of PDE4B2 in the contraction of myometrial strips (Mehats et al., 2002). In addition, PDE4D knockout mice display altered responses to muscarinic cholinergic stimulation (Hansen et al., 2000)

Further work in chapter 4 identified a role of the extracellular regulated kinase (ERK) in the increased levels of cAMP observed in hypoxia. ERK has previously been shown to act in an autocrine manner to stimulate production of PGE_2 and raise cAMP levels in human aortic smooth muscle cells (Baillie et al 2001; see figure 1.3). As cAMP has been demonstrated to regulate ERK activity within this loop (Pinelli et al., 1999), this provides a mechanism whereby cAMP levels and ERK activity can directly influence one another and themselves. Such a mechanism should enable the cell to regulate proliferative signals

correctly through inhibition of proliferation by cAMP and stimulation of proliferation through ERK activation, for example. In chapter 6, I demonstrated that hypoxic hPASMC have an increased proliferation rate, therefore the balance between cAMP signalling and ERK activation is altered. In fact, ERK was demonstrated to be directly involved in the hypoxia-induced increase in cAMP levels as MEK inhibitors used on hypoxic cells reduced cAMP levels to that of normoxic cells. It appears that the aforementioned components of the autocrine loop are increased in response to hypoxic stimulus. An increased ERK activity is well documented, and PGE₂ and COX-2 levels correspondingly have been noted to increase in hypoxia (Shaul et al., 1991; Jin et al., 2000; Bradbury et al., 2002; Yang et al., 2002). Interestingly, the increased ERK activity observed in hypoxia is transient (Jin et al., 2000; Scott et al., 1998). This is also suggestive of ERK regulation of cAMP levels in hypoxia as I have proposed that the increased cAMP observed is also temporary.

A recent study has shown that cAMP also regulates the autocrine loop albeit in a negative fashion. In human bronchial epithelial cells, stimulation of ERK leads to the increased production of prostaglandins in a manner which was potentiated by using a protein kinase A inhibitor (Pinelli et al., 1999). This indicates cAMP has an inhibitory effect on the ERK pathway in human bronchial epithelial cells thus providing a mechanism to down regulate the production of prostaglandins and thus cAMP.

In addition, the increased ERK activity in hypoxia regulates cAMP through its actions on the PDE4 family (Baillie et al., 2000). Dependant on isoform type, ERK can either inhibit or promote PDE4 activity (Houslay 2003; Baillie et al., 2000; Hoffmann et al., 1999), thus providing a cell specific ERK regulation of cAMP which is dependant on the PDE4 isoform profile within the cell. This is further complicated with the tight compartmentalisation of cAMP and ERK signalling within cells (Zaccolo et al., 2002; Pouyssegur et al., 2002). Pools of cAMP exist in regions of cells and can thus be localised away from the degrading action of ERK activated PDE4 short forms, or next to ERK inhibited PDE4 long forms within the cell and vice versa.

In combination, these results suggest the observed increase in cAMP levels in hypoxia may be an initial attempt by the cell to down regulate the enhanced ERK activity (Jin et al., 2000; Scott et al., 1998) and prostaglandin production (Shaul et al., 1991) witnessed in hypoxia that leads to increased smooth muscle cell proliferation (Yau & Zahradka 2003). Instead, the increased ERK maintains the increased level of cAMP through PGE₂

production and PDE4 inhibition, and levels continue to rise until a 'threshold' is breached whereby PKA phosphorylation of PDE4 overturns the inhibitory effect of ERK and cAMP is degraded at a faster rate than synthesised. ERK would then be reduced as a result. The decrease in ERK activity would completely alleviate inhibition of the PDE4 long forms present in hPASMC and increase the rate of cAMP hydrolysis even more, resulting in a desensitisation of the cell to cAMP raising agents. Indeed a negative feedback of cAMP has been demonstrated before and is thought to exist to protect against excessive accumulation of cAMP (Moon et al., 2002; Degerman et al., 1997).

Unfortunately, detecting the effect of ERK on PDE4 activity is compromised by the opposing modes of regulation ERK exerts on PDE4 long forms and short forms. To address this in future, immunoprecipitated PDE4 subfamilies should be assayed for PDE4 activity to determine the extent of inhibition. In addition, multiple time points must be used to gain a clearer understanding.

In conclusion, the first part of this study indicates a complex method of regulation of both cAMP and ERK. cAMP and ERK directly influence each other in a manner that has PDE4 at its centre. With the change in signalling that occurs in hypoxia, the balance that exists between cAMP and ERK is thrown into confusion and as a result, aids the development of pulmonary remodelling and desensitisation of the cAMP pathway.

The second part of this study examined the effect of hypoxia on the proliferative capacity of hPASMC. It was observed that hypoxia induces a significant increase in hPASMC proliferation, both in response to serum and in serum starved cells. This suggests the increased proliferation occurs in response to increased mitogenic signals produced in hypoxia. In addition, results gained using specific PKA agonists indicate the involvement of cAMP signalling. It appears that the increase in cAMP levels and proliferation witnessed in hypoxia contrast each other, suggesting the cAMP pathway is severely impaired in hypoxia.

Chapter 6 also revealed a novel role for the recently discovered cAMP substrate, Epac (de Rooij et al., 1998). Previous studies investigating the role of cAMP and cell proliferation have attributed all effects to PKA, although PKA-independent pathways are being uncovered (Cass et al., 1999). Surprisingly, the role of Epac in cAMP-mediated inhibition or stimulation of proliferation has not been investigated. Using an Epac-specific agonist, it was discovered that Epac selectively inhibits serum-induced proliferation. Following on

from this, identification of cell-cycle proteins altered in response to this agonist should be investigated to determine how Epac affects proliferation. In addition, utilising this agonist in cells with PKA RI or PKA RII knocked out could be useful as the results presented here show that Epac and PKA both serve the same function when overstimulated in these cells, indicating a balancing mechanism between PKA and Epac signalling. The result of this is either PKA or Epac inhibit cell proliferation in the absence of equal stimulation of the other. Also, the effect of stimulating Epac in other cells should be examined to determine if this is specific to smooth muscle cells.

The majority of the work presented here adds a further dimension to the role and regulation of cAMP in pulmonary arterial hypertension. It is obvious that PAH is a complex process, inducing the alteration of signalling pathways in an attempt to restore normal tone to the pulmonary circulation. Unfortunately, these alterations can directly and indirectly affect other pathways leading to enhanced signalling and the resultant remodelling of the pulmonary arterial tree. Due to the numerous signalling pathways involved in the hypoxic response, it is apparent that multiple drug therapies are required to attenuate the development of PAH. The results presented in this study further establish the potential of prostanoid therapy and PDE inhibition as a treatment for PAH. As both can impact upon cAMP and ERK pathways, it is possible these therapies can affect upon remodelling and hypoxia-induced transcription. Further studies examining the time points over the development of PAH should uncover a more precise characterisation of the effects of hypoxia on cAMP signalling and affected pathways, allowing the identification of specific therapeutic targets to potentiate the beneficial effects of such treatments and reduce the side effects associated with the non-specificity of the treatments used today.

The proliferation studies carried out here identify an exciting role for Epac in the regulation of proliferation. Although the Epac response is equal in both normoxic and hypoxic cells, a better understanding of the mechanisms controlling this response could help elucidate the alterations occurring in hypoxia to induce proliferation.

References

Alessi, D. R., Cuenda, A., Cohen, P., Dudley, D. T., and Saltiel, A. R. (1995). PD 098059 is a specific inhibitor of the activation of mitogen-activated protein kinase kinase in vitro and in vivo. J Biol Chem 270, 27489-27494.

Alvarez, R., Sette, C., Yang, D., Eglen, R. M., Wilhelm, R., Shelton, E. R., and Conti, M. (1995). Activation and selective inhibition of a cyclic AMP-specific phosphodiesterase, PDE-4D3. Mol Pharmacol 48, 616-622.

Andrews, G. K. (2000). Regulation of metallothionein gene expression by oxidative stress and metal ions. Biochem Pharmacol 59, 95-104.

Antoni, F. A. (2000). Molecular diversity of cyclic AMP signalling. Front Neuroendocrinol 21, 103-132.

Arany, Z., Huang, L. E., Eckner, R., Bhattacharya, S., Jiang, C., Goldberg, M. A., Bunn,
H. F., and Livingston, D. M. (1996). An essential role for p300/CBP in the cellular response to hypoxia. Proc Natl Acad Sci U S A 93, 12969-12973.

Archer, S., and Rich, S. (2000). Primary pulmonary hypertension: a vascular biology and translational research "Work in progress". Circulation *102*, 2781-2791.

Archer, S. L., Djaballah, K., Humbert, M., Weir, K. E., Fartoukh, M., Dall'ava-Santucci, J., Mercier, J. C., Simonneau, G., and Dinh-Xuan, A. T. (1998). Nitric oxide deficiency in fenfluramine- and dexfenfluramine-induced pulmonary hypertension. Am J Respir Crit Care Med 158, 1061-1067.

Arcot, S. S., Fagerland, J. A., Lipke, D. W., Gillespie, M. N., and Olson, J. W. (1995). Basic fibroblast growth factor alterations during development of monocrotaline-induced pulmonary hypertension in rats. Growth Factors 12, 121-130.

Arcot, S. S., Lipke, D. W., Gillespie, M. N., and Olson, J. W. (1993). Alterations of growth factor transcripts in rat lungs during development of monocrotaline-induced pulmonary

hypertension. Biochem Pharmacol 46, 1086-1091.

Badesch, D. B., McLaughlin, V. V., Delcroix, M., Vizza, C. D., Olschewski, H., Sitbon, O., and Barst, R. J. (2004). Prostanoid therapy for pulmonary arterial hypertension. J Am Coll Cardiol 43, 56S-61S.

Bae, M. K., Ahn, M. Y., Jeong, J. W., Bae, M. H., Lee, Y. M., Bae, S. K., Park, J. W., Kim, K. R., and Kim, K. W. (2002). Jab1 interacts directly with HIF-1alpha and regulates its stability. J Biol Chem 277, 9-12.

Baillie, G., MacKenzie, S. J., and Houslay, M. D. (2001). Phorbol 12-myristate 13-acetate triggers the protein kinase A-mediated phosphorylation and activation of the PDE4D5 cAMP phosphodiesterase in human aortic smooth muscle cells through a route involving extracellular signal regulated kinase (ERK). Mol Pharmacol *60*, 1100-1111.

Baillie, G. S., Huston, E., Scotland, G., Hodgkin, M., Gall, I., Peden, A. H., MacKenzie, C., Houslay, E. S., Currie, R., Pettitt, T. R., *et al.* (2002). TAPAS-1, a novel microdomain within the unique N-terminal region of the PDE4A1 cAMP-specific phosphodiesterase that allows rapid, Ca2+-triggered membrane association with selectivity for interaction with phosphatidic acid. J Biol Chem 277, 28298-28309.

Baillie, G. S., MacKenzie, S. J., McPhee, I., and Houslay, M. D. (2000). Sub-family selective actions in the ability of Erk2 MAP kinase to phosphorylate and regulate the activity of PDE4 cyclic AMP-specific phosphodiesterases. Br J Pharmacol *131*, 811-819. Baillie, G. S., Sood, A., McPhee, I., Gall, I., Perry, S. J., Lefkowitz, R. J., and Houslay, M. D. (2003). beta-Arrestin-mediated PDE4 cAMP phosphodiesterase recruitment regulates beta-adrenoceptor switching from Gs to Gi. Proc Natl Acad Sci U S A *100*, 940-945.

Balmanno, K., Millar, T., McMahon, M., and Cook, S. J. (2003). DeltaRaf-1:ER* bypasses the cyclic AMP block of extracellular signal-regulated kinase 1 and 2 activation but not CDK2 activation or cell cycle reentry. Mol Cell Biol *23*, 9303-9317.

Banner, K. H., Dimitriou, G., Kinali, M., Page, C. P., and Greenough, A. (2000). Evidence to suggest that the phosphodiesterase 4 isoenzyme is present and involved in the proliferation of umbilical cord blood mononuclear cells. Clin Exp Allergy *30*, 706-712.

Banner, K. H., Hoult, J. R., Taylor, M. N., Landells, L. J., and Page, C. P. (1999). Possible Contribution of Prostaglandin E2 to the antiproliferative effect of phosphodiesterase 4 inhibitors in human mononuclear cells. Biochem Pharmacol 58, 1487-1495.

Barber, R., Baillie, G. S., Bergmann, R., Shepherd, M. C., Sepper, R., Houslay, M. D., and Heeke, G. V. (2004). Differential expression of PDE4 cAMP phosphodiesterase isoforms in inflammatory cells of smokers with COPD, smokers without COPD, and nonsmokers. Am J Physiol Lung Cell Mol Physiol 287, L332-343.

Bardou, M., Goirand, F., Bernard, A., Guerard, P., Gatinet, M., Devillier, P., Dumas, J. P., Morciilo, E. J., Rochette, L., and Dumas, M. (2002). Relaxant effects of selective phosphodiesterase inhibitors on U46619 precontracted human intralobar pulmonary arteries and role of potassium channels. J Cardiovasc Pharmacol 40, 153-161.

Bardou, M., Goirand, F., Marchand, S., Rouget, C., Devillier, P., Dumas, J. P., Morcillo, E. J., Rochette, L., and Dumas, M. (2001). Hypoxic vasoconstriction of rat main pulmonary artery: role of endogenous nitric oxide, potassium channels, and phosphodiesterase inhibition. J Cardiovase Pharmacol *38*, 325-334.

Barnard, J. W., Barman, S. A., Adkins, W. K., Longenecker, G. L., and Taylor, A. E. (1991). Sustained effects of endothelin-1 on rabbit, dog, and rat pulmonary circulations. Am J Physiol 261, H479-486.

Bauer, M., Reibiger, I., and Spanel-Borowski, K. (2001). Leucocyte proliferation in the bovine corpus luteum. Reproduction 121, 297-305.

Beard, M. B., Huston, E., Campbell, L., Gall, I., McPhee, I., Yarwood, S., Scotland, G., and Houslay, M. D. (2002). In addition to the SH3 binding region, multiple regions within the N-terminal noncatalytic portion of the cAMP-specific phosphodicsterase, PDE4A5, contribute to its intracellular targeting. Cell Signal *14*, 453-465.

Beard, M. B., O'Connell, J. C., Bolger, G. B., and Houslay, M. D. (1999). The unique N-terminal domain of the cAMP phosphodiesterase PDE4D4 allows for interaction with specific SH3 domains. FEBS Lett *460*, 173-177.

Beard, M. B., Oisen, A. E., Jones, R. E., Erdogan, S., Houslay, M. D., and Bolger, G. B. (2000). UCR1 and UCR2 domains unique to the cAMP-specific phosphodiesterase family form a discrete module via electrostatic interactions. J Biol Chem 275, 10349-10358.

Beitner-Johnson, D., and Millhorn, D. E. (1998). Hypoxia induces phosphorylation of the cyclic AMP response element-binding protein by a novel signaling mechanism. J Biol Chem 273, 19834-19839.

Belknap, J. K., Orton, E. C., Ensley, B., Tucker, A., and Stenmark, K. R. (1997). Hypoxia increases bromodeoxyuridine labeling indices in bovine neonatal pulmonary arteries. Am J Respir Cell Mol Biol *16*, 366-371.

Bernardelli, P., Lorthiois, E., Vergne, F., Oliveira, C., Mafroud, A. K., Proust, E., Pham, N., Ducrot, P., Moreau, F., Idrissi, M., *et al.* (2004). Spiroquinazolinones as novel, potent, and selective PDE7 inhibitors. Part 2: Optimization of 5,8-disubstituted derivatives. Bioorg Med Chem Lett *14*, 4627-4631.

Bill, H. M., Knudsen, B., Moores, S. L., Muthuswamy, S. K., Rao, V. R., Brugge, J. S., and Miranti, C. K. (2004). Epidermal growth factor receptor-dependent regulation of integrin-mediated signaling and cell cycle entry in epithelial cells. Mol Cell Biol 24, 8586-8599.

Blagosklonny, M. V., An, W. G., Romanova, L. Y., Trepel, J., Fojo, T., and Neckers, L. (1998). p53 inhibits hypoxia-inducible factor-stimulated transcription. J Biol Chem 273, 11995-11998.

Bohm, M., Flesch, M., and Schnabel, P. (1997). Beta-adrenergic signal transduction in the failing and hypertrophied myocardium. J Mol Med 75, 842-848.

Bolger, G. B., Erdogan, S., Jones, R. E., Loughney, K., Scotland, G., Hoffmann, R., Wilkinson, I., Farrell, C., and Houslay, M. D. (1997). Characterization of five different proteins produced by alternatively spliced mRNAs from the human cAMP-specific phosphodiesterase PDE4D gene. Biochem J 328 (*Pt 2*), 539-548.

Bolger, G. B., McPhee, I., and Houslay, M. D. (1996). Alternative splicing of cAMP-

specific phosphodiesterase mRNA transcripts. Characterization of a novel tissuespecific isoform, RNPDE4A8. J Biol Chem 271, 1065-1071.

Bornfeldt, K. E., and Krebs, E. G. (1999). Crosstalk between protein kinase A and growth factor receptor signaling pathways in arterial smooth muscle. Cell Signal 11, 465-477.

Bos, J. L., de Bruyn, K., Enserink, J., Kuiperij, B., Rangarajan, S., Rehmann, H., Riedl, J., de Rooij, J., van Mansfeld, F., and Zwartkruis, F. (2003). The role of Rap1 in integrinmediated cell adhesion. Biochem Soc Trans 31, 83-86.

Bos, R., van Diest, P. J., van der Groep, P., Shvarts, A., Greijer, A. E., and van der Wall, E. (2004). Expression of hypoxia-inducible factor-1alpha and cell cycle proteins in invasive breast cancer are estrogen receptor related. Breast Cancer Res *6*, R450-459.

Boucher, M. J., Duchesne, C., Laine, J., Morisset, J., and Rivard, N. (2001). cAMP protection of pancreatic cancer cells against apoptosis induced by ERK inhibition. Biochem Biophys Res Commun 285, 207-216.

Boudreau, N., and Rabinovitch, M. (1991). Developmentally regulated changes in extracellular matrix in endothelial and smooth muscle cells in the ductus arteriosus may be related to intimal proliferation. Lab Invest 64, 187-199.

Boynton, A. J., Whitfield, J. F., and Kleine, L. P. (1983). Ca2+/phospholipid-dependent protein kinase activity correlates to the ability of transformed liver cells to proliferate in Ca2+-deficient medium. Biochem Biophys Res Commun 115, 383-390.

Bradbury, D. A., Newton, R., Zhu, Y. M., Stocks, J., Corbett, L., Holland, E. D., Pang, L. H., and Knox, A. J. (2002). Effect of bradykinin, TGF-beta1, IL-1beta, and hypoxia on COX-2 expression in pulmonary artery smooth muscle cells. Am J Physiol Lung Cell Mol Physiol 283, L717-725.

Brij, S. O., and Peacock, A. J. (1998). Cellular responses to hypoxia in the pulmonary circulation. Thorax 53, 1075-1079.

Burgun, C., Esteve, L., Humblot, N., Aunis, D., and Zwiller, J. (2000). Cyclic AMPelevating agents induce the expression of MAP kinase phosphatase-1 in PC12 cells. FEBS The state of the second second

Burnouf, C., and Pruniaux, M. P. (2002). Recent advances in PDE4 inhibitors as immunoregulators and anti-inflammatory drugs. Curr Pharm Des 8, 1255-1296.

Cacoub, P., Dorent, R., Nataf, P., Carayon, A., Riquet, M., Noe, E., Piette, J. C., Godeau, P., and Gandjbakhch, I. (1997). Endothelin-1 in the lungs of patients with pulmonary hypertension. Cardiovasc Res 33, 196-200.

Cali, J. J., Zwaagstra, J. C., Mons, N., Cooper, D. M., and Krupinski, J. (1994). Type VIII adenylyl cyclase. A Ca2+/calmodulin-stimulated enzyme expressed in discrete regions of rat brain. J Biol Chem 269, 12190-12195.

Cass, L. A., Summers, S. A., Prendergast, G. V., Backer, J. M., Birnbaum, M. J., and Meinkoth, J. L. (1999). Protein kinase A-dependent and -independent signaling pathways contribute to cyclic AMP-stimulated proliferation. Mol Cell Biol *19*, 5882-5891.

Cass, L. A., Summers, S. A., Prendergast, G. V., Backer, J. M., Birnbaum, M. J., and Meinkoth, J. L. (1999). Protein kinase A-dependent and -independent signaling pathways contribute to cyclic AMP-stimulated proliferation. Mol Cell Biol *19*, 5882-5891.

Channick, R. N., Sitbon, O., Barst, R. J., Manes, A., and Rubin, L. J. (2004). Endothelin receptor antagonists in pulmonary arterial hypertension. J Am Coll Cardiol 43, 62S-67S. Chen, J., and Iyengar, R. (1994). Suppression of Ras-induced transformation of NIII 3T3 cells by activated G alpha s. Science 263, 1278-1281.

Chen, S. J., Chen, Y. F., Opgenorth, T. J., Wessale, J. L., Meng, Q. C., Durand, J., DiCarlo, V. S., and Oparil, S. (1997). The orally active nonpeptide endothelin A-receptor antagonist A-127722 prevents and reverses hypoxia-induced pulmonary hypertension and pulmonary vascular remodeling in Sprague-Dawley rats. J Cardiovasc Pharmacol 29, 713-725.

Chen, T. C., Wadsten, P., Su, S., Rawlinson, N., Hofman, F. M., Hill, C. K., and Schonthal, A. H. (2002). The type IV phosphodiesterase inhibitor rolipram induces expression of the cell cycle inhibitors p21(Cip1) and p27(Kip1), resulting in growth inhibition, increased differentiation, and subsequent apoptosis of malignant A-172 glioma cells. Cancer Biol Ther 1, 268-276.

Chiarugi, V., Magnelli, L., Chiarugi, A., and Gallo, O. (1999). Hypoxia induces pivotal tumor angiogenesis control factors including p53, vascular endothelial growth factor and the NFkappaB-dependent inducible nitric oxide synthase and cyclooxygenase-2. J Cancer Res Clin Oncol *125*, 525-528.

Cho-Chung, Y. S., and Clair, T. (1993). The regulatory subunit of cAMP-dependent protein kinase as a target for chemotherapy of cancer and other cellular dysfunctional-related diseases. Pharmacol Ther *60*, 265-288.

Cho-Chung, Y. S., Pepe, S., Clair, T., Budillon, A., and Nesterova, M. (1995). cAMPdependent protein kinase: role in normal and malignant growth. Crit Rev Oncol Hernatol 21, 33-61.

Christman, B. W. (1998). Lipid mediator dysregulation in primary pulmonary hypertension. Chest 114, 205S-207S.

Christman, B. W., McPherson, C. D., Newman, J. H., King, G. A., Bernard, G. R., Groves, B. M., and Loyd, J. E. (1992). An imbalance between the excretion of thromboxane and prostacyclin metabolites in pulmonary hypertension. N Engl J Med *327*, 70-75.

Cody, R. J., Haas, G. J., Binkley, P. F., Capers, Q., and Kelley, R. (1992). Plasma endothelin correlates with the extent of pulmonary hypertension in patients with chronic congestive heart failure. Circulation *85*, 504-509.

Cohen, A. H., Hanson, K., Morris, K., Fouty, B., McMurty, I. F., Clarke, W., and Rodman, D. M. (1996). Inhibition of cyclic 3'-5'-guanosine monophosphate-specific phosphodiesterase selectively vasodilates the pulmonary circulation in chronically hypoxic rats. J Clin Invest 97, 172-179.

Compton, C. H., Gubb, J., Nieman, R., Edelson, J., Amit, O., Bakst, A., Ayres, J. G., Creemers, J. P., Schultze-Werninghaus, G., Brambilla, C., and Barnes, N. C. (2001). Cilomilast, a selective phosphodiesterase-4 inhibitor for treatment of patients with chronic obstructive pulmonary disease: a randomised, dose-ranging study. Lancet *358*, 265-270.

Conti, M. (2000). Phosphodiesterases and cyclic nucleotide signaling in endocrine cells. Mol Endocrinol 14, 1317-1327.

Cook, S. J., and McCormick, F. (1993). Inhibition by cAMP of Ras-dependent activation of Raf. Science 262, 1069-1072.

Cooper, D. M., Mons, N., and Karpen, J. W. (1995). Adenylyl cyclases and the interaction between calcium and cAMP signalling. Nature *374*, 421-424.

Corbin, J. D., and Francis, S. H. (1999). Cyclic GMP phosphodiesterase-5: target of sildenafil. J Biol Chem 274, 13729-13732.

Coso, O. A., Chiariello, M., Kalinec, G., Kyriakis, J. M., Woodgett, J., and Gutkind, J. S. (1995). Transforming G protein-coupled receptors potently activate JNK (SAPK). Evidence for a divergence from the tyrosine kinase signaling pathway. J Biol Chem 270, 5620-5624.

D'Angelo, G., Lee, H., and Weiner, R. I. (1997). cAMP-dependent protein kinase inhibits the mitogenic action of vascular endothelial growth factor and fibroblast growth factor in capillary endothelial cells by blocking Raf activation. J Cell Biochem 67, 353-366.

Davies, S. P., Reddy, H., Caivano, M., and Cohen, P. (2000). Specificity and mechanism of action of some commonly used protein kinase inhibitors. Biochem J *351*, 95-105.

Davis, R. L., and Dauwalder, B. (1991). The Drosophila dunce locus: learning and memory genes in the fly. Trends Genet 7, 224-229.

de Rooij, J., Zwartkruis, F. J., Verheijen, M. H., Cool, R. H., Nijman, S. M., Wittinghofer, A., and Bos, J. L. (1998). Epac is a Rap1 guanine-nucleotide-exchange factor directly activated by cyclic AMP. Nature *396*, 474-477.

Degerman, E., Belfrage, P., and Manganiello, V. C. (1997). Structure, localization, and regulation of cGMP-inhibited phosphodiesterase (PDE3). J Biol Chem 272, 6823-6826.

Deng, Z., Haghighi, F., Helleby, L., Vanterpool, K., Horn, E. M., Barst, R. J., Hodge, S. E., Morse, J. H., and Knowles, J. A. (2000). Fine mapping of PPH1, a gene for familial primary pulmonary hypertension, to a 3-cM region on chromosome 2q33. Am J Respir Crit Care Med 161, 1055-1059.

DiCarlo, V. S., Chen, S. J., Meng, Q. C., Durand, J., Yano, M., Chen, Y. F., and Oparil, S. (1995). ETA-receptor antagonist prevents and reverses chronic hypoxia-induced pulmonary hypertension in rat. Am J Physiol 269, L690-697.

Dodge, K., and Scott, J. D. (2000). AKAP79 and the evolution of the AKAP model. FEBS Lett 476, 58-61.

Dodge, K. L., Khouangsathiene, S., Kapiloff, M. S., Mouton, R., Hill, E. V., Houslay, M. D., Langeberg, L. K., and Scott, J. D. (2001). mAKAP assembles a protein kinase A/PDE4 phosphodiesterase cAMP signaling module. Embo J 20, 1921-1930.

Doherty, A. M. (1999). Phosphodiesterase 4 inhibitors as novel anti-inflammatory agents. Curr Opin Chem Biol 3, 466-473.

Dousa, T. P. (1999). Cyclic-3',5'-nucleotide phosphodiesterase isozymes in cell biology and pathophysiology of the kidney. Kidney Int 55, 29-62.

Duncia, J. V., Santella, J. B., 3rd, Higley, C. A., Pitts, W. J., Wityak, J., Frietze, W. E., Rankin, F. W., Sun, J. H., Earl, R. A., Tabaka, A. C., *et al.* (1998). MEK inhibitors: the chemistry and biological activity of U0126, its analogs, and cyclization products. Bioorg Med Chem Lett *8*, 2839-2844.

Dykxhoorn, D. M., Novina, C. D., and Sharp, P. A. (2003). Killing the messenger: short RNAs that silence gene expression. Nat Rev Mol Cell Biol *4*, 457-467.

Eckly-Michel, A., Martin, V., and Lugnier, C. (1997). Involvement of cyclic nucleotidedependent protein kinases in cyclic AMP-mediated vasorelaxation. Br J Pharmacol *122*, 158-164.

Eddahibi, S., Chaouat, A., Morrell, N., Fadel, E., Fuhrman, C., Bugnet, A. S., Dartevelle, P., Housset, B., Hamon, M., Weitzenblum, E., and Adnot, S. (2003). Polymorphism of the serotonin transporter gene and pulmonary hypertension in chronic obstructive pulmonary disease. Circulation *108*, 1839-1844.

Eddahibi, S., Fabre, V., Boni, C., Martres, M. P., Raffestin, B., Hamon, M., and Adnot, S. (1999). Induction of serotonin transporter by hypoxia in pulmonary vascular smooth muscle cells. Relationship with the mitogenic action of serotonin. Circ Res *84*, 329-336.

Eddahibi, S., Hanoun, N., Lanfumey, L., Lesch, K. P., Raffestin, B., Hamon, M., and Adnot, S. (2000). Attenuated hypoxic pulmonary hypertension in mice lacking the 5-hydroxytryptamine transporter gene. J Clin Invest *105*, 1555-1562.

Eddahibi, S., Humbert, M., Fadel, E., Raffestin, B., Darmon, M., Capron, F., Simonneau, G., Dartevelle, P., Hamon, M., and Adnot, S. (2001). Serotonin transporter overexpression is responsible for pulmonary artery smooth muscle hyperplasia in primary pulmonary hypertension. J Clin Invest 108, 1141-1150.

Eddahibi, S., Raffestin, B., Braquet, P., Chabrier, P. E., and Adnot, S. (1991). Pulmonary vascular reactivity to endothelin-1 in normal and chronically pulmonary hypertensive rats. J Cardiovasc Pharmacol *17 Suppl 7*, S358-361.

Eddahibi, S., Raffestin, B., Le Monnier de Gouville, A. C., and Adnot, S. (1998). Effect of DMPPO, a phosphodiesterase type 5 inhibitor, on hypoxic pulmonary hypertension in rats. Br J Pharmacol *125*, 681-688.

Eddahibi, S., Raffestin, B., Pham, I., Launay, J. M., Aegerter, P., Sitbon, M., and Adnot, S. (1997). Treatment with 5-HT potentiates development of pulmonary hypertension in chronically hypoxic rats. Am J Physiol 272, H1173-1181.

El Jamali, A., Freund, C., Rechner, C., Scheidereit, C., Dietz, R., and Bergmann, M. W. (2004). Reoxygenation after severe hypoxia induces cardiomyocyte hypertrophy in vitro: activation of CREB downstream of GSK3beta. Faseb J *18*, 1096-1098.

Enserink, J. M., Christensen, A. E., de Rooij, J., van Triest, M., Schwede, F., Genieser, H. G., Doskeland, S. O., Blank, J. L., and Bos, J. L. (2002). A novel Epac-specific cAMP analogue demonstrates independent regulation of Rap1 and ERK. Nat Cell Biol 4, 901-906.

Essayan, D. M. (1999). Cyclic nucleotide phosphodiesterase (PDE) inhibitors and

Essayan, D. M. (2001). Cyclic nucleotide phosphodiesterases. J Allergy Clin Immunol 108, 671-680.

Fagan, K. A., Fouty, B. W., Tyler, R. C., Morris, K. G., Jr., Hepler, L. K., Sato, K., LeCras, T. D., Abman, S. H., Weinberger, H. D., Huang, P. L., *et al.* (1999). The pulmonary circulation of homozygous or heterozygous eNOS-null mice is hyperresponsive to mild hypoxia. J Clin Invest 103, 291-299.

Faller, D. V. (1999). Endothelial cell responses to hypoxic stress. Clin Exp Pharmacol Physiol 26, 74-84.

Feliciello, A., Gottesman, M. E., and Avvedimento, E. V. (2001). The biological functions of A-kinase anchor proteins. J Mol Biol *308*, 99-114.

Ferguson, S. S. (2001). Evolving concepts in G protein-coupled receptor endocytosis: the role in receptor desensitization and signaling. Pharmacol Rev 53, 1-24.

Fisher, D. A., Smith, J. F., Pillar, J. S., St Denis, S. H., and Cheng, J. B. (1998). Isolation and characterization of PDE9A, a novel human cGMP-specific phosphodiesterase. J Biol Chem 273, 15559-15564.

Fishman, A. P. (1999). Aminorex to fen/phen: an epidemic foretold. Circulation 99, 156-161.

Fishman, A. P. (2004). Primary pulmonary arterial hypertension: a look back. J Am Coll Cardiol 43, 2S-4S.

Forrest, I. A., Small, T., and Corris, P. A. (1999). Effect of nebulized epoprostenol (prostacyclin) on exhaled nitric oxide in patients with pulmonary hypertension due to congenital heart disease and in normal controls. Clin Sci (Lond) *97*, 99-102.

Frid, M. G., Dempsey, E. C., Durmowicz, A. G., and Stenmark, K. R. (1997). Smooth muscle cell heterogeneity in pulmonary and systemic vessels. Importance in vascular disease. Arterioscler Thromb Vasc Biol 17, 1203-1209.

Fujita, M., Urano, T., Horie, K., Ikeda, K., Tsukui, T., Fukuoka, H., Tsutsumi, O., Ouchi, Y., and Inoue, S. (2002). Estrogen activates cyclin-dependent kinases 4 and 6 through induction of cyclin D in rat primary osteoblasts. Biochem Biophys Res Commun 299, 222-228.

Fukumoto, S., Koyama, H., Hosoi, M., Yamakawa, K., Tanaka, S., Morii, H., and Nishizawa, Y. (1999). Distinct role of cAMP and cGMP in the cell cycle control of vascular smooth muscle cells: cGMP delays cell cycle transition through suppression of cyclin D1 and cyclin-dependent kinase 4 activation. Circ Res *85*, 985-991.

Geraci, M. W., Gao, B., Shepherd, D. C., Moore, M. D., Westcott, J. Y., Fagan, K. A., Alger, L. A., Tuder, R. M., and Voelkel, N. F. (1999). Pulmonary prostacyclin synthase overexpression in transgenic mice protects against development of hypoxic pulmonary hypertension. J Clin Invest *103*, 1509-1515.

Gerber, H. P., Condorelli, F., Park, J., and Ferrara, N. (1997). Differential transcriptional regulation of the two vascular endothelial growth factor receptor genes. Flt-1, but not Flk-1/KDR, is up-regulated by hypoxia. J Biol Chem 272, 23659-23667.

Ghofrani, H. A., Pepke-Zaba, J., Barbera, J. A., Channick, R., Kcogh, A. M., Gomez-Sanchez, M. A., Kneussl, M., and Grimminger, F. (2004). Nitric oxide pathway and phosphodiesterase inhibitors in pulmonary arterial hypertension. J Am Coll Cardiol 43, 68S-72S.

Giaid, A., and Saleh, D. (1995). Reduced expression of endothelial nitric oxide synthase in the lungs of patients with pulmonary hypertension. N Engl J Med 333, 214-221.

Giaid, A., Salch, D., Yanagisawa, M., and Forbes, R. D. (1995). Endothelin-1 immunoreactivity and mRNA in the transplanted human heart. Transplantation *59*, 1308-1313.

Giaid, A., Yauagisawa, M., Langleben, D., Michel, R. P., Levy, R., Shennib, H., Kimura, S., Masaki, T., Duguid, W. P., and Stewart, D. J. (1993). Expression of endothelin-1 in the lungs of patients with pulmonary hypertension. N Engl J Med *328*, 1732-1739. Giembycz, M. A. (2001). Cilomilast: a second generation phosphodiesterase 4 inhibitor for

asthma and chronic obstructive pulmonary disease. Expert Opin Investig Drugs 10, 1361-1379.

Gillespie, M. N., Rippetoc, P. E., Haven, C. A., Shiao, R. T., Orlinska, U., Maley, B. E., and Olson, J. W. (1989). Polyamines and epidermal growth factor in monocrotaline-induced pulmonary hypertension. Am Rev Respir Dis 140, 1463-1466.

Giordano, D., De Stefano, M. E., Citro, G., Modica, A., and Giorgi, M. (2001). Expression of cGMP-binding cGMP-specific phosphodiesterase (PDE5) in mouse tissues and cell lines using an antibody against the enzyme amino-terminal domain. Biochim Biophys Acta *1539*, 16-27.

Goirand, F., Bardou, M., Dumas, J., Rochette, L., and Dumas, M. (2001). Effects of phosphodiesterase inhibitors on hypoxic pulmonary vasoconstriction. Influence of K(+) channels and nitric oxide. Eur J Pharmacol *417*, 141-148.

Goncharova, E. A., Billington, C. K., Irani, C., Vorotnikov, A. V., Tkachuk, V. A., Penn, R. B., Krymskaya, V. P., and Panettieri, R. A., Jr. (2003). Cyclic AMP-mobilizing agents and glucocorticoids modulate human smooth muscle cell migration. Am J Respir Cell Mol Biol 29, 19-27.

Graeber, T. G., Peterson, J. F., Tsai, M., Monica, K., Fornace, A. J., Jr., and Giaccia, A. J. (1994). Hypoxia induces accumulation of p53 protein, but activation of a G1-phase checkpoint by low-oxygen conditions is independent of p53 status. Mol Cell Biol 14, 6264-6277.

Grainger, D. J., Kemp, P. R., Witchell, C. M., Weissberg, P. L., and Metcalfe, J. C. (1994). Transforming growth factor beta decreases the rate of proliferation of rat vascular smooth muscle cells by extending the G2 phase of the cell cycle and delays the rise in cyclic AMP before entry into M phase. Biochem J 299 (*Pt 1*), 227-235.

Grange, M., Sette, C., Cuomo, M., Conti, M., Lagarde, M., Prigent, A. F., and Nemoz, G. (2000). The cAMP-specific phosphodiesterase PDE4D3 is regulated by phosphatidic acid binding. Consequences for cAMP signaling pathway and characterization of a phosphatidic acid binding site. J Biol Chem 275, 33379-33387.

Gretarsdottir, S., Thorleifsson, G., Reynisdottir, S. T., Manolescu, A., Jonsdottir, S., Jonsdottir, T., Gudmundsdottir, T., Bjarnadottir, S. M., Einarsson, O. B., Gudjonsdottir, H. M., *et al.* (2003). The gene encoding phosphodiesterase 4D confers risk of ischemic stroke. Nat Genet 35, 131-138.

Grootendorst, D. C., Gauw, S. A., Benschop, N., Sterk, P. J., Hiemstra, P. S., and Rabe, K. F. (2003). Efficacy of the novel phosphodiesterase-4 inhibitor BAY 19-8004 on lung function and airway inflammation in asthma and chronic obstructive pulmonary disease (COPD). Pulm Pharmacol Ther *16*, 341-347.

Guldemeester, H. A., Stenmark, K. R., Brough, G., Tuder, R. M., and Stevens, T. (1998). Role of adenylyl cyclase in proliferation of neonatal pulmonary artery smooth muscle cells. Chest *114*, 38S-39S.

Gutkind, J. S. (2000). Regulation of mitogen-activated protein kinase signaling networks by G protein-coupled receptors. Sci STKE 2000, RE1.

Hakonarson, H., and Grunstein, M. M. (1998). Regulation of second messengers associated with airway smooth muscle contraction and relaxation. Am J Respir Crit Care Med 158, \$115-122.

Hakonarson, H., Halapi, E., Whelan, R., Gulcher, J., Stefansson, K., and Grunstein, M. M. (2001). Association between IL-1beta/TNF-alpha-induced glucocorticoid-sensitive changes in multiple gene expression and altered responsiveness in airway smooth muscle. Am J Respir Cell Mol Biol 25, 761-771.

Hanoune, J., and Defer, N. (2001). Regulation and role of adenylyl cyclase isoforms. Annu Rev Pharmacol Toxicol 41, 145-174.

Hanoune, J., Pouille, Y., Tzavara, E., Shen, T., Lipskaya, L., Miyamoto, N., Suzuki, Y., and Defer, N. (1997). Adenylyl cyclases: structure, regulation and function in an enzyme superfamily. Mol Cell Endocrinol *128*, 179-194.

Hansen, G., Jin, S., Umetsu, D. T., and Conti, M. (2000). Absence of muscarinic cholinergic airway responses in mice deficient in the cyclic nucleotide phosphodicsterase PDE4D. Proc Natl Acad Sci U S A 97, 6751-6756.

229

Hansen, P. A., Wang, W., Marshall, B. A., Holloszy, J. O., and Mueckler, M. (1998). Dissociation of GLUT4 translocation and insulin-stimulated glucose transport in transgenic mice overexpressing GLUT1 in skeletal muscle. J Biol Chem 273, 18173-18179.

Hanze, J., Eul, B. G., Savai, R., Krick, S., Goyal, P., Grimminger, F., Seeger, W., and Rose, F. (2003). RNA interference for HIF-1alpha inhibits its downstream signalling and affects cellular proliferation. Biochem Biophys Res Commun *312*, 571-577.

Harbeck, B., Huttelmaier, S., Schluter, K., Jockusch, B. M., and Illenberger, S. (2000). Phosphorylation of the vasodilator-stimulated phosphoprotein regulates its interaction with actin. J Biol Chem 275, 30817-30825.

Hartsough, M. T., and Mulder, K. M. (1995). Transforming growth factor beta activation of p44mapk in proliferating cultures of epithelial cells. J Biol Chem 270, 7117-7124.

Hayashi, M., Shimada, Y., Nishimura, Y., Hama, T., and Tanaka, T. (2002). Genomic organization, chromosomal localization, and alternative splicing of the human phosphodiesterase 8B gene. Biochem Biophys Res Commun 297, 1253-1258.

Hayashi, S., Morishita, R., Matsushita, H., Nakagami, H., Taniyama, Y., Nakamura, T., Aoki, M., Yamamoto, K., Higaki, J., and Ogihara, T. (2000). Cyclic AMP inhibited proliferation of human aortic vascular smooth muscle cells, accompanied by induction of p53 and p21. Hypertension *35*, 237-243.

Haynes, J., Jr., Killilea, D. W., Peterson, P. D., and Thompson, W. J. (1996). Erythro-9-(2hydroxy-3-nonyl)adenine inhibits cyclic-3',5'-guanosine monophosphate-stimulated phosphodiesterase to reverse hypoxic pulmonary vasoconstriction in the perfused rat lung. J Pharmacol Exp Ther 276, 752-757.

Hecquet, C., Lefevre, G., Valtink, M., Engelmann, K., and Mascarelli, F. (2002). cAMP inhibits the proliferation of retinal pigmented epithelial cells through the inhibition of ERK1/2 in a PKA-independent manner. Oncogene 21, 6101-6112.

Herve, P., Launay, J. M., Scrobohaci, M. L., Brenot, F., Simonneau, G., Petitpretz, P., Poubeau, P., Cerrina, J., Duroux, P., and Drouet, L. (1995). Increased plasma serotonin in primary pulmonary hypertension. Am J Med 99, 249-254.

Hetman, J. M., Soderling, S. H., Giavas, N. A., and Beavo, J. A. (2000). Cloning and characterization of PDE7B, a cAMP-specific phosphodiesterase. Proc Natl Acad Sci U S A 97, 472-476.

Hoffmann, R., Baillie, G. S., MacKenzie, S. J., Yarwood, S. J., and Houslay, M. D. (1999). The MAP kinase ERK2 inhibits the cyclic AMP-specific phosphodiesterase HSPDE4D3 by phosphorylating it at Ser579. Embo J 18, 893-903.

Hoffmann, R., Wilkinson, I. R., McCallum, J. F., Engels, P., and Houslay, M. D. (1998). cAMP-specific phosphodiesterase HSPDE4D3 mutants which mimic activation and changes in rolipram inhibition triggered by protein kinase A phosphorylation of Ser-54: generation of a molecular model. Biochem J *333 (Pt 1)*, 139-149.

Hofmann, F., Ammendola, A., and Schlossmann, J. (2000). Rising behind NO: cGMPdependent protein kinases. J Cell Sci 113 (*Pt* 10), 1671-1676.

Horstman, D. J., Frank, D. U., and Rich, G. F. (1998). Prolonged inhaled NO attenuates hypoxic, but not monocrotaline-induced, pulmonary vascular remodeling in rats. Anesth Analg 86, 74-81.

Houslay, M. D. (2001). PDE4 cAMP-specific phosphodiesterases. Prog Nucleic Acid Res Mol Biol 69, 249-315.

Houslay, M. D., and Adams, D. R. (2003). PDE4 cAMP phosphodiesterases: modular enzymes that orchestrate signalling cross-talk, desensitization and compartmentalization. Biochem J 370, 1-18.

Houslay, M. D., and Baillie, G. S. (2003). The role of ERK2 docking and phosphorylation of PDE4 cAMP phosphodiesterase isoforms in mediating cross-talk between the cAMP and ERK signalling pathways. Biochem Soc Trans *31*, 1186-1190.

Houslay, M. D., and Kolch, W. (2000). Cell-type specific integration of cross-talk between extracellular signal-regulated kinase and cAMP signaling. Mol Pharmacol 58, 659-668.

Houslay, M. D., and Milligan, G. (1997). Tailoring cAMP-signalling responses through isoform multiplicity. Trends Biochem Sci 22, 217-224.

Houslay, M. D., Scotland, G., Erdogan, S., Huston, E., Mackenzie, S., McCallum, J. F., McPhee, I., Pooley, L., Rena, G., Ross, A., *et al.* (1997). Intracellular targeting, interaction with Src homology 3 (SH3) domains and rolipram-detected conformational switches in cAMP-specific PDE4A phosphodiesterase. Biochem Soc Trans 25, 374-381.

Houslay, M. D., Scotland, G., Pooley, L., Spence, S., Wilkinson, I., McCallum, F., Julien, P., Rena, N. G., Michie, A. M., and Erdogan, S. (1995). Alternative splicing of the type-IVA cyclic AMP phosphodicsterase gene provides isoform variants with distinct Nterminal domains fused to a common, soluble catalytic unit: 'designer' changes in Vmax, stability and membrane association. Biochem Soc Trans 23, 393-398.

Houslay, M. D., Sullivan, M., and Bolger, G. B. (1998). The multienzyme PDE4 cyclic adenosine monophosphate-specific phosphodiesterase family: intracellular targeting, regulation, and selective inhibition by compounds exerting anti-inflammatory and antidepressant actions. Adv Pharmacol *44*, 225-342.

Howe, P. H., Dobrowolski, S. F., Reddy, K. B., and Stacey, D. W. (1993). Release from G1 growth arrest by transforming growth factor beta 1 requires cellular ras activity. J Biol Chem 268, 21448-21452.

Hrbasova, M., Novotny, J., Hejnova, L., Kolar, F., Neckar, J., and Svoboda, P. (2003). Altered myocardial Gs protein and adenylyl cyclase signaling in rats exposed to chronic hypoxia and normoxic recovery. J Appl Physiol 94, 2423-2432.

Huang, L. E., and Bunn, H. F. (2003). Hypoxia-inducible factor and its biomedical relevance. J Biol Chem 278, 19575-19578.

Huang, L. J., Durick, K., Weiner, J. A., Chun, J., and Taylor, S. S. (1997). D-AKAP2, a novel protein kinase A anchoring protein with a putative RGS domain. Proc Natl Acad Sci U S A 94, 11184-11189.

Humbert, M., Morrell, N. W., Archer, S. L., Stenmark, K. R., MacLean, M. R., Lang, I. M., Christman, B. W., Weir, E. K., Eickelberg, O., Voelkel, N. F., and Rabinovitch, M.
(2004). Cellular and molecular pathobiology of pulmonary arterial hypertension. J Am Coll Cardiol 43, 13S-24S.

Huston, E., Pooley, L., Julien, P., Scotland, G., McPhee, I., Sullivan, M., Bolger, G., and Houslay, M. D. (1996). The human cyclic AMP-specific phosphodiesterase PDE-46 (HSPDE4A4B) expressed in transfected COS7 cells occurs as both particulate and cytosolic species that exhibit distinct kinetics of inhibition by the antidepressant rolipram. J Biol Chem 271, 31334-31344.

Iacovelli, L., Capobianco, L., Salvatore, L., Sallese, M., D'Ancona, G. M., and De Blasi, A. (2001). Thyrotropin activates mitogen-activated protein kinase pathway in FRTL-5 by a cAMP-dependent protein kinase A-independent mechanism. Mol Pharmacol *60*, 924-933.

Igari, H., Tatsumi, K., Sugito, K., Kasahara, Y., Saito, M., Tani, T., Kimura, H., and Kuriyama, T. (1998). Role of EDRF in pulmonary circulation during sustained hypoxia. J Cardiovasc Pharmacol *31*, 299-305.

Indolfi, C., Avvedimento, E. V., Di Lorenzo, E., Esposito, G., Rapacciuolo, A., Giuliano, P., Grieco, D., Cavuto, L., Stingone, A. M., Ciullo, I., *et al.* (1997). Activation of cAMP-PKA signaling in vivo inhibits smooth muscle cell proliferation induced by vascular injury. Nat Med *3*, 775-779.

Indolfi, C., Stabile, E., Coppola, C., Gallo, A., Perrino, C., Allevato, G., Cavuto, L., Torella, D., Di Lorenzo, E., Troncone, G., *et al.* (2001). Membrane-bound protein kinase A inhibits smooth muscle cell proliferation in vitro and in vivo by amplifying cAMP-protein kinase A signals. Circ Res *88*, 319-324.

Inoue, Y., Toga, K., Sudo, T., Tachibana, K., Tochizawa, S., Kimura, Y., Yoshida, Y., and Hidaka, H. (2000). Suppression of arterial intimal hyperplasia by cilostamide, a cyclic nucleotide phosphodiesterase 3 inhibitor, in a rat balloon double-injury model. Br J Pharmacol *130*, 231-241.

Iwami, G., Kawabe, J., Ebina, T., Cannon, P. J., Homey, C. J., and Ishikawa, Y. (1995). Regulation of adenylyl cyclase by protein kinase A. J Biol Chem 270, 12481-12484.

Jacobowitz, O., and Iyengar, R. (1994). Phorbol ester-induced stimulation and

phosphorylation of adenylyl cyclase 2. Proc Natl Acad Sci U S A 91, 10630-10634.

Janakidevi, K., Fisher, M. A., Del Vecchio, P. J., Tiruppathi, C., Figge, J., and Malik, A. B. (1992). Endothelin-1 stimulates DNA synthesis and proliferation of pulmonary artery smooth muscle cells. Am J Physiol *263*, C1295-1301.

Jaski, B. E., Fifer, M. A., Wright, R. F., Braunwald, E., and Colucci, W. S. (1985). Positive inotropic and vasodilator actions of milrinone in patients with severe congestive heart failure. Dose-response relationships and comparison to nitroprusside. J Clin Invest 75, 643-649.

Jin, N., Hatton, N., Swartz, D. R., Xia, X., Harrington, M. A., Larsen, S. H., and Rhoades, R. A. (2000). Hypoxia activates jun-N-terminal kinase, extracellular signal-regulated protein kinase, and p38 kinase in pulmonary arteries. Am J Respir Cell Mol Biol 23, 593-601.

Jin, S. L., Bushnik, T., Lan, L., and Conti, M. (1998). Subcellular localization of rolipramsensitive, cAMP-specific phosphodiesterases. Differential targeting and activation of the splicing variants derived from the PDE4D gene. J Biol Chem 273, 19672-19678.

Jin, S. L., Richard, F. J., Kuo, W. P., D'Ercole, A. J., and Conti, M. (1999). Impaired growth and fertility of cAMP-specific phosphodiesterase PDE4D-deficient mice. Proc Natl Acad Sci U S A 96, 11998-12003.

Johannessen, M., Delghandi, M. P., and Moens, U. (2004). What turns CREB on? Cell Signal 16, 1211-1227.

Johnson-Mills, K., Arauz, E., Coffey, R. G., Krzanowski, J. J., Jr., and Polson, J. B. (1998). Effect of CI-930 [3-(2H)-pyridazinone-4,5-dihydro-6-[4-(1H-imidazolyl) phenyl}-5-methyl-monohydrochloride] and rolipram on human coronary artery smooth muscle cell proliferation. Biochem Pharmacol *56*, 1065-1073.

Johnston, L. A., Erdogan, S., Cheung, Y. F., Sullivan, M., Barber, R., Lynch, M. J., Baillie, G. S., Van Heeke, G., Adams, D. R., Huston, E., and Houslay, M. D. (2004). Expression, intracellular distribution and basis for lack of catalytic activity of the PDE4A7 isoform encoded by the human PDE4A cAMP-specific phosphodiesterase gene. Biochem J 380,

371-384.

Jourdan, K. B., Mason, N. A., Long, L., Philips, P. G., Wilkins, M. R., and Morrell, N. W. (2001). Characterization of adenylyl cyclase isoforms in rat peripheral pulmonary arteries. Am J Physiol Lung Cell Mol Physiol 280, L1359-1369.

Kaji, T., Inada, M., Yamamoto, C., Fujiwara, Y., and Koizumi, F. (1996). Cyclic AMPdependent pathway that mediates suppressive regulation of glycosaminoglycan production in cultured vascular endothelial cells. Thromb Res *82*, 389-397.

Kakkar, R., Raju, R. V., and Sharma, R. K. (1999). Calmodulin-dependent cyclic nucleotide phosphodiesterase (PDE1). Cell Mol Life Sci 55, 1164-1186.

Kallio, P. J., Okamoto, K., O'Brien, S., Carrero, P., Makino, Y., Tanaka, H., and Poellinger, L. (1998). Signal transduction in hypoxic cells: inducible nuclear translocation and recruitment of the CBP/p300 coactivator by the hypoxia-inducible factor-lalpha. Embo *J 17*, 6573-6586.

Kambayashi, T., Jacob, C. O., Zhou, D., Mazurek, N., Fong, M., and Strassmann, G. (1995). Cyclic nucleotide phosphodiesterase type IV participates in the regulation of IL-10 and in the subsequent inhibition of TNF-alpha and IL-6 release by endotoxin-stimulated macrophages. J Immunol *155*, 4909-4916.

Kato, J. Y., Matsuoka, M., Polyak, K., Massague, J., and Sherr, C. J. (1994). Cyclic AMPinduced G1 phase arrest mediated by an inhibitor (p27Kip1) of cyclin-dependent kinase 4 activation. Cell 79, 487-496.

Kato, R., Sato, J., and Nishino, T. (1998). Milrinone decreases both pulmonary arterial and venous resistances in the hypoxic dog. Br J Anaesth 81, 920-924.

Kaufmann, R., Junker, U., Junker, K., Nuske, K., Ranke, C., Zieger, M., and Scheele, J. (2002). The serine proteinase thrombin promotes migration of human renal carcinoma cells by a PKA-dependent mechanism. Cancer Lett *180*, 183-190.

Kitamura, T., Kitamura, Y., Kuroda, S., Hino, Y., Ando, M., Kotani, K., Konishi, H., Matsuzaki, H., Kikkawa, U., Ogawa, W., and Kasuga, M. (1999). Insulin-induced

phosphorylation and activation of cyclic nucleotide phosphodiesterase 3B by the serine-threonine kinase Akt. Mol Cell Biol 19, 6286-6296.

Klemm, D. J., Watson, P. A., Frid, M. G., Dempsey, E. C., Schaack, J., Colton, L. A., Nesterova, A., Stenmark, K. R., and Reusch, J. E. (2001). cAMP response element-binding protein content is a molecular determinant of smooth muscle cell proliferation and migration. J Biol Chem 276, 46132-46141.

Klinger, M., Kudlacek, O., Seidel, M. G., Freissmuth, M., and Sexl, V. (2002). MAP kinase stimulation by cAMP does not require RAP1 but SRC family kinases. J Biol Chem 277, 32490-32497.

Kolch, W. (2000). Meaningful relationships: the regulation of the Ras/Raf/MEK/ERK pathway by protein interactions. Biochem J *351 Pt* 2, 289-305.

Komas, N., Lugnier, C., and Stoclet, J. C. (1991). Endothelium-dependent and independent relaxation of the rat aorta by cyclic nucleotide phosphodiesterase inhibitors. Br J Pharmacol 104, 495-503.

Koyama, H., Bornfeldt, K. E., Fukumoto, S., and Nishizawa, Y. (2001). Molecular pathways of cyclic nucleotide-induced inhibition of arterial smooth muscle cell proliferation. J Cell Physiol *186*, 1-10.

Kronemann, N., Nockher, W. A., Busse, R., and Schini-Kerth, V. B. (1999). Growthinhibitory effect of cyclic GMP- and cyclic AMP-dependent vasodilators on rat vascular smooth muscle cells: effect on cell cycle and cyclin expression. Br J Pharmacol *126*, 349-357.

Laderoute, K. R., Mendonca, H. L., Calaoagan, J. M., Knapp, A. M., Giaccia, A. J., and Stork, P. J. (1999). Mitogen-activated protein kinase phosphatase-1 (MKP-1) expression is induced by low oxygen conditions found in solid tumor microenvironments. A candidate MKP for the inactivation of hypoxia-inducible stress-activated protein kinase/c-Jun Nterminal protein kinase activity. J Biol Chem 274, 12890-12897.

Lai, H. L., Lin, T. H., Kao, Y. Y., Lin, W. J., Hwang, M. J., and Chern, Y. (1999). The N terminus domain of type VI adenylyl cyclase mediates its inhibition by protein kinase C.

Mol Pharmacol 56, 644-650.

L'Allemain, G., Lavoie, J. N., Rivard, N., Baldin, V., and Pouyssegur, J. (1997). Cyclin D1 expression is a major target of the cAMP-induced inhibition of cell cycle entry in fibroblasts. Oncogene 14, 1981-1990.

Landells, L. J., Szilagy, C. M., Jones, N. A., Banner, K. H., Allen, J. M., Doherty, A., O'Connor, B. J., Spina, D., and Page, C. P. (2001). Identification and quantification of phosphodiesterase 4 subtypes in CD4 and CD8 lymphocytes from healthy and asthmatic subjects. Br J Pharmacol *133*, 722-729.

Lane, K. B., Machado, R. D., Pauciulo, M. W., Thomson, J. R., Phillips, J. A., 3rd, Loyd, J. E., Nichols, W. C., and Trembath, R. C. (2000). Heterozygous germline mutations in BMPR2, encoding a TGF-beta receptor, cause familial primary pulmonary hypertension. The International PPH Consortium. Nat Genet 26, 81-84.

Lang, D., and Lewis, M. J. (1991). Endothelium-derived relaxing factor inhibits the endothelin-1-induced increase in protein kinase C activity in rat aorta. Br J Pharmacol 104, 139-144.

Lavoie, J. N., L'Allemain, G., Brunet, A., Muller, R., and Pouyssegur, J. (1996). Cyclin D1 expression is regulated positively by the p42/p44MAPK and negatively by the p38/HOGMAPK pathway. J Biol Chem 271, 20608-20616.

Le Cras, T. D., Xue, C., Rengasamy, A., and Johns, R. A. (1996). Chronic hypoxia upregulates endothelial and inducible NO synthase gene and protein expression in rat lung. Am J Physiol *270*, L164-170.

Le Jenne, I. R., Shepherd, M., Van Heeke, G., Houslay, M. D., and Hall, I. P. (2002). Cyclic AMP-dependent transcriptional up-regulation of phosphodiesterase 4D5 in human airway smooth muscle cells. Identification and characterization of a novel PDE4D5 promoter. J Biol Chem 277, 35980-35989.

Lee, S. L., Wang, W. W., Finlay, G. A., and Fanburg, B. L. (1999). Scrotonin stimulates mitogen-activated protein kinase activity through the formation of superoxide anion. Am J Physiol 277, L282-291.

Leon-Velarde, F., Bourin, M. C., Germack, R., Mohammadi, K., Crozatier, B., and Richalet, J. P. (2001). Differential alterations in cardiac adrenergic signaling in chronic hypoxia or norepinephrine infusion. Am J Physiol Regul Integr Comp Physiol 280, R274-281.

Leroy, M. J., Mehats, C., Duc-Goiran, P., Tanguy, G., Robert, B., Dałlot, E., Mignot, T. M., Grange, G., and Ferre, F. (1999). Effect of pregnancy on PDE4 cAMP-specific phosphodiesterase messenger ribonucleic acid expression in human myometrium. Cell Signal 11, 31-37.

Li, D., Zhou, N., and Johns, R. A. (1999). Soluble guanylate cyclase gene expression and localization in rat lung after exposure to hypoxia. Am J Physiol 277, L841-847.
Li, H., Chen, S. J., Chen, Y. F., Meng, Q. C., Durand, J., Oparil, S., and Elton, T. S. (1994). Enhanced endothelin-1 and endothelin receptor gene expression in chronic hypoxia. J Appl Physiol 77, 1451-1459.

Li, H. T., Honbo, N. Y., and Karliner, J. S. (1996). Chronic hypoxia increases beta 1adrenergic receptor mRNA and density but not signaling in neonatal rat cardiac myocytes. Circulation 94, 3303-3310.

Lim, J., Pahlke, G., and Conti, M. (1999). Activation of the cAMP-specific phosphodiesterase PDE4D3 by phosphorylation. Identification and function of an inhibitory domain. J Biol Chem 274, 19677-19685.

Limsuwan, A., Platoshyn, O., Yu, Y., Rubin, L. J., Rothman, A., and Yuan, J. X. (2001). Inhibition of K(+) channel activity in human pulmonary artery smooth muscle cells by serum from patients with pulmonary hypertension secondary to congenital heart disease. Pediatr Res 50, 23-28.

Lindgren, S., Andersson, K. E., Belfrage, P., Degerman, E., and Manganiello, V. C. (1989). Relaxant effects of the selective phosphodiesterase inhibitors milrinone and OPC 3911 on isolated human mesenteric vessels. Pharmacol Toxicol 64, 440-445.

Lindquist, J. M., Fredriksson, J. M., Rehnmark, S., Cannon, B., and Nedergaard, J. (2000). Beta 3- and alphal-adrenergic Erk1/2 activation is Src- but not Gi-mediated in Brown adipocytes. J Biol Chem 275, 22670-22677.

Littler, C. M., Morris, K. G., Jr., Fagan, K. A., McMurtry, I. F., Messing, R. O., and Dempsey, E. C. (2003). Protein kinase C-epsilon-null mice have decreased hypoxic pulmonary vasoconstriction. Am J Physiol Heart Circ Physiol 284, H1321-1331.

Liu, H., and Maurice, D. H. (1998). Expression of cyclic GMP-inhibited phosphodicsterases 3A and 3B (PDE3A and PDE3B) in rat tissues: differential subcellular localization and regulated expression by cyclic AMP. Br J Pharmacol *125*, 1501-1510.

Liu, H., and Maurice, D. H. (1999). Phosphorylation-mediated activation and translocation of the cyclic AMP-specific phosphodiesterase PDE4D3 by cyclic AMP-dependent protein kinase and mitogen-activated protein kinases. A potential mechanism allowing for the coordinated regulation of PDE4D activity and targeting. J Biol Chem 274, 10557-10565.

Lohmann, S. M., Vaandrager, A. B., Smolenski, A., Walter, U., and De Jonge, H. R. (1997). Distinct and specific functions of cGMP-dependent protein kinases. Trends Biochem Sci 22, 307-312.

Lorthiois, E., Bernardelli, P., Vergne, F., Oliveira, C., Mafroud, A. K., Proust, E., Heuze, L., Moreau, F., Idrissi, M., Tertre, A., *et al.* (2004). Spiroquinazolinones as novel, potent, and selective PDE7 inhibitors. Part 1. Bioorg Med Chem Lett *14*, 4623-4626.

Loughney, K., Hill, T. R., Florio, V. A., Uher, L., Rosman, G. J., Wolda, S. L., Jones, B. A., Howard, M. L., McAllister-Lucas, L. M., Sonnenburg, W. K., *et al.* (1998). Isolation and characterization of cDNAs encoding PDE5A, a human cGMP-binding, cGMP-specific 3',5'-cyclic nucleotide phosphodiesterase. Gene *216*, 139-147.

Luttrell, L. M., and Lefkowitz, R. J. (2002). The role of beta-arrestins in the termination and transduction of G-protein-coupled receptor signals. J Cell Sci 115, 455-465.

Ma, D., Wu, P., Egan, R. W., Billah, M. M., and Wang, P. (1999). Phosphodiesterase 4B gene transcription is activated by lipopolysaccharide and inhibited by interleukin-10 in human monocytes. Mol Pharmacol 55, 50-57.

Machado, R. D., Pauciulo, M. W., Thomson, J. R., Lane, K. B., Morgan, N. V., Wheeler,

L., Phillips, J. A., 3rd, Newman, J., Williams, D., Galie, N., *et al.* (2001). BMPR2 haploinsufficiency as the inherited molecular mechanism for primary pulmonary hypertension. Am J Hum Genet 68, 92-102.

MacKenzie, S. J., Baillie, G. S., McPhee, I., Bolger, G. B., and Houslay, M. D. (2000). ERK2 mitogen-activated protein kinase binding, phosphorylation, and regulation of the PDE4D cAMP-specific phosphodiesterases. The involvement of COOH-terminal docking sites and NH2-terminal UCR regions. J Biol Chem 275, 16609-16617.

MacKenzie, S. J., Baillie, G. S., McPhec, I., MacKenzie, C., Seamons, R., McSorley, T., Milien, J., Beard, M. B., van Heeke, G., and Houslay, M. D. (2002). Long PDE4 cAMP specific phosphodiesterases are activated by protein kinase A-mediated phosphorylation of a single serine residue in Upstream Conserved Region 1 (UCR1). Br J Pharmacol 136, 421-433.

MacKenzie, S. J., and Houslay, M. D. (2000). Action of rolipram on specific PDE4 cAMP phosphodiesterase isoforms and on the phosphorylation of cAMP-response-elementbinding protein (CREB) and p38 mitogen-activated protein (MAP) kinase in U937 monocytic cells. Biochem J 347, 571-578.

MacKenzie, S. J., Yarwood, S. J., Peden, A. H., Bolger, G. B., Vernon, R. G., and Houslay, M. D. (1998). Stimulation of p70S6 kinase via a growth hormone-controlled phosphatidylinositol 3-kinase pathway leads to the activation of a PDE4A cyclic AMPspecific phosphodiesterase in 3T3-F442A preadipocytes. Proc Natl Acad Sci U S A 95, 3549-3554.

MacLean, M. R. (1998). Endothelin-1: a mediator of pulmonary hypertension? Pulm Pharmacol Ther 11, 125-132.

MacLean, M. R. (1999). Endothelin-1 and scrotonin: mediators of primary and secondary pulmonary hypertension? J Lab Clin Med 134, 105-114.

MacLean, M. R., Deuchar, G. A., Hicks, M. N., Morecroft, I., Shen, S., Sheward, J., Colston, J., Loughlin, L., Nilsen, M., Dempsie, Y., and Harmar, A. (2004). Overexpression of the 5-hydroxytryptamine transporter gene: effect on pulmonary hemodynamics and hypoxia-induced pulmonary hypertension. Circulation *109*, 2150-2155.

MacLean, M. R., Herve, P., Eddahibi, S., and Adnot, S. (2000). 5-hydroxytryptamine and the pulmonary circulation: receptors, transporters and relevance to pulmonary arterial hypertension. Br J Pharmacol *131*, 161-168.

Maclean, M. R., Johnston, E. D., McCulloch, K. M., Pooley, L., Houslay, M. D., and Sweeney, G. (1997). Phosphodicsterase isoforms in the pulmonary arterial circulation of the rat: changes in pulmonary hypertension. J Pharmacol Exp Ther 283, 619-624.

MacLean, M. R., McCulloch, K. M., and Baird, M. (1995). Effects of pulmonary hypertension on vasoconstrictor responses to endothelin-1 and sarafotoxin S6C and on inherent tone in rat pulmonary arteries. J Cardiovasc Pharmacol *26*, 822-830.

MacLean, M. R., Sweeney, G., Baird, M., McCulloch, K. M., Houslay, M., and Morecroft, I. (1996). 5-Hydroxytryptamine receptors mediating vasoconstriction in pulmonary arteries from control and pulmonary hypertensive rats. Br J Pharmacol *119*, 917-930.

MacNicol, M. C., and MacNicol, A. M. (1999). Nerve growth factor-stimulated B-Raf catalytic activity is refractory to inhibition by cAMP-dependent protein kinase. J Biol Chem 274, 13193-13197.

Madden, J. A., Ray, D. E., Keller, P. A., and Kleinman, J. G. (2001). Ion exchange activity in pulmonary artery smooth muscle cells: the response to hypoxia. Am J Physiol Lung Cell Mol Physiol 280, L264-271.

Mandegar, M., Fung, Y. C., Huang, W., Remillard, C. V., Rubin, L. J., and Yuan, J. X. (2004). Cellular and molecular mechanisms of pulmonary vascular remodeling: role in the development of pulmonary hypertension. Microvasc Res *68*, 75-103.

Manning, C. D., Burman, M., Christensen, S. B., Cieslinski, L. B., Essayan, D. M., Grous, M., Torphy, T. J., and Barnette, M. S. (1999). Suppression of human inflammatory cell function by subtype-selective PDE4 inhibitors correlates with inhibition of PDE4A and PDE4B. Br J Pharmacol *128*, 1393-1398.

Masaki, T., Kimura, S., Yanagisawa, M., and Goto, K. (1991). Molecular and cellular mechanism of endothelin regulation. Implications for vascular function. Circulation 84,

1457-1468.

Mason, N. A., Springali, D. R., Burke, M., Pollock, J., Mikhail, G., Yacoub, M. H., and Polak, J. M. (1998). High expression of endothelial nitric oxide synthase in plexiform lesions of pulmonary hypertension. J Pathol 185, 313-318.

Maurice, D. H., Palmer, D., Tilley, D. G., Dunkerley, H. A., Netherton, S. J., Raymond, D. R., Elbatarny, H. S., and Jimmo, S. L. (2003). Cyclic nucleotide phosphodiesterase activity, expression, and targeting in cells of the cardiovascular system. Mol Pharmacol *64*, 533-546.

Mayr, B., and Montminy, M. (2001). Transcriptional regulation by the phosphorylationdependent factor CREB. Nat Rev Mol Cell Biol 2, 599-609.

McCulloch, K. M., Docherty, C., and MacLean, M. R. (1998). Endothelin receptors mediating contraction of rat and human pulmonary resistance arteries: effect of chronic hypoxia in the rat. Br J Pharmacol *123*, 1621-1630.

McDaid, II. M., Cairns, M. T., Atkinson, R. J., McAleer, S., Harkin, D. P., Gilmore, P., and Johnston, P. G. (1999). Increased expression of the RIalpha subunit of the cAMP-dependent protein kinase A is associated with advanced stage ovarian cancer. Br J Cancer 79, 933-939.

McPhee, I., Yarwood, S. J., Scotland, G., Huston, E., Beard, M. B., Ross, A. H., Houslay, E. S., and Houslay, M. D. (1999). Association with the SRC family tyrosyl kinase LYN triggers a conformational change in the catalytic region of human cAMP-specific phosphodicsterase HSPDE4A4B. Consequences for rolipram inhibition. J Biol Chem 274, 11796-11810.

Mehats, C., Andersen, C. B., Filopanti, M., Jin, S. L., and Conti, M. (2002). Cyclic nucleotide phosphodiesterases and their role in endocrine cell signaling. Trends Endocrinol Metab 13, 29-35.

Mehats, C., Jin, S. L., Wahlstrom, J., Law, E., Umetsu, D. T., and Conti, M. (2003). PDE4D plays a critical role in the control of airway smooth muscle contraction. Faseb J 17, 1831-1841.

Mehats, C., Tanguy, G., Dallot, E., Cabrol, D., Ferre, F., and Leroy, M. J. (2001). Is upregulation of phosphodiesterase 4 activity by PGE2 involved in the desensitization of betamimetics in late pregnancy human myometrium? J Clin Endocrinol Metab *86*, 5358-5365.

Mehats, C., Tanguy, G., Dallot, E., Robert, B., Robourcet, R., Ferre, F., and Leroy, M. J. (1999). Selective up-regulation of phosphodiesterase-4 cyclic adenosine 3',5'monophosphate (cAMP)-specific phosphodiesterase variants by elevated cAMP content in human myometrial cells in culture. Endocrinology 140, 3228-3237.

Mehats, C., Tanguy, G., Paris, B., Robert, B., Pernin, N., Ferre, F., and Leroy, M. J. (2000). Pregnancy induces a modulation of the cAMP phosphodiesterase 4-conformers ratio in human myometrium: consequences for the utero-relaxant effect of PDE4-selective inhibitors. J Pharmacol Exp Ther 292, 817-823.

Meyrick, B., and Reid, L. (1978). The effect of continued hypoxia on rat pulmonary arterial circulation. An ultrastructural study. Lab Invest *38*, 188-200.

Michel, J. J., and Scott, J. D. (2002). AKAP mediated signal transduction. Annu Rev Pharmacol Toxicol 42, 235-257.

Michelakis, E., Tymchak, W., Lien, D., Webster, L., Hashimoto, K., and Archer, S. (2002). Oral sildenafil is an effective and specific pulmonary vasodilator in patients with pulmonary arterial hypertension: comparison with inhaled nitric oxide. Circulation *105*, 2398-2403.

Michelakis, E. D., Tymchak, W., Noga, M., Webster, L., Wu, X. C., Lien, D., Wang, S. H., Modry, D., and Archer, S. L. (2003). Long-term treatment with oral sildenafil is safe and improves functional capacity and hemodynamics in patients with pulmonary arterial hypertension. Circulation *108*, 2066-2069.

Miller, W. R. (2002). Regulatory subunits of PKA and breast cancer. Ann N Y Acad Sci 968, 37-48.

Minet, E., Arnould, T., Michel, G., Roland, I., Mottet, D., Raes, M., Remacle, J., and Michiels, C. (2000). ERK activation upon hypoxia: involvement in HIF-1 activation. FEBS

Lett 468, 53-58.

Minet, E., Michel, G., Remacle, J., and Michiels, C. (2000). Role of HIF-1 as a transcription factor involved in embryonic development, cancer progression and apoptosis (review). Int J Mol Med 5, 253-259.

Mingone, C. J., Gupte, S. A., Iesaki, T., and Wolin, M. S. (2003). Hypoxia enhances a cGMP-independent nitric oxide relaxing mechanism in pulmonary arteries. Am J Physiol Lung Cell Mol Physiol 285, L296-304.

Mischak, H., Seitz, T., Janosch, P., Eulitz, M., Steen, H., Schellerer, M., Philipp, A., and Kolch, W. (1996). Negative regulation of Raf-1 by phosphorylation of serine 621. Mol Cell Biol 16, 5409-5418.

Montminy, M. (1997). Transcriptional regulation by cyclic AMP. Annu Rev Biochem 66, 807-822.

Moon, E., Lee, R., Near, R., Weintraub, L., Wolda, S., and Lerner, A. (2002). Inhibition of PDE3B augments PDE4 inhibitor-induced apoptosis in a subset of patients with chronic lymphocytic leukemia. Clin Cancer Res 8, 589-595.

Mullaney, I., Vaughan, D. M., and MacLean, M. M. (1998). Endothelin-1 modulation of cAMP in rat pulmonary arteries: effect of chronic hypoxia. J Cardiovasc Pharmacol 31 Suppl 1, S112-114.

Mullaney, I., Vaughan, D. M., and MacLean, M. R. (2000). Regional modulation of cyclic nucleotides by endothelin-1 in rat pulmonary arteries: direct activation of G(i)2-protein in the main pulmonary artery. Br J Pharmacol *129*, 1042-1048.

Muller, T., Engels, P., and Fozard, J. R. (1996). Subtypes of the type 4 cAMP phosphodiesterases: structure, regulation and selective inhibition. Trends Pharmacol Sci 17, 294-298.

Murata, T., Sato, K., Hori, M., Ozaki, H., and Karaki, H. (2002). Decreased endothelial nitric-oxide synthase (eNOS) activity resulting from abnormal interaction between eNOS and its regulatory proteins in hypoxia-induced pulmonary hypertension. J Biol Chem 277,

44085-44092.

Murray, F., MacLean, M. R., and Pyne, N. J. (2002). Increased expression of the cGMPinhibited cAMP-specific (PDE3) and cGMP binding cGMP-specific (PDE5) phosphodiesterases in models of pulmonary hypertension. Br J Pharmacol 137, 1187-1194.

Murray, F., MacLean, M. R., and Pyne, N. J. (2003). An assessment of the role of the inhibitory gamma subunit of the retinal cyclic GMP phosphodiesterase and its effect on the p42/p44 mitogen-activated protein kinase pathway in animal and cellular models of pulmonary hypertension. Br J Pharmacol *138*, 1313-1319.

Murray, K. J. (1990). Cyclic AMP and mechanisms of vasodilation. Pharmacol Ther 47, 329-345.

Murray, T. R., Chen, L., Marshall, B. E., and Macarak, E. J. (1990). Hypoxic contraction of cultured pulmonary vascular smooth muscle cells. Am J Respir Cell Mol Biol *3*, 457-465.

Murthy, K. S., Zhou, H., and Makhlouf, G. M. (2002). PKA-dependent activation of PDE3A and PDE4 and inhibition of adenylyl cyclase V/VI in smooth muscle. Am J Physiol Cell Physiol 282, C508-517.

Musat, M., Vax, V. V., Borboli, N., Gueorguiev, M., Bonner, S., Korbonits, M., and Grossman, A. B. (2004). Cell cycle dysregulation in pituitary oncogenesis. Front Horm Res 32, 34-62.

Nemoz, G., Prigent, A. F., Moueqqit, M., Fougier, S., Macovschi, O., and Pacheco, H. (1985). Selective inhibition of one of the cyclic AMP phosphodiesterases from rat brain by the neurotropic compound rolipram. Biochem Pharmacol *34*, 2997-3000.

Nemoz, G., Sette, C., and Conti, M. (1997). Selective activation of rolipram-sensitive, cAMP-specific phosphodiesterase isoforms by phosphatidic acid. Mol Pharmacol 51, 242-249.

Newman, W. H., Castresana, M. R., Webb, J. G., and Wang, Z. (2003). Cyclic AMP inhibits production of interleukin-6 and migration in human vascular smooth muscle cells.

J Surg Res 109, 57-61.

Obernolte, R., Ratzliff, J., Baecker, P. A., Daniels, D. V., Zuppan, P., Jarnagin, K., and Shelton, E. R. (1997). Multiple splice variants of phosphodiesterase PDE4C cloned from human lung and testis. Biochim Biophys Acta 1353, 287-297.

Ogawa, R., Streiff, M. B., Bugayenko, A., and Kato, G. J. (2002). Inhibition of PDE4 phosphodiesterase activity induces growth suppression, apoptosis, glucocorticoid sensitivity, p53, and p21(WAF1/CIP1) proteins in human acute lymphoblastic leukemia cells. Blood 99, 3390-3397.

Oger, S., Mehats, C., Dallot, E., Ferre, F., and Leroy, M. J. (2002). Interleukin-Ibeta induces phosphodicsterase 4B2 expression in human myometrial cells through a prostaglandin E2- and cyclic adenosine 3',5'-monophosphate-dependent pathway. J Clin Endocrinol Metab 87, 5524-5531.

Oki, N., Takahashi, S. I., Hidaka, H., and Conti, M. (2000). Short term feedback regulation of cAMP in FRTL-5 thyroid cells. Role of PDE4D3 phosphodiesterase activation. J Biol Chem 275, 10831-10837.

Osinski, M. T., Rauch, B. H., and Schror, K. (2001). Antimitogenic actions of organic nitrates are potentiated by sildenafil and mediated via activation of protein kinase A. Mol Pharmacol *59*, 1044-1050.

Osinski, M. T., and Schror, K. (2000). Inhibition of platelet-derived growth factor-induced mitogenesis by phosphodiesterase 3 inhibitors: role of protein kinase A in vascular smooth muscle cell mitogenesis. Biochem Pharmacol *60*, 381-387.

Osinski, M. T., Weber, A., and Schror, K. (2000). Complex actions of protein kinase A inhibitors on mitogenesis of bovine coronary artery smooth muscle cells. Eur J Pharmacol *395*, 173-176.

Palmer, D., and Maurice, D. H. (2000). Dual expression and differential regulation of phosphodiesterase 3A and phosphodiesterase 3B in human vascular smooth muscle: implications for phosphodiesterase 3 inhibition in human cardiovascular tissues. Mol Pharmacol 58, 247-252,

Palmer, D., Tsoi, K., and Maurice, D. H. (1998). Synergistic inhibition of vascular smooth muscle cell migration by phosphodiesterase 3 and phosphodiesterase 4 inhibitors. Circ Res *82*, 852-861.

Pan, X., Arauz, E., Krzanowski, J. J., Fitzpatrick, D. F., and Polson, J. B. (1994). Synergistic interactions between selective pharmacological inhibitors of phosphodiesterase isozyme families PDE III and PDE IV to attenuate proliferation of rat vascular smooth muscle cells. Biochem Pharmacol *48*, 827-835.

Pastan, I. H., Johnson, G. S., and Anderson, W. B. (1975). Role of cyclic nucleotides in growth control. Annu Rev Biochem 44, 491-522.

Pauvert, O., Salvail, D., Rousseau, E., Lugnier, C., Marthan, R., and Savineau, J. P. (2002). Characterisation of cyclic nucleotide phosphodiesterase isoforms in the media layer of the main pulmonary artery. Biochem Pharmacol *63*, 1763-1772.

Pei, J. M., Yu, X. C., Fung, M. L., Zhou, J. J., Cheung, C. S., Wong, N. S., Leung, M. P., and Wong, T. M. (2000). Impaired G(s)alpha and adenylyl cyclase cause beta-adrenoceptor desensitization in chronically hypoxic rat hearts. Am J Physiol Cell Physiol 279, C1455-1463.

Perkett, E. A., Badesch, D. B., Roessler, M. K., Stenmark, K. R., and Meyrick, B. (1992). Insulin-like growth factor I and pulmonary hypertension induced by continuous air embolization in sheep. Am J Respir Cell Mol Biol 6, 82-87.

Perry, S. J., Baillie, G. S., Kohout, T. A., McPhee, I., Magiera, M. M., Ang, K. L., Miller, W. E., McLean, A. J., Conti, M., Houslay, M. D., and Lefkowitz, R. J. (2002). Targeting of cyclic AMP degradation to beta 2-adrenergic receptors by beta-arrestins. Science 298, 834-836.

Peyssonnaux, C., and Eychene, A. (2001). The Raf/MEK/ERK pathway: new concepts of activation. Biol Cell 93, 53-62.

Phillips, B. G., Kato, M., Pesek, C. A., Winnicki, M., Narkiewicz, K., Davison, D., and Somers, V. K. (2000). Sympathetic activation by sildenafil. Circulation 102, 3068-3073.

Pilz, R. B., and Casteel, D. E. (2003). Regulation of gene expression by cyclic GMP. Circ Res 93, 1034-1046.

Pinelli, E., Poux, N., Garren, L., Pipy, B., Castegnaro, M., Miller, D. J., and Pfohl-Leszkowicz, A. (1999). Activation of mitogen-activated protein kinase by fumonisin B(1) stimulates cPLA(2) phosphorylation, the arachidonic acid cascade and cAMP production. Carcinogenesis 20, 1683-1688.

Platoshyn, O., Yu, Y., Golovina, V. A., McDaniel, S. S., Krick, S., Li, L., Wang, J. Y., Rubin, L. J., and Yuan, J. X. (2001). Chronic hypoxia decreases K(V) channel expression and function in pulmonary artery myocytes. Am J Physiol Lung Cell Mol Physiol 280, L801-812.

Podzuweit, T., Nennstiel, P., and Muller, A. (1995). Isozyme selective inhibition of cGMPstimulated cyclic nucleotide phosphodiesterases by erythro-9-(2-hydroxy-3-nonyl) adenine. Cell Signal 7, 733-738.

Polson, J. B., and Strada, S. J. (1996). Cyclic nucleotide phosphodiesterases and vascular smooth muscle. Annu Rev Pharmacol Toxicol *36*, 403-427.

Pooley, L., Shakur, Y., Rena, G., and Houslay, M. D. (1997). Intracellular localization of the PDE4A cAMP-specific phosphodiesterase splice variant RD1 (RNPDE4A1A) in stably transfected human thyroid carcinoma FTC cell lines. Biochem J 321 (*Pt 1*), 177-185.

Pouyssegur, J., Volmat, V., and Lenormand, P. (2002). Fidelity and spatio-temporal control in MAP kinase (ERKs) signalling. Biochem Pharmacol 64, 755-763.

Prasad, S., Wilkinson, J., and Gatzoulis, M. A. (2000). Sildenafil in primary pulmonary hypertension. N Engl J Med 343, 1342.

Prowse, C. N., and Lew, J. (2001). Mechanism of activation of ERK2 by dual phosphorylation. J Biol Chem 276, 99-103.

Qiao, J., Mei, F. C., Popov, V. L., Vergara, L. A., and Cheng, X. (2002). Cell cycledependent subcellular localization of exchange factor directly activated by cAMP. J Biol Chem 277, 26581-26586. Rabe, K. F., Tenor, H., Dent, G., Schudt, C., Nakashima, M., and Magnussen, H. (1994). Identification of PDE isozymes in human pulmonary artery and effect of selective PDE inhibitors. Am J Physiol 266, L536-543.

Rabinovitch, M., Gamble, W., Nadas, A. S., Miettinen, O. S., and Reid, L. (1979). Rat pulmonary circulation after chronic hypoxia: hemodynamic and structural features. Am J Physiol 236, H818-827.

Raj, U., and Shimoda, L. (2002). Oxygen-dependent signaling in pulmonary vascular smooth muscle. Am J Physiol Lung Cell Mol Physiol 283, L671-677.

Rangarajan, S., Enserink, J. M., Kuiperij, H. B., de Rooij, J., Price, L. S., Schwede, F., and Bos, J. L. (2003). Cyclic AMP induces integrin-mediated cell adhesion through Epac and Rap1 upon stimulation of the beta 2-adrenergic receptor. J Cell Biol *160*, 487-493.

Rena, G., Begg, F., Ross, A., MacKenzie, C., McPhee, I., Campbell, L., Huston, E., Sullivan, M., and Houslay, M. D. (2001). Molecular cloning, genomic positioning, promoter identification, and characterization of the novel cyclic amp-specific phosphodiesterase PDE4A10. Mol Pharmacol *59*, 996-1011.

Ribeiro-Neto, F., Urbani, J., Lemee, N., Lou, L., and Altschuler, D. L. (2002). On the mitogenic properties of Rap1b: cAMP-induced G(1)/S entry requires activated and phosphorylated Rap1b. Proc Natl Acad Sci U S A 99, 5418-5423.

Rich, S., Dantzker, D. R., Ayres, S. M., Bergofsky, E. H., Brundage, B. H., Detre, K. M., Fishman, A. P., Goldring, R. M., Groves, B. M., and Koerner, S. K. (1987). Primary pulmonary hypertension. A national prospective study. Ann Intern Med *107*, 216-223.

Richard, D. E., Berra, E., Gothie, E., Roux, D., and Pouyssegur, J. (1999). p42/p44 mitogen-activated protein kinases phosphorylate hypoxia-inducible factor lalpha (HIF-lalpha) and enhance the transcriptional activity of HIF-1. J Biol Chem 274, 32631-32637.

Rothman, A., Wolner, B., Button, D., and Taylor, P. (1994). Immediate-early gene expression in response to hypertrophic and proliferative stimuli in pulmonary arterial smooth muscle cells. J Biol Chem 269, 6399-6404.

Ruth, P. (1999). Cyclic GMP-dependent protein kinases: understanding in vivo functions by gene targeting. Pharmacol Ther 82, 355-372.

Rybalkin, S. D., and Bornfeldt, K. E. (1999). Cyclic nucleotide phosphodiesterases and human arterial smooth muscle cell proliferation. Thromb Haemost 82, 424-434.

Rybalkin, S. D., Rybalkina, I., Beavo, J. A., and Bornfeldt, K. E. (2002). Cyclic nucleotide phosphodiesterase IC promotes human arterial smooth muscle cell proliferation. Circ Res 90, 151-157.

Rybalkin, S. D., Yan, C., Bornfeldt, K. E., and Beavo, J. A. (2003). Cyclic GMP phosphodiesterases and regulation of smooth muscle function. Circ Res 93, 280-291.

Saeki, K., Yuo, A., and Takaku, F. (1999). Cell-cycle-regulated phosphorylation of cAMP response element-binding protein: identification of novel phosphorylation sites. Biochem J *338 (Pt 1)*, 49-54.

Salanova, M., Jin, S. C., and Conti, M. (1998). Heterologous expression and purification of recombinant rolipram-sensitive cyclic AMP-specific phosphodiesterases. Methods 14, 55-64.

Savage, A., Zeng, L., and Houslay, M. D. (1995). A role for protein kinase C-mediated phosphorylation in eliciting glucagon desensitization in rat hepatocytes. Biochem J 307 (*Pt 1*), 281-285.

Savini, F., Berardi, S., Tatone, D., and Spoto, G. (1995). Phosphodiesterase in human colon carcinoma cell line CaCo-2 in culture. Life Sci 56, PL421-425.

Schmidt, C. M., McKillop, I. H., Cahill, P. A., and Sitzmann, J. V. (1999). The role of cAMP-MAPK signalling in the regulation of human hepatocellular carcinoma growth in vitro. Eur J Gastroenterol Hepatol 11, 1393-1399.

Schmitt, J. M., and Stork, P. J. (2001). Cyclic AMP-mediated inhibition of cell growth requires the small G protein Rap1. Mol Cell Biol 21, 3671-3683.

Schmitt, J. M., and Stork, P. J. (2002). PKA phosphorylation of Src mediates cAMP's inhibition of cell growth via Rap1. Mol Ceil *9*, 85-94.

Schudt, C., Gantner, F., Tenors, H., and Hatzelmann, A. (1999). Therapeutic potential of selective PDE inhibitors in asthma. Pulm Pharmacol Ther *12*, 123-129.

Schutte, H., Witzenrath, M., Mayer, K., Weissmann, N., Schell, A., Rosseau, S., Seeger, W., and Grimminger, F. (2000). The PDE inhibitor zaprinast enhances NO-mediated protection against vascular leakage in reperfused lungs. Am J Physiol Lung Cell Mol Physiol 279, L496-502.

Schutze, N. (2004). siRNA technology. Mol Cell Endocrinol 213, 115-119.

Scott, P. H., Paul, A., Belham, C. M., Peacock, A. J., Wadsworth, R. M., Gould, G. W., Welsh, D., and Plevin, R. (1998). Hypoxic stimulation of the stress-activated protein kinases in pulmonary artery fibroblasts. Am J Respir Crit Care Med 158, 958-962.

Sebkhi, A., Strange, J. W., Phillips, S. C., Wharton, J., and Wilkins, M. R. (2003). Phosphodiesterase type 5 as a target for the treatment of hypoxia-induced pulmonary hypertension. Circulation *107*, 3230-3235.

Semenza, G. L. (1999). Regulation of mammalian O2 homeostasis by hypoxia-inducible factor 1. Annu Rev Cell Dev Biol 15, 551-578.

Semenza, G. L. (2000). HIF-1 and human disease: one highly involved factor. Genes Dev 14, 1983-1991.

Sette, C., and Conti, M. (1996). Phosphorylation and activation of a cAMP-specific phosphodiesterase by the cAMP-dependent protein kinase. Involvement of serine 54 in the enzyme activation. J Biol Chem 271, 16526-16534.

Shabb, J. B. (2001). Physiological substrates of cAMP-dependent protein kinase. Chem Rev 101, 2381-2411.

Shaul, P. W., Kinaue, B., Farrar, M. A., Buja, L. M., and Magness, R. R. (1991). Prostacyclin production and mediation of adenylate cyclase activity in the pulmonary artery. Alterations after prolonged hypoxia in the rat. J Clin Invest 88, 447-455.

Shaul, P. W., Muntz, K. H., and Buja, L. M. (1990). Comparison of beta adrenergic receptor binding characteristics and coupling to adenylate cyclase in rat pulmonary artery versus aorta. J Pharmacol Exp Ther 252, 86-92.

Shaul, P. W., Wells, L. B., and Horning, K. M. (1993). Acute and prolonged hypoxia attenuate endothelial nitric oxide production in rat pulmonary arteries by different mechanisms. J Cardiovase Pharmacol 22, 819-827.

Shaywitz, A. J., and Greenberg, M. E. (1999). CREB: a stimulus-induced transcription factor activated by a diverse array of extracellular signals. Annu Rev Biochem 68, 821-861.

Showers, M. O., and Maurer, R. A. (1986). A cloned bovine cDNA encodes an alternate form of the catalytic subunit of cAMP-dependent protein kinase. J Biol Chem 261, 16288-16291.

Silver, P. J. (1985). Regulation of contractile activity in vascular smooth muscle by protein kinases. Rev Clin Basic Pharm 5, 341-395.

Simonneau, G., Galie, N., Rubin, L. J., Langleben, D., Seeger, W., Domenighetti, G., Gibbs, S., Lebrec, D., Speich, R., Beghetti, M., *et al.* (2004). Clinical classification of pulmonary hypertension. J Am Coll Cardiol 43, 5S-12S.

Singh, S., and Evans, T. W. (1997). Nitric oxide, the biological mediator of the decade: fact or fiction? Eur Respir J 10, 699-707.

Singhal, S., Henderson, R., Horsfield, K., Harding, K., and Cumming, G. (1973). Morphometry of the human pulmonary arterial tree. Circ Res *33*, 190-197.

Skalhegg, B. S., and Tasken, K. (2000). Specificity in the cAMP/PKA signaling pathway. Differential expression, regulation, and subcellular localization of subunits of PKA. Front Biosci *5*, D678-693.

Soderling, S. H., Bayuga, S. J., and Beavo, J. A. (1998). Cloning and characterization of a

cAMP-specific cyclic nucleotide phosphodiesterase. Proc Natl Acad Sci U S A 95, 8991-8996.

Soderling, S. H., Bayuga, S. J., and Beavo, J. A. (1999). Isolation and characterization of a dual-substrate phosphodiesterase gene family: PDE10A. Proc Natl Acad Sci U S A 96, 7071-7076.

Soderling, S. H., and Beavo, J. A. (2000). Regulation of cAMP and cGMP signaling: new phosphodiesterases and new functions. Curr Opin Cell Biol *12*, 174-179.

Sonnenburg, W. K., Rybalkin, S. D., Bornfeldt, K. E., Kwak, K. S., Rybalkina, I. G., and Beavo, J. A. (1998). Identification, quantitation, and cellular localization of PDE1 calmodulin-stimulated cyclic nucleotide phosphodiesterases. Methods *14*, 3-19.

Souness, J. E., Hassall, G. A., and Parrott, D. P. (1992). Inhibition of pig aortic smooth muscle cell DNA synthesis by selective type III and type IV cyclic AMP phosphodicsterase inhibitors. Biochem Pharmacol 44, 857-866.

Souness, J. E., and Rao, S. (1997). Proposal for pharmacologically distinct conformers of PDE4 cyclic AMP phosphodiesterases. Cell Signal 9, 227-236.

Spina, D. (2003). Phosphodiesterase-4 inhibitors in the treatment of inflammatory lung disease. Drugs 63, 2575-2594.

Spina, D., Landells, L. J., and Page, C. P. (1998). The role of phosphodiesterase enzymes in allergy and asthma. Adv Pharmacol 44, 33-89.

Stelzner, T. J., O'Brien, R. F., Yanagisawa, M., Sakurai, T., Sato, K., Webb, S., Zamora, M., McMurtry, I. F., and Fisher, J. H. (1992). Increased lung endothelin-1 production in rats with idiopathic pulmonary hypertension. Am J Physiol 262, L614-620.

Stenmark, K. R., Bouchey, D., Nemenoff, R., Dempsey, E. C., and Das, M. (2000). Hypoxia-induced pulmonary vascular remodeling: contribution of the adventitial fibroblasts. Physiol Res 49, 503-517.

Stenmark, K. R., Gerasimovskaya, E., Nemenoff, R. A., and Das, M. (2002). Hypoxic

activation of adventitial fibroblasts: role in vascular remodeling. Chest 122, 326S-334S.

Stergiopoulos, S. G., and Stratakis, C. A. (2003). Human tumors associated with Carney complex and germline PRKAR1A mutations: a protein kinase A disease! FEBS Lett 546, 59-64.

Stewart, A. G., Harris, T., Fernandes, D. J., Schachte, L. C., Koutsoubos, V., Guida, E., Ravenhall, C. E., Vadiveloo, P., and Wilson, J. W. (1999). Beta2-adrenergic receptor agonists and cAMP arrest human cultured airway smooth muscle cells in the G(1) phase of the cell cycle: role of proteasome degradation of cyclin D1. Mol Pharmacol *56*, 1079-1086.

Stewart, D. J., Levy, R. D., Cernacek, P., and Langleben, D. (1991). Increased plasma endothelin-1 in pulmonary hypertension: marker or mediator of disease? Ann Intern Med *114*, 464-469.

Stork, P. J., and Schmitt, J. M. (2002). Crosstalk between cAMP and MAP kinase signaling in the regulation of cell proliferation. Trends Cell Biol *12*, 258-266.

Strange, J. W., Wharton, J., Phillips, P. G., and Wilkins, M. R. (2002). Recent insights into the pathogenesis and therapeutics of pulmonary hypertension. Clin Sci (Lond) *102*, 253-268.

Sun, J., Sui, X., Bradbury, J. A., Zeldin, D. C., Conte, M. S., and Liao, J. K. (2002). Inhibition of vascular smooth muscle cell migration by cytochrome p450 epoxygenase-derived cicosanoids. Circ Res *90*, 1020-1027.

Sutherland, E. W., and Rall, T. W. (1958). Fractionation and characterization of a cyclic adenine ribonucleotide formed by tissue particles. J Biol Chem 232, 1077-1091.

Tasken, K. A., Collas, P., Kemmner, W. A., Witczak, O., Conti, M., and Tasken, K. (2001). Phosphodiesterase 4D and protein kinase a type II constitute a signaling unit in the centrosomal area. J Biol Chem 276, 21999-22002.

Taylor, C. T., Furuta, G. T., Synnestvedt, K., and Colgan, S. P. (2000). Phosphorylationdependent targeting of cAMP response element binding protein to the

ubiquitin/proteasome pathway in hypoxia. Proc Natl Acad Sci U S A 97, 12091-12096.

Terry, R., Cheung, Y. F., Praestegaard, M., Baillie, G. S., Huston, E., Gall, I., Adams, D. R., and Houslay, M. D. (2003). Occupancy of the catalytic site of the PDE4A4 cyclic AMP phosphodiesterase by rolipram triggers the dynamic redistribution of this specific isoform in living cells through a cyclic AMP independent process. Cell Signal *15*, 955-971.

Thomas, M. K., Francis, S. H., and Corbin, J. D. (1990). Substrate- and kinase-directed regulation of phosphorylation of a cGMP-binding phosphodiesterase by cGMP. J Biol Chem 265, 14971-14978.

Thomson, J. R., Machado, R. D., Pauciulo, M. W., Morgan, N. V., Humbert, M., Elliott, G. C., Ward, K., Yacoub, M., Mikhail, G., Rogers, P., *et al.* (2000). Sporadic primary pulmonary hypertension is associated with germline mutations of the gene encoding BMPR-II, a receptor member of the TGF-beta family. J Med Genet *37*, 741-745.

Tibbles, L. A., and Woodgett, J. R. (1999). The stress-activated protein kinase pathways. Cell Mol Life Sci 55, 1230-1254.

Tilley, D. G., and Maurice, D. H. (2002). Vascular smooth muscle cell phosphodiesterase (PDE) 3 and PDE4 activities and levels are regulated by cyclic AMP in vivo. Mol Pharmacol 62, 497-506.

Timmer, W., Leclerc, V., Birraux, G., Neuhauser, M., Hatzelmann, A., Bethke, T., and Wurst, W. (2002). The new phosphodiesterase 4 inhibitor roflumilast is efficacious in exercise-induced asthma and leads to suppression of LPS-stimulated TNF-alpha ex vivo. J Clin Pharmacol *42*, 297-303.

Tomlinson, P. R., Wilson, J. W., and Stewart, A. G. (1995). Salbutamol inhibits the proliferation of human airway smooth muscle cells grown in culture: relationship to elevated cAMP levels. Biochem Pharmacol 49, 1809-1819.

Torphy, T. J. (1998). Phosphodiesterase isozymes: molecular targets for novel antiasthma agents. Am J Respir Crit Care Med 157, 351-370.

Torphy, T. J., Barnette, M. S., Underwood, D. C., Griswold, D. E., Christensen, S. B.,

Murdoch, R. D., Nieman, R. B., and Compton, C. H. (1999). Ariflo (SB 207499), a second generation phosphodiesterase 4 inhibitor for the treatment of asthma and COPD: from concept to clinic. Pulm Pharmacol Ther *12*, 131-135.

Torphy, T. J., and Page, C. (2000). Phosphodiesterases: the journey towards therapeutics. Trends Pharmacol Sci 21, 157-159.

Tsuchikane, E., Fukuhara, A., Kobayashi, T., Kirino, M., Yamasaki, K., Izumi, M., Otsuji, S., Tateyama, H., Sakurai, M., and Awata, N. (1999). Impact of cilostazol on restenosis after percutaneous coronary balloon angioplasty. Circulation *100*, 21-26.

Tuder, R. M., Cool, C. D., Geraci, M. W., Wang, J., Abman, S. H., Wright, L., Badesch, D., and Voelkel, N. F. (1999). Prostacyclin synthase expression is decreased in lungs from patients with severe pulmonary hypertension. Am J Respir Crit Care Med 159, 1925-1932.

Tuder, R. M., Flook, B. E., and Voelkel, N. F. (1995). Increased gene expression for VEGF and the VEGF receptors KDR/Flk and Flt in lungs exposed to acute or to chronic hypoxia. Modulation of gene expression by nitric oxide. J Clin Invest *95*, 1798-1807.

Uhler, M. D., Chrivia, J. C., and McKnight, G. S. (1986). Evidence for a second isoform of the catalytic subunit of cAMP-dependent protein kinase. J Biol Chem 261, 15360-15363.

Underwood, D. C., Bochnowicz, S., Osborn, R. R., Louden, C. S., Hart, T. K., Ohlstein, E. H., and Hay, D. W. (1998). Chronic hypoxia-induced cardiopulmonary changes in three rat strains: inhibition by the endothelin receptor antagonist SB 217242. J Cardiovasc Pharmacol *31 Suppl 1*, S453-455.

van Oirschot, B. A., Stahl, M., Lens, S. M., and Medema, R. H. (2001). Protein kinase A regulates expression of p27(kip1) and cyclin D3 to suppress proliferation of leukemic T cell lines. J Biol Chem 276, 33854-33860.

Vender, R. L. (1994). Chronic hypoxic pulmonary hypertension. Cell biology to pathophysiology. Chest 106, 236-243.

Verde, I., Pahlke, G., Salanova, M., Zhang, G., Wang, S., Coletti, D., Onuffer, J., Jin, S. L., and Conti, M. (2001). Myomegalin is a novel protein of the golgi/centrosome that interacts

with a cyclic nucleotide phosphodiesterase. J Biol Chem 276, 11189-11198.

Veugelers M, W. D., Burton K, McDermott DA, Yan S, Vaughan CJ, Hahn R, Goldstein MM, La Perle K, McKnight S, Basson CT (2003). Comparative genotype-phenotype analyses of human carney complex and murine prkarlalpha haploinsufficiency. CIRCULATION 108, 387 Suppl.

Vicencio, A. G., Eickelberg, O., Stankewich, M. C., Kashgarian, M., and Haddad, G. G. (2002). Regulation of TGF-beta ligand and receptor expression in neonatal rat lungs exposed to chronic hypoxia. J Appl Physiol *93*, 1123-1130.

Vicini, E., and Conti, M. (1997). Characterization of an intronic promoter of a cyclic adenosine 3',5'-monophosphate (cAMP)-specific phosphodiesterase gene that confers hormone and cAMP inducibility. Mol Endocrinol 11, 839-850.

Vignola, A. M. (2004). PDE4 inhibitors in COPD--a more selective approach to treatment. Respir Med 98, 495-503.

Voelkel, N. F. (1997). Appetite suppressants and pulmonary hypertension. Thorax 52 Suppl 3, S63-67.

Voelkel, N. F., Hegstrand, L., Reeves, J. T., McMurty, I. F., and Molinoff, P. B. (1981). Effects of hypoxia on density of beta-adrenergic receptors. J Appl Physiol 50, 363-366.

Voelkel, N. F., and Tuder, R. M. (1995). Cellular and molecular mechanisms in the pathogenesis of severe pulmonary hypertension. Eur Respir J 8, 2129-2138.

Voeikel, N. F., and Tuder, R. M. (1997). Cellular and molecular biology of vascular smooth muscle cells in pulmonary hypertension. Pulm Pharmacol Ther 10, 231-241.

Volmat, V., Camps, M., Arkinstall, S., Pouyssegur, J., and Lenormand, P. (2001). The nucleus, a site for signal termination by sequestration and inactivation of p42/p44 MAP kinases. J Cell Sci *114*, 3433-3443.

Wagner, R. S., Smith, C. J., Taylor, A. M., and Rhoades, R. A. (1997). Phosphodiesterase inhibition improves agonist-induced relaxation of hypertensive pulmonary arteries. J

Pharmacol Exp Ther 282, 1650-1657.

Wang, J., Juhaszova, M., Rubin, L. J., and Yuan, X. J. (1997). Hypoxia inhibits gene expression of voltage-gated K+ channel alpha subunits in pulmonary artery smooth muscle cells. J Clin Invest *100*, 2347-2353.

Wang, P., Wu, P., Ohleth, K. M., Egan, R. W., and Billah, M. M. (1999). Phosphodiesterase 4B2 is the predominant phosphodiesterase species and undergoes differential regulation of gene expression in human monocytes and neutrophils. Mol Pharmacol 56, 170-174.

Wang, W., Rastinejad, F., and El-Deiry, W. S. (2003). Restoring p53-dependent tumor suppression. Cancer Biol Ther 2, S55-63.

Wang, Y., Chen, X., and Colvin, R. A. (2000). Expression of the Na(+)/Ca(2+) exchanger ameliorates ionomycin-induced cell death. Biochem Biophys Res Commun 276, 93-96.

Wang, Z., Jin, N., Ganguli, S., Swartz, D. R., Li, L., and Rhoades, R. A. (2001). Rhokinase activation is involved in hypoxia-induced pulmonary vasoconstriction. Am J Respir Cell Mol Biol 25, 628-635.

Ward, Y., Gupta, S., Jensen, P., Wartmann, M., Davis, R. J., and Kelly, K. (1994). Control of MAP kinase activation by the mitogen-induced threonine/tyrosine phosphatase PAC1. Nature *367*, 651-654.

Watson, S. J., Meng, F., Thompson, R. C., and Akil, H. (2000). The "chip" as a specific genetic tool. Biol Psychiatry 48, 1147-1156.

West, J., Fagan, K., Steudel, W., Fouty, B., Lane, K., Harral, J., Hoedt-Miller, M., Tada, Y., Ozimek, J., Tuder, R., and Rodman, D. M. (2004). Pulmonary hypertension in transgenic mice expressing a dominant-negative BMPRII gene in smooth muscle. Circ Res 94, 1109-1114.

Wildman, S. S., King, B. F., and Burnstock, G. (1999). Modulation of ATP-responses at recombinant rP2X4 receptors by extracellular pH and zinc. Br J Pharmacol *126*, 762-768. Wohrley, J. D., Frid, M. G., Moiseeva, E. P., Orton, E. C., Belknap, J. K., and Stenmark,

K. R. (1995). Hypoxia selectively induces proliferation in a specific subpopulation of smooth muscle cells in the bovine neonatal pulmonary arterial media. J Clin Invest 96, 273-281.

Wojnowski, L., Zimmer, A. M., Beck, T. W., Hahn, H., Bernal, R., Rapp, U. R., and Zimmer, A. (1997). Endothelial apoptosis in Braf-deficient mice. Nat Genet 16, 293-297. Wong, W. S., and Koh, D. S. (2000). Advances in immunopharmacology of asthma. Biochem Pharmacol 59, 1323-1335.

Xiao, F., Che, D. Y., and Zhang, W. R. (1993). A study of endothelium-dependent proliferation of pulmonary smooth muscle cells in vitro. J Tongji Med Univ 13, 10-13. Yanagisawa, M., Kurihara, H., Kimura, S., Goto, K., and Masaki, T. (1988). A novel peptide vasoconstrictor, endothelin, is produced by vascular endothelium and modulates smooth muscle Ca2+ channels. J Hypertens Suppl 6, S188-191.

Yang, X., Sheares, K. K., Davie, N., Upton, P. D., Taylor, G. W., Horsley, J., Wharton, J., and Morrell, N. W. (2002). Hypoxic induction of cox-2 regulates proliferation of human pulmonary artery smooth muscle cells. Am J Respir Cell Mol Biol 27, 688-696.

Yarwood, S. J., Steele, M. R., Scotland, G., Houslay, M. D., and Bolger, G. B. (1999). The RACK1 signaling scaffold protein selectively interacts with the cAMP-specific phosphodiesterase PDE4D5 isoform. J Biol Chem 274, 14909-14917.

Yau, L., and Zahradka, P. (2003). PGE(2) stimulates vascular smooth muscle cell proliferation via the EP2 receptor. Mol Cell Endocrinol 203, 77-90.

Yonehara, T., Kanasaki, H., Yamamoto, H., Fukunaga, K., Miyazaki, K., and Miyamoto, E. (2001). Involvement of mitogen-activated protein kinase in cyclic adenosine 3',5'monophosphate-induced hormone gene expression in rat pituitary GH(3) cells. Endocrinology *142*, 2811-2819.

Yu, A. Y., Shimoda, L. A., Iyer, N. V., Huso, D. L., Sun, X., McWilliams, R., Beaty, T., Sham, J. S., Wiener, C. M., Sylvester, J. T., and Semenza, G. L. (1999). Impaired physiological responses to chronic hypoxia in mice partially deficient for hypoxia-inducible factor lalpha. J Clin Invest *103*, 691-696.

Yu, S. M., Hung, L. M., and Lin, C. C. (1997). cGMP-elevating agents suppress proliferation of vascular smooth muscle cells by inhibiting the activation of epidermal growth factor signaling pathway. Circulation 95, 1269-1277.

Yuan, J. Q., Xu, J. Y., Zhang, J., He, Q. C., Zhu, J., and Sheng, C. X. (2004). Expression and significance of cyclin D1, p27kip1 protein in bronchioloalveolar carcinoma. J Zhejiang Univ Sci 5, 235-241.

Yuan, J. X., Aldinger, A. M., Juhaszova, M., Wang, J., Conte, J. V., Jr., Gaine, S. P., Orens, J. B., and Rubin, L. J. (1998). Dysfunctional voltage-gated K+ channels in pulmonary artery smooth muscle cells of patients with primary pulmonary hypertension. Circulation 98, 1400-1406.

Yuan, X. J., Wang, J., Juhaszova, M., Golovina, V. A., and Rubin, L. J. (1998). Molecular basis and function of voltage-gated K+ channels in pulmonary arterial smooth muscle cells. Am J Physiol 274, L621-635.

Yuasa, K., Kanoh, Y., Okumura, K., and Omori, K. (2001). Genomic organization of the human phosphodiesterase PDE11A gene. Evolutionary relatedness with other PDEs containing GAF domains. Eur J Biochem 268, 168-178.

Yuasa, K., Kotera, J., Fujishige, K., Michibata, H., Sasaki, T., and Omori, K. (2000). Isolation and characterization of two novel phosphodiesterase PDE11A variants showing unique structure and tissue-specific expression. J Biol Chem 275, 31469-31479.

Zaccolo, M., Magalhaes, P., and Pozzan, T. (2002). Compartmentalisation of cAMP and Ca(2+) signals. Curr Opin Cell Biol 14, 160-166.

Zamora, M. A., Dempsey, E. C., Walchak, S. J., and Stelzner, T. J. (1993). BQ123, an ETA receptor antagonist, inhibits endothelin-1-mediated proliferation of human pulmonary artery smooth muscle cells. Am J Respir Cell Mol Biol 9, 429-433.

Zhang, B. H., Tang, E. D., Zhu, T., Greenberg, M. E., Vojtek, A. B., and Guan, K. L. (2001). Serum- and glucocorticoid-inducible kinase SGK phosphorylates and negatively regulates B-Raf. J Biol Chem 276, 31620-31626.

Zhao, L., Mason, N. A., Morrell, N. W., Kojonazarov, B., Sadykov, A., Maripov, A., Mirrakhimov, M. M., Aldashev, A., and Wilkins, M. R. (2001). Sildenafil inhibits hypoxia-induced pulmonary hypertension. Circulation 104, 424-428.

Zimmermann, S., and Moelling, K. (1999). Phosphorylation and regulation of Raf by Akt (protein kinase B). Science 286, 1741-1744.

Zwartkruis, F. J., Wolthuis, R. M., Nabben, N. M., Franke, B., and Bos, J. L. (1998). Extracellular signal-regulated activation of Rap1 fails to interfere in Ras effector signalling. Embo J 17, 5905-5912.

. 1